$\boxtimes$ 

# **UNITED STATES SECURITIES AND EXCHANGE COMMISSION**

	Washington, D.C. 20549
	FORM 10-K
ANNUAL REPORT PURSUANT TO SEC	TION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
For the Fiscal Year Ended December 28, 2007	
	or
TRANSITION REPORT PURSUANT TO OF 1934	SECTION 13 OR 15(d) OF THE SECURITIES AND EXCHANGE AC
For the transition period from to	
	Commission File No. 1-13881
	INTERNATIONAL, INC. ct name of registrant as specified in its charter)
<b>Delaware</b> (State or other jurisdiction of incorporation or organization)	52-2055918 (IRS Employer Identification No.)
10400 Fernwood Road, Bethesda, Maryland (Address of Principal Executive Offices)	20817 (Zip Code)
Registrant's Telen	hone Number Including Area Code (301) 380-3000

Securities registered pursuant to Section 12(b) of the Act:		
Title of Each Class	Name of Each Exchange on Which Registered	
Class A Common Stock, \$0.01 par value	New York Stock Exchange	
(354,043,159 shares outstanding as of January 25, 2008)	Chicago Stock Exchange	
Securities registered pursuant to Section 12(g) of the Act: NONE		
Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in rule	405 of the Securities Act. Yes $\boxtimes$ No $\square$	
Indicate by check mark if the registrant is not required to file reports pursuant to Section 13	3 or Section 15(d) of the Act. Yes $\square$ No $\boxtimes$	
Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by the preceding 12 months (or for such shorter period that the registrant was required to file so the past 90 days. Yes $\boxtimes$ No $\square$	• • • • • • • • • • • • • • • • • • • •	
Indicate by check mark if disclosure by delinquent filers pursuant to Item 405 of Regulation the registrant's knowledge, in definitive proxy or information statements incorporated by result. $\Box$		
Indicate by check mark whether the registrant is a large accelerated filer, an accelerated file large accelerated filer" in Rule 12b-2 of the Exchange Act. (Check one):	er or a non-accelerated filer. See definition of "accelerated filer and	
Large accelerated filer $oxed{\boxtimes}$ Accelerated filer $oxed{\Box}$	Non-accelerated filer □	
Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2	of the Exchange Act). Yes $\square$ No $\boxtimes$	

## DOCUMENTS INCORPORATED BY REFERENCE

The aggregate market value of shares of common stock held by non-affiliates at June 15, 2007, was \$13,599,596,837.

Portions of the Proxy Statement prepared for the 2008 Annual Meeting of Shareholders are incorporated by reference into Part III of this report.

# MARRIOTT INTERNATIONAL, INC.

# FORM 10-K TABLE OF CONTENTS

# FISCAL YEAR ENDED DECEMBER 28, 2007

Page No.

Part I.		
Item 1.	<u>Business</u>	3
Item 1A.	Risk Factors	15
Item 1B.	<u>Unresolved Staff Comments</u>	19
Item 2.	<u>Properties</u>	19
Item 3.	<u>Legal Proceedings</u>	19
Item 4.	Submission of Matters to a Vote of Security Holders	19
Part II.		
Item 5.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	20
Item 6.	Selected Financial Data	21
Item 7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	22
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	58
Item 8.	Financial Statements and Supplementary Data	60
Item 9.	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	105
Item 9A.	Controls and Procedures	105
Item 9B.	Other Information	105
Part III.		
Item 10.	Directors, Executive Officers and Corporate Governance	106
Item 11.	Executive Compensation	106
Item 12.	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	106
Item 13.	Certain Relationships and Related Transactions, and Director Independence	106
Item 14.	Principal Accounting Fees and Services	106
Part IV.		
Item 15.	Exhibits and Financial Statement Schedules	110
	<u>Signatures</u>	113

Throughout this report, we refer to Marriott International, Inc., together with its subsidiaries, as "we," "us," or "the Company." Unless otherwise specified, each reference to a particular year means the fiscal year ended on the date shown in the table below, rather than the corresponding calendar year:

Fiscal Year	Fiscal Year-end Date	Fiscal Year	Fiscal Year-end Date
2007	December 28, 2007	2003	January 2, 2004
2006	December 29, 2006	2002	January 3, 2003
2005	December 30, 2005	2001	December 28, 2001
2004	December 31, 2004	2000	December 29, 2000

## PART I

## Item 1. Business.

We are a worldwide operator and franchisor of hotels and related lodging facilities. We were organized as a corporation in Delaware in 1997 and became a public company in 1998 when we were "spun off" as a separate entity by the company formerly named "Marriott International, Inc." Our operations are grouped into the following five business segments:

<u>Segment</u>	Percentage of 2007 Total Revenues
North American Full-Service Lodging Segment	42%
North American Limited-Service Lodging Segment	17%
International Lodging Segment	12%
Luxury Lodging Segment	12%
Timeshare Segment	16%
Other unallocated corporate	1%

Prior to November 3, 2007, our operations also included our synthetic fuel business, which we now classify as discontinued operations.

Our business includes our North American Full-Service, North American Limited-Service, International, Luxury, and Timeshare segments. We develop, operate, and franchise hotels and corporate housing properties under 15 separate brand names, and we develop, operate, and market timeshare, fractional ownership, and residential properties under four separate brand names. We also provide services to home/condominium owner associations for projects associated with several of our brands.

Financial information by industry segment and geographic area as of and for the 2007, 2006, and 2005 fiscal years then ended appears in Footnote No. 19, "Business Segments," of the Notes to our Consolidated Financial Statements included in this annual report.

## Lodging

We operate or franchise 2,999 lodging properties worldwide, with 535,093 rooms as of year-end 2007 inclusive of 21 home and condominium products (1,916 units) for which we manage the related owners' associations. In addition, we provided 2,156 furnished corporate housing rental units, which are not included in the totals. We believe that our portfolio of lodging brands is the broadest of any company in the world and that we are the leader in the quality tier of the vacation timesharing business. Consistent with our focus on management and franchising, we own very few of our lodging properties. We manage and franchise lodging properties employing the following brands:

## North American Full-Service Lodging Segment

- Marriott® Hotels & Resorts
- JW Marriott® Hotels & Resorts
- Renaissance® Hotels & Resorts
- Renaissance ClubSport®

## North American Limited-Service Lodging Segment

- Courtyard by Marriott® ("Courtyard")
- Fairfield Inn by Marriott® ("Fairfield Inn")
- SpringHill Suites by Marriott ® ("SpringHill Suites")
- Residence Inn by Marriott® ("Residence Inn")
- TownePlace Suites by Marriott® ("TownePlace Suites")
- Marriott ExecuStay®

## **Luxury Segment**

- The Ritz-Carlton®
- Bulgari Hotels & Resorts<sup>®</sup>

## **International Lodging Segment**

- Marriott® Hotels & Resorts
- JW Marriott® Hotels & Resorts
- Renaissance® Hotels & Resorts
- Courtyard by Marriott®
- Fairfield Inn by Marriott®
- Residence Inn by Marriott<sup>®</sup>
- Marriott Executive Apartments<sup>®</sup>

## **Timeshare Segment**

- Marriott Vacation Club SM
- The Ritz-Carlton Club®
- Grand Residences by Marriott®
- · Horizons by Marriott Vacation Club®

The North American Full-Service segment and the North American Limited-Service segment include properties located in the continental United States and Canada. The Luxury segment includes worldwide properties. The International segment includes full-service and limited-service properties located outside the continental United States and Canada. Unless otherwise indicated, our references to Marriott Hotels & Resorts throughout this report include Marriott Conference Centers and JW Marriott Hotels & Resorts, references to Renaissance Hotels & Resorts include Renaissance ClubSport, and references to Fairfield Inn include Fairfield Inn & Suites.

In addition to the brands noted above, in 2007 we announced our new brand of family-friendly resorts and spas, "Nickelodeon Resorts by Marriott" and a new brand of lifestyle boutique hotels, "Edition." At year-end 2007, no properties were yet open under either brand.

## Company-Operated Lodging Properties

At year-end 2007, we operated 1,040 properties (266,751 rooms) under long-term management agreements with property owners, 31 properties (7,729 rooms) under long-term lease agreements with property owners (management and lease agreements together, "the Operating Agreements"), and six properties (1,316 rooms) as owned. The figures noted for properties operated under long-term management agreements include 21 residential products (1,916 units) for which we manage the related owners' associations.

Terms of our management agreements vary, but typically we earn a management fee, which comprises a base management fee, which is a percentage of the revenues of the hotel and an incentive management fee, which is based on the profits of the hotel. Our management agreements also typically include reimbursement of costs (both direct and indirect) of operations. Such agreements are generally for initial periods of 20 to 30 years, with options to renew for up to 50 or more additional years. Our lease agreements also vary, but may include fixed annual rentals plus additional rentals based on a percentage of annual revenues in excess of a fixed amount. Many of the Operating Agreements are subordinated to mortgages or other liens securing indebtedness of the owners. Additionally, many of our Operating Agreements permit the owners to terminate the agreement if certain performance metrics are not met and financial returns fail to meet defined levels for a period of time and we have not cured such deficiencies.

For lodging facilities that we operate, we generally are responsible for hiring, training, and supervising the managers and employees required to operate the facilities and for purchasing supplies, both for which we generally are reimbursed by the owners. We provide centralized reservation services and national advertising, marketing and promotional services, as well as various accounting and data processing services. We are generally reimbursed by owners for the cost of providing these services.

## Franchised Lodging Properties

We have franchising programs that permit the use of certain of our brand names and our lodging systems by other hotel owners and operators. Under these programs, we generally receive an initial application fee and continuing royalty fees, which typically range from 4 percent to 6 percent of room revenues for all brands, plus 2 percent to 3 percent of food and beverage revenues for certain full-service hotels. In addition, franchisees contribute to our national marketing and advertising programs and pay fees for use of our centralized reservation systems. At year-end 2007, we had 1,922 franchised properties (259,297 rooms).

## Timeshare and Residential

We develop, operate, market, and sell timeshare interval, fractional ownership, and residential properties under four brand names and generate revenues from three primary sources: (1) selling fee simple and other forms of timeshare intervals and personal residences; (2) financing consumer purchases; and (3) operating the resorts. Many resorts are located adjacent to company-operated hotels, such as Marriott Hotels & Resorts and The Ritz-Carlton, and owners have access to certain hotel facilities during their vacation. Owners can trade their annual interval for intervals at other Marriott timesharing resorts or for intervals at certain timesharing resorts not otherwise sponsored by Marriott through a third-party exchange company. Owners can also trade their unused interval for points in the Marriott Rewards® frequent stay program, enabling them to stay at over 2,900 company-operated or franchised properties worldwide.

We sell residential real estate in conjunction with luxury hotel development (Ritz-Carlton-Residential) and Timeshare segment projects (Ritz-Carlton Club-Residential and Grand Residences by Marriott-Residential). Our Timeshare segment residential projects are typically opened over time with limited inventory available at any one time. Residences developed in conjunction with hotels are typically constructed and sold by hotel owners with limited amounts, if any, of our capital at risk. While the worldwide residential market is very large, the luxurious nature of our residential properties, the quality and exclusivity associated with our brands, and the hospitality services that we provide, all serve to make our residential products distinctive.

## Seasonality

In general, business at company-operated and franchised properties is relatively stable and includes only moderate seasonal fluctuations. Business at some resort properties may be seasonal depending on location.

#### Relationship with Major Customer

We operate a number of properties under long-term management agreements that are owned or leased by Host Hotels & Resorts, Inc. ("Host"). In addition, Host is a partner in several partnerships that own properties operated by us under long-term management agreements. See Footnote No. 22, "Relationship with Major Customer," in the Notes to our Consolidated Financial Statements included in this annual report for more information.

Summary of Properties by Brand

At year-end 2007, we operated or franchised the following properties by brand (excluding 2,156 corporate housing rental units):

Brand	Company- Properties	Operated Rooms	Franc Properties	hised Rooms
U.S. Locations	rroperties	Roulis	rioperties	Kooms
Marriott Hotels & Resorts	146	73,937	167	50,647
Marriott Conference Centers	13	3,476	_	
JW Marriott Hotels & Resorts	11	6,736	5	1,552
Renaissance Hotels & Resorts	36	16,198	34	9,744
Renaissance ClubSport	_		1	175
The Ritz-Carlton	36	11,627	_	_
The Ritz-Carlton-Residential (1)	16	1,614	_	_
Courtyard	272	42,429	421	54,330
Fairfield Inn	2	855	527	46,075
SpringHill Suites	24	3,700	152	16,745
Residence Inn	138	18,693	390	44,112
TownePlace Suites	34	3,661	107	10,461
Marriott Vacation Club (2)	37	8,987	_	_
The Ritz-Carlton Club-Fractional (2)	5	283	_	_
The Ritz-Carlton Club-Residential (1), (2)	2	138	_	_
Grand Residences by Marriott-Fractional (2)	1	199	_	_
Grand Residences by Marriott-Residential (1), (2)	1	65	_	_
Horizons by Marriott Vacation Club (2)	2	444	_	_
Non-U.S. Locations				
Marriott Hotels & Resorts	122	34,016	34	9,936
JW Marriott Hotels & Resorts	21	8,183	1	61
Renaissance Hotels & Resorts	53	17,493	17	5,324
The Ritz-Carlton	34	9,978	_	_
The Ritz-Carlton-Residential (1)	1	93	_	_
Bulgari Hotels & Resorts	2	117	_	_
Marriott Executive Apartments	17	2,806	1	99
Courtyard	35	7,477	39	6,544
Fairfield Inn	_	_	8	947
SpringHill Suites			1	124
Residence Inn	1	190	17	2,421
Ramada International	2	332		
Marriott Vacation Club (2)	9	1,909	_	_
The Ritz-Carlton Club-Fractional (2)	2	105		
The Ritz-Carlton Club-Residential (1), (2)	1	6	_	_
Grand Residences by Marriott-Fractional (2)	1	49		
Total	1,077	275,796	1,922	259,297

Represents projects where we manage the related owners association. Residential products are included once they possess a certificate of occupancy. Indicates a Timeshare product. Includes products in active sales as well as those that are sold out.

<sup>(1)</sup> (2)

The following table provides additional detail by brand as of year-end 2007, for our Timeshare properties:

Active Sales <sup>(2)</sup>
24
4
3
2
1
4
38

Includes products that are in active sales as well as those that are sold out. Residential products are included once they possess a certificate of occupancy. Products in active sales may not be ready for occupancy.

# Summary of Properties by Country

At year-end 2007, we operated or franchised properties in the following 68 countries and territories:

	Country	Properties (1)	Rooms (1)
Ame	ericas		
	Argentina	1	320
	Aruba	4	1,637
	Bahamas	1	6
	Brazil	6	1,609
	Canada	56	11,838
	Cayman Islands	4	929
	Chile	2	485
	Costa Rica	3	620
	Curacao	1	247
	Dominican Republic	3	583
	Ecuador	1	257
	El Salvador	1	133
	Guatemala	1	385
	Honduras	1	153
	Jamaica	1	427
	Mexico	16	3,833
	Panama	2	415
	Peru	1	300
	Puerto Rico	5	1,585
	Saint Kitts and Nevis	2	559
	Trinidad and Tobago	1	119
	United States	2,580	426,883
	U.S. Virgin Islands	5	862
	Venezuela	1	269
	Total Americas	2,699	454,454

<sup>(2)</sup> 

Ameria   1	Middle East and Africa	Country	Properties (1)	Rooms (1)
Bahrain         1         2         5         5         5         5         5         5         5         5         5         5         5         5         6         4         8         3         5         6         4         8         6         6         6         6         6         6         7         9         9         2         6         1         1         2         2         1         2         2         1         3         9         1         3         1         2         2         1         1         2         2         1         1         2         2         1         1         2         2         1         1         2         2         1         1         3         3         1         1         3         3         1         1         3         3         1         1         3         3         1         1         3         3         1         1         3         3         1         1         3         3         1         2         2         1         1         3         3         1         2         4         4         3         3         3 </td <td></td> <td></td> <td>1</td> <td>226</td>			1	226
Egypt         8         3.5           Israel         2         3.5           Jordan         3         3         4           Kuwalt         2         60           Quar         3         14           Sould Ambin         3         14           Turkey         5         1.47           United Acab Emirates         5         1.75           China         3         2         1.6           Guam         1         3         2         1.6           Guam         1         3         1.1         3         1.1         3         1.1         3         1.1         3.2         1.4         1.2         3.1         1.2         3.1         3.2         1.4         1.2         3.1         3.2         1.4         1.2         3.1         3.2         1.4         1.2         3.2         1.4         1.2         3.2         1.4         1.2         3.2         1.4         1.2         3.2         1.4         1.2         3.2         1.2         3.2         1.2         3.2         1.2         1.2         3.2         1.2         3.2         1.2         3.2         3.2         3.2         3.2				264
İsrael         2         50           Jordan         3         64           Kuwait         2         60           Qatar         3         94           Saudi Arabia         1         12           Turisis         1         12           User Middle Least and Africa         5         1,47           Asia           Asia         1         3         12           Caram         1         1         3         14         1         3         14         1         3         14         1         3         14         1         3         1         1         3         1         1         1         3         1         1         3         1         1         3         1         1         3         1         1         1         3         1         1         3         1         1         1         3         1         1         3         1         1         1         3         1         1         1         3         1         1         1         3         1         1         1         1         1         1         1         1				3,562
Jordam				966
Kusweit         2         50           Qaar         3         91           Sandi Arabia         1         12           Tunkis         5         1.47           Uniced Ana Emirates         5         1.47           Uniced Ana Emirates         35         10.77           Asia           Asia         1         1         2           Guam         1         1         3         1         3         1         1         3         1         1         1         3         1         1         1         3         1         1         3         1         1         1         3         1         1         1         3         1				644
Qafar         3         91           Saudi Arabia         3         34           Tunisa         1         22           Turkey         5         1,47           United Arab Emirates         6         1,15           Total Middle East and Africa         3         12,66           Asia         1         35           China         3         12,46           Guam         1         15           Indio         6         1,52           Indio         6         1,52           Indionsia         8         1,90           Japan         10         3,15           Melaysia         2         2,00           Pakistan         2         2,00           Soingapore         2         10           Soingapore         2         4           Total Asia         8         2,35           Astralia         8         2,35           Cortinated Fuore         8         2,35           Cortinated Europe         9         2,33           Demanted Fuore         1         2,35           Cortinated Europe         1         2,32           Georgia </td <td></td> <td></td> <td></td> <td>604</td>				604
Sandi Arabia         3         74           Tunkis         1         22           Turkey         5         1,47           United Arab Emirates         5         1,57           China         35         10,77           Asia         8         1,26           China         1         35           China         1         35           Lough         1         35           India         6         1,52           India         1         3,5           Appan         10         3,5           Philippines         2         2           Singapore         2         1,0           South Korea         2         1,0           Notea         2         2,0           Vienam         2         2,0           Nustralis         8         2,3           Australis         7         1,68				910
Tunisis				749
Tarkey         5         1,47           United Arab Eminates         6         1,15           Stal         Total Middle East and Africa         35         10,77           Assian         3         1,26           Chlina         1         35         1,26           Glam         1         35         1,26				
United Areb Emirees         6         1,15           Total Middle East and Africa         35         10,75           Asia         China         32         12,46           China         1         35         10,42           Goam         1         13         15         11,22         10,22 <td></td> <td></td> <td></td> <td></td>				
Total Middle East and Africa         35         10,77           Asia				
Asia         12 (Ania)         3 (Ania) <t< td=""><td></td><td>ut on</td><td></td><td></td></t<>		ut on		
China         32         12.46           Guam         1         35           India         6         1,52           Indonesia         8         1,80           Japan         10         3,15           Malaysia         7         3,01           Palistam         2         9           Philippines         2         10           Singapore         2         10           South Korea         4         1,50           Thailand         2         28           Wastrali         8         2,35           Continental Europe         8         2,35           Continental Europe         7         1,00           Austria         7         1,00           Belgium         5         87           Czech Republic         5         80           Czech Republic         5         93           Demark         1         3           France         10         2,55           Georgia         2         24           Gerece         1         3           Greece         1         3           Greece         1         3		rica	35	10,//2
Guam       1       35         India       6       1,52         Indonesia       10       3,15         Ajapan       7       3,01         Pakistan       2       50         Philippines       2       80         Singapore       2       1,00         South Korea       4       1,50         Tailand       9       2,33         Vietnam       8       2,33         Australk       7       1,60         Belgium       5       87         Czech Republic       5       87         Georgia       1       33         France       10       2,55         Georgia       2       4         Gerece       1       3       3         Gerece       1       1,1       1,1         Hungary       2       3       7         Netherlands       3       3       2         Polond       2       3       3         Polond       2       3       3         Polond       3       3       3         Polond       3       3       3         Polo	Asia			
India         6         1,52           Indonesia         8         1,90           Japan         7         3,01           Pakistan         2         50           Philippines         2         1,00           Singapore         2         1,00           South Korea         4         1,50           Thalland         9         2,33           Vietnam         8         2,35           Cottlean         8         2,35           Cottlean         7         1,68           Belgium         7         1,68           Cech Republic         5         3           Cech Republic         5         3           Cengia         1         1,9           France         10         2,55           Georgia         2         2           Greece         1         31           Hungary         2         3           Italy         3         3           Netherlands         2         2           Poland         2         74           Hortugal         3         3           Russia         6         2,10				
Inlonesia       8       1,90         Japan       10       3,15         Malaysia       2       50         Philippines       2       100         Singapore       2       1,00         South Korea       4       1,50         Thailand       9       2,33         Vietnam       2       6         Australia       7       1,68         Belgium       7       1,68         Belgium       5       87         Czech Republic       5       33         Denmark       1       19         France       1       13         Gerogia       2       2         Gerece       1       31         Hungary       2       3         Italy       3       3         Netherlands       2       3         Netherlands       2       3         Poland       2       7         Poland       2       7         Poland       2       7         Portugal       3       3         Romania       1       1         Romania       3       3      <				
Japan       10       3,15         Malaysia       7       3,01         Pakistan       2       80         Philippines       2       180         Singapore       4       1,50         South Korea       4       1,50         Thailand       9       2,33         Vietnam       85       29,43         Australi       7       1,68         Belgium       5       93         Coctinental Europe       5       87         Ceck Republic       5       87         Ceck Republic       1       3         Denmark       1       33         France       10       2,55         Georgia       1       3         Gerece       1       3         Hungary       8       1,7         Italy       8       1,7         Italy       8       1,7         Netherlands       2       2         Poland       2       7         Poland       2       7         Portugal       3       8         Russia       8       1,1         Portugal       3				
Malaysia       7       3,01         Pakistan       2       50         Singapore       2       1,00         South Korea       4       1,10         Thailland       9       2,33         Vietnam       2       87         Total Asia       7       1,68         Austrial       7       1,68         Belgium       7       1,88         Belgium       5       87         Czech Republic       5       93         Dennark       1       2,3         France       10       2,5         Georgia       2       24         Greece       1       13         Hungary       2       14         Italy       8       1,7         Kazakhstan       2       3         Netherlands       3       3         Poland       2       3         Portugal       3       3         Russia       1       1         Spain       9       2,3         Spain       9       2,3         Spain       9       2,3         Spain       9       2,3				
Palistam         2         50           Philippines         2         89           Singapore         2         1,00           South Korea         4         1,50           Thailand         9         2,33           Vietnam         2         87           Total Asia         8         2,35           Currental Eurore         8         2,35           Cothienetal Eurore         5         87           Austria         7         1,68           Belgium         5         87           Czech Republic         5         93           Czech Republic         5         93           Georgia         1         1,39           Georgia         2         24           Gereacy         1         3           Hungary         2         47           Italy         8         1,11           Kazakhstan         2         3           Poland         2         74           Hugary         3         3           Poland         2         72           Polang         3         3           Rossia         1         1 </td <td></td> <td></td> <td></td> <td>3,155</td>				3,155
Philippines         2         89           Singapore         2         1,00           South Korea         4         1,50           Thailand         9         2,33           Vietnam         2         88           Australia         8         2,35           Corbinental Europe         7         1,68           Belgium         5         87           Czech Republic         5         93           Denmark         1         39           Frace         10         2,55           Georgia         2         24           Germany         36         8,24           Greece         1         31           Hungary         2         14           Italy         8         1,7           Italy         8         1,7           Italy         8         1,7           Rosece         1         3         92           Poland         2         2           Poland         2         2           Potugal         3         3           Russia         8         2,10           Spain         9         2,3 <td></td> <td></td> <td>7</td> <td>3,019</td>			7	3,019
Singapore       2       1,00         South Korea       4       1,50         Thailand       9       2,33         Vetnam       2       87         Total Asia       8       2,35         Continental Europe       7       1,68         Belgium       7       1,68         Belgium       5       87         Czech Republic       5       93         Denmark       1       39         France       10       2,55         Georgia       2       24         Gereany       3       3,24         Greece       1       31         Hungary       2       47         Italy       8       1,7         Kazakhstan       2       3         Poland       2       74         Potugal       3       3         Romania       1       4         Rusia       3       9         Spain       9       2,35         Sylicerland       3       6         Spain       9       2,35         Sylicerland       3       6         Spain       9       2,35			2	504
Suh Korea       4       1,50         Thailand       9       2,33         Vetnam       85       29,43         Australia       8       2,35         Continetal Europe	Philippines		2	899
Thailand       9       2,33         Vienam       2       87         Total Asia       8       2,33         Astrala       8       2,33         Continental Europe         Austria       7       1,68         Belgium       5       87         Czech Republic       5       93         Czech Republic       1       39         Pance       10       2,55         Georgia       1       3       9         Georgia       1       3       9       2,24         Germany       36       8,24       3       9       2,24         Geregia       1       3       1,71       4       1,71       4       1,71       4       1,71       4       1,71       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4       4       1,71       4			2	1,002
Vienam         2         87           Total Asia         85         29,43           Austrial         8         23,55           Continetal Europe         9         1,68           Belgium         5         87           Czech Republic         5         89           Denmark         1         39           France         10         2,55           Georgia         2         24           Germany         3         8,24           Greece         1         31           Hungary         2         47           Islay         8         1,71           Kazakhstan         2         2,72           Poland         2         7,42           Kazakhstan         3         92           Poland         2         7,4           Kazakhstan         3         92           Poland         2         7,4           Kazakhstan         3         93           Romania         3         93           Romania         4         4           Russia         8         1,1           Spain         9         2,35 <t< td=""><td></td><td></td><td>4</td><td>1,501</td></t<>			4	1,501
Vietnam       2       87         Total Asia       85       29,33         Austria       8       2,35         Continetal Europe         Austria       7       1,68         Belgium       5       87         Czech Republic       5       83         Denmark       1       39         France       10       2,55         Georgia       2       24         Germany       3       82         Gerece       1       3       13         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       3       3         Netherlands       3       92         Poland       2       74         Portugal       3       92         Romania       1       4         Russia       8       1,1         Spain       9       2,35         Switzeland       3       6         Spain       9       2,35         Switzeland       1       4         Spain       9       2,35         Switzeland       1 <t< td=""><td>Thailand</td><td></td><td>9</td><td>2,330</td></t<>	Thailand		9	2,330
Total Asia         85         29,43           Australia         8         2,35           Continental Europe         1         3           Austria         7         1,68           Belgium         5         87           Czech Republic         5         87           Czech Republic         5         93           Denmark         1         39           France         10         2,55           Georgia         2         24           Germany         36         8,24           Greece         1         31         34           Hungary         2         47           I kazakhstan         2         3         3         92           Poland         2         3         9           Potrugal         3         92           Potrugal         3         92           Spain         3         2,25           Switzerland         3         62           Spain         3         62           Spain         3         62           Spain         3         62           Switzerland         3         62	Vietnam		2	874
Australia       8       2,35         Continental Europe       3       1,68         Austria       7       1,68         Belgium       5       87         Czech Republic       5       93         Denmark       1       39         France       10       2,55         Georgia       2       2       2         Germany       36       8,24         Greece       1       31       31       31       31       31       32       47       42       47       43       42       47       43       43       42	Total Asia			29,439
Austria       7       1,68         Belgium       5       87         Belgium       5       87         Czech Republic       1       39         Denmark       1       39         France       10       2,55         Georgia       2       24         Germany       3       8,24         Greece       1       31         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       14         Russia       8       2,10         Spain       9       2,35         Switzerland       8       2,10         Spair       9       2,35         Switzerland       3       62         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       1,11         Total United Kingdom and Ireland       57       1,11	Australia		8	2,354
Austria       7       1,68         Belgium       5       87         Belgium       5       87         Czech Republic       1       39         Denmark       1       39         France       10       2,55         Georgia       2       24         Germany       3       8,24         Greece       1       31         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       14         Russia       8       2,10         Spain       9       2,35         Switzerland       8       2,10         Spair       9       2,35         Switzerland       3       62         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       1,11         Total United Kingdom and Ireland       57       1,11	Continental Europe			
Belgium       5       87         Czech Republic       5       93         Denmark       1       39         France       10       2,55         Georgia       2       24         Germany       36       8,24         Greece       1       31         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       8       2,10         Spain       9       2,35         Switzerland       9       2,35         Witted Kingdom al Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       1,11         United Kingdom (England, Scotland, and Ireland       57       1,11         Total United Kingdom and Ireland       64       1,21			7	1,686
Czech Republic       5       93         Denmark       1       39         France       10       2,55         Georgia       2       24         Germany       36       8,24         Greece       1       31         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       9       2,35         Switzerland       9       2,35         Total Europe       10       5         United Kingdom ard Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       1,11         Total United Kingdom and Ireland       57       1,11         Total United Kingdom and Ireland       64       1,223				878
Denmark       1       39         France       10       2,55         Georgia       2       24         Germany       36       8,24         Greece       1       31         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       3       93         Rownatia       1       40         Russia       8       2,10         Spain       3       62         Switzerland       8       2,10         Spain       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       1,11         Total United Kingdom (England, Scotland, and Wales)       64       1,223				
France       10       2,55         Georgia       2       24         Germany       36       8,24         Greece       1       31         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       36         Total Europe       3       36         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       1,11         Total United Kingdom and Ireland       64       12,23				
Georgia       2       24         Germany       36       8,24         Greece       1       31         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Germany       36       8,24         Greece       1       31         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Greece       1       31         Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Hungary       2       47         Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       1,12         Total United Kingdom and Ireland       64       12,23				
Italy       8       1,71         Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Kazakhstan       2       32         Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Netherlands       3       92         Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland         Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Poland       2       74         Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Portugal       3       93         Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Romania       1       40         Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Russia       8       2,10         Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Spain       9       2,35         Switzerland       3       62         Total Europe       108       25,84         United Kingdom and Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Switzerland         3         62           Total Europe         108         25,84           United Kingdom and Ireland         7         1,11           United Kingdom (England, Scotland, and Wales)         57         11,12           Total United Kingdom and Ireland         64         12,23				
Total Europe         108         25,84           United Kingdom and Ireland         7         1,11           Ireland         7         1,11           United Kingdom (England, Scotland, and Wales)         57         11,12           Total United Kingdom and Ireland         64         12,23				
United Kingdom and Ireland         Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23				
Ireland       7       1,11         United Kingdom (England, Scotland, and Wales)       57       11,12         Total United Kingdom and Ireland       64       12,23			108	25,840
United Kingdom (England, Scotland, and Wales)5711,12Total United Kingdom and Ireland6412,23	United Kingdom and Ireland			
Total United Kingdom and Ireland 64 12,23		1 15.71		
Total-All Countries and Territories 2,999 535,09		d Ireland		12,234
	Total-All Countries and Territories		2,999	535,093

Includes Timeshare products that are in active sales as well as those that are sold out. Products in active sales may not be ready for occupancy.

## **Descriptions of our Brands**

# North American Full-Service Segment, North American Limited-Service Segment, and International Segment Lodging Products

*Marriott Hotels & Resorts* is our global flagship brand, primarily serving business and leisure upper-upscale travelers and meeting groups. Marriott properties are located in downtown, urban, and suburban areas, near airports, and at resort locations. Marriott inspires your best performance during travel, engineering an experience with the services and human touches that help you achieve and revive.

Typically, properties contain from 300 to 700 well-appointed rooms, the Revive® bedding package, in-room high-speed Internet access, swimming pools, convention and banquet facilities, destination-driven restaurants and lounges, room service, concierge lounges, wireless Internet access in public places, and parking facilities. Sixteen properties have over 1,000 rooms. Many resort properties have additional recreational and entertainment facilities, such as tennis courts, golf courses, additional restaurants and lounges, and spa facilities. New and renovated properties typically reflect the new Marriott guest room which features contemporary residential design, warm colors, rich woods and architectural detail, flat-screen high-definition televisions, "plug and play" technology, and bathrooms reflecting spa-like luxury. At year-end 2007, there were 469 Marriott Hotels & Resorts properties (168,536 rooms), not including JW Marriott Hotels & Resorts or Marriott Conference Centers.

JW Marriott Hotels & Resorts is the Marriott brand's collection of luxurious properties and resorts that cater to accomplished, discerning travelers seeking an elegant environment and personal service. At year-end 2007 there were 38 properties (16,532 rooms) primarily located in gateway cities and upscale locations throughout the world. JW Marriott Resorts offer attentive service and exceptional amenities, many with world-class golf and spa facilities. In addition to the features found in a typical Marriott full-service property, the facilities and amenities at JW Marriott Hotels & Resorts properties normally include larger guest rooms, higher end décor and furnishings, upgraded in-room amenities, upgraded business centers and fitness centers, and 24-hour room service.

At year-end 2007, there were 13 *Marriott Conference Centers* (3,476 rooms) throughout the United States. Some of the centers are used exclusively by employees of sponsoring organizations, while others are marketed to outside meeting groups and individuals. In addition to the features found in a typical Marriott full-service property, the centers typically include expanded meeting room space, banquet and dining facilities, and recreational facilities.

# Marriott Hotels & Resorts, JW Marriott Hotels & Resorts, and Marriott

Geographic Distribution at Year-end 2007	Properties	
United States (42 states and the District of Columbia)	342	(136,348 rooms)
Non-U.S. (49 countries and territories)		
Americas	39	
Continental Europe	37	
United Kingdom and Ireland	50	
Asia	29	
Middle East and Africa	18	
Australia	5	
Total Non-U.S.	178	(52,196 rooms)

**Renaissance Hotels & Resorts** is a distinctive and global quality-tier full-service brand that targets individual business and leisure travelers and group meetings seeking to discover stylish and personalized environments. Renaissance helps keep life on the road interesting with "Savvy Services," and touches like destination restaurants and stylish designs that help guests enjoy access to new, ever-changing experiences on and off property.

Renaissance Hotels & Resorts properties are generally located at downtown locations in major cities, in suburban office parks, near major gateway airports, and in destination resorts. Most properties contain from 300 to 500 rooms. Renaissance properties and services typically feature distinctive décor, in-room high-speed Internet access, restaurants and lounges, room service, swimming pools, gift shops, concierge lounges, and meeting and banquet facilities. New and renovated properties typically reflect the new Renaissance guest room that blends fashion and provocative design with local touches. Rooms also feature sophisticated lighting, architectural detail, flat-screen high-definition televisions, and "plug and play" technology. Resort properties typically have additional recreational and entertainment facilities and services, including golf courses, tennis courts, water sports, additional restaurants, and spa facilities. At year-end 2007, there were 141 Renaissance Hotels & Resorts properties (48,934 rooms), including one Renaissance ClubSport property (175 rooms).

Renaissance Hotels & Resorts Geographic Distribution at Year-end 2007	Properties	
United States (26 states and the District of Columbia)	71	(26,117 rooms)
Non-U.S. (30 countries and territories)		
Americas	8	
Continental Europe	30	
United Kingdom	5	
Asia	22	
Middle East and Africa	5	
Total Non-U.S.	70	(22,817 rooms)

Courtyard is our upper-moderate price select-service hotel product aimed primarily at transient business travel. Courtyard hotels maintain a residential atmosphere and typically contain 90 to 150 rooms in suburban locales and 140 to 340 rooms in downtown domestic and international locales. Well-landscaped grounds typically include a courtyard with a pool and social areas. Hotels feature functionally designed quality guest rooms and meeting rooms, free in-room and lobby high-speed Internet access (in North America), limited restaurant facilities, a swimming pool, an exercise room, and *The Market* (a self-serve food store open 24 hours a day). The operating systems developed for these hotels allow Courtyard to be price-competitive while providing better value through superior facilities and guest service. At year-end 2007, there were 767 Courtyards (110,780 rooms) operating in 29 countries and territories.

Courtyard Geographic Distribution at Year-end 2007	Properties	
United States (47 states and the District of Columbia)	693	(96,759 rooms)
Non-U.S. (28 countries and territories)	<u></u>	
Americas	30	
Continental Europe	31	
Asia	7	
Middle East and Africa	3	
Australia	3	
Total Non-U.S.	74	(14,021 rooms)

*Fairfield Inn* is our hotel brand that competes in the lower-moderate price tier and is primarily aimed at value-conscious individual business travelers. Fairfield Inn (which includes Fairfield Inn & Suites) offers free in-room and lobby high-speed Internet access, a swimming pool, complimentary "Early Eats<sup>TM</sup>" continental breakfast and free local phone calls. A typical Fairfield Inn or Fairfield Inn & Suites property has 60 to 140 rooms in suburban locations and up to 200 rooms in urban destinations. At year-end 2007, there were 334 Fairfield Inn properties and 203 Fairfield Inn & Suites properties (537 hotels total), operating in the United States, Canada, and Mexico.

Fairfield Inn and Fairfield Inn & Suites		
Geographic Distribution at Year-end 2007	Properties	
United States (47 states)	529	(46,930 rooms)
Non-U.S. Americas (Canada and Mexico)	8	(947 rooms)

*SpringHill Suites* is our all-suite brand in the upper-moderate-price tier primarily targeting business travelers. SpringHill Suites properties typically have 90 to 165 studio suites that are more spacious than a typical hotel guest room. The brand offers a broad range of amenities, including free in-room and lobby high-speed Internet access, *The Market* (a self-serve food store open 24 hours a day), complimentary "Suite Seasons®" hot breakfast buffet, exercise facilities, and a swimming pool. There were 177 properties (20,569 rooms) located in the United States and Canada at year-end 2007.

**Residence Inn** is North America's leading extended-stay brand. For today's marathon business traveler, Residence Inn allows guests on long-term trips to experience all the comforts of home while traveling so that they can 'Thrive on Long Stays' in a particular location. Residence Inn provides connectivity to home and office, exercise options and comfortable places to work or relax. Residence Inn also provides upscale design and style, spacious suites with full kitchens, separate sleeping areas and "real" food at social events or in suite. At year-end 2007, there were 546 Residence Inn properties (65,416 rooms) located in the United States, Canada, and Mexico.

lence	

Geographic Distribution at Year-end 2007	Properties	
United States (47 states and the District of Columbia)	528	(62,805 rooms)
Non-U.S. Americas (Canada and Mexico)	18	(2,611 rooms)

TownePlace Suites is a moderately priced extended-stay hotel product that is designed to appeal to business and leisure travelers who stay for five nights or more. Designed for the self-sufficient, value-conscious traveler, each suite generally provides functional spaces for living and working, including a full kitchen and a home office. TownePlace Suites associates are trained to provide insightful local knowledge, and each hotel specializes in delivering service that helps guests settle in to the local area. Additional amenities include housekeeping services, on-site exercise facilities, an outdoor pool, 24-hour staffing, free in-room high-speed Internet access, and laundry facilities. At year-end 2007, 141 TownePlace Suites properties (14,122 rooms) were located in 37 states.

*Marriott ExecuStay* provides furnished corporate apartments primarily for long-term stays nationwide. ExecuStay owns no residential real estate and provides units primarily through short-term lease agreements with apartment owners and managers and franchise agreements. At year-end 2007, Marriott leased approximately 2,100 apartments and our 11 franchisees leased over 2,300 apartments. Apartments are located in 44 different markets in the United States, of which 34 are franchised.

*Marriott Executive Apartments*. We provide temporary housing ("Serviced Apartments") for business executives and others who need quality accommodations outside their home country, usually for 30 or more days. Some Serviced Apartments operate under the *Marriott Executive Apartments* brand, which is designed specifically for the long-term international traveler. At year-end 2007, 15 Marriott Executive Apartments and three other Serviced Apartments properties (2,905 rooms total) were located in 12 countries and territories. All Marriott Executive Apartments are located outside the United States.

## **Luxury Segment Lodging Products**

The Ritz-Carlton is a leading global luxury lifestyle brand of hotels and resorts renowned for their distinctive architecture and for the high quality level of their facilities, dining options, and exceptional personalized guest service. Most of The Ritz-Carlton hotels have 250 to 400 guest rooms and typically include meeting and banquet facilities, a variety of restaurants and lounges, a club level, gift shops, high-speed Internet access, flat-screen high-definition televisions, swimming pools, and parking facilities. Guests at most of The Ritz-Carlton resorts have access to additional recreational amenities, such as tennis courts, golf courses, and health spas.

#### The Ritz-Carlton

THE THE CHIRCH		
Geographic Distribution at Year-end 2007 <sup>(1)</sup>	Properties	
United States (16 states and the District of Columbia)	52	(13,241 rooms)
Non-U.S. (22 countries and territories)		
Americas	7	
Continental Europe	8	
Asia	14	
Middle East and Africa	6	
Total Non-U.S.	35	(10,071 rooms)

Includes 17 home and condominium projects (1,707 units) for which we manage the related owners' associations.

**Bulgari Hotels & Resorts.** Through a joint venture with jeweler and luxury goods designer Bulgari SpA we operate distinctive luxury hotel properties in prime locations under the name Bulgari Hotels & Resorts. The first property (58 rooms), the Bulgari Hotel Milano, opened in Milan, Italy, in 2004. The second property, the Bulgari Resort Bali, opened in late 2006 and includes 59 private villas, two restaurants, and comprehensive spa facilities. In November 2007, we opened two new restaurants in Tokyo, Japan, which we operate in connection with two new Bulgari retail stores. Other projects are currently in various stages of development in Europe, Asia, and North America.

## **Timeshare Segment Lodging Products**

The *Marriott Vacation Club* ("MVC") brand offers full-service villas featuring living and dining areas, one-, two-, and three-bedroom options, a full kitchen, and washer/dryer units. Customers may purchase a one-week interval or more at each resort. In 46 locations worldwide, this brand draws United States and international customers who vacation regularly with a focus on family, relaxation, and recreational activities. In the United States, in addition to other areas, MVC is located in Las Vegas, Nevada, in beach and/or golf communities in

Arizona, California, South Carolina, Florida, and Hawaii and in ski resorts in California, Colorado, and Utah. Internationally, MVC has resorts in Aruba, France, Spain, St. Thomas (U.S. Virgin Islands), the West Indies, and Thailand.

The Ritz-Carlton Club brand is a luxury-tier real estate fractional ownership and personal residence brand that combines the benefits of second-home ownership with personalized services and amenities. This brand is designed as a private club whose members have access to all Ritz-Carlton Clubs, and is offered in ski, golf, and beach destinations in the Bahamas, California, Colorado, St. Thomas (U.S. Virgin Islands), Florida, and Hawaii. Customers typically purchase three-to-five-week intervals, but may also purchase a residence outright.

*Grand Residences by Marriott* is an upper-quality-tier fractional ownership and personal residence brand for corporate and leisure customers. This brand is currently offering ownership in projects located in Lake Tahoe, California, Panama City, Florida, and London, England. Customers typically purchase three-to-13-week intervals.

Horizons by Marriott Vacation Club is Marriott Vacation Club's moderately priced timeshare brand whose product offerings and customer base are currently focused on offering family vacations in entertainment communities. Horizons resorts are located in Orlando, Florida and Branson, Missouri. Customers may purchase a one-week interval or more at each resort.

We expect that our long-term timeshare growth will reflect opportunities presented by various third-party business structures, as well as a continued focus on our core and predominantly wholly owned Marriott Vacation Club brand. We also anticipate that residential and fractional products, especially of the luxury The Ritz-Carlton Club brand, will offer the fastest growth opportunity over the short-term within the Timeshare segment.

The Timeshare segment's owner base continues to expand, with approximately 371,000 owners at year-end 2007, compared to approximately 343,000 at year-end 2006.

Timeshare (all brands)		
Geographic Distribution at Year-end 2007	Resorts	Units
Continental United States	43	Units 8,572
Hawaii	5	1,544
Caribbean	7	993
Europe	5	932
Asia	1	144
Total	61	12,185

## **New Lodging Products**

*Nickelodeon Resorts by Marriott.* In May 2007, we announced our new brand of family-friendly resorts and spas, "Nickelodeon Resorts by Marriott." These upscale, self-contained resorts are expected to feature Nickelodeon signature activities, state-of-the-art pools, water parks, entertainment starring Nickelodeon's characters such as Dora the Explorer and SpongeBob SquarePants, and spas for adults and kids. Projects are in various stages of development in the United States, and we expect to open the first property in San Diego in 2010. Other potential resort locations being considered for development include major family-oriented destinations in the United States, Caribbean and Mexican resort areas, the United Kingdom and Europe, Asia, Australia, and the Middle East.

*Edition.* In June 2007, we announced that we had joined forces with hotel innovator Ian Schrager to create next-generation lifestyle boutique hotels. Edition combines a personal, intimate, individualized, and unique lodging experience on a global scale. These new lifestyle boutique hotels attempt to push the boundaries, break new ground, and take the hotel industry to a new level. Plans include building and operating up to 100 distinct hotels around the world.

## Other Activities

Marriott Golf manages 44 golf course facilities as part of our management of hotels and for other golf course owners.

We operate 14 systemwide hotel reservation centers, eight in the United States and Canada and six in other countries and territories, which handle reservation requests for our lodging brands worldwide, including franchised properties. We own one of the U.S. facilities and lease the others.

Additionally, we focus on increasing value for the consumer and "selling the way the customer wants to buy." Our Look No Further® Best Rate Guarantee, which assures customers that they won't find cheaper rates than those available through our telephone reservation system, our Web site or any other Marriott reservation channel, our

strong Marriott Rewards loyalty program, and our information-rich and easy-to-use <a href="https://www.Marriott.com">www.Marriott.com</a> Web site all encourage customers to make reservations using the Marriott Web site. Our reservation system manages and controls inventory availability and pricing set by our hotels, and allows us to utilize online and offline agents where cost effective. With almost 3,000 hotels economies of scale enable us to minimize costs per occupied room, drive profits for our owners, and maximize our fee revenue.

Our Architecture and Construction ("A&C") division provides design, development, construction, refurbishment, and procurement services to owners and franchisees of lodging properties on a voluntary basis outside the scope of and separate from our management or franchise contracts. Similar to third-party contractors, A&C provides these services for owners and franchisees of Marriott-branded properties on a fee basis.

#### Competition

We encounter strong competition both as a lodging operator and as a franchisor. We believe that by operating a number of hotels among our brands, we stay in direct touch with customers and react to changes in the marketplace more quickly than chains that rely exclusively on franchising. There are approximately 770 lodging management companies in the United States, including several that operate more than 100 properties. These operators are primarily private management firms, but also include several large national chains that own and operate their own hotels and also franchise their brands. Our management contracts are typically long-term in nature, but most allow the hotel owner to replace the management firm if certain financial or performance criteria are not met.

Affiliation with a national or regional brand is prevalent in the U.S. lodging industry. In 2007, approximately two-thirds of U.S. hotel rooms were brand-affiliated. Most of the branded properties are franchises, under which the operator pays the franchisor a fee for use of its hotel name and reservation system. The franchising business is fairly concentrated, with the six largest franchisors operating multiple brands accounting for a significant proportion of all U.S. rooms.

Outside the United States, branding is much less prevalent and most markets are served primarily by independent operators, although branding is more common for new hotel development. We believe that chain affiliation will increase in overseas markets as local economies grow, trade barriers are reduced, international travel accelerates and hotel owners seek the economies of centralized reservation systems and marketing programs.

Based on lodging industry data, we have a 9 percent share of the U.S. hotel market (based on number of rooms) and we estimate less than a 1 percent share of the lodging market outside the United States. We believe that our hotel brands are attractive to hotel owners seeking a management company or franchise affiliation because our hotels typically generate higher occupancies and Revenue per Available Room ("RevPAR") than direct competitors in most market areas. We attribute this performance premium to our success in achieving and maintaining strong customer preference. We believe that the location and quality of our lodging facilities, our marketing programs, our reservation systems and our emphasis on guest service and satisfaction are contributing factors across all of our brands.

Properties that we operate or franchise are regularly upgraded to maintain their competitiveness. Most of our management agreements provide for the allocation of funds, generally a fixed percentage of revenue, for periodic renovation of buildings and replacement of furnishings. These ongoing refurbishment programs, along with periodic brand initiatives, are generally adequate to preserve or enhance the competitive position and earning power of the hotels and timeshare properties.

While service excellence is Marriott's hallmark, we continually look for new ways to delight our guests. Currently, we are focused on elevating the Marriott experience beyond that of a traditional hotel stay to a total guest experience that encompasses exceptional style, personal luxury, and superior service. This approach to hospitality, "The New Look and Feel of Marriott Now," is influenced by the world's foremost innovations in design, technology, culinary expertise, service, and comfort. This evolution can be seen across all of our brands, in new and stylish hotel designs, luxurious bedding, exotic destinations, world-class spas and fitness centers, and inspired cuisine. Each brand, from luxury to moderately priced, will be more upscale and attuned to customer needs than ever before.

Our At Your Service® program focuses on the total guest experience from point of reservation to checkout. As part of the pre-arrival planning service, guests receive a personalized email prior to check-in that includes local transportation, weather, and restaurant information, as well as directions and maps. At a growing number of hotels and resorts, service has been expanded to include a "virtual concierge." Guests are able to reserve spa treatments, room service for delivery upon arrival, and other amenities specific to each property. Guests may also request complimentary amenities such as extra pillows, miniature refrigerators, and early check-in or late checkout. We have also streamlined the travel planning and booking process with the addition of Marriott Flexrez to At Your Service. In addition to booking hotel rooms on <a href="https://www.Marriott.com">www.Marriott.com</a>, our online customers can now find competitive airfare and car rental rates there as well.

The vacation ownership industry is one of the fastest growing segments in hospitality and is comprised of a number of highly competitive companies including several branded hotel companies. Since entering the timeshare industry over 20 years ago, we have become a recognized leader in vacation ownership worldwide. Competition in the timeshare interval, fractional, and residential business is based primarily on the quality and location of timeshare resorts, trust in the brand, the pricing of product offerings, and the availability of program benefits, such as exchange programs. We believe that our focus on offering distinct vacation experiences, combined with our financial strength, diverse market presence, strong brands, and well-maintained properties, will enable us to remain competitive. Approximately 40 percent of our timeshare ownership resort sales come from additional purchases by or referrals from existing owners.

Marriott Rewards is a frequent guest program with over 28 million members and nine participating Marriott brands. The Marriott Rewards program yields repeat guest business by rewarding frequent stays with points toward free hotel stays and other rewards, or airline miles with any of 29 participating airline programs. We believe that Marriott Rewards generates substantial repeat business that might otherwise go to competing hotels. In 2007, over 50 percent of our room nights were purchased by Marriott Rewards members. In addition, the ability of timeshare owners to convert unused intervals into Marriott Rewards points enhances the competitive position of our timeshare brand.

## **Discontinued Operations**

Synthetic Fuel

Our synthetic fuel operation consisted of four coal-based synthetic fuel production facilities (the "Facilities"). Because tax credits under Section 45K of the IRC are not available for the production and sale of synthetic fuel produced from coal after calendar year-end 2007, and because high oil prices during 2007 will result in the phase-out of a significant portion of the tax credits available for synthetic fuel produced and sold in 2007, on November 3, 2007, we shut down the Facilities and permanently ceased production of synthetic fuel. Accordingly, we now report this business segment as a discontinued operation. The book value of the Facilities was zero at year-end 2007, as the Facilities have been transferred to third parties. Under the site leases for the Facilities, we were required to restore the leased premises to substantially the condition the premises were in when the leases were originally executed. However, we executed agreements with the lessors of the sites pursuant to which we transferred the Facilities to the lessors in exchange for the release of our obligations to restore the leased premises to their original conditions. Costs associated with shutting down the synthetic fuel operation and transferring the Facilities to the site lessors were not material.

## **Employee Relations**

At year-end 2007, we had approximately 151,000 employees, approximately 8,700 of which were represented by labor unions. We believe relations with our employees are positive.

#### **Environmental Compliance**

Our compliance with laws and regulations relating to environmental protection and discharge of hazardous materials has not had a material impact on our capital expenditures, earnings or competitive position, and we do not anticipate any material impact from such compliance in the future.

## **Internet Address and Company SEC Filings**

Our Internet address is <a href="www.Marriott.com">www.Marriott.com</a>. On the investor relations portion of our Web site, <a href="www.Marriott.com/investor">www.Marriott.com/investor</a>, we provide a link to our electronic SEC filings, including our annual report on Form 10-K, our quarterly reports on Form 10-Q, our current reports on Form 8-K and any amendments to these reports. All such filings are available free of charge and are available as soon as reasonably practicable after filing. The information found on our Web site is not part of this or any other report we file with or furnish to the United States Securities and Exchange Commission (the "SEC").

#### Item 1A. Risk Factors

## Forward-Looking Statements

We make forward-looking statements in Management's Discussion and Analysis of Financial Condition and Results of Operations and elsewhere in this report based on the beliefs and assumptions of our management and on information currently available to us. Forward-looking statements include information about our possible or assumed future results of operations which follow under the headings "Business and Overview," "Liquidity and Capital Resources," and other statements throughout this report preceded by, followed by or that include the words "believes," "expects," "anticipates," "intends," "plans," "estimates" or similar expressions.

Forward-looking statements are subject to a number of risks and uncertainties that could cause actual results to differ materially from those expressed in these forward-looking statements, including the risks and uncertainties described below and other factors we describe from time to time in our periodic filings with the SEC. We therefore caution you not to rely unduly on any forward-looking statements. The forward-looking statements in this report speak only as of the date of this report, and we undertake no obligation to update or revise any forward-looking statement, whether as a result of new information, future developments or otherwise.

## **Risks and Uncertainties**

We are subject to various risks that could have a negative effect on the Company and its financial condition. You should understand that these risks could cause results to differ materially from those expressed in forward-looking statements contained in this report and in other Company communications. Because there is no way to determine in advance whether, or to what extent, any present uncertainty will ultimately impact our business, you should give equal weight to each of the following:

## **Lodging Industry Risks**

The lodging industry is highly competitive, which may impact our ability to compete successfully with other hotel and timeshare properties for customers. We generally operate in markets that contain numerous competitors. Each of our hotel and timeshare brands competes with major hotel chains in national and international venues and with independent companies in regional markets. Our ability to remain competitive and to attract and retain business and leisure travelers depends on our success in distinguishing the quality, value, and efficiency of our lodging products and services from those offered by others. If we are unable to compete successfully in these areas, this could limit our operating margins, diminish our market share, and reduce our earnings.

We are subject to the range of operating risks common to the hotel, timeshare, and corporate apartment industries. The profitability of the hotels, vacation timeshare resorts, and corporate apartments that we operate or franchise may be adversely affected by a number of factors, including:

- (1) the availability of and demand for hotel rooms, timeshare interval, fractional ownership and residential products, and apartments;
- (2) international, national, and regional economic and geopolitical conditions;
- (3) the impact of war, actual or threatened terrorist activity and heightened travel security measures instituted in response to war, terrorist activity or threats;
- (4) the desirability of particular locations and changes in travel patterns;
- (5) travelers' fears of exposure to contagious diseases, such as Avian Flu and Severe Acute Respiratory Syndrome ("SARS");
- (6) the occurrence of natural disasters, such as earthquakes, tsunamis, and hurricanes;
- (7) taxes and government regulations that influence or determine wages, prices, interest rates, construction procedures, and costs;
- (8) the availability and cost of capital to allow us and potential hotel owners and joint venture partners to fund investments;
- (9) regional and national development of competing properties;
- (10) increases in wages and other labor costs, energy, healthcare, insurance, transportation and fuel, and other expenses central to the conduct of our business, including recent increases in energy costs; and
- (11) organized labor activities, which could cause the diversion of business from hotels involved in labor negotiations, loss of group business, and/or increased labor costs.

Any one or more of these factors could limit or reduce the demand or the prices we are able to obtain for hotel rooms, timeshare units, residential units, and corporate apartments or could increase our costs and therefore reduce the profit of our lodging businesses. Reduced demand for hotels could also give rise to losses under loans, guarantees, and minority equity investments that we have made in connection with hotels that we manage. Even where such factors do not reduce demand, our profit margins may suffer if we are unable to fully recover increased operating costs from our customers.

The uncertain environment in the lodging industry will continue to impact our financial results and growth. Both the Company and the lodging industry were hurt by several events occurring over the last several years, including the global economic downturn, the terrorist attacks on New York and Washington in September 2001, the global outbreak of SARS in 2003, and military action in Iraq. Although by 2007 both the lodging and travel industries had recovered from the depressed levels during those years, recent concerns over the possibility of an economic slowdown have left it unclear whether the recent growth environment will continue. Accordingly, our financial results and growth could be harmed if the industry recovery stalls or is reversed.

## Operational Risks

Our new branded hotel products may not be successful. We recently announced two new branded hotel products, Nickelodeon Resorts by Marriott® and Edition, and may launch additional branded hotel products in the future. We cannot assure that these brands will be accepted by hotel owners, potential franchisees, or the traveling public, that we will recover the costs we incurred in developing the brands, or that the brands will be successful. In addition, each of these new brands involves cooperation and/or consultation with a third party, including some shared control over product design and development, sales and marketing and brand standards. Disagreements between us and these third parties regarding areas of consultation or shared control could slow the development of these new brands and/or impair Marriott's ability to take actions it believes to be advisable for the success and profitability of such brands.

Our lodging operations are subject to international, national, and regional conditions. Because we conduct our business on a national and international platform, our activities are susceptible to changes in the performance of regional and global economies. In recent years, our business was hurt by decreases in travel resulting from recent economic conditions, the military action in Iraq, and the heightened travel security measures that have resulted from the threat of further terrorism. Our future economic performance is similarly subject to the uncertain economic environment in the United States and other regions, the resulting unknown pace of business travel, and the occurrence of any future incidents in the countries where we operate.

Risks relating to natural disasters, contagious disease, terrorist activity, and war could reduce the demand for lodging, which may adversely affect our revenues. So called "Acts of God," such as hurricanes, earthquakes and other natural disasters, and the spread of contagious diseases, such as Avian Flu and SARS, in locations where we own, manage or franchise significant properties, and areas of the world from which we draw a large number of customers can cause a decline in the level of business and leisure travel and reduce the demand for lodging. Actual or threatened war, terrorist activity, political unrest, civil strife, and other geopolitical uncertainty can have a similar effect. Any one or more of these events may reduce the overall demand for hotel rooms, timeshare units, and corporate apartments or limit the prices that we are able to obtain for them, both of which could adversely affect our profits.

We may have disputes with the owners of the hotels that we manage or franchise. Consistent with our focus on management and franchising, we own very few of our lodging properties. The nature of our responsibilities under our management agreements to manage each hotel and enforce the standards required for our brands under both management and franchise agreements may be subject to interpretation and may give rise to disagreements in some instances. We seek to resolve any disagreements in order to develop and maintain positive relations with current and potential hotel owners and joint venture partners but have not always been able to do so. Failure to resolve such disagreements has in the past resulted in litigation, and could do so in the future.

Damage to or other potential losses involving properties that we own, manage or franchise may not be covered by insurance. We have comprehensive property and liability insurance policies with coverage features and insured limits that we believe are customary. Market forces beyond our control may nonetheless limit the scope of insurance coverage that we can obtain and our ability to obtain coverage at reasonable rates. Certain types of losses, generally of a catastrophic nature, such as earthquakes, hurricanes and floods, or terrorist acts, may be uninsurable or too expensive to justify obtaining insurance. As a result, we may not be successful in obtaining insurance without increases in cost or decreases in coverage levels. In addition, in the event of a substantial loss, the insurance coverage we carry may not be sufficient to pay the full market value or replacement cost of our lost investment or that of hotel owners or in some cases could result in certain losses being totally uninsured. As a result, we could lose some or all of the capital we have invested in a property, as well as the anticipated future revenue from the property, and we could remain obligated for guarantees, debt, or other financial obligations related to the property.

## **Development and Financing Risks**

Our growth strategy depends upon third-party owners/operators, and future arrangements with these third parties may be less favorable. Our present growth strategy for development of additional lodging facilities entails entering into and maintaining various arrangements with property owners. The terms of our management agreements, franchise agreements, and leases for each of our lodging facilities are influenced by contract terms offered by our competitors, among other things. We cannot assure you that any of our current arrangements will continue or that we will be able to enter into future collaborations, renew agreements, or enter into new agreements in the future on terms that are as favorable to us as those that exist today.

Our ability to grow our management and franchise systems is subject to the range of risks associated with real estate investments. Our ability to sustain continued growth through management or franchise agreements for new hotels and the conversion of existing facilities to managed or franchised Marriott brands is affected, and may potentially be limited, by a variety of factors influencing real estate development generally. These include site availability, financing, planning, zoning and other local approvals, and other limitations that may be imposed by market and submarket factors, such as projected room occupancy, changes in growth in demand compared to projected supply, territorial restrictions in our management and franchise agreements, costs of construction, and anticipated room rate structure.

We depend on capital to buy and maintain hotels, and hotel owners or we may be unable to access capital when necessary. In order to fund new hotel investments, as well as refurbish and improve existing hotels, both the Company and current and potential hotel owners must periodically spend money. The availability of funds for new investments and maintenance of existing hotels depends in large measure on capital markets and liquidity factors over which we can exert little control. Our ability to recover loan and guarantee advances from hotel operations or from owners through the proceeds of hotel sales, refinancing of debt or otherwise may also affect our ability to recycle and raise new capital. In addition, downgrades of our public debt ratings by Standard & Poor's, Moody's Investor Service or similar companies could increase our cost of capital.

Further volatility in the credit markets could adversely impact our ability to sell the loans that our Timeshare business generates. Our Timeshare business provides financing to purchasers of our timeshare and fractional properties, and we periodically sell interests in those loans in the securities markets. While we do not believe that recent volatility in the credit markets will prevent us from continuing to sell those notes on attractive terms, further volatility could cause future sale terms to be materially less favorable to us or prevent us from selling our timeshare notes entirely, which in turn would reduce future gains and could result in increased borrowings to provide capital to replace anticipated proceeds from such sales.

Our development activities expose us to project cost, completion, and resale risks. We develop new hotel, timeshare interval, fractional ownership, and residential properties, both directly and through partnerships, joint ventures, and other business structures with third parties. Our involvement in the development of properties presents a number of risks, including that: (1) construction delays, cost overruns, or so called "Acts of God" such as earthquakes, hurricanes, floods or fires may increase overall project costs or result in project cancellations; (2) we may be unable to recover development costs we incur for projects that are not pursued to completion; (3) conditions within capital markets may limit our ability, or that of third parties with whom we do business, to raise capital for completion of projects that have commenced or development of future properties; and (4) properties that we develop could become less attractive due to changes in mortgage rates, market absorption or oversupply, with the result that we may not be able to sell such properties for a profit or at the prices or selling pace we anticipate.

Development activities that involve our co-investment with third parties may result in disputes that could increase project costs, impair project operations, or increase project completion risks. Partnerships, joint ventures, and other business structures involving our co-investment with third parties generally include some form of shared control over the operations of the business and create additional risks, including the possibility that other investors in such ventures could become bankrupt or otherwise lack the financial resources to meet their obligations, or could have or develop business interests, policies or objectives that are inconsistent with ours. Although we actively seek to minimize such risks before investing in partnerships, joint ventures or similar structures, actions by another investor may present additional risks of project delay, increased project costs, or operational difficulties following project completion.

Risks associated with development and sale of residential properties that are associated with our lodging and timeshare properties or brands may reduce our profits. In certain hotel and timeshare projects we participate, through minority interests and/or licensing fees, in the development and sale of residential properties associated with

our brands, including luxury residences, and condominiums under our Ritz-Carlton and Marriott brands. Such projects pose additional risks beyond those generally associated with our lodging and timeshare businesses, which may reduce our profits or compromise our brand equity, including the following:

- Decreases in residential real estate and vacation home prices or demand generally, which have historically been cyclical, could reduce our profits or even result in losses on residential sales, result in significant carrying costs if the pace of sales is slower than we anticipate, or make it more difficult to convince future hotel development partners of the value added by our brands;
- Increases in interest rates, reductions in mortgage availability, or increases in the costs of residential ownership could prevent potential customers from buying residential products or reduce the prices they are willing to pay; and
- Residential construction may be subject to warranty and liability claims, and the costs of resolving such claims may be significant.

## Technology, Information Protection, and Privacy Risks

A failure to keep pace with developments in technology could impair our operations or competitive position. The lodging and timeshare industries continue to demand the use of sophisticated technology and systems, including those used for our reservation, revenue management and property management systems, our Marriott Rewards program, and technologies we make available to our guests. These technologies and systems must be refined, updated, and/or replaced with more advanced systems on a regular basis. If we are unable to do so as quickly as our competitors or within budgeted costs and time frames, our business could suffer. We also may not achieve the benefits that we anticipate from any new technology or system, and a failure to do so could result in higher than anticipated costs or could impair our operating results.

An increase in the use of third-party Internet reservation services could adversely impact our revenues. Some of our hotel rooms are booked through Internet travel intermediaries, such as Expedia.com®, Travelocity.com®, and Priceline.com®, serving both the leisure and, increasingly, the corporate travel and group meeting sectors. While Marriott's Look No Further® Best Rate Guarantee has greatly reduced the ability of these Internet travel intermediaries to undercut the published rates at our hotels, these intermediaries continue their attempts to commoditize hotel rooms by aggressively marketing to price-sensitive travelers and corporate accounts and increasing the importance of general indicators of quality (such as "three-star downtown hotel") at the expense of brand identification. These agencies hope that consumers will eventually develop brand loyalties to their travel services rather than to our lodging brands. Although we expect to continue to maintain and even increase the strength of our brands in the online marketplace, if the amount of sales made through Internet intermediaries increases significantly, our business and profitability may be harmed.

Failure to maintain the integrity of internal or customer data could result in faulty business decisions, damage of reputation and/or subject us to costs, fines or lawsuits. Our businesses require collection and retention of large volumes of internal and customer data, including credit card numbers and other personally identifiable information of our customers as they are entered into, processed by, summarized by, and reported by our various information systems. We also maintain personally identifiable information about our employees. The integrity and protection of that customer, employee, and company data is critical to us. If that data is not accurate or complete we could make faulty decisions. Our customers also have a high expectation that we will adequately protect their personal information, and the regulatory environment surrounding information security and privacy is increasingly demanding, both in the U.S. and other international jurisdictions in which we operate. A significant theft, loss or fraudulent use of customer, employee or company data could adversely impact our reputation and could result in remedial and other expenses, fines and litigation.

Changes in privacy law could adversely affect our ability to market our products effectively. Our Timeshare segment, and to a lesser extent our other lodging segments, rely on a variety of direct marketing techniques, including telemarketing, email marketing, and postal mailings. Any further restrictions in laws such as the Telemarketing Sales Rule, CANSPAM Act, and various U.S. state laws, or new federal laws, regarding marketing and solicitation or international data protection laws that govern these activities could adversely affect the continuing effectiveness of telemarketing, email, and postal mailing techniques and could force further changes in our marketing strategy. If this occurs, we may not be able to develop adequate alternative marketing strategies, which could impact the amount and timing of our sales of timeshare units and other products. We also obtain access to potential customers from travel service providers or other companies with whom we have substantial relationships and market to some individuals on these lists directly or by including our marketing message in the other company's marketing materials. If the acquisition of these lists was prohibited or otherwise restricted, our ability to develop new customers and introduce them to our products could be impaired.

#### Other Risks

If we cannot attract and retain talented associates our business could suffer. We compete with other companies both within and outside of our industry for talented personnel. If we are not able to recruit, train, develop and retain sufficient numbers of talented associates, we could experience increased associate turnover, decreased guest satisfaction, low morale, inefficiency or internal control failures. Insufficient numbers of talented associates could also limit our ability to grow and expand our businesses.

Delaware law and our governing corporate documents contain, and our board of directors could implement, anti-takeover provisions that could deter takeover attempts. Under the Delaware business combination statute, a stockholder holding 15 percent or more of our outstanding voting stock could not acquire us without board of director's consent for at least three years after the date the stockholder first held 15 percent or more of the voting stock. Our governing corporate documents also, among other things, require supermajority votes in connection with mergers and similar transactions. In addition, our Board of Directors could, without stockholder approval, implement other anti-takeover defenses, such as a stockholder rights plan to replace the existing stockholder's rights plan that will expire in March 2008 and which we do not presently plan to renew.

#### Item 1B. Unresolved Staff Comments.

None.

## Item 2. Properties.

Company-operated properties are described in Part I, Item 1, "Business," earlier in this report. We believe our properties are in generally good physical condition with the need for only routine repairs and maintenance and periodic capital improvements. Most of our regional offices and reservation centers, both domestically and internationally, are located in leased facilities. We also lease space in six office buildings with combined space of approximately 1.3 million square feet in Maryland and Florida where our corporate, Ritz-Carlton, and Marriott Vacation Club International headquarters are located.

## Item 3. Legal Proceedings.

From time to time, we are subject to certain legal proceedings and claims in the ordinary course of business, including adjustments proposed during governmental examinations of the various tax returns we file. We currently are not aware of any legal proceedings or claims that we believe will have, individually or in aggregate, a material adverse effect on our business, financial condition, or operating results.

## Item 4. Submission of Matters to a Vote of Security Holders.

No matters were submitted to a vote of shareholders during the fourth quarter of the fiscal year covered by this report.

## **Executive Officers of the Registrant**

See Part III, Item 10 of this report for information about our executive officers.

## Part II

## Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities.

## **Market Information and Dividends**

On June 9, 2006, we completed a two-for-one stock split that was effected in the form of a stock dividend. For periods prior to the stock split, all share and per share data in this Annual Report on Form 10-K have been retroactively adjusted to reflect the stock split.

The range of prices of our common stock and dividends declared per share for each quarterly period within the last two years are as follows:

		Stock	Stock Price		
		High	Low	Declared Per Share	
2006	First Quarter	\$35.35	\$32.32	\$ 0.0525	
	Second Quarter	38.38	34.20	0.0625	
	Third Quarter	38.44	34.30	0.0625	
	Fourth Quarter	48.31	37.19	0.0625	
		Stock I		Dividends Declared Per	
		High	Low	Share	
2007	First Overton	¢ = 1 = 0	¢ 44 70	¢ 0.000F	
	First Quarter	\$51.50	\$44.79	\$ 0.0625	
	Second Quarter	\$51.50 52.00	43.90	0.0625	
	•				

At January 25, 2008, there were 354,043,159 shares of Class A Common Stock outstanding held by 48,288 shareholders of record. Our Class A Common Stock is traded on the New York Stock Exchange and the Chicago Stock Exchange. The year-end closing price for our stock was \$34.12 on December 28, 2007, and \$47.72 on December 29, 2006. All prices are reported on the consolidated transaction reporting system.

## Fourth Quarter 2007 Issuer Purchases of Equity Securities

(in millions, except per share amounts)	Total Number of Shares	Average Price per	Total Number of Shares Purchased as Part of Publicly Announced Plans	Maximum Number of Shares That May Yet Be Purchased Under the Plans or
Period	Purchased	Share	or Programs <sup>(1)</sup>	Programs (1)
September 8, 2007-October 5, 2007	1.9	\$ 42.97	1.9	43.4
October 6, 2007-November 2, 2007	4.6	40.50	4.6	38.8
November 3, 2007-November 30, 2007	0.9	36.72	0.9	37.9
December 1, 2007-December 28, 2007	4.7	34.12	4.7	33.2

On August 2, 2007, we announced that our Board of Directors increased, by 40 million shares, the authorization to repurchase our Class A Common Stock for a total outstanding authorization of approximately 51 million shares on that date. We repurchase shares in the open market and in privately negotiated transactions.

#### Selected Financial Data. Item 6.

The following table presents a summary of selected historical financial data for the Company derived from our financial statements as of and for our last eight fiscal years.

Since the information in this table is only a summary and does not provide all of the information contained in our financial statements, including the related notes, you should read "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our Consolidated Financial Statements.

				Fiscal	l Year <sup>(1)</sup>			
(\$ in millions, except per share data)	2007	2006	2005	2004	2003	2002	2001	2000
Income Statement Data:	#42.000	Ф44 OOF	Ф44.4D0	A 0 770	A 0 710	ф. о ppp	A = = = 0	Φ 5 044
Revenues (2)	\$12,990	\$11,995	\$11,129	\$ 9,778	\$ 8,712	\$ 8,222	\$ 7,768	\$ 7,911
Operating income (2)	\$ 1,188	\$ 1,087	\$ 699	\$ 575	\$ 481	\$ 455	\$ 420	\$ 762
Income from continuing operations	\$ 697	\$ 712	\$ 543	\$ 487	\$ 380	\$ 365	\$ 269	\$ 490
Cumulative effect of change in accounting principle (3)	_	(109)	_	_	_	_	_	_
Discontinued operations (4)	(1)	5	126	109	122	(88)	(33)	(11)
Net income	\$ 696	\$ 608	\$ 669	\$ 596	\$ 502	\$ 277	\$ 236	\$ 479
Per Share Data:			<del></del>					
Diluted earnings per share from continuing operations	\$ 1.75	\$ 1.65	\$ 1.17	\$ 1.01	\$ 0.77	\$ 0.72	\$ 0.52	\$ 0.96
Diluted loss per share from cumulative effect of								
accounting change		(0.25)	_	_			_	_
Diluted earnings (loss) per share from discontinued								
operations		0.01	0.28	0.23	0.25	(0.17)	(0.06)	(0.02)
Diluted earnings per share	\$ 1.75	\$ 1.41	\$ 1.45	\$ 1.24	\$ 1.02	\$ 0.55	\$ 0.46	\$ 0.94
Cash dividends declared per share	\$0.2875	\$0.2400	\$0.2000	\$0.1650	\$0.1475	\$0.1375	\$0.1275	\$0.1175
Balance Sheet Data (at end of year):								
Total assets	\$ 8,942	\$ 8,588	\$ 8,530	\$ 8,668	\$ 8,177	\$ 8,296	\$ 9,107	\$ 8,237
Long-term debt (2)	2,790	1,818	1,681	836	1,391	1,553	2,708	1,908
Shareholders' equity	1,429	2,618	3,252	4,081	3,838	3,573	3,478	3,267
Other Data:								
Base management fees (2)	\$ 620	\$ 553	\$ 497	\$ 435	\$ 388	\$ 379	\$ 372	\$ 383
Franchise fees (2)	439	390	329	296	245	232	220	208
Incentive management fees (2)	369	281	201	142	109	162	202	316
Total fees	\$ 1,428	\$ 1,224	\$ 1,027	\$ 873	\$ 742	\$ 773	\$ 794	\$ 907
Fee Revenue-Source:								
North America (5)	\$ 1,115	\$ 955	\$ 809	\$ 682	\$ 592	\$ 626	\$ 664	\$ 779
Outside North America	313	269	218	191	150	147	130	128
Total fees	\$ 1,428	\$ 1,224	\$ 1,027	\$ 873	\$ 742	\$ 773	\$ 794	\$ 907

All fiscal years included 52 weeks, except for 2002, which included 53 weeks.

<sup>(2)</sup> 

Balances do not reflect the impact of discontinued operations.

We adopted Statement of Position 04-2 "Accounting for Real Estate Time-Sharing Transactions," in our 2006 first quarter which we reported in our Consolidated Statements of Income as a cumulative effect of change in accounting principle.

In 2002, we announced our intent to sell, and subsequently did sell, our Senior Living Services business and exited our Distribution Services business. In 2007, we exited our synthetic fuel business. These businesses are now reflected as discontinued operations.

Includes the continental United States and Canada.

## Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations.

## **BUSINESS AND OVERVIEW**

We are a worldwide operator and franchisor of 2,999 properties (535,093 rooms) and related facilities. The figures in the preceding sentence are as of year-end 2007 and include 21 home and condominium projects (1,916 units) for which we manage the related owners' associations. In addition, we provided 2,156 furnished corporate housing rental units, which are not included in the totals.

Our operations are grouped into five business segments: North American Full-Service Lodging, North American Limited-Service Lodging, International Lodging, Luxury Lodging, and Timeshare. We operate, develop, and franchise under 19 separate brand names in 68 countries and territories.

We earn base, incentive, and franchise fees based upon the terms of our management and franchise agreements. Revenues are also generated from the following sources associated with our Timeshare segment: (1) selling timeshare interval, fractional ownership, and residential properties; (2) operating the resorts and residential properties; and (3) financing customer purchases of timesharing intervals. We earn revenues from the limited number of hotels we own or lease. Finally, we earn fees in association with affinity card endorsements and the sale of branded residential real estate.

We sell residential real estate in conjunction with luxury hotel development (Ritz-Carlton-Residential) and Timeshare segment projects (Ritz-Carlton Club-Residential and Grand Residences by Marriott-Residential). Our Timeshare segment residential projects are typically opened over time with limited inventory available at any one time. Residences developed in conjunction with hotels are typically constructed and sold by hotel owners with limited amounts, if any, of our capital at risk. While the worldwide residential market is very large, the luxurious nature of our residential properties, the quality and exclusivity associated with our brands, and the hospitality services that we provide, all serve to make our residential products distinctive.

Generally, lodging demand remained strong through 2007, driven by continued strength associated with worldwide business travel. In general, luxury, international, and full-service properties experienced stronger demand than limited-service properties. Strong demand enabled us to increase rates at the property level, which resulted in solid year-over-year RevPAR increases. Revenue mix improvement is a function of the strong demand environment and results as some less profitable business is limited in favor of more profitable business, such as fewer discounted leisure packages in favor of more corporate business. This strategy of shifting business to higher rated tiers, yielded strong year-over-year average daily rate growth and only modest occupancy declines. In addition, group rates continue to increase as business negotiated in earlier years at lower rates is replaced with business negotiated at higher rates.

Demand for our brands is strong in many markets around the world. For our North American comparable properties, RevPAR increases in 2007, as compared to the year-ago period, were particularly strong in Dallas, New York City, Los Angeles, and San Francisco. Internationally, 2007 RevPAR increases as compared to the prior year were particularly strong in Central and South East Asia, South America, the Middle East, and Eastern Europe.

Our approach to improving property-level and above-property productivity has benefited our profitability, as well as that of owners and franchisees. Driving room rate improvement, benchmarking successful performance, and leveraging our size have all contributed to property-level margin improvements and higher incentive management fees to us. We continue to enhance the appeal of our proprietary Web site, <a href="https://www.Marriott.com">www.Marriott.com</a>, through functionality and service improvements, and we continue to capture an increasing proportion of property-level reservations via this cost-efficient channel.

Our brands are strong as a result of superior customer service with an emphasis on guest and associate satisfaction, the worldwide presence and quality of our brands, our Marriott Rewards loyalty program, an information-rich and easy-to-use Web site, a multi-channel central reservations system, and desirable property amenities. We, along with owners and franchisees, continue to invest in our brands by means of new, refreshed, and reinvented properties, new room and public space designs and enhanced amenities and technology offerings.

In 2007, we announced our agreement with the Nickelodeon division of Viacom, Inc. and Miller Global Properties, LLC to co-develop a new lodging resort brand and concept for travelers seeking fun and adventure, "Nickelodeon Resorts by Marriott." Also during 2007, we announced our agreement with Ian Schrager to create a global boutique lifestyle hotel brand "Edition" on a large scale.

See Part I, Item 1A., "Risk Factors," of this report for important information regarding forward-looking statements made in this report and risks and uncertainties that the Company faces.

## CONSOLIDATED RESULTS

The following discussion presents an analysis of results of our operations for 2007, 2006, and 2005.

## **Continuing Operations**

#### Revenues

2007 Compared to 2006

Revenues increased by \$995 million (8 percent) to \$12,990 million in 2007 from \$11,995 million in 2006, as a result of stronger demand for hotel rooms worldwide, which allowed us to increase room rates and favorable exchange rates worldwide. Base management and franchise fees increased by \$116 million as a result of stronger RevPAR and unit growth, as we opened 203 properties (29,200 rooms) throughout 2007. In 2006, we recognized \$5 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006, versus no similar fees in 2007. Incentive management fees increased by \$88 million due to stronger RevPAR and property-level margin improvements associated with room rate increases and productivity improvements. Incentive management fees included \$17 million and \$10 million for 2007 and 2006, respectively, that were calculated based on prior periods' results, but not earned and due until the periods in which they were recognized. Furthermore, incentive management fees for 2007 included \$13 million and base management fees for 2007 included \$6 million of business interruption insurance proceeds associated with hurricanes in prior years compared to \$1 million of business interruption insurance proceeds received in 2006. Stronger catering, food and beverage, spa, and other profits also drove property-level margins higher. Year-over-year RevPAR increases were driven primarily by rate increases.

Timeshare sales and services revenue increased by \$170 million (11 percent) in 2007 over the prior year. The increase largely reflected development revenue increases over the prior year as some newer projects reached revenue recognition thresholds. In 2006, some projects were in the early stages of development and did not reach revenue recognition thresholds until 2007. The increase in revenue also reflects increased services and financing revenue.

Owned, leased, corporate housing and other revenue increased by \$121 million (11 percent). The increase largely reflected stronger RevPAR and the mix of owned and leased properties in 2007, as compared to 2006, and to a lesser extent, higher fees associated with affinity card endorsements and the sale of branded residential real estate, offset by the recognition of \$19 million of hotel management and franchise agreement termination fees in 2007, as compared to the recognition of \$26 million of such fees in 2006.

The \$995 million increase in total revenue includes \$500 million of increased cost reimbursements revenue, to \$8,575 million in 2007 from \$8,075 million in the prior year. This revenue represents reimbursements of costs incurred on behalf of managed and franchised properties and relates, predominantly, to payroll costs at managed properties where we are the employer. As we record cost reimbursements based upon the costs incurred with no added markup, this revenue and related expense have no impact on either our operating income or net income. The increase in reimbursed costs is primarily attributable to wage increases, sales growth, and the growth in the number of properties we manage. We added 15 managed properties (4,870 rooms) and 138 franchised properties (15,963 rooms) to our system in 2007, net of properties exiting the system.

## 2006 Compared to 2005

Revenues increased by \$866 million (8 percent) to \$11,995 million in 2006 from \$11,129 million in 2005, as a result of stronger demand for hotel rooms worldwide. Base management and franchise fees increased by \$117 million as a result of stronger RevPAR and unit growth. In 2006, we recognized \$5 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006. Incentive management fees increased by \$80 million due to stronger RevPAR and property-level margin improvements associated with room rate increases and productivity improvements. Incentive management fees include \$10 million and \$14 million for 2006 and 2005, respectively, that were calculated based on prior periods' earnings but not earned and due until the periods in which they were recognized. Stronger catering, food and beverage, spa, and other profits also drove property-level margins higher. Year-over-year RevPAR increases were driven primarily by rate increases. Owned and leased revenue increased significantly, primarily as a result of our purchase, early in the second half of 2005, of 13 formerly managed properties from CTF Holdings Ltd. ("CTF"). See Footnote No. 8, "Acquisitions and Dispositions," later in this report for a detailed description of the CTF transaction. As planned, eight of the CTF properties were sold during 2006: one property was sold in the first quarter; five properties were sold in the second quarter; and two properties were sold in the third quarter.

Timeshare sales and services revenue increased by \$90 million (6 percent) over the prior year. The increase largely reflects \$77 million of revenue in 2006 from note securitization gains. As detailed later in the "Cumulative Effect of Change in Accounting Principle" narrative, note securitization gains of \$69 million for 2005 are not reflected in revenue, but instead are a component of gains and other income. Additionally, financing and services revenue increased in 2006 versus the prior year, as did villa rental revenue. Partially offsetting these increases, development revenue declined due to projects in the early stages of development that did not reach revenue recognition thresholds and limited available inventory associated with projects that sold out or were nearing sell-out.

The \$866 million increase in total revenue includes \$404 million of increased cost reimbursements revenue, to \$8,075 million in 2006 from \$7,671 million in the prior year. The increase in reimbursed costs is primarily attributable to the growth in the number of properties we manage and to wage increases. We added 13 managed properties (4,126 rooms) and 77 franchised properties (11,286 rooms) to our system in 2006, net of properties exiting the system.

## **Operating Income**

2007 Compared to 2006

Operating income increased by \$101 million (9 percent) to \$1,188 million in 2007 from \$1,087 million in the prior year. The increase in operating income reflects stronger combined base management, incentive management, and franchise fees of \$204 million, partially offset by higher general, administrative, and other expenses of \$91 million, lower Timeshare sales and services revenue net of direct expenses of \$7 million, and lower owned, leased, corporate housing, and other revenue net of direct expenses of \$5 million.

The combined base management, incentive management, and franchise fees increase of \$204 million reflected strong RevPAR growth, unit growth, and property-level margin improvements and favorable exchange rates worldwide. In 2006, we recognized \$5 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006, versus no similar fees in 2007. Incentive management fees included \$17 million and \$10 million for 2007 and 2006, respectively, that were calculated based on prior periods' results, but not earned and due until the periods in which they were recognized. Furthermore, incentive management fees for 2007 included \$13 million and base management fees included \$6 million of business interruption insurance proceeds associated with hurricanes in prior years compared to \$1 million of business interruption insurance proceeds received in 2006.

As compared to the year-ago period, general, administrative, and other expenses increased by \$91 million (13 percent) from \$677 million in 2006 to \$768 million in 2007. In 2007, we incurred a \$35 million charge related to excise taxes associated with the settlement of issues raised during the Internal Revenue Service and Department of Labor examination of our employee stock ownership plan ("ESOP") feature of our Employees' Profit Sharing, Retirement and Savings Plan and Trust (the "Plan"). See Footnote No. 3, "Income Taxes," for additional information on the ESOP settlement. Additionally, the increase was attributable to, among other things, increased costs related to our unit growth, development and systems improvements, increased litigation expenses, and increased other administrative costs. Also unfavorably impacting general, administrative, and other expenses, when compared to the prior year, were foreign exchange losses totaling \$2 million in 2007 as compared to foreign exchange gains of \$6 million in 2006 and \$1 million of guarantee charges in 2007 while the year-ago period reflected \$6 million of guarantee reversals. Partially offsetting the aforementioned increases were \$4 million of lower hotel operating agreement performance cure payments in 2007 as compared to 2006 and a \$9 million reversal in 2007 of reserves that were no longer required. Increased legal expenses in 2007 include charges associated with litigation and other legal matters. Of the \$91 million increase in total general, administrative, and other expenses, an increase of \$13 million was attributable to our Lodging segments and a \$78 million increase was unallocated.

Timeshare sales and services revenue net of direct expenses of \$350 million decreased by \$7 million, as compared to the prior year, primarily reflecting flat development revenue net of product costs and marketing and selling costs and \$12 million of increased financing revenue net of financing expenses, more than offset by the reversal in 2006 of \$15 million reversal of contingency reserves and \$4 million of lower services revenue net of services expenses. Flat development revenue net of product costs and marketing and selling costs reflected newer projects that reached reportability thresholds in 2007, offset by several other projects that were approaching sell-out. The increase in financing revenue net of financing costs primarily reflects increased accretion, interest income, and higher note sale gains in 2007, as compared to 2006.

The \$5 million decrease in owned, leased, corporate housing, and other revenue net of direct expenses reflected \$19 million in hotel management and franchise agreement termination fees received in 2007, as compared to \$26 million in 2006. Depreciation charges totaling \$8 million were recorded in 2007 associated with one owned

property that was reclassified from "held for sale" to "held and used" during 2007 as compared to depreciation charges recorded in 2006 of \$7 million associated with two properties that were reclassified from "held for sale" to "held and used." Partially offsetting the aforementioned decreases in owned, leased, corporate housing, and other revenue net of direct expenses was the favorable impact of \$4 million associated with both the stronger demand environment in 2007 and the impact of the sale and purchase of several properties.

#### 2006 Compared to 2005

Operating income increased by \$388 million (56 percent) to \$1,087 million in 2006 from \$699 million in the prior year. The increase in 2006 is, in part, due to a combined base management, franchise, and incentive management fee increase of \$197 million, reflecting stronger RevPAR growth, unit growth, and property-level margin improvements. Stronger owned, leased, corporate housing, and other revenue net of direct expenses contributed \$17 million of the improvement in operating income and reflected the strong demand environment in 2006 and the impact of the CTF hotel properties acquired in 2005. The \$17 million improvement in 2006 versus the prior year reflects increased owned and leased results of \$20 million, a \$12 million increase in hotel management and franchise agreement termination fees received and \$5 million of higher other income, partially offset by \$20 million of lower land lease income. Also reflected in the year-over-year change in owned and leased results are depreciation charges totaling \$7 million recorded in 2006 associated with two properties that were reclassified in 2006 from "held for sale" to "held and used." The depreciation charges totaling \$7 million represented the amount that would have been recognized had the two properties been continuously classified as "held and used."

Timeshare sales and services revenue net of direct expenses increased by \$98 million (38 percent) in 2006 and largely reflects \$77 million of revenue in 2006 from note securitization gains. As noted earlier in the "Revenue" discussion, note securitization gains of \$69 million for 2005 are not reflected in revenue, but instead are a component of gains and other income. Also reflected in the \$98 million favorable variance, development revenue net of expenses increased by \$14 million and financing, services and other revenue net of expenses increased by \$7 million. Increased development revenue net of expenses primarily reflects lower development expenses associated with projects in 2006 in the early stages of development that did not reach revenue recognition thresholds and the timing of expenses associated with Statement of Position 04-2, "Accounting for Real Estate Time-Sharing Transactions" ("SOP 04-2"), implemented in 2006. Increased financing, services and other revenue net of expenses reflects a \$15 million reversal of marketing related contingency reserves in 2006.

As compared to the year-ago period, general, administrative, and other expenses decreased by \$76 million (10 percent) to \$677 million in 2006 from \$753 million in 2005. In 2005, we incurred general, administrative, and other expenses of \$94 million primarily due to the non-cash write-off of deferred contract acquisition costs associated with the termination of management agreements resulting from the CTF transaction. We also incurred general, administrative, and other expenses of \$30 million in 2005 associated with our bedding incentive program. We implemented the bedding incentive program in the 2005 second quarter to help ensure that guests could enjoy the comfort and luxury of our new bedding by year-end 2005. Further impacting general, administrative, and other expenses, 2005 reflected hotel operating agreement performance cure payments of \$15 million versus a \$6 million similar payment in 2006, and 2005 also reflected \$9 million of guarantee charges associated with three properties versus the reversal of an additional \$5 million of guarantee charges in 2006. Additionally, impacting the year-over-year general, administrative, and other expenses variance were foreign exchange gains totaling \$6 million in 2006 as compared to losses of \$5 million in 2005. Also impacting the year-over-year change in general, administrative, and other expenses, an additional \$22 million reflects unit growth, systems improvements, higher program and joint venture development expenses, and customary increases in ordinary costs such as wages and benefits. Development expenses and deferred compensation expenses were higher in 2006 by \$15 million and \$5 million, respectively. As noted under the heading "New Accounting Standards" in the "Management's Discussion and Analysis of Financial Condition and Results of Operations," section of our 2006 Form 10-K, we adopted a new accounting standard in 2006 associated with share-based compensation. This new standard resulted in incremental general, administrative, and other expenses of \$39 million versus 2005, primarily impacting the unallocated portion of our general, administrative, and other expenses. Of the \$76 million decrease in total general, administrative, and other expenses, a decrease of \$124 million was attributable to our Lodging segments and a \$48 million increase was unallocated.

## Gains and Other Income (Expense)

The table below shows our gains and other income for fiscal years 2007, 2006, and 2005:

(\$ in millions)	2007	2006	2005
Timeshare note sale gains	\$	\$	\$ 69
Gains on sales of real estate and other	39	26	34
Loss on expected land sale	_	(37)	_
Other note sale/repayment gains	1	2	25
Gains on forgiveness of debt	12	_	_
Gain on sale/income on redemption of joint venture and other investments	31	68	7
Income from cost method joint ventures	14	15	14
	\$ 97	\$ 74	\$149

## 2007 Compared to 2006

The \$12 million gain on forgiveness of debt for 2007 was associated with government incentives. The loans were forgiven in recognition of our contribution to job growth and economic development. Gain on sale/income on redemption of joint venture and other investments of \$31 million in 2007 reflected an \$18 million gain associated with the sale of stock we held and net gains totaling \$13 million on the sale of joint venture investments. Gain on sale/income on redemption of joint venture and other investments of \$68 million in 2006 comprised \$43 million of net gains associated with the sale of joint venture investments and the redemption of preferred stock we held in one investee which generated a gain of \$25 million.

## 2006 Compared to 2005

Gain on sale/income on redemption of joint venture and other investments of \$68 million in 2006 represents \$43 million of net gains associated with the sale of joint venture and other investments and \$25 million of income associated with the redemption of preferred stock we held in one investee. As further explained in the earlier "Revenues" discussion for 2006, Timeshare segment note sale gains of \$77 million in 2006 are presented in the "Timeshare sales and services" revenue caption.

## Interest Expense

## 2007 Compared to 2006

Interest expense increased by \$60 million (48 percent) to \$184 million for 2007 from \$124 million in 2006. Of the \$60 million increase over 2006, \$78 million was due to: \$26 million of higher interest reflecting a higher outstanding commercial paper balance, primarily due to increased share repurchases and the ESOP settlement payments, and related interest rates; \$25 million of interest associated with our Series H Senior Notes issuance which occurred late in 2006 and our Series I and Series J Senior Notes issuances which occurred in 2007; a charge of \$13 million in 2007 related to the ESOP settlement; charges totaling \$53 million and \$46 million in 2007 and 2006, respectively, relating to interest on accumulated cash inflows in advance of our cash outflows for various programs that we operate on the owners' behalf including Marriott Rewards, gift certificates, and self-insurance programs; interest totaling \$5 million associated with other additional debt; and the write off of \$2 million of deferred financing costs associated with the refinancing of our revolving credit agreement in 2007. See Footnote No. 3, "Income Taxes," for additional information on the ESOP settlement. The increase in interest on the programs we operate on behalf of the owners over the year-ago period is attributable to higher liability balances and higher interest rates. Partially offsetting the \$78 million interest expense increases over 2006 was an \$18 million favorable variance to last year for higher capitalized interest associated with construction projects.

## 2006 Compared to 2005

Interest expense increased by \$18 million (17 percent) to \$124 million in 2006 from \$106 million in 2005. Included within interest expense for 2006 and 2005 are charges totaling \$46 million and \$29 million, respectively, relating to interest on accumulated cash inflows, in advance of our cash outflows for various programs that we operate on the owners' behalf. The increase in interest on these programs over 2005 is related to higher liability balances and higher interest rates. Interest expense also increased in 2006, due to our June 2005 Series F Notes issuance, our June 2006 Series H Notes issuance, and higher commercial paper balances coupled with higher rates. Partially offsetting these increases were interest expense declines associated with the payoff, at maturity, of both our Series D Notes in April 2005 and Series B Notes in November 2005, and the exchange of our Series C and Series E Notes for lower interest rate Series G Notes in 2005.

## Interest Income, Provision for Loan Losses, and Income Tax

2007 Compared to 2006

Interest income, before the provision for loan losses, decreased by \$11 million (22 percent) to \$38 million in 2007 from \$49 million in 2006, primarily reflecting lower interest income associated with loans that have been repaid to us, partially offset by the impact associated with new loan fundings.

Loan loss provisions increased by \$20 million versus the prior year primarily reflecting a \$12 million charge associated with one property and a \$5 million charge to write off our remaining exposure associated with our investment in a Delta Airlines, Inc. ("Delta") lease versus loan loss reversals of \$3 million in 2006. For additional information regarding the Delta lease investment write-off, see the "Investment in Leveraged Lease" caption later in this report.

Our tax provision increased by \$61 million (16 percent) from a tax provision of \$380 million in 2006 to a tax provision of \$441 million in 2007 and reflected higher pretax income from our Lodging segments as well as a higher tax rate in 2007, primarily reflecting both increased taxes associated with our international operations and a less favorable mix of taxable earnings between countries. Increased taxes also reflect a charge for a German legislative tax change in 2007, which had a one-time impact and \$6 million of taxes in 2007 associated with additional interest on the ESOP settlement. See Footnote No. 3, "Income Taxes," for additional information on the ESOP settlement.

## 2006 Compared to 2005

Interest income, before the provision for loan losses, decreased by \$30 million (38 percent) to \$49 million in 2006 from \$79 million in 2005, primarily reflecting the impact of loans repaid to us in 2005. Loan loss provisions decreased by \$31 million versus the prior year reflecting the reversal of loan loss provisions totaling \$3 million in 2006 compared to a charge of \$17 million in 2005 due to the impairment of our Delta lease, see the "Investment in Leveraged Lease" caption later in this report for additional information and an \$11 million loan loss provision in 2005 associated with one property.

Our tax provision totaled \$380 million in 2006 compared to a tax provision of \$284 million in 2005. The difference of \$96 million is attributable to higher taxes in 2006 associated with higher pretax income from our lodging operations.

## **Equity in Earnings**

2007 Compared to 2006

Equity in earnings increased by \$12 million from \$3 million in 2006 to \$15 million in 2007 and reflected the mix of investments, compared to 2006, and stronger results at several joint ventures reflecting the strong lodging demand environment in 2007, for one joint venture, the reopening of a hotel, late in 2006, in Mexico, which had been closed following a hurricane in 2005 and strong demand in 2007 for our timeshare products in Hawaii.

#### 2006 Compared to 2005

The \$33 million decline from earnings of \$36 million in 2005 to earnings of \$3 million in 2006, attributable to our equity investments, reflected the recognition in 2005 of \$30 million of equity earnings from the sale of hotels by three equity joint ventures in which we had equity interests. In addition, since 2005 we have sold several equity joint ventures.

## **Income from Continuing Operations**

2007 Compared to 2006

Compared to 2006, income from continuing operations decreased by \$15 million (2 percent) to \$697 million in 2007, and diluted earnings per share from continuing operations increased by \$0.10 (6 percent) to \$1.75. As discussed in more detail in the preceding sections beginning with "Operating Income," the decrease versus the prior year is due to higher general, administrative, and other expenses (\$91 million), higher taxes (\$61 million), higher interest expense (\$60 million), higher loan loss provision (\$20 million), lower interest income (\$11 million), lower timeshare sales and services revenue net of direct expenses (\$7 million), and lower owned, leased, corporate housing, and other revenue net of direct expenses (\$5 million). Partially offsetting these unfavorable variances were higher fee income (\$204 million), higher gains and other income (\$23 million), higher equity investment results (\$12 million), and a higher minority interest benefit (\$1 million).

## 2006 Compared to 2005

Compared to 2005, income from continuing operations increased by \$169 million (31 percent) to \$712 million in 2006, and diluted earnings per share from continuing operations increased by \$0.48 (41 percent) to \$1.65. As discussed in more detail in the preceding sections beginning with "Operating Income," the increase versus the prior year is due to higher fee income (\$197 million), higher timeshare sales and services revenue net of direct expenses (\$98 million), lower general, administrative, and other expenses (\$76 million), a lower loan loss provision (\$31 million), higher owned, leased, corporate housing, and other revenue net of direct expenses (\$17 million), and lower minority interest expense (\$2 million). Partially offsetting these favorable variances, were significantly higher taxes (\$96 million), lower gains and other income (\$75 million), lower equity investment results (\$33 million), lower interest income (\$30 million), and higher interest expense (\$18 million).

## **Cumulative Effect of Change in Accounting Principle**

2006

Statement of Position 04-2, "Accounting for Real Estate Time-Sharing Transactions"

In December 2004, the Financial Accounting Standards Board ("FASB") issued Financial Accounting Standards ("FAS") No. 152, "Accounting for Real Estate Time-Sharing Transactions-an amendment of FASB Statements No. 66 and 67," and the American Institute of Certified Public Accountants issued SOP 04-2. Additionally, the FASB amended FAS No. 66, "Accounting for Sales of Real Estate," and FAS No. 67, "Accounting for Costs and Initial Rental Operations of Real Estate Projects," to exclude accounting for real estate time-sharing transactions from these statements. We adopted SOP 04-2 at the beginning of the 2006 first quarter.

Under SOP 04-2, we charge the majority of sales and marketing costs we incur to sell timeshares to expense when incurred. We also record an estimate of expected uncollectibility on notes receivable that we receive from timeshare purchasers as a reduction in revenue at the time that we recognize profit on a timeshare sale. We also account for rental and other operations during holding periods as incidental operations, which requires us to record any excess of revenues over costs as a reduction of inventory costs.

The adoption of SOP 04-2 in 2006, which we reported as a cumulative effect of change in accounting principle in our Consolidated Statements of Income, resulted in a non-cash after-tax charge of \$109 million (\$0.25 per diluted share). The pretax charge totaled \$173 million and comprised a \$130 million inventory write-down, the establishment of a \$25 million notes receivable reserve and an increase in current liabilities of \$18 million.

We estimate that, for the 20-year period from 2008 through 2027, the cost of completing improvements and currently planned amenities for our owned timeshare properties will be approximately \$1.7 billion.

## **Business Segments**

We are a diversified hospitality company with operations in five business segments:

- North American Full-Service Lodging, which includes Marriott Hotels & Resorts, Marriott Conference Centers, JW Marriott Hotels & Resorts, Renaissance Hotels & Resorts, and Renaissance ClubSport properties located in the continental United States and Canada;
- North American Limited-Service Lodging, which includes Courtyard, Fairfield Inn, SpringHill Suites, Residence Inn, TownePlace Suites, and Marriott ExecuStay properties located in the continental United States and Canada;
- *International Lodging*, which includes Marriott Hotels & Resorts, JW Marriott Hotels & Resorts, Renaissance Hotels & Resorts, Courtyard, Fairfield Inn, Residence Inn, Ramada International, and Marriott Executive Apartments properties located outside the continental United States and Canada;
- · Luxury Lodging, which includes The Ritz-Carlton and Bulgari Hotels & Resorts properties worldwide; and
- *Timeshare*, which includes the development, marketing, operation, and sale of timeshare, fractional ownership, and residential properties worldwide under Marriott Vacation Club, The Ritz-Carlton Club, Grand Residences by Marriott, and Horizons by Marriott Vacation Club.

In addition to the segments above, in 2007 we exited the synthetic fuel business, which was formerly a separate segment but which we now report under discontinued operations.

In addition to the brands noted above, in 2007 we announced our new brand of family-friendly resorts and spas, "Nickelodeon Resorts by Marriott" and a new brand of lifestyle boutique hotels, "Edition." At year-end 2007, no properties were yet open under either brand.

In 2006, we analyzed our internal reporting process and implemented changes in the fourth quarter that were designed to improve efficiency and, as part of this process, we evaluated the impact on segment reporting. Accordingly, we now report five operating segments, and no longer allocate indirect administrative expenses to our segments. We reflect this revised segment reporting throughout this report for all periods presented, and present historical figures in a manner that is consistent with the revised segment reporting. See also the Form 8-K that we filed on March 19, 2007, furnishing quarterly Revenues and Income from Continuing Operations for each of 2006 and 2005 in the new segment format.

We evaluate the performance of our segments based primarily on the results of the segment without allocating corporate expenses, interest expense, indirect general, administrative, and other expenses, or income taxes. With the exception of the Timeshare segment, we do not allocate interest income to our segments. Because note sales are an integral part of the Timeshare segment, we include note sale gains in our Timeshare segment results. We include interest income associated with Timeshare segment notes in our Timeshare segment results because financing sales are an integral part of that segment's operations. We also allocate other gains or losses as well as equity in earnings or losses from our joint ventures and divisional general, administrative, and other expenses to each of our segments. "Other unallocated corporate" represents that portion of our revenues, general, administrative, and other expenses, equity in earnings or losses, and other gains or losses that are not allocable to our segments.

We aggregate the brands presented within our North American Full-Service, North American Limited-Service, International, Luxury, and Timeshare segments considering their similar economic characteristics, types of customers, distribution channels, the regulatory business environment of the brands and operations within each segment, and our organizational and management reporting structure.

# **Total Lodging Products by Segment**

At year-end 2007, we operated or franchised the following properties by segment (excluding 2,156 corporate housing rental units):

		Total Lodging Products				
	TI C	Properties	m . 1	TI 0	Rooms	m . 1
North American Full-Service Lodging Segment (1)	U.S.	Non-U.S.	Total	U.S.	Non-U.S.	Total
Marriott Hotels & Resorts	309	12	321	121,842	4,556	126,398
Marriott Conference Centers	13		13	3,476	<del>4</del> ,550	3,476
JW Marriott Hotels & Resorts	15	_	15	7,901	<u> </u>	7,901
Renaissance Hotels & Resorts	70	3	73	25,942	1,034	26,976
Renaissance ClubSport	1	_	1	175		175
Renaissance Glubsport	408	15	423	159,336	5,590	164,926
	400	13	723	155,550	3,330	104,520
North American Limited-Service Lodging Segment (1)	600	10	700	00.750	2.047	00.000
Courtyard	693	16	709	96,759	2,847	99,606
Fairfield Inn	529	7	536	46,930	741	47,671
SpringHill Suites	176	1	177	20,445	124	20,569
Residence Inn	528	17	545	62,805	2,536	65,341
TownePlace Suites	141		141	14,122	<u> </u>	14,122
	2,067	41	2,108	241,061	6,248	247,309
International Lodging Segment (1)						
Marriott Hotels & Resorts	4	144	148	2,742	39,396	42,138
JW Marriott Hotels & Resorts	1	22	23	387	8,244	8,631
Renaissance Hotels & Resorts	_	67	67		21,783	21,783
Courtyard	_	58	58	_	11,174	11,174
Fairfield Inn	_	1	1		206	206
Residence Inn	_	1	1	_	<i>7</i> 5	75
Marriott Executive Apartments	_	18	18	_	2,905	2,905
Ramada International		2	2		332	332
	5	313	318	3,129	84,115	87,244
Luxury Lodging Segment						
The Ritz-Carlton	36	34	70	11,627	9,978	21,605
Bulgari Hotels & Resorts	_	2	2	_	117	117
The Ritz-Carlton-Residential (2)	16	1	17	1,614	93	1,707
	52	37	89	13,241	10,188	23,429
Timeshare Lodging Segment (3)						
Marriott Vacation Club	37	9	46	8,987	1,909	10,896
The Ritz-Carlton Club-Fractional	5	2	7	283	105	388
The Ritz-Carlton Club-Residential (2)	2	1	3	138	6	144
Grand Residences by Marriott-Fractional	1	1	2	199	49	248
Grand Residences by Marriott-Residential (1), (2)	1	_	1	65		65
Horizons by Marriott Vacation Club	2	_	2	444	_	444
	48	13	61	10,116	2,069	12,185
Total	2,580	419	2,999	426,883	108,210	535,093

<sup>(1)</sup> North American includes properties located in the continental United States and Canada. International includes properties located outside the continental United States and Canada.

Represents projects where we manage the related owners' association. Residential products are included once they possess a certificate of occupancy. Includes resorts that are in active sales as well as those that are sold out. Products in active sales may not be ready for occupancy. (2)

#### Revenues

(\$ in millions)	2007	2006	2005
North American Full-Service Segment	\$ 5,476	\$ 5,196	\$ 5,116
North American Limited-Service Segment	2,198	2,060	1,886
International Segment	1,594	1,411	1,017
Luxury Segment	1,576	1,423	1,333
Timeshare Segment	2,065	1,840	1,721
Total segment revenues	12,909	11,930	11,073
Other unallocated corporate	81	65	56
	\$12,990	\$11,995	\$11,129

## **Income from Continuing Operations**

(\$ in millions)	2	2007		2006		2005
North American Full-Service Segment	\$	478	\$	455	\$	349
North American Limited-Service Segment		461		380		303
International Segment		271		237		133
Luxury Segment		72		63		45
Timeshare Segment		306		280		271
Total segment financial results		1,588	1	l,415		1,101
Other unallocated corporate		(287)		(251)		(219)
Interest income, provision for loan losses and interest						
expense		(163)		(72)		(55)
Income taxes		(441)		(380)		(284)
	\$	697	\$	712	\$	543

## **Equity in Earnings (Losses) of Equity Method Investees**

(\$ in millions)	2007	2006	2005
North American Full-Service Segment	\$ 3	\$ 2	\$ 21
North American Limited-Service Segment	2	_	(6)
International Segment	3	_	20
Luxury Segment	(4)	(2)	(1)
Timeshare Segment	10	(2)	1
Total segment equity in earnings (losses)	14	(2)	35
Other unallocated corporate	1	5	1
	\$ 15	\$ 3	\$ 36

Our business includes our North American Full-Service, North American Limited-Service, International, Luxury, and Timeshare segments. We consider total segment revenues and total segment financial results to be meaningful indicators of our performance because they measure our growth in profitability and enable investors to compare the revenues and results of our operations to those of other lodging companies.

We consider RevPAR to be a meaningful indicator of our performance because it measures the period-over-period change in room revenues for comparable properties. We calculate RevPAR by dividing room sales for comparable properties by room nights available to guests for the period. RevPAR may not be comparable to similarly titled measures, such as revenues.

Company-operated house profit margin is the ratio of property-level gross operating profit (also known as house profit) to total property-level revenue. This ratio measures our overall ability as the operator to produce property-level profits by generating sales and controlling the operating expenses over which we have the most direct control. Gross operating profit includes room, food and beverage, and other revenue and the related expenses including payroll and benefits expenses, as well as repairs and maintenance, utility, general and administrative, and sales and marketing expenses. Gross operating profit does not include the impact of management fees, furniture, fixtures and equipment replacement reserves, insurance, taxes, or other fixed expenses.

#### 2007 Compared to 2006

We added 203 properties (29,200 rooms) and 55 properties (9,722 rooms) exited the system in 2007, not including residential products. We also added three residential properties (347 units).

Total segment financial results increased by \$173 million (12 percent) to \$1,588 million in 2007 from \$1,415 million in 2006, and total segment revenues increased by \$979 million to \$12,909 million in 2007, an 8 percent increase from revenues of \$11,930 million in 2006. The results, as compared to the prior year, reflect a \$204 million (17 percent) increase in combined base management, franchise, and incentive management fees from \$1,224 million in 2006 to \$1,428 million in 2007, a \$16 million increase in earnings associated with equity investments, and a \$1 million minority interest benefit. Partially offsetting these favorable variances was a decrease of \$18 million in owned, leased, corporate housing, and other revenue net of direct expenses, \$13 million of increased general, administrative, and other expenses, a decrease of \$10 million in gains and other income, and a decrease of \$7 million in timeshare sales and services revenue net of direct expenses.

Higher RevPAR for comparable rooms, resulting from both domestic and international rate increases, higher property-level food and beverage and other revenue, and new unit growth, drove the increase in base management and franchise fees. In 2006, we recognized \$5 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006, versus no similar fees in 2007. Incentive management fees increased by \$88 million (31 percent) during 2007, reflecting the impact of strong room rate increases and property-level margin improvements associated with productivity improvements. Incentive management fees included \$17 million and \$10 million for 2007 and 2006, respectively, that were calculated based on prior periods' earnings, but not earned and due until the periods in which they were recognized. Furthermore, incentive management fees for 2007 also included \$13 million of business interruption insurance proceeds associated with hurricanes in prior years and base management fees included \$6 million and \$1 million of business interruption insurance proceeds, also associated with hurricanes in prior years, received in 2007 and 2006, respectively. In 2007, 67 percent of our managed properties paid incentive management fees to us versus 62 percent in 2006.

Systemwide RevPAR, which includes data from our franchised properties, in addition to our owned, leased, and managed properties, for comparable North American properties increased by 6.0 percent over 2006, and RevPAR for our comparable North American company-operated properties increased by 6.2 percent over 2006.

Systemwide RevPAR for comparable international properties increased by 8.4 percent, and RevPAR for comparable international company-operated properties increased by 8.9 percent. Worldwide RevPAR for comparable systemwide properties increased by 6.5 percent (7.6 percent using actual dollars) while worldwide RevPAR for comparable company-operated properties increased by 7.0 percent.

As compared to 2006, 2007 worldwide company-operated house profit margins increased by 150 basis points, while North American company-operated house profit margins improved by 160 basis points versus 2006. For 2007, as compared to 2006, house profit per available room ("HP-PAR") at our full-service managed properties in North America increased by 10.6 percent. HP-PAR at our North American limited-service managed properties increased by 8.9 percent, and worldwide HP-PAR for all our brands increased by 11.0 percent on a constant dollar basis.

## 2006 Compared to 2005

We added 136 properties (23,466 rooms) and 45 properties (8,616 rooms) left the system in 2006. Most of the properties that left the system were limited-service properties.

Total segment financial results increased by \$314 million (29 percent) to \$1,415 million in 2006 from \$1,101 million in 2005, and total segment revenues increased by \$857 million to \$11,930 million in 2006, an 8 percent increase from revenues of \$11,073 million in 2005. The results as compared to the prior year reflect a \$197 million (19 percent) increase in combined base management, franchise, and incentive management fees from \$1,027 million in 2005 to \$1,224 million in 2006, \$124 million of lower general, administrative, and other expenses, \$8 million of increased owned, leased, corporate housing, and other revenue net of direct expenses, a \$98 million increase in timeshare sales and services revenue net of direct expenses, and an increase in minority interest of \$2 million, partially offset by \$78 million of lower gains and other income and a \$37 million decline in earnings associated with equity investments. Higher RevPAR for comparable rooms, resulting from both domestic and international rate increases, higher property-level food and beverage and other revenue, and new unit growth drove the increase in base management and franchise fees. In 2006, we recognized \$5 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006. Incentive management fees increased by \$80 million (40 percent) during 2006, reflecting the impact of strong room rate improvement and property-level margin improvements. Incentive management fees include \$10 million and \$14 million for 2006 and 2005,

respectively, that were calculated based on prior periods' earnings, but not earned and due until the periods in which they were recognized. In 2006, 62 percent of our managed properties paid incentive management fees to us versus 51 percent in 2005.

Systemwide RevPAR for comparable North American properties increased by 9.1 percent over 2005 and RevPAR for our comparable North American company-operated properties increased by 8.9 percent.

Systemwide RevPAR for comparable international properties, increased by 10.5 percent, and RevPAR for comparable international company-operated properties increased by 11.1 percent. Worldwide RevPAR for comparable systemwide properties increased by 9.4 percent while worldwide RevPAR for comparable company-operated properties increased by 9.5 percent.

In addition, worldwide comparable company-operated house profit margins increased by 230 basis points, while North American company-operated house profit margins improved by 240 basis points versus 2005. For 2006, as compared to 2005, HP-PAR at our full-service managed properties in North America increased by 16.3 percent. HP-PAR at our North American limited-service managed properties increased by 12.6 percent, and worldwide HP-PAR for all our brands increased by 15.4 percent on a constant dollar basis.

#### **Lodging Development**

We opened 203 properties, totaling 29,200 rooms, across our brands in 2007 and 55 properties (9,722 rooms) left the system, not including residential products. We also added three residential properties (347 units). Highlights of the year included:

- Converting 26 properties (5,618 rooms), or 18 percent of our gross room additions for the year, from other brands, and 21 percent of the rooms were located in international markets;
- Opening approximately 16 percent of all the new rooms outside the United States;
- Adding 156 properties (17,517 rooms) to our North American Limited-Service brands; and
- Opening two new Marriott Vacation Club properties in the West Indies on the island of St. Kitts, and the U.S. Virgin Islands on the island of St. Thomas, as well as one new Ritz-Carlton Club and residences property in San Francisco, California.

We currently have more than 125,000 hotel rooms under construction, awaiting conversion, or approved for development in our hotel development pipeline and we expect to add over 30,000 hotel rooms and timeshare units to our system in 2008. We expect to remove approximately 5,000 rooms from our system during the 2008 full year.

We believe that we have access to sufficient financial resources to finance our growth, as well as to support our ongoing operations and meet debt service and other cash requirements. Nonetheless, our ability to sell properties that we develop and the ability of hotel developers to build or acquire new Marriott-branded properties, both of which are important parts of our growth plan, depend in part on capital access, availability and cost for other hotel developers and third-party owners. These growth plans are subject to numerous risks and uncertainties, many of which are outside of our control. See the "Forward-Looking Statements" and "Risks and Uncertainties" captions earlier in this report and the "Liquidity and Capital Resources" caption later in this report.

Statistics. The following tables show occupancy, average daily rate, and RevPAR for comparable properties, for each of the brands in our North American Full-Service and North American Limited-Service segments, for our International segment by region, and the principal brand in our Luxury segment, The Ritz-Carlton. We have not presented statistics for company-operated Fairfield Inn properties in these tables because we operate only a limited number of properties, as the brand is predominantly franchised and such information would not be meaningful (identified as "nm" in the tables that follow). Systemwide statistics include data from our franchised properties, in addition to our owned, leased, and managed properties.

The occupancy, average daily rate, and RevPAR statistics used throughout this report for 2007 include the period from December 30, 2006, through December 28, 2007, the statistics for 2006 include the period from December 31, 2005, through December 29, 2006, and the statistics for 2005 include the period from January 1, 2005, through December 30, 2005 (except in each case, for The Ritz-Carlton brand properties and properties located outside of the continental United States and Canada, which for them includes the period from January 1 through December 31 for each year).

		Comparable Company-Operated North American Properties <sup>(1)</sup>		Comparable Systemwide North American Properties (1)		
	2007	Change vs.	2006	2007	Change vs	s. 2006
Marriott Hotels & Resorts (2)	<b>TO</b> 00/	1.00/		<b>=</b> 0.007	0.607	
Occupancy	72.6%	1.0%	pts.	70.8%	0.6%	pt
Average Daily Rate	\$ 175.41	5.4%		\$160.61	5.5%	
RevPAR	\$ 127.43	6.9%		\$113.66	6.4%	
Renaissance Hotels & Resorts						
Occupancy	73.1%	0.4%	pts.	71.8%	-0.4%	pt
Average Daily Rate	\$ 169.93	5.4%		\$157.29	5.8%	
RevPAR	\$ 124.17	5.9%		\$112.96	5.1%	
Composite North American Full-Service (3)						
Occupancy	72.7%	0.9%	pts.	70.9%	0.5%	pt
Average Daily Rate	\$ 174.54	5.4%		\$160.10	5.5%	
RevPAR	\$ 126.92	6.8%		\$113.56	6.2%	
The Ritz-Carlton North America						
Occupancy	72.3%	0.1%	pts.	72.3%	0.1%	pt
Average Daily Rate	\$ 331.48	7.3%	•	\$331.48	7.3%	
RevPAR	\$ 239.67	7.5%		\$239.67	7.5%	
Composite North American Full-Service and Luxury (4)						
Occupancy	72.7%	0.8%	pts.	71.0%	0.4%	pt
Average Daily Rate	\$ 189.41	5.7%	pts.	\$169.92	5.7%	ρι
RevPAR	\$ 137.66	6.9%		\$100.65	6.4%	
	\$ 157.00	0.570		Ψ120.05	0.170	
Residence Inn	77. 70/	0.50/		70.20/	0.00/	
Occupancy	77.7%	-0.5%	pts.	78.2%	-0.8%	pt
Average Daily Rate RevPAR	\$ 124.24 \$ 96.53	4.6% 3.9%		\$122.44 \$ 95.80	6.1% 5.1%	
REVPAR	\$ 90.33	3.970		\$ 95.60	3.170	
Courtyard						
Occupancy	70.4%	-0.4%	pts.	72.1%	-0.2%	pt
Average Daily Rate	\$ 127.34	5.6%		\$124.12	5.9%	
RevPAR	\$ 89.69	4.9%		\$ 89.53	5.6%	
Fairfield Inn						
Occupancy	nm	nm		70.5%	-0.3%	pt
Average Daily Rate	nm	nm		\$ 88.19	7.2%	
RevPAR	nm	nm		\$ 62.17	6.7%	
TownePlace Suites						
Occupancy	74.2%	-1.1%	pts.	73.5%	-2.4%	pt
Average Daily Rate	\$ 85.65	8.9%	•	\$ 86.93	8.4%	
RevPAR	\$ 63.56	7.2%		\$ 63.89	5.0%	
SpringHill Suites						
Occupancy	72.6%	0.6%	pts.	73.2%	-0.6%	pt
Average Daily Rate	\$ 107.86	4.2%	pto.	\$106.49	6.5%	Pt
RevPAR	\$ 78.27	5.0%		\$ 77.97	5.7%	
Composite North American Limited-Service (5) Occupancy	72.7%	-0.4%	ptc	73.6%	-0.5%	n+
Average Daily Rate	\$ 122.63	-0.4% 5.4%	pts.	\$113.34	6.3%	pt
RevPAR	\$ 122.03	4.8%		\$ 83.37	5.6%	
	ψ 05.10	1.070		ψ 00.07	3.070	
Composite North American (6)	E0 E0/	0.207		FD 604	0.207	
Occupancy Average Deily Bets	72.7%	0.3%	pts.	72.6%	-0.2%	pt
Average Daily Rate	\$ 159.01	5.8%		\$134.62	6.2%	
RevPAR	\$ 115.60	6.2%		\$ 97.70	6.0%	

Statistics are for the fifty-two weeks ended December 28, 2007, and December 29, 2006, except for Ritz-Carlton for which the statistics are for the twelve months ended December 31, 2007, and December 31, 2006. For properties located in Canada the comparison to 2006 is on a constant U.S. dollar basis.

<sup>(2)</sup> 

Marriott Hotels & Resorts includes JW Marriott Hotels & Resorts.

Composite North American Full-Service statistics include properties located in the continental United States and Canada for Marriott Hotels & Resorts and (3) Renaissance Hotels & Resorts.

Composite North American Full-Service and Luxury includes Marriott Hotels & Resorts, Renaissance Hotels & Resorts, and Ritz-Carlton.
Composite North American Limited-Service statistics include properties located in the continental United States and Canada for Residence Inn, Courtyard, (5) Fairfield Inn, TownePlace Suites, and SpringHill Suites.

Composite North American statistics include properties located in the continental United States and Canada for Marriott Hotels & Resorts, Renaissance Hotels & Resorts, Residence Inn, Courtyard, Fairfield Inn, TownePlace Suites, SpringHill Suites, and The Ritz-Carlton.

		Comparable Company-Operated Properties (1)		Comparable Systemwide Properties <sup>(1)</sup>	
	2007	Change vs. 2006	2007	Change vs	s. 2006
Caribbean and Latin America (2)					
Occupancy	76.5%	2.4% pts	. 74.8%	2.8%	pts.
Average Daily Rate	\$ 167.56	9.2%	\$156.76	7.5%	1
RevPAR	\$ 128.25	12.7%	\$117.20	11.7%	
Continental Europe (2)					
Occupancy	74.4%	1.3% pts	. 72.0%	1.2%	pts.
Average Daily Rate	\$ 173.92	5.8%	\$174.93	5.4%	
RevPAR	\$ 129.34	7.7%	\$126.01	7.3%	
United Kingdom (2)					
Occupancy	77.8%	0.1% pts	. 77.4%	0.3%	pts.
Average Daily Rate	\$ 203.27	5.2%	\$200.65	4.7%	
RevPAR	\$ 158.08	5.2%	\$155.27	5.1%	
Middle East and Africa (2)					
Occupancy	73.3%	4.8% pts	. 72.4%	4.9%	pts.
Average Daily Rate	\$ 135.74	9.1%	\$133.98	9.0%	
RevPAR	\$ 99.57	16.8%	\$ 96.95	17.0%	
Asia Pacific (2), (3)					
Occupancy	75.2%	-0.8% pts	. 75.3%	-0.5%	pts.
Average Daily Rate	\$ 147.79	8.9%	\$148.67	7.7%	
RevPAR	\$ 111.15	7.8%	\$111.92	7.0%	
Regional Composite (4), (5)					
Occupancy	75.7%	0.8% pts	. 74.6%	1.0%	pts.
Average Daily Rate	\$ 168.30	7.0%	\$166.03	6.3%	
RevPAR	\$ 127.44	8.2%	\$123.78	7.8%	
International Luxury (6)					
Occupancy	72.7%	4.0% pts	. 72.7%	4.0%	pts.
Average Daily Rate	\$ 292.24	6.1%	\$292.24	6.1%	
RevPAR	\$ 212.54	12.2%	\$212.54	12.2%	
Total International (7)					
Occupancy	75.4%	1.2% pts	. 74.4%	1.3%	pts.
Average Daily Rate	\$ 180.73	7.2%	\$176.57	6.6%	
RevPAR	\$ 136.29	8.9%	\$131.36	8.4%	

Financial results for all properties are reported on a period-end basis, while statistics for properties located outside the continental United States and Canada are reported on a month-end basis. The statistics are for January through December. For the properties located in countries that use currencies other than the U.S. dollar, the comparison to 2006 is on a constant U.S. dollar basis.

Regional information includes Marriott Hotels & Resorts, Renaissance Hotels & Resorts, and Courtyard properties located outside of the continental United States and Canada.

Excludes Hawaii. (3)

Includes Hawaii. (4)

<sup>(5)</sup> Regional Composite statistics include all properties located outside of the continental United States and Canada for Marriott Hotels & Resorts, Renaissance Hotels & Resorts, and Courtyard.

Includes The Ritz-Carlton properties located outside of North America and Bulgari Hotels & Resorts.

Total International includes Regional Composite statistics and statistics for The Ritz-Carlton International and Bulgari Hotels & Resorts. (7)

		Comparable Company-Operated Properties (1)  2007 Change vs. 2006		Comparable Systemwide Properties (1)		ide
	2007			2007	Change vs. 2006	
Composite Luxury (2)						
Occupancy	72.5%	1.8%	pts.	72.5%	1.8%	pts.
Average Daily Rate	\$ 314.36	6.6%		\$314.36	6.6%	
RevPAR	\$ 227.87	9.4%		\$227.87	9.4%	
Total Worldwide (3)						
Occupancy	73.5%	0.5%	pts.	72.9%	0.1%	pts.
Average Daily Rate	\$ 165.19	6.2%		\$141.60	6.4%	
RevPAR	\$ 121.34	7.0%		\$103.19	6.5%	

<sup>(1)</sup> Financial results for all properties are reported on a period-end basis, while statistics for properties located outside the continental United States and Canada are reported on a month-end basis. For the properties located in countries that use currencies other than the U.S. dollar, the comparison to 2006 is on a constant U.S. dollar basis.

(2) Composite Luxury includes worldwide properties for The Ritz-Carlton and Bulgari Hotels & Resorts.

Total Worldwide statistics include all properties worldwide for Marriott Hotels & Resorts, Renaissance Hotels & Resorts, Residence Inn, Courtyard, Fairfield Inn, TownePlace Suites, SpringHill Suites, and The Ritz-Carlton. Statistics for properties located in the continental United States and Canada (except for The Ritz-Carlton) represent the fifty-two weeks ended December 28, 2007, and December 29, 2006. Statistics for all The Ritz-Carlton properties and properties located outside of the continental United States and Canada represent the twelve months ended December 31, 2007, and December 31, 2006.

		Company-Operated rican Properties (1) Change vs. 2005		nble Systemwide rican Properties <sup>(1)</sup> Change vs. 2005	
Marriott Hotels & Resorts (2)					
Occupancy	72.3%	-0.6% pts	. 70.9%	0.2%	pts
Average Daily Rate	\$ 168.11	9.1%	\$154.37	8.3%	
RevPAR	\$ 121.58	8.3%	\$109.48	8.6%	
Renaissance Hotels & Resorts					
Occupancy	72.8%	0.6% pts	. 72.2%	1.1%	pts
Average Daily Rate	\$ 162.96	8.7%	\$151.91	8.4%	
RevPAR	\$ 118.57	9.6%	\$109.75	10.0%	
Composite North American Full-Service (3)					
Occupancy	72.4%	-0.4% pts	. 71.1%	0.3%	pts
Average Daily Rate	\$ 167.27	9.0%	\$153.99	8.3%	
RevPAR	\$ 121.10	8.5%	\$109.52	8.9%	
The Ritz-Carlton North America					
Occupancy	72.9%	2.1% pts	. 72.9%	2.1%	pts.
Average Daily Rate	\$ 307.20	7.1%	\$307.20	7.1%	Pto
RevPAR	\$ 223.88	10.3%	\$223.88	10.3%	
C					
Composite North American Full-Service and Luxury (4) Occupancy	72.4%	-0.1% pts	. 71.2%	0.5%	nte
Average Daily Rate	\$ 180.17	-0.1% pts	\$163.03	8.3%	pts.
RevPAR	\$ 100.17 \$ 130.52	8.7%	\$105.05	9.0%	
	\$ 150.52	0.7 70	\$110.11	3.070	
Residence Inn	<b>=0</b> 00/	4.00/	<b>=0</b> 00/	0.50/	
Occupancy	78.3%	-1.8% pts		-0.5%	pts.
Average Daily Rate	\$ 117.99	9.2%	\$113.85	8.2%	
RevPAR	\$ 92.35	6.8%	\$ 90.15	7.6%	
Courtyard					
Occupancy	70.9%	-0.4% pts		0.1%	pts.
Average Daily Rate	\$ 119.30	10.9%	\$116.67	9.5%	
RevPAR	\$ 84.62	10.3%	\$ 84.57	9.7%	
Fairfield Inn					
Occupancy	nm	nm	70.7%	1.0%	pts.
Average Daily Rate	nm	nm	\$ 82.05	9.1%	
RevPAR	nm	nm	\$ 58.01	10.6%	
TownePlace Suites					
Occupancy	75.3%	-0.3% pts	. 75.7%	-0.2%	pts.
Average Daily Rate	\$ 78.68	10.9%		10.2%	•
RevPAR	\$ 59.28	10.4%	\$ 60.35	9.9%	
SpringHill Suites					
Occupancy	72.3%	-2.0% pts	. 74.1%	0.2%	pts.
Average Daily Rate	\$ 102.86	10.7%	\$ 98.76	9.8%	Pto.
RevPAR	\$ 74.42	7.8%	\$ 73.16	10.0%	
Composite North American Limited-Service (5)					
Occupancy	73.3%	-0.8% pts	. 74.2%	0.1%	pts.
Average Daily Rate	\$ 115.24	10.4%	\$105.65	9.0%	pts.
RevPAR	\$ 84.41	9.1%	\$ 78.34	9.2%	
	Ψ 0-1,-+1	J.170	Ψ , υ.υ	J. Z / U	
Composite North American (6)		0.407		0.507	
Occupancy	72.8%	-0.4% pts		0.3%	pts.
Average Daily Rate	\$ 152.14	9.5%	\$128.07	8.7%	
RevPAR	\$ 110.74	8.9%	\$ 93.47	9.1%	

Statistics are for the fifty-two weeks ended December 29, 2006, and December 30, 2005, except for Ritz-Carlton for which the statistics are for the twelve months ended December 31, 2006, and December 31, 2005. For properties located in Canada the comparison to 2005 is on a constant U.S. dollar basis.

<sup>(2)</sup> Marriott Hotels & Resorts includes JW Marriott Hotels & Resorts.

<sup>(3)</sup> Composite North American Full-Service statistics include properties located in the continental United States and Canada for Marriott Hotels & Resorts and Renaissance Hotels & Resorts.

<sup>(4)</sup> Composite North American Full-Service and Luxury includes Marriott Hotels & Resorts, Renaissance Hotels & Resorts, and Ritz-Carlton.

<sup>(5)</sup> Composite North American Limited-Service statistics include properties located in the continental United States and Canada for Residence Inn, Courtyard, Fairfield Inn, TownePlace Suites, and SpringHill Suites.

<sup>(6)</sup> Composite North American statistics include properties located in the continental United States and Canada for Marriott Hotels & Resorts, Renaissance Hotels & Resorts, Residence Inn, Courtyard, Fairfield Inn, TownePlace Suites, SpringHill Suites, and The Ritz-Carlton.

		Company-Ope operties <sup>(1)</sup>	rated		ble Systemwi perties <sup>(1)</sup>	de
	2006	Change vs.	2005	2006	Change vs	. 2005
Caribbean and Latin America (2)	T4.00/	4.50/		<b>72.00</b> /	0.50/	
Occupancy	74.9%	1.5%	pts.	73.0%	0.5%	pts.
Average Daily Rate	\$ 159.93	9.7%		\$150.93	9.2%	
RevPAR	\$ 119.81	12.0%		\$110.11	9.9%	
Continental Europe (2)						
Occupancy	72.6%	2.0%	pts.	70.7%	1.9%	pts.
Average Daily Rate	\$ 147.28	6.6%	_	\$150.58	7.2%	_
RevPAR	\$ 106.95	9.6%		\$106.53	10.2%	
United Kingdom (2)						
Occupancy	79.6%	3.5%	pts.	75.1%	3.1%	pts.
Average Daily Rate	\$ 225.38	9.5%		\$204.99	8.6%	
RevPAR	\$ 179.44	14.6%		\$153.94	13.3%	
Middle East and Africa (2)						
Occupancy	68.9%	-0.3%	pts.	69.0%	-0.7%	pts.
Average Daily Rate	\$ 143.12	11.0%		\$134.95	11.5%	
RevPAR	\$ 98.58	10.5%		\$ 93.05	10.3%	
Asia Pacific (2), (3)						
Occupancy	75.8%	1.0% pts.		76.2%	0.9%	pts.
Average Daily Rate	\$ 127.09	11.3%		\$129.26	9.8%	
RevPAR	\$ 96.28	12.7%		\$ 98.46	11.2%	
Regional Composite (4), (5)						
Occupancy	74.6%	1.3%	pts.	73.6%	1.1%	pts.
Average Daily Rate	\$ 148.13	9.5%		\$147.12	9.0%	
RevPAR	\$ 110.53	11.4%		\$108.32	10.7%	
International Luxury (6)						
Occupancy	71.7%	-0.1%	pts.	71.7%	-0.1%	pts.
Average Daily Rate	\$ 241.90	9.2%		\$241.90	9.2%	
RevPAR	\$ 173.35	9.1%		\$173.55	9.1%	
Total International <sup>(7)</sup>						
Occupancy	74.4%	1.2%	pts.	73.5%	1.0%	pts.
Average Daily Rate	\$ 153.99	9.4%		\$152.02	9.0%	
RevPAR	\$ 114.61	11.1%		\$111.78	10.5%	

Financial results for all properties are reported on a period-end basis, while statistics for properties located outside the continental United States and Canada are reported on a month-end basis. The statistics are for January through December. For the properties located in countries that use currencies other than the U.S. dollar, the comparison to 2005 is on a constant U.S. dollar basis.

(2) Regional information includes Marriott Hotels & Resorts, Renaissance Hotels & Resorts, and Courtyard properties located outside of the continental United States and Canada.

(6)

<sup>(3)</sup> 

Excludes Hawaii. Includes Hawaii. (4)

Regional Composite statistics include all properties located outside of the continental United States and Canada for Marriott Hotels & Resorts, Renaissance (5) Hotels & Resorts, and Courtyard.

Includes The Ritz-Carlton properties located outside of North America and Bulgari Hotels & Resorts.

Total International includes Regional Composite statistics and statistics for The Ritz-Carlton International and Bulgari Hotels & Resorts.

<sup>(7)</sup> 

		Comparable Company-Operated Properties <sup>(1)</sup>			Comparable Systemwic Properties <sup>(1)</sup>	
	2006	Change v	s. 2005	2006	Change v	s. 2005
Composite Luxury (2)						
Occupancy	72.5%	1.4%	pts.	72.5%	1.4%	pts.
Average Daily Rate	\$ 289.50	7.5%		\$289.50	7.5%	
RevPAR	\$ 209.88	9.7%		\$209.88	9.7%	
Total Worldwide (3)						
Occupancy	73.2%	0.0%	pts.	73.1%	0.4%	pts.
Average Daily Rate	\$ 152.63	9.5%		\$131.92	8.8%	
RevPAR	\$ 111.75	9.5%		\$ 96.39	9.4%	

<sup>(1)</sup> Financial results for all properties are reported on a period-end basis, while statistics for properties located outside the continental United States and Canada are reported on a month-end basis. For the properties located in countries that use currencies other than the U.S. dollar, the comparison to 2005 is on a constant U.S. dollar basis.

(2) Composite Luxury includes worldwide properties for The Ritz-Carlton and Bulgari Hotels & Resorts.

Total Worldwide statistics include all properties worldwide for Marriott Hotels & Resorts, Renaissance Hotels & Resorts, Residence Inn, Courtyard, Fairfield Inn, TownePlace Suites, SpringHill Suites, and The Ritz-Carlton. Statistics for properties located in the continental United States and Canada (except for The Ritz-Carlton) represent the fifty-two weeks ended December 29, 2006, and December 30, 2005. Statistics for all The Ritz-Carlton properties and properties located outside of the continental United States and Canada represent the twelve months ended December 31, 2006, and December 31, 2005.

**North American Full-Service Lodging** includes *Marriott Hotels & Resorts*, *Marriott Conference Centers*, *JW Marriott Hotels & Resorts*, Renaissance *Hotels & Resorts*, and *Renaissance ClubSport*.

				Annual (	Change
(\$ in millions)	2007	2006	2005	2007/2006	2006/2005
Segment revenues	\$5,476	\$5,196	\$5,116	5%	2%
Segment results	\$ 478	\$ 455	\$ 349	5%	30%

# 2007 Compared to 2006

In 2007, across our North American Full-Service Lodging segment, we added 13 properties (3,947 rooms) and six properties (2,853 rooms) left the system.

In 2007, RevPAR for comparable company-operated North American full-service properties increased by 6.8 percent to \$126.92. Occupancy for these properties increased by 0.9 percentage points to 72.7 percent, and average daily rates increased by 5.4 percent to \$174.54.

The \$23 million increase in segment results compared to 2006 reflects a \$65 million increase in base management, incentive management, and franchise fees, partially offset by a \$22 million decline in gains and other income and a \$20 million decrease in owned, leased, and other revenue net of direct expenses.

The \$65 million increase in fees was largely due to stronger RevPAR and property-level margins, driven primarily by rate increases and productivity improvements. Incentive management fees for 2007 reflect the receipt of \$12 million and base management fees for 2007 reflect the receipt of \$2 million, both of which were for business interruption insurance proceeds associated with Hurricane Katrina, and no similar fees in 2006. Additionally, 2006 reflected the recognition of \$3 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006, compared to the recognition of no similar fees in 2007.

Gains and other income was \$22 million lower in 2007, as compared to 2006, and reflected a \$37 million charge in 2006 associated with a straight-line rent receivable, partially offset by the redemption of preferred stock in a cost method investee that generated income of \$25 million in 2006 and \$34 million of lower other gains in 2007, as compared to 2006. The \$34 million decrease in other gains compared to the prior year reflects gains in 2006 associated with the sale of joint venture investments and real estate, while 2007 reflects limited similar activity. Owned, leased, and other revenue net of direct expenses decreased by \$20 million and reflected properties sold and \$10 million of hotel management and franchise agreement termination fees received in 2006, compared to \$3 million of such fees received in 2007. General, administrative, and other expenses for 2007 were unchanged as compared to 2006 primarily reflecting \$5 million of hotel management agreement performance cure payments in 2006 offset by a guarantee charge in 2007 associated with one property and the write-off of a deferred contract acquisition cost associated with another property that left the system.

# 2006 Compared to 2005

In 2006, across our North American Full-Service Lodging segment, we added 15 properties (4,971 rooms) and six properties (1,604 rooms) left the system.

In 2006, RevPAR for comparable company-operated North American full-service properties increased by 8.5 percent to \$121.10. Occupancy for these properties decreased by 0.4 percentage points, and average daily rates increased by 9.0 percent to \$167.27.

Compared to the prior year, our 2006 results reflect a \$59 million increase in base management, incentive management, and franchise fees. The increase in fees is largely due to stronger RevPAR, driven primarily by rate increases and, to a lesser extent, higher food and beverage, meeting room rental and other revenue, and productivity improvements, all of which favorably impacted property-level house profit margins. The growth in the number of rooms, year-over-year, also contributed to the increase in fees. Incentive management fees included \$10 million for 2005 that were calculated based on prior periods' earnings but not earned and due until they were recognized. Similarly, base management fees for 2006 included \$4 million of fees that were calculated based on prior periods' results, but not earned and due until 2006. Owned, leased, and other revenue net of direct expenses decreased by \$8 million primarily as a result of properties sold in 2006 and the receipt in 2005 of a \$10 million hotel agreement termination fee associated with one property, partially offset by the receipt in 2006 of \$10 million of hotel management and franchise agreement termination fees.

General, administrative, and other expenses decreased by \$69 million as a result of, among other things, a \$60 million charge in 2005 associated with the CTF transaction, more fully discussed in the previous "Operating Income" discussion, as well as expenses of \$14 million in 2005 related to our bedding incentive program, partially offset by increased expenses in 2006 reflecting costs related to unit growth and development, systems improvements, and increases in ordinary costs such as wages and benefits. In 2005, general, administrative, and other expenses included \$3 million of hotel management agreement performance cure payments associated with one property. In 2006, general, administrative, and other expenses included a \$5 million hotel management agreement performance cure payment.

Gains and other income was \$3 million higher than the prior year and reflected the redemption of preferred stock in a cost method investee that generated income of \$25 million in 2006 and \$15 million of higher net gains and other income in 2006 reflecting gains in 2006 associated with the sale of joint venture investments and real estate that were partially offset by lower gains in 2006 associated with the sale or repayment before maturity of loans receivable associated with several properties. Gains and other income for 2006 reflected a \$37 million non-cash charge to adjust the carrying amount to net realizable value associated with land we own and lease, as further described in the "Expected Land Sale" caption in the "Management's Discussion and Analysis of Financial Condition and Results of Operations" section of our 2006 Form 10-K. Equity results decreased by \$19 million versus the prior year and reflected the recognition in 2005 of \$16 million in equity earnings from two joint ventures as a result of the ventures' sale of hotels and our sale of some joint venture investments in 2005 and 2006, offset to some extent by improved equity joint venture results reflecting the stronger demand environment.

North American Limited-Service Lodging includes Courtyard, Fairfield Inn, SpringHill Suites, Residence Inn, TownePlace Suites, and Marriott ExecuStay.

				Annual (	Change
(\$ in millions)	2007	2006	2005	2007/2006	2006/2005
Segment revenues	\$2,198	\$2,060	\$1,886	7%	9%
Segment results	\$ 461	\$ 380	\$ 303	21%	25%

#### 2007 Compared to 2006

Across our North American Limited-Service Lodging segment, we added 156 properties (17,517 rooms) and 16 properties (1,853 rooms) left the system in 2007. The properties that left the system were primarily associated with our Fairfield Inn brand.

In 2007, RevPAR for comparable company-operated North American limited-service properties increased by 4.8 percent to \$89.18. Occupancy for these properties decreased by 0.4 percentage points to 72.7 percent, and average daily rates increased by 5.4 percent to \$122.63.

The \$81 million increase in segment results, as compared to 2006, primarily reflects an \$83 million increase in base management, incentive management, and franchise fees, a \$5 million increase in owned, leased, and other revenue net of direct expenses, a \$2 million increase in gains and other income, and a \$2 million increase in joint venture equity earnings, partially offset by \$11 million of higher general, administrative, and other expenses.

In 2007, we recognized \$15 million of incentive management fees that were calculated based on prior years' results but not earned and due until 2007 as compared to the recognition of no similar fees in 2006. The additional increase in fees is largely due to higher RevPAR, driven by rate increases, which increased base management and franchise fees, and to productivity improvements, which increased property-level margins and incentive management fees. Additionally, growth in the number of rooms contributed to the increase in base management and franchise fees. The \$5 million increase in owned, leased, and other revenue net of direct expenses is primarily a result of hotel franchise agreement termination fees of \$6 million received in 2007, which were associated with properties, primarily our Fairfield Inn brand, that left our system and \$1 million of similar fees in 2006. The \$11 million increase in general, administrative, and other expenses in 2007 primarily reflects the impact of unit growth and development and the write-off of \$3 million of deferred contract acquisition costs.

# 2006 Compared to 2005

Across our North American Limited-Service Lodging segment, we added 91 properties (11,329 rooms) and 28 properties (3,647 rooms) left the system in 2006. The properties that left the system were primarily associated with our Fairfield Inn brand.

In 2006, RevPAR for comparable company-operated North American limited-service properties increased by 9.1 percent to \$84.41. Occupancy for these properties decreased by 0.8 percentage points to 73.3 percent, and average daily rates increased by 10.4 percent to \$115.24.

The \$77 million increase in segment results for 2006 primarily reflects a \$77 million increase in base management, incentive management, and franchise fees, \$27 million of lower general, administrative, and other expenses, and \$6 million of increased equity results, partially offset by a \$14 million decrease in owned, leased, and other revenue net of direct expenses and a \$19 million decrease in gains and other income. The increase in fees is largely due to higher RevPAR, driven by rate increases, which impacted property-level house profits and, to a lesser extent, to 2006 productivity improvements and the growth in the number of rooms. Stronger performance at our renovated Courtyard properties, versus nonrenovated properties, also contributed to the increase in segment results over the prior year. The decrease in owned, leased, and other revenue net of direct expenses reflects lower lease revenue as a result of our sale, late in 2005, of a portfolio of land underlying 75 Courtyard hotels, partially offset by improved owned and leased results in 2006 reflecting stronger demand.

The decrease in general, administrative, and other expenses of \$27 million is attributable to the recognition of expenses in 2005 totaling \$11 million associated with our bedding incentive program, a \$6 million litigation charge and a \$3 million guarantee charge. Improved equity results in 2006 versus 2005 reflect the impact of a stronger demand environment in 2006, new joint ventures and the impact of owning a 50 percent interest in the Courtyard Joint Venture through the first quarter of 2005 versus owning a 21 percent interest thereafter. For additional information regarding the Courtyard Joint Venture, see the "Courtyard Joint Venture" caption in the "Liquidity and Capital Resources" section in "Management's Discussion and Analysis of Financial Condition and Results of Operations" of our 2005 Form 10-K. Gains and other income decreased by \$19 million in 2006 as a result of the 2005 sale of a portfolio of land that generated gains and other income of \$17 million, a \$10 million gain in 2005 associated with the repayment, before maturity, to us of the loan we made to the Courtyard Joint Venture, partially offset by higher other real estate and other gains of \$4 million in 2006 and increased income of \$4 million in 2006 associated with cost method joint ventures.

**International Lodging** includes International Marriott Hotels & Resorts, International JW Marriott Hotels & Resorts, International Renaissance Hotels & Resorts, International Courtyard, International Fairfield Inn, International Residence Inn, Ramada International, and Marriott Executive Apartments.

				Annual (	Change
(\$ in millions)	2007	2006	2005	2007/2006	2006/2005
Segment revenues	\$1,594	\$1,411	\$1,017	13%	39%
Segment results	\$ 271	\$ 237	\$ 133	14%	78%

# 2007 Compared to 2006

Across our International Lodging segment, we added 20 properties (4,686 rooms) and 31 properties (4,678 rooms) left the system in 2007.

In 2007, RevPAR for comparable company-operated international properties increased by 8.2 percent to \$127.44. Occupancy for these properties increased by 0.8 percentage points to 75.7 percent, and average daily rates increased by 7.0 percent to \$168.30. Results for our international operations were strong across most regions. RevPAR increases as compared to the prior year were particularly strong in Central and South East Asia, South America, the Middle East, and Eastern Europe.

The \$34 million increase in segment results in 2007 as compared to 2006, reflects a \$27 million increase in base management, incentive management, and franchise fees, an \$11 million increase in gains and other income, and a \$3 million increase in joint venture equity earnings, partially offset by \$7 million of higher general, administrative, and other expenses. Incentive management fees included \$2 million for 2007 that were calculated based on prior periods' earnings, but not earned and due until recognized, compared to \$10 million of similar fees in 2006.

The increase in fees is largely due to strong demand and higher RevPAR, driven by rate increases and favorable exchange rates. The \$11 million increase in gains and other income reflects higher gains in 2007 on real estate sales, as compared to the year-ago period, offset by lower gains on the sale of joint ventures. The \$7 million increase in general, administrative, and other expenses from 2006 is primarily attributable to costs related to our unit growth and development. Owned, leased, and other revenue net of direct expenses remained flat primarily reflecting hotel management and franchise agreement termination fees of \$8 million received in 2007 and \$12 million of similar fees in 2006, entirely offset by the improved performance of our owned and leased properties in 2007.

#### 2006 Compared to 2005

Across our International Lodging segment, we added 22 properties (5,242 rooms) and 10 properties (3,136 rooms) left the system in 2006.

In 2006, RevPAR for comparable company-operated international properties increased by 11.1 percent to \$114.61. Occupancy for these properties increased by 1.2 percentage points to 74.4 percent, and average daily rates increased by 9.4 percent to \$153.99. Results for our international operations were strong across most regions. China, Mexico, Australia, Germany, France, the United Kingdom, and certain Middle Eastern countries all had strong RevPAR increases.

The \$104 million increase in segment results for 2006 primarily reflects a \$43 million increase in base management, incentive management, and franchise fees, \$37 million of lower general, administrative, and other expenses, \$29 million of increased owned, leased, and other revenue net of expenses, and a \$15 million increase in gains and other income, partially offset by \$20 million of lower equity results. Incentive management fees included \$10 million and \$4 million for 2006 and 2005, respectively, that were calculated based on prior periods' earnings, but not earned and due until the periods they were recognized. The increase in fees is largely due to higher RevPAR, driven by rate increases, which impacted property-level house profits and, to a lesser extent, to productivity improvements and the growth in the number of rooms.

Equity results decreased by \$20 million in 2006 versus 2005 and reflected the recognition in 2005 of \$14 million in equity earnings from a joint venture as a result of the venture's sale of a hotel and \$4 million of increased equity losses in 2006 associated with one joint venture's hotel that was closed for renovation. The increase of \$29 million in owned, leased, and other revenue net of direct expenses reflects \$19 million of improved results for owned and leased properties primarily due to the stronger demand environment and an increase of \$12 million for hotel agreement termination fees for several properties. The \$19 million improvement in owned and leased results also reflects the impact of a \$5 million charge in 2006 for depreciation expense associated with one property that was reclassified from "held for sale" to "held and used" as it was not sold within one year of its classification as "held for sale," as had been expected.

The \$15 million increase in gains and other income is primarily attributable to gains on the sale of various joint ventures throughout 2006, partially offset by a decrease in other income associated with one of the sold cost method joint venture investments. The decrease in general, administrative, and other expenses of \$37 million is attributable to the recognition of expenses in 2005 totaling \$5 million associated with our bedding incentive program as well as a \$34 million charge in 2005 associated with the CTF transaction discussed more fully in the previous operating income disclosure. Also impacting general, administrative, and other expenses, in 2005 we recorded expenses totaling \$6 million associated with two guarantees and 2006 reflected \$7 million of increased expenses related to unit growth and development, systems improvements, and increased ordinary costs such as wages and benefits.

Luxury Lodging includes The Ritz-Carlton and Bulgari Hotels & Resorts.

				Annual (	Change
(\$ in millions)	2007	2006	2005	2007/2006	2006/2005
Segment revenues	\$1,576	\$1,423	\$1,333	11%	7%
Segment results	\$ 72	\$ 63	\$ 45	14%	40%

# 2007 Compared to 2006

Across our Luxury Lodging segment, we added 11 properties (2,529 rooms) and one property (273 rooms) left the system in 2007. In addition, we added three residential products (347 units) in 2007.

In 2007, RevPAR for comparable company-operated luxury properties increased by 9.4 percent to \$227.87. Occupancy for these properties increased by 1.8 percentage points to 72.5 percent, and average daily rates increased by 6.6 percent to \$314.36.

The \$9 million increase in segment results, as compared to 2006, reflected a \$20 million increase in base management and incentive management fees, partially offset by \$3 million of lower owned, leased, and other revenue net of direct expenses, \$6 million of increased general, administrative, and other expenses, and \$2 million of lower equity joint venture results. Fiscal year 2006 included \$2 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006, as compared to no similar fees in 2007. The increase in fees over the prior year reflects stronger RevPAR driven by rate increases, new properties added to

the system, the receipt in 2007 of \$5 million of business interruption insurance proceeds associated with hurricanes in prior years, the year-over-year favorable impact associated with the reopening, late in 2006, of two properties impacted by the same aforementioned hurricanes and increased branding fees. The \$6 million increase in general, administrative, and other expenses is primarily attributable to costs related to our unit growth and development. The \$3 million decrease in owned, leased, and other revenue net of direct expenses reflected charges totaling \$8 million in 2007 for depreciation expense associated with one property that was reclassified from "held and used," as the property no longer satisfied the criteria to be classified as "held for sale," partially offset by a favorable \$3 million impact associated with the opening of a new leased property in 2007.

# 2006 Compared to 2005

Across our Luxury Lodging segment, we added three properties (424 rooms) and one property (229 rooms) left the system in 2006.

In 2006, RevPAR for comparable company-operated luxury properties increased by 9.7 percent to \$209.88. Occupancy for these properties increased by 1.4 percentage points to 72.5 percent, and average daily rates increased by 7.5 percent to \$289.50.

The \$18 million increase in segment results for 2006 primarily reflects a \$16 million increase in base management and incentive management fees, \$7 million of lower general, administrative, and other expenses and \$1 million of higher owned, leased, and other revenue net of direct expenses, partially offset by \$5 million of lower gains and other income. The increase in fees is largely due to higher RevPAR, driven by rate increases, which impacted property-level house profits and, to a lesser extent, to productivity improvements, the growth in the number of rooms, and the recognition of \$2 million of base management fees that were calculated based on prior periods' results, but not earned and due until 2006.

The decrease in general, administrative, and other expenses of \$7 million is attributable to a 2005 hotel management agreement performance termination cure payment of \$12 million associated with one property, partially offset by a \$5 million increase in expenses in 2006 reflecting costs related to unit growth and development and an increase in ordinary costs such as wages and benefits. Gains and other income decreased by \$5 million in 2006 as a result of a \$9 million gain in 2005 associated with the repayment before maturity of a note receivable associated with one property, partially offset by gains totaling \$3 million associated with the 2006 sale of a preferred interest in one property.

Timeshare includes our Marriott Vacation Club, The Ritz-Carlton Club, Grand Residences by Marriott, and Horizons by Marriott Vacation Club brands.

				Annual C	Change
(\$ in millions)	2007	2006	2005	2007/2006	2006/2005
Segment Revenues					
Segment revenues	\$2,065	\$1,840	\$1,721	12%	7%
Segment Results					
Base management fee revenue	\$ 43	\$ 34	\$ 32		
Timeshare sales and services, net	350	357	259		
Joint venture equity	10	(2)	1		
Minority interest	1	_	_		
General, administrative, and other expense	(98)	(109)	(93)		
Gains and other income			72		
Segment results	\$ 306	\$ 280	\$ 271	9%	3%
Sales and Services Revenue					
Development	\$1,208	\$1,112	\$1,208		
Services	315	286	151		
Financing	195	171	72		
Other revenue	29	8	56		
Sales and services revenue	\$1,747	\$1,577	\$1,487	11%	6%
Contract Sales					
Timeshare	\$1,221	\$1,207	\$1,187		
Fractional	44	42	101		
Residential	(9)	5	22		
Total company	1,256	1,254	1,310		
Timeshare	33	28	49		
Fractional	54	68	17		
Residential	58	282	24		
Total joint venture	145	378	90		
Total contract sales	\$1,401	\$1,632	\$1,400	-14%	17%

# 2007 Compared to 2006

Timeshare contract sales, including sales made by our timeshare joint venture projects, represent sales of timeshare interval, fractional ownership, and residential ownership products before the adjustment for percentage-of-completion accounting. Timeshare contract sales decreased by 14 percent as compared to 2006, reflecting fewer residential and fractional sales, partially offset by increased timeshare sales. Contract sales in 2006 reflected particularly strong joint venture residential sales associated with the launch of our San Francisco and Kapalua, Hawaii products.

The \$225 million increase in Timeshare segment revenues from \$1,840 million to \$2,065 million reflected a \$170 million increase in timeshare sales and services revenue, a \$46 million increase in cost reimbursements revenue, and \$9 million of increased base management fees. The increase in timeshare sales and services revenue primarily reflects newer projects that reached reportability thresholds in 2007 and increased services and financing revenue. Higher base management fees reflect the growing number of timeshare resorts under management. Timeshare segment revenues include \$50 million and \$41 million of interest income for 2007 and 2006, respectively, and note sale gains of \$81 million and \$77 million for 2007 and 2006, respectively, recorded in our Consolidated Statements of Income on the "Timeshare sales and services" revenue line, associated with Timeshare segment notes receivable.

Segment results of \$306 million in 2007 increased by \$26 million over 2006 and primarily reflected \$9 million of increased base management fees, \$12 million of increased joint venture equity results, and \$11 million of lower general, administrative, and other expenses, partially offset by \$7 million of lower timeshare sales and services revenue net of expenses. Timeshare sales and services revenue net of direct expenses of \$350 million decreased by \$7 million, as compared to the prior year, primarily reflecting flat development revenue net of product costs and marketing and selling costs and \$12 million of increased financing revenue net of financing expenses, partially offset by the \$15 million reversal of contingency reserves in 2006 and \$4 million of lower services revenue net of services expenses. Flat development revenue net of product costs and marketing and selling costs reflected newer projects that reached reportability thresholds in 2007,

offset by several other projects that were approaching sell-out. The increase in financing revenue net of financing costs primarily reflects increased accretion, interest income, and higher note sale gains in 2007, as compared to 2006. As compared to the prior year, the \$12 million increase in joint venture equity results primarily reflects strong demand in 2007 for our products in Kapalua, Hawaii and start-up costs in 2006 associated with that joint venture. The \$11 million decrease in general, administrative, and other expenses reflected lower program and systems expenses in 2007.

# 2006 Compared to 2005

Timeshare contract sales increased by 17 percent in 2006. Timeshare segment revenues of \$1,840 million and \$1,721 million in 2006 and 2005, respectively, include \$41 million and \$38 million for 2006 and 2005, respectively, of interest income, recorded in our Consolidated Statements of Income on the "Timeshare sales and services" revenue line, associated with Timeshare segment notes receivable. The \$119 million increase in Timeshare segment revenues reflects a \$90 million increase in Timeshare sales and services revenue, a \$27 million increase in cost reimbursements revenue, and \$2 million of increased base management fees. The \$90 million increase in Timeshare sales and services revenue primarily reflects \$77 million of revenue in 2006 from note securitization gains. As detailed earlier in the "Cumulative Effect of Change in Accounting Principle" narrative, note securitization gains of \$69 million for 2005 are not reflected in revenue, but instead are a component of gains and other income.

Segment results of \$280 million in 2006 increased by \$9 million over 2005. The \$9 million increase includes an increase of \$98 million for Timeshare sales and services revenue net of direct expenses and a \$2 million increase in base management fees, partially offset by a decline of \$72 million in gains and other income, a \$16 million increase in general, administrative, and other expenses, and \$3 million of lower joint venture equity earnings. The increase in Timeshare sales and services revenue net of direct expenses of \$98 million largely reflects \$77 million of revenue in 2006 from note securitization gains. Note securitization gains of \$69 million for 2005 are not reflected in revenue, but instead are a component of gains and other income. Also reflected in the \$98 million variance, development revenue net of expenses increased by \$14 million primarily reflecting lower development expenses in 2006 associated with projects in the early stages of development that did not reach revenue recognition thresholds and the timing of expenses associated with SOP 04-2, implemented in 2006.

Additionally, the \$98 million variance reflected a \$15 million reversal of marketing related contingency reserves and \$12 million of higher financing revenue net of expense, partially offset by \$21 million of lower services revenue net of expenses. The \$3 million decline in equity income primarily reflects start-up losses associated with several new joint ventures, partially offset by improved performance at other joint ventures. The \$72 million decline in gains and other income primarily reflected \$69 million of note securitization gains in 2005 versus the recording of note securitization gains in revenue for 2006, while the \$16 million increase in general, administrative, and other expenses reflected higher program and joint venture development expenses, and customary increases in ordinary costs such as wages and benefits.

# **Investment in Leveraged Lease**

Historically, we had a \$23 million investment in an aircraft leveraged lease with Delta, which we acquired in 1994. The gross investment was comprised of rentals receivable and the residual value of the aircraft offset by unearned income. On September 14, 2005, Delta filed for bankruptcy protection under Chapter 11 of the U.S. Bankruptcy Code and informed us that it wished to restructure the lease. As a result, our investment was impaired and we had recorded pretax charges of approximately \$18 million through 2006. We recorded an additional \$5 million loss related to this investment in fiscal 2007. We have no remaining exposure related to this historical investment.

# **Effective Tax Rate**

Tax credits contributed by our synthetic fuel operations have significantly reduced our effective tax rate during the last several years. As we exited the business in November 2007 our future effective tax rate is likely to increase significantly, thereby reducing our after-tax profits.

# DISCONTINUED OPERATIONS

# **Synthetic Fuel**

The tax credits provided under Internal Revenue Code Section 45K were only available for the production and sale of synthetic fuels produced from coal through December 31, 2007. Given high oil prices during 2007 and the anticipated and related phase-out of a significant portion of tax credits available for synthetic fuel produced and sold in 2007, we permanently ceased operations at our synthetic fuel facilities on November 3, 2007 and report this business as a discontinued operation. See Footnote No. 2, "Discontinued Operations," and Footnote No. 17, "Contingencies," in this report for additional information regarding the Synthetic Fuel segment.

# 2007 Compared to 2006

For 2007, the synthetic fuel operation generated revenue of \$352 million versus revenue of \$165 million for the prior year, primarily reflecting higher production in 2007. Production in 2006 reflected production suspensions instituted in response to high oil prices. Income from the Synthetic Fuel segment declined from \$5 million in 2006 to a loss of \$1 million in 2007, primarily reflecting increased operating losses associated with higher production in 2007, partially offset by increased revenue and increased tax credits associated with higher production and, on the increased operating losses, a higher tax benefit. Results for 2007 also reflect an estimated 70.71 percent phase-out of tax credits due to high oil prices versus a phase-out that was estimated at year-end 2006 to be 39 percent for 2006. Additionally, results in 2007 reflect interest costs of \$8 million associated with hedges entered into in response to high oil prices.

# 2006 Compared to 2005

For 2006, the synthetic fuel operation generated revenue of \$165 million versus revenue of \$421 million for the prior year, primarily reflecting significantly lower production in 2006 as a result of production suspensions instituted in response to high oil prices. Income from the Synthetic Fuel segment declined from \$125 million in 2005 to \$5 million in 2006, primarily as a result of both lower production in 2006 and the estimated 39 percent phase-out of tax credits at year-end 2006 due to high oil prices in 2006. Additionally, results in 2006 reflect interest costs of \$4 million associated with hedges entered into in response to high oil prices. Segment results for 2006 also reflect a \$5 million charge reflecting the write-down of assets at the Alabama production facility as the adjacent mine was closed at year-end and we did not anticipate operating the facility again at this location.

# **Distribution Services**

In 2005, we had income, net of tax, of \$1 million associated with the distribution services business that we exited in 2002.

# **New Accounting Standards**

FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes"

We adopted FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes," ("FIN 48") on December 30, 2006, the first day of our 2007 fiscal year. FIN 48 is an interpretation of FASB Statement No. 109, "Accounting for Income Taxes," which seeks to standardize practices associated with certain aspects of measurement and recognition in accounting for income taxes. FIN 48 prescribes a recognition threshold and measurement requirement for the financial statement recognition of a tax position taken or expected to be taken on a tax return. Additionally, FIN 48 provides guidance on de-recognition, classification, interest and penalties, accounting in interim periods, disclosure, and transition. Under FIN 48, an entity may only recognize or continue to recognize tax positions that meet a "more likely than not" threshold. We recorded the cumulative effect of applying FIN 48 of \$155 million as an adjustment to the opening balance of retained earnings and additional paid-in-capital on December 30, 2006, the first day of our 2007 fiscal year. See Footnote No. 3, "Income Taxes," for additional information.

Financial Accounting Standards No. 156, "Accounting for Servicing of Financial Assets-an Amendment of FASB Statement No. 140"

We adopted FASB's FAS No. 156 on December 30, 2006, the first day of our 2007 fiscal year. FAS No. 156 requires that all separately recognized servicing assets and liabilities initially be measured at fair value, if practicable. It also allows an entity to subsequently elect fair value measurement for its servicing assets and liabilities. We recorded the cumulative effect of applying FAS No. 156, of \$1 million, net of tax, as an adjustment to the opening balance of retained earnings on December 30, 2006. See Footnote No. 11, "Asset Securitizations," for additional information.

# **Future Adoption of Accounting Standards**

EITF Issue No. 06-8, "Applicability of the Assessment of a Buyer's Continuing Investment under FASB Statement No. 66, Accounting for Sales of Real Estate, for Sales of Condominiums"

In November 2006, the Emerging Issues Task Force of FASB ("EITF") reached a consensus on EITF Issue No. 06-8, "Applicability of the Assessment of a Buyer's Continuing Investment under FASB Statement No. 66, Accounting for Sales of Real Estate, for Sales of Condominiums" ("EITF 06-8"). EITF 06-8 will require condominium sales to meet the continuing investment criterion in FAS No. 66 in order to recognize profit under the

percentage of completion method. EITF 06-8 will be effective for annual reporting periods beginning after March 15, 2007, which for us begins with our 2008 fiscal year. The cumulative effect of applying EITF 06-8, if any, will be recorded as an adjustment to the opening balance of retained earnings in the year of adoption. We do not expect the impact of adoption of EITF 06-8 to be material.

FAS No. 157, "Fair Value Measurements"

In September 2006, the FASB issued FAS No. 157, "Fair Value Measurements" ("FAS No. 157"). This standard defines fair value, establishes a methodology for measuring fair value, and expands the required disclosure for fair value measurements. FAS No. 157 is effective for fiscal years beginning after November 15, 2007, which for us begins with our 2008 fiscal year. Provisions of FAS No. 157 must be applied prospectively as of the beginning of the first fiscal year in which FAS No. 157 is applied. In November 2007, the FASB agreed to partially defer the effective date, for one year, of FAS No. 157, for non-financial assets and liabilities, except those that are recognized or disclosed at fair value in the financial statements on a recurring basis. We are currently evaluating the impact that FAS No. 157 will have on our financial statements.

Financial Accounting Standards No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities Including an Amendment of FASB Statement No. 115"

In February 2007, the FASB issued FAS No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities Including an Amendment of FASB Statement No. 115" ("FAS No. 159"). This standard permits entities to choose to measure many financial instruments and certain other items at fair value and is effective for the first fiscal year beginning after November 15, 2007, which for us begins with our 2008 fiscal year. We do not expect to elect the fair value measurement option for any financial assets or liabilities at the present time.

EITF Issue No. 07-6, "Accounting for Sales of Real Estate Subject to the Requirements of FASB Statement No. 66, 'Accounting for Sales of Real Estate,' When the Agreement Includes a Buy-Sell Clause"

In December 2007, the EITF reached a consensus on EITF Issue No. 07-6, "Accounting for Sales of Real Estate Subject to the Requirements of FASB Statement No. 66, 'Accounting for Sales of Real Estate,' When the Agreement Includes a Buy-Sell Clause" ("EITF 07-6"). EITF 07-6 clarifies whether a buy-sell clause is a prohibited form of continuing involvement that would preclude partial sales treatment under FAS No. 66. EITF 07-6 is effective for new arrangements entered into and assessments of existing transactions originally accounted for under the deposit, profit-sharing, leasing, or financing methods for reasons other than the exercise of a buy-sell clause performed in fiscal years beginning after December 15, 2007, which for us begins with our 2008 fiscal year. We do not expect EITF 07-6 to have a material impact on our financial statements.

Financial Accounting Standards No. 141 (Revised 2007), "Business Combinations"

On December 4, 2007, the FASB issued FAS No. 141 (Revised 2007), "Business Combinations" ("FAS No. 141(R)"). FAS No. 141(R) will significantly change the accounting for business combinations. Under FAS No. 141(R), an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. FAS No. 141(R) also includes a substantial number of new disclosure requirements. FAS No. 141(R) applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008, which for us begins with our 2009 fiscal year. We are currently evaluating the impact that FAS No. 141(R) will have on our financial statements.

Financial Accounting Standards No. 160, "Non-controlling Interests in Consolidated Financial Statements-an Amendment of ARB No. 51"

On December 4, 2007, the FASB issued FAS No. 160, "Non-controlling Interests in Consolidated Financial Statements-an Amendment of Accounting Research Bulletin ("ARB") No. 51" ("FAS No. 160"). FAS No. 160 establishes new accounting and reporting standards for a non-controlling interest in a subsidiary and for the deconsolidation of a subsidiary. Specifically, this statement requires the recognition of a non-controlling interest (minority interest) as equity in the consolidated financial statements separate from the parent's equity. The amount of net income attributable to the non-controlling interest will be included in consolidated net income on the face of the income statement. FAS No. 160 clarifies that changes in a parent's ownership interest in a subsidiary that do not result in deconsolidation are equity transactions if the parent retains its controlling financial interest. In addition, this statement requires that a parent recognize a gain or loss in net income when a subsidiary is deconsolidated. Such gain or loss will be measured using the fair value of the non-controlling equity investment on the deconsolidation date. FAS No. 160 also includes expanded disclosure requirements regarding the interests of the

parent and its non-controlling interest. FAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008, which for us begins with our 2009 fiscal year. We are currently evaluating the impact that FAS No. 160 will have on our financial statements.

# LIQUIDITY AND CAPITAL RESOURCES

Cash Requirements and Our Credit Facilities

We are party to a multicurrency revolving credit agreement that provides for borrowings and letters of credit and supports our commercial paper program. On May 14, 2007, we amended and restated this facility to increase the aggregate borrowings available under the facility from \$2 billion to \$2.5 billion and extended the expiration of the facility from 2011 to 2012. The material terms of the amended and restated credit agreement were otherwise unchanged. Borrowings under the facility bear interest at the London Interbank Offered Rate (LIBOR) plus a spread based on our public debt rating. Additionally, we pay annual fees on the facility at a rate also based on our public debt rating. We do not anticipate that fluctuations in the availability of the commercial paper market will affect our liquidity because of the flexibility provided by our credit facility. We classify commercial paper as long-term debt based on our ability and intent to refinance it on a long-term basis.

At year-end 2007, our available borrowing capacity amounted to \$2.151 billion and reflected borrowing capacity of \$2.5 billion under the credit facility, plus our cash balance of \$332 million, less letters of credit outstanding totaling \$96 million, and less \$585 million of outstanding commercial paper supported by the facility. We consider these resources, together with cash we expect to generate from operations, adequate to meet our short-term and long-term liquidity requirements, finance our long-term growth plans, meet debt service, and fulfill other cash requirements. We periodically evaluate opportunities to issue and sell additional debt or equity securities, obtain credit facilities from lenders, or repurchase, refinance, or otherwise restructure our long-term debt for strategic reasons, or to further strengthen our financial position.

We issue short-term commercial paper in the United States and in Europe. Our commercial paper issuances are subject to the then-current demand for our commercial paper in each of these markets, as we have no commitments from buyers to purchase our commercial paper. We reserve unused capacity under our credit facility to repay outstanding commercial paper borrowings in the event that the commercial paper market is not available to us for any reason when outstanding borrowings mature.

We monitor the status of the capital markets and regularly evaluate the effect that changes in capital market conditions may have on our ability to execute our announced growth plans. We expect that part of our financing and liquidity needs will continue to be met through commercial paper borrowings and access to long-term committed credit facilities. If conditions in the lodging industry deteriorate, or if disruptions in the commercial paper market take place as they did in the immediate aftermath of September 11, 2001, we may be unable to place some or all of our commercial paper on a temporary or extended basis and may have to rely more on borrowings under the credit facility, which may or may not carry a higher cost than commercial paper.

#### Cash from Operations

Cash from operations, depreciation expense, and amortization expense for the last three fiscal years are as follows:

(\$ in millions)	2007	2006	2005
Cash from operations	\$778	\$970	\$840
Depreciation expense	162	155	156
Amortization expense	35	33	28

Our ratio of current assets to current liabilities was roughly 1.2 to 1.0 at year-end 2007 and 1.3 to 1.0 at year-end 2006. We minimize working capital through cash management, strict credit-granting policies, aggressive collection efforts, and high inventory turnover. We also have significant borrowing capacity under our revolving credit facility should we need additional working capital.

Our ratios of earnings to fixed charges for the last five fiscal years, the calculations of which are detailed in Exhibit 12 to this report, are as follows:

		Fiscal Years		
2007	2006	2005	2004	2003
4.3x	5.3x	4.6x	4.9x	3.9x

While our Timeshare segment generates strong operating cash flow, year-to-year cash flow varies based on the timing of both cash outlays for the acquisition and development of new resorts and cash received from purchaser financing. We include timeshare reportable sales we finance in cash from operations when we collect cash payments or the notes are sold for cash. The following table shows the net operating activity from our Timeshare segment (which does not include the portion of income from continuing operations from our Timeshare segment):

(\$ in millions)	2007	2006	2005
Timeshare segment development (in excess of) less than cost of sales	\$ (55)	\$ (83)	\$ 40
New Timeshare segment mortgages, net of collections	(559)	(537)	(441)
Note repurchases	(30)	(55)	(23)
Financially reportable sales (in excess of) less than closed sales	(16)	61	(57)
Note sale gains	(81)	(77)	(69)
Note sale proceeds	515	508	399
Collection on retained interests in notes sold and servicing fees	106	96	90
Other cash (outflows) inflows	(35)	(17)	26
Net cash outflows from Timeshare segment activity	\$(155)	\$(104)	\$ (35)

Our ability to sell Timeshare segment notes is dependent upon continued liquidity within the asset-backed and broader structured credit markets. While the recent dislocation in the capital markets has resulted in a general reduction in liquidity for new asset-backed issuances, we believe that demand remains sufficient for our continued sale of Timeshare segment notes, albeit at somewhat less favorable terms. If the liquidity of the markets decreases further, or the underlying quality of any future Timeshare segment notes we originate were to deteriorate, we might have increased difficulty or be unable to consummate such sales, although we do not expect such deterioration.

# Investing Activities Cash Flows

Capital Expenditures and Other Investments. Capital expenditures of \$671 million in 2007, \$529 million in 2006, and \$780 million in 2005 included expenditures related to the development and construction of new hotels and acquisitions of hotel properties, as well as improvements to existing properties and systems initiatives. Timeshare segment development expenditures, which are included in "Cash from Operations," as noted in that section's narrative, are not reflected in these numbers. Over time, we have sold lodging properties under development subject to long-term management agreements. The ability of third-party purchasers to raise the necessary debt and equity capital depends in part on the perceived risks inherent in the lodging industry and other constraints inherent in the capital markets as a whole. Although we expect to continue to consummate such real estate sales, if we were unable to do so, our liquidity could decrease and we could have increased exposure to the operating risks of owning real estate. We monitor the status of the capital markets and regularly evaluate the potential impact on our business operations of changes in capital market conditions. We also expect to continue to make other investments in connection with adding units to our lodging business. These investments include loans and minority equity investments.

Fluctuations in the values of hotel real estate generally have little impact on the overall results of our Lodging segments because: (1) we own less than 1 percent of the total number of hotels that we operate or franchise; (2) management and franchise fees are generally based upon hotel revenues and profits rather than current hotel property values; and (3) our management agreements generally do not terminate upon hotel sale.

At the end of 2007, we were party to a venture that developed and marketed fractional ownership and residential interests. Subsequent to year-end 2007, we purchased our partner's interest in the joint venture. Concurrent with this transaction, we purchased additional land from our partner as well. Cash consideration for this transaction totaled \$37 million and we acquired assets and liabilities totaling \$74 million and \$37 million, respectively, on the date of purchase.

**Dispositions.** Property and asset sales generated cash proceeds of \$745 million in 2007, \$798 million in 2006, and \$298 million in 2005. In 2007, we closed on the sales of 13 properties and five joint venture investments. Cash proceeds of \$90 million for a parcel of land sold in 2007 are not reflected in the \$745 million as the proceeds were initially recorded as a deposit because of a contingency. Accordingly, these proceeds impacted the "Other investing activities" section of our Consolidated Statements of Cash Flows rather than "Dispositions."

**Loan Activity.** We have made loans to owners of hotels that we operate or franchise, typically to facilitate the development of a new hotel. Based on historical experience, over time we expect these owners to repay the loans in accordance with the loan agreements, or earlier as the hotels mature and capital markets permit. Loan collections and sales, net of advances during 2007, amounted to \$75 million. Lodging senior loans outstanding totaled

\$7 million (which included a current portion of \$4 million) at year-end 2007 and \$9 million (which included a current portion of \$1 million) at year-end 2006. Lodging mezzanine and other loans totaled \$206 million (which included a current portion of \$18 million) at year-end 2007 and \$268 million (which included a current portion of \$32 million) at year-end 2006. In 2007 our notes receivable balance associated with Lodging senior loans and Lodging mezzanine and other loans, declined by \$64 million and primarily reflects the repayment or sale of several loans and the reserve against an underperforming loan.

**Equity and Cost Method Investments.** Cash outflows of \$40 million in 2007 associated with equity and cost method investments primarily reflects our investments in two joint ventures. Cash outflows of \$95 million in 2006 associated with equity and cost method investments primarily reflects our investments in three joint ventures. Cash outflows of \$216 million in 2005 associated with equity and cost method investments primarily reflects our establishment in 2005 of a 50/50 joint venture with Whitbread PLC ("Whitbread"). For additional information regarding Whitbread see Footnote No. 10, "Marriott and Whitbread Joint Venture" in our 2006 Form 10-K.

#### Cash from Financing Activities

**Debt.** Debt increased by \$1,132 million in 2007, from \$1,833 million to \$2,965 million at year-end 2007, due to the issuance of \$346 million (book value at issuance) of Series I Senior Notes (described more fully below), the issuance of \$397 million (book value at issuance) of Series J Senior Notes (described more fully below), a net increase in commercial paper outstanding of \$270 million and other debt increases of \$119 million. Debt increased by \$96 million in 2006, from \$1,737 million to \$1,833 million at year-end 2006, due to the issuance of \$349 million (book value at issuance) of Series H Senior Notes, partially offset by a net reduction in commercial paper outstanding of \$184 million, and other debt decreases of \$69 million.

In 2007, we issued \$350 million of aggregate principal amount of 6.375 percent Series I Senior Notes due 2017. The offering of the notes closed on June 25, 2007. We received net proceeds before expenses of approximately \$346 million from this offering, after deducting the underwriting discount and estimated expenses of the offering. We used these proceeds for general corporate purposes, including the repayment of commercial paper borrowings. Interest on these notes will be paid on June 15 and December 15 of each year, and commenced on December 15, 2007. The notes will mature on June 15, 2017, and are redeemable, in whole or in part, at any time and from time to time under the terms provided in the form of note.

Also in 2007, we issued \$400 million of aggregate principal amount of 5.625 percent Series J Senior Notes due 2013. The offering of the notes closed on October 19, 2007. We received net proceeds before expenses of approximately \$396 million from this offering, after deducting the underwriting discount and estimated expenses of the offering. We used these proceeds for general corporate purposes, including working capital, acquisitions, stock repurchases and the repayment of commercial paper borrowings. Interest on these notes will be paid on February 15 and August 15 of each year, and commenced on February 15, 2008. The notes will mature on February 15, 2013, and are redeemable, in whole or in part, at any time and from time to time under the terms provided in the form of note.

Both the Series I Senior Notes and the Series J Senior Notes were issued under an indenture with The Bank of New York, successor to JPMorgan Chase Bank, N.A. (formerly known as The Chase Manhattan Bank), as trustee, dated as of November 16, 1998.

Our financial objectives include diversifying our financing sources, optimizing the mix and maturity of our long-term debt and reducing our working capital. At year-end 2007, our long-term debt had an average interest rate of 6.0 percent and an average maturity of approximately 5.9 years. The ratio of fixed-rate long-term debt to total long-term debt was 0.8 to 1.0 at year-end 2007. At the end of 2007, we had long-term public debt ratings of BBB from Standard and Poor's and Baa2 from Moody's.

Subsequent to year-end 2007, on January 15, 2008, we made a \$94 million cash payment of principal and interest to retire, at maturity, all of our outstanding Series E Senior Notes.

**Share Repurchases.** We purchased 41.0 million shares of our Class A Common Stock in 2007 at an average price of \$43.32 per share, 41.5 million shares of our Class A Common Stock in 2006 at an average price of \$38.13 per share, and 51.4 million shares of our Class A Common Stock in 2005 at an average price of \$32.12 per share. We purchase shares in the open market and in privately negotiated transactions. As of year-end 2007, 33.2 million shares remained available for repurchase under authorizations from our Board of Directors.

**Dividends.** In April 2007, our Board of Directors increased the quarterly cash dividend by 2 percent to \$0.0750 per share.

Contractual Obligations and Off Balance Sheet Arrangements

The following table summarizes our contractual obligations as of year-end 2007:

#### **Contractual Obligations**

			Payments Due by Period							
			Less	Than						After
(\$ in millions)		Total	1`	Year	1-3	Years	3-5	Years	_ 5	Years
Debt (1)	\$	3,855	\$	300	\$	405	\$	596	\$	2,554
Capital lease obligations (1)		12		1		2		2		7
Operating leases where we are the primary obligor:										
Recourse		1,394		121		256		219		798
Non-recourse		439		21		30		30		358
Operating leases where we are secondarily liable		166		30		60		43		33
Other long-term liabilities		110		3		14		5		88
Total contractual obligations	\$	5,976	\$	476	\$	767	\$	895	\$	3,838

<sup>(1)</sup> Includes principal as well as interest payments.

The total amount of unrecognized tax benefits as of year-end 2007 was \$132 million and is not reflected in the Contractual Obligations table. As a large taxpayer, we are under continual audit by the Internal Revenue Service and other taxing authorities on several open tax positions, and it is possible that the amount of the liability for unrecognized tax benefits could change during the next 52-week period. While it is possible that one or more of these examinations may be resolved in the next year, it is not anticipated that a significant impact to the unrecognized tax benefit balance will occur. See Footnote No. 3, "Income Taxes," for additional information.

The following table summarizes our commitments as of year-end 2007:

# **Other Commercial Commitments**

				Amo	unt of C	ommitmei	nt Expira	ition Per	Period	
	Total .	Amounts	Less	Than					I	After
(\$ in millions)	Con	ımitted	1 7	Year	1-3	Years	3-5	Years	_5	Years
Total guarantees where we are the primary obligor	\$	315	\$	21	\$	89	\$	49	\$	156
Total guarantees where we are secondarily liable		322		46		91		112		73
Total other commercial commitments	\$	637	\$	67	\$	180	\$	161	\$	229

Our guarantees where we are the primary obligor of \$315 million listed in the preceding table include \$41 million of operating profit guarantees that will not be in effect until the underlying properties open and we begin to operate the properties.

The guarantees where we are secondarily liable of \$322 million in the preceding table include \$245 million of guarantees that expire in the years 2011 through 2013, related to Senior Living Services lease obligations and lifecare bonds for which we are secondarily liable. Sunrise Senior Living, Inc. ("Sunrise") is the primary obligor of the leases and a portion of the lifecare bonds, and CNL Retirement Properties, Inc. ("CNL"), which subsequently merged with Health Care Property Investors, Inc., is the primary obligor of the remainder of the lifecare bonds. Prior to our sale of the Senior Living Services business in 2003, these preexisting guarantees were guarantees by us of obligations of consolidated Senior Living Services subsidiaries. Sunrise and CNL indemnified us for any guarantee fundings we may be called on to make in connection with these lease obligations and lifecare bonds. We do not expect to fund under the guarantees.

The guarantees in the preceding table for which we are secondarily liable include lease obligations for which we became secondarily liable when we acquired the Renaissance Hotel Group N.V. in 1997, consisting of annual rent payments of approximately \$7 million and total remaining rent payments through the initial term of approximately \$77 million. Most of these obligations expire at the end of the 2023 calendar year. CTF has made available €35 million in cash collateral in the event that we are required to fund under such guarantees (approximately €7 million [\$11 million] remained at year-end 2007). As CTF obtains releases from the landlords and these hotels exit the system, our contingent liability exposure of approximately \$77 million will decline. Since the time we assumed these guarantees, we have not funded any amounts and we do not expect to fund any amounts under these guarantees in the future.

Furthermore, in addition to the guarantees noted in the preceding table, we have provided a project completion guarantee to a lender for a project with an estimated aggregate total cost of \$586 million. Payments for cost overruns for this project will be satisfied by the joint venture via contributions from the partners, and we are liable on a several basis with our partners in an amount equal to our pro rata ownership in the joint venture, which is

34 percent. We do not expect to fund under this guarantee. We have also provided a project completion guarantee to another lender for a project with an estimated aggregate total cost of \$80 million. Payments for cost overruns for this project will be satisfied by the joint venture via contributions from the partners, and we are liable on a several basis with our partners in an amount equal to our pro rata ownership in the joint venture, which is 25 percent. We do not expect to fund under this guarantee. The carrying value of the liabilities associated with these two project completion guarantees is \$7 million.

In addition to the guarantees described above, in conjunction with financing obtained for specific projects or properties owned by joint ventures in which we are a party, we may provide industry standard indemnifications to the lender for loss, liability or damage occurring as a result of the actions of the other joint venture owner or our own actions.

In addition to the guarantees noted previously, as of year-end 2007, we had extended approximately \$4 million of loan commitments to owners of lodging properties, under which we expect to fund approximately \$2 million within one year. We do not expect to fund the remaining \$2 million of commitments, which expire after five years.

At year-end 2007, we also have commitments to invest up to \$44 million of equity for minority interests in partnerships that plan to purchase North American full-service and limited-service properties or purchase or develop hotel anchored mixed-use real estate projects, which expire as follows: \$14 million in one to two years; and \$30 million in three to five years. As of year-end 2007, we also have a commitment to invest up to \$25 million in a joint venture of which we have funded \$12 million and have \$13 million remaining that we expect to fund within one year. As of year-end 2007, we also had a commitment to invest up to \$29 million (€20 million) in a joint venture in which we are an investor. We currently do not expect to fund under this commitment.

At year-end 2007 we had \$96 million of letters of credit outstanding on our behalf, the majority of which related to our self-insurance programs. Surety bonds issued on our behalf as of year-end 2007 totaled \$468 million, the majority of which were requested by federal, state or local governments related to our lodging operations, including our Timeshare segment and self-insurance programs.

In the normal course of the hotel management business, we enter into purchase commitments to manage the daily operating needs of hotels we manage for owners. Since we are reimbursed from the cash flows of the hotels, these obligations have minimal impact on our net income and cash flow.

# RELATED PARTY TRANSACTIONS

# **Equity Method Investments**

We have equity method investments in entities that own properties for which we provide management and/or franchise services and receive fees. In addition, in some cases we provide loans, preferred equity or guarantees to these entities. Our ownership interest in these equity method investments generally varies from 10 to 50 percent. The amount of consolidated retained earnings that represents undistributed earnings attributable to our equity investments totaled \$9 million at year-end 2007.

The following tables present financial data resulting from transactions with these related parties:

# Income Statement Data

(\$ in millions)	 2007	2	2006	2005
Base management fees	\$ 56	\$	62	\$ 83
Franchise fees	1		2	2
Incentive management fees	26		22	14
Cost reimbursements	510		649	936
Owned, leased, corporate housing, and other revenue	_		_	19
Total revenue	\$ 593	\$	735	\$1,054
General, administrative, and other	\$ (4)	\$	(1)	\$ (19)
Reimbursed costs	(510)		(649)	(936)
Gains and other income	25		28	54
Interest expense	(1)		(1)	_
Interest income	4		4	31
(Provision for) reversal of provision for loan losses	(12)		1	_
Equity in earnings	15		3	36

# **Balance Sheet Data**

(\$ in millions)	At Year-end 2007	At Year-end 2006
Current assets-accounts and notes receivable	\$ 42	\$ 76
Deferred development	2	_
Contract acquisition costs	33	34
Equity and cost method investments	316	377
Loans to equity method investees	21	27
Long-term deferred tax asset, net	1	4
Current liabilities:		
Other payables and accruals	(2)	(2)
Other long-term liabilities	(16)	(13)

# CRITICAL ACCOUNTING ESTIMATES

The preparation of financial statements in accordance with U.S. generally accepted accounting principles ("GAAP") requires management to make estimates and assumptions that affect reported amounts and related disclosures. Management considers an accounting estimate to be critical if:

- · it requires assumptions to be made that were uncertain at the time the estimate was made; and
- changes in the estimate, or different estimates that could have been selected, could have a material effect on our consolidated results of operations or financial condition.

Management has discussed the development and selection of its critical accounting estimates with the Audit Committee of the Board of Directors, and the Audit Committee has reviewed the disclosure presented below relating to them.

#### Marriott Rewards

Marriott Rewards is our frequent guest loyalty program. Marriott Rewards members earn points based on their monetary spending at our lodging operations, purchases of timeshare interval, fractional ownership, and residential products and, to a lesser degree, through participation in affiliated partners' programs, such as those offered by car rental and credit card companies. Points, which we track on members' behalf, can be redeemed for stays at most of our lodging operations, airline tickets, airline frequent flyer program miles, rental cars, and a variety of other awards; however, points cannot be redeemed for cash. We provide Marriott Rewards as a marketing program to participating properties. We charge the cost of operating the program, including the estimated cost of award redemption, to properties based on members' qualifying expenditures.

We defer revenue received from managed, franchised, and Marriott-owned/leased hotels and program partners equal to the fair value of our future redemption obligation. We determine the fair value of the future redemption obligation based on statistical formulas that project timing of future point redemption based on historical levels, including an estimate of the "breakage" for points that will never be redeemed, and an estimate of the points that will eventually be redeemed. These judgment factors determine the required liability for outstanding points.

Our management and franchise agreements require that we be reimbursed currently for the costs of operating the program, including marketing, promotion, communication with, and performing member services for the Marriott Rewards members. Due to the requirement that properties reimburse us for program operating costs as incurred, we receive and recognize the balance of the revenue from properties in connection with the Marriott Rewards program at the time such costs are incurred and expensed. We recognize the component of revenue from program partners that corresponds to program maintenance services over the expected life of the points awarded. Upon the redemption of points, we recognize as revenue the amounts previously deferred and recognize the corresponding expense relating to the costs of the awards redeemed.

# Valuation of Goodwill

We evaluate the fair value of goodwill to assess potential impairments on an annual basis, or during the year if an event or other circumstance indicates that we may not be able to recover the carrying amount of the asset. We evaluate the fair value of goodwill at the reporting unit level and make that determination based upon future cash flow projections that assume certain growth projections which may or may not occur. We record an impairment loss for goodwill when the carrying value of the intangible asset is less than its estimated fair value.

#### Loan Loss Reserves

# Lodging Senior Loans and Lodging Mezzanine and Other Loans

We measure loan impairment based on the present value of expected future cash flows discounted at the loan's original effective interest rate or the estimated fair value of the collateral. For impaired loans, we establish a specific impairment reserve for the difference between the recorded investment in the loan and the present value of the expected future cash flows, that assumes certain growth projections which may or may not occur, or the estimated fair value of the collateral. We apply our loan impairment policy individually to all loans in the portfolio and do not aggregate loans for the purpose of applying such policy. Where we determine that a loan is impaired, we recognize interest income on a cash basis. At year-end 2007, our recorded investment in impaired loans was \$112 million. We had a \$92 million allowance for credit losses, leaving \$20 million of our investment in impaired loans for which there was no related allowance for credit losses. At year-end 2006, our recorded investment in impaired loans was \$92 million. We had a \$70 million allowance for credit losses, leaving \$22 million of our investment in impaired loans for which there was no related allowance for credit losses. During 2007 and 2006, our average investment in impaired loans totaled \$102 million and \$138 million, respectively.

# Loans to Timeshare Owners

In accordance with the adoption of SOP 04-2 in 2006, we record an estimate of expected uncollectibility on notes receivable that we receive from timeshare purchasers as a reduction of revenue at the time we recognize profit on a timeshare sale. We assess uncollectibility based on pools of receivables, because we hold large numbers of homogeneous timeshare notes receivable. We estimate uncollectibles based on historical activity for similar timeshare notes receivable over the past three years. We use a technique referred to as static pool analysis, which tracks uncollectibles for each year's sales over the life of those notes. At year-end 2007 and year-end 2006, our allowance for credit losses associated with "Loans to timeshare owners" totaled \$19 million and \$29 million, respectively.

# Legal Contingencies

We are subject to various legal proceedings and claims, the outcomes of which are subject to significant uncertainty. We record an accrual for loss contingencies when a loss is probable and the amount of the loss can be reasonably estimated. We review these accruals each reporting period and make revisions based on changes in facts and circumstances.

# Income Taxes

We record the current year amounts payable or refundable, as well as the consequences of events that give rise to deferred tax assets and liabilities based on differences in how those events are treated for tax purposes. We base our estimate of deferred tax assets and liabilities on current tax laws and rates and, in certain cases, business plans and other expectations about future outcomes.

Changes in existing laws and rates and their related interpretations, and future business results may affect the amount of deferred tax liabilities or the valuation of deferred tax assets over time. Our accounting for deferred tax consequences represents management's best estimate of future events that can be appropriately reflected in the accounting estimates.

# OTHER MATTERS

# Inflation

Inflation has been moderate in recent years and has not had a significant impact on our businesses.

# Item 7A. Quantitative and Qualitative Disclosures About Market Risk.

We are exposed to market risk from changes in interest rates, foreign exchange rates, and debt and equity prices, and with respect to our synthetic fuel business, changes in oil prices. We manage our exposure to these risks by monitoring available financing alternatives, through development and application of credit granting policies and by entering into derivative arrangements. We do not foresee any significant changes in either our exposure to fluctuations in interest rates or foreign exchange rates or how such exposure is managed in the future.

We are exposed to interest rate risk on our floating-rate notes receivable, our residual interests retained in connection with the sale of Timeshare segment notes receivable and the fair value of our fixed-rate notes receivable. Changes in interest rates also impact our floating-rate long-term debt and the fair value of our fixed-rate long-term debt.

We are also subject to risk from changes in debt and equity prices from our investments in debt securities and common stock, which have a carrying value of \$55 million at year-end 2007, which we account for as available-for-sale securities under FAS No. 115, "Accounting for Certain Investments in Debt and Equity Securities."

We use derivative instruments as part of our overall strategy to manage our exposure to market risks associated with fluctuations in interest rates and foreign currency exchange rates. As a matter of policy, we do not use derivatives for trading or speculative purposes.

At year-end 2007, we were party to the following derivative instruments:

- An interest rate swap agreement under which we receive a floating rate of interest and pay a fixed rate of interest. The swap modifies our interest rate exposure by effectively converting a note receivable with a fixed rate to a floating rate. The aggregate notional amount of the swap was \$92 million and it matures in 2010.
- Four outstanding interest rate swap agreements to manage interest rate risk associated with the residual interests we retain in conjunction with our timeshare note sales. Historically, we have been required by purchasers and/or rating agencies to utilize interest rate swaps to protect the excess spread within our sold-note pools. The aggregate notional amount of the swaps was \$157 million, and they expire through 2022.
- Option contracts to hedge the potential volatility of earnings and cash flows associated with variations in foreign exchange rates during 2008. The aggregate dollar equivalent of the notional amounts of the contracts was approximately \$86 million, and they expire throughout 2008.
- Forward contracts to hedge forecasted transactions for contracts denominated in foreign currencies. The aggregate dollar equivalent of the notional amounts was approximately \$58 million, and they expire in 2008 and 2010.
- Forward foreign exchange contracts to manage the foreign currency exposure related to certain monetary assets. The aggregate dollar equivalent of the notional amounts of the forward contracts was \$133 million at year-end 2007.
- Oil price hedges to manage the volatility associated with oil prices and the uncertainty surrounding the potential phase-out of tax credits in 2007. The hedges cover 19 million barrels of oil and expired on December 31, 2007.

The following table sets forth the scheduled maturities and the total fair value of our derivatives and other financial instruments as of year-end 2007:

				Matu	rities by Pe	riod				
(\$ in millions)	2008	2009	2010	2011	2012		here- fter	Total Carrying Amount		Total Fair Value
Assets-Maturities represent expected principal receipts, fair values represent	t assets.							<u> </u>		
Timeshare segment notes receivable Average interest rate	\$ 68	\$ 47	\$ 42	\$ 38	\$ 38	\$	243	\$ 476 12.70%	\$	476
Fixed-rate notes receivable Average interest rate	\$ 17	\$ 4	\$ 88	\$—	\$ —	\$	25	\$ 134 12.05%	\$	138
Floating-rate notes receivable Average interest rate	\$ 4	\$ —	\$ 4	\$ 20	\$ 19	\$	32	\$ 79 7.04%	\$	79
Residual interests Average interest rate	\$ 81	\$ 55	\$ 36	\$ 24	\$ 16	\$	26	\$ 238 9.02%	\$	238
Liabilities-Maturities represent expected principal payments, fair values rep	resent liab	ilities.								
Fixed-rate debt Average interest rate	\$(137)	\$(113)	\$ (47)	\$ (14)	\$(354)	\$(1	1,639)	\$(2,304) 6.12%	\$(	(2,338)
Floating-rate debt Average interest rate	\$ (34)	\$ —	\$—	\$—	\$ —	\$	(585)	\$ (619) 5.53%	\$	(619)
NOTE: We classify commercial paper as long-term debt based on our ability	y and inten	t to refinar	ice it on a	a long-ter	m basis.					
<b>Derivatives</b> -Maturities represent notional amounts; fair values represent ass	ets (liabili	ties).								
Interest Rate Swaps:										
Fixed to variable Average pay rate Average receive rate	\$ —	\$ —	\$ 93	\$—	\$ 238	\$	123	\$ (7) 4.02% 3.64%	\$	(7)
Variable to fixed	\$ —	\$ —	¢	<b>\$</b> —	\$ —	\$	34	\$ —	\$	
Average pay rate Average receive rate	ψ —	φ —	Φ—	<b>J</b> —	φ —	Φ	54	5.23% 4.01%	Ф	_
Forward Foreign Exchange Contracts:										
Fixed (EUR) to Fixed (USD)  Average exchange rate	\$ 69	\$ —	\$—	\$—	\$ —	\$	_	\$ — 1.47	\$	_
Fixed (GBP) to Fixed (USD)  Average exchange rate	\$ 46	\$ —	\$—	\$—	\$ —	\$	_	\$ — 2.00	\$	_
Fixed (HKD) to Fixed (USD) Average Exchange Rate	\$ 38	\$ —	\$—	\$—	\$ —	\$	_	\$ — 0.13	\$	_
Fixed (MXN) to Fixed (USD) Average Exchange Rate	\$ 13	\$ —	\$—	\$—	\$ —	\$	_	\$ — 0.09	\$	_
Fixed (JPY) to Fixed (USD) Average Exchange Rate	\$ 3	\$ —	\$—	\$—	\$ —	\$	_	\$ — 0.01	\$	_
Fixed (CAD) to Fixed (USD) Average Exchange Rate	\$ 43	\$ —	\$—	\$—	\$ —	\$	_	\$ — 1.02	\$	_
Fixed (THB) to Fixed (USD) Average Exchange Rate	\$ 39	\$ —	\$ 25	\$—	\$ —	\$	_	\$ — 0.03	\$	_
Oil Price Hedges:										
Contract Amount (Contract volume is 19 million barrels)	\$ 14	\$ —	\$—	\$—	\$ —	\$	_	\$ 4	\$	4

# Item 8. Financial Statements and Supplementary Data.

The following financial information is included on the pages indicated:

Management's Report on Internal Control Over Financial Reporting	61
Report of Independent Registered Public Accounting Firm on Internal Control Over Financial Reporting	62
Report of Independent Registered Public Accounting Firm	63
Consolidated Statements of Income	64
Consolidated Balance Sheets	65
Consolidated Statements of Cash Flows	66
Consolidated Statements of Comprehensive Income	67
Consolidated Statements of Shareholders' Equity	68
Notes to Consolidated Financial Statements	69

# MANAGEMENT'S REPORT ON INTERNAL CONTROL OVER FINANCIAL REPORTING

Management of Marriott International, Inc. (the "Company") is responsible for establishing and maintaining adequate internal control over financial reporting and for the assessment of the effectiveness of internal control over financial reporting. As defined by the Securities and Exchange Commission, internal control over financial reporting is a process designed by, or under the supervision of, the Company's principal executive and principal financial officers and effected by the Company's Board of Directors, management, and other personnel, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of the consolidated financial statements in accordance with U.S. generally accepted accounting principles.

The Company's internal control over financial reporting is supported by written policies and procedures that: (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the Company's transactions and dispositions of the Company's assets; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of the consolidated financial statements in accordance with U.S. generally accepted accounting principles, and that receipts and expenditures of the Company are being made only in accordance with authorizations of the Company's management and directors; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company's assets that could have a material effect on the consolidated financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

In connection with the preparation of the Company's annual consolidated financial statements, management has undertaken an assessment of the effectiveness of the Company's internal control over financial reporting as of December 28, 2007, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (the "COSO Framework"). Management's assessment included an evaluation of the design of the Company's internal control over financial reporting and testing of the operational effectiveness of those controls.

Based on this assessment, management has concluded that as of December 28, 2007, the Company's internal control over financial reporting was effective to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with U.S. generally accepted accounting principles.

Ernst & Young LLP, the independent registered public accounting firm that audited the Company's consolidated financial statements included in this report, has issued an attestation report on internal control over financial reporting, a copy of which appears on the next page of this annual report.

# REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM ON INTERNAL CONTROL OVER FINANCIAL REPORTING

To the Board of Directors and Shareholders of Marriott International, Inc.:

We have audited Marriott International, Inc.'s internal control over financial reporting as of December 28, 2007, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission ("the COSO criteria"). Marriott International, Inc.'s management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting - included in the accompanying Management's Report on Internal Control Over Financial Reporting. Our responsibility is to express an opinion on the company's internal control over financial reporting based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control based on the assessed risk, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

In our opinion, Marriott International, Inc. maintained in all material respects, effective internal control over financial reporting as of December 28, 2007, based on the COSO criteria.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheets of Marriott International, Inc. as of December 28, 2007 and December 29, 2006, and the related consolidated statements of income, cash flows, comprehensive income and shareholders' equity for each of the three fiscal years in the period ended December 28, 2007, of Marriott International, Inc. and our report dated February 14, 2008 expressed an unqualified opinion thereon.

/s/ Ernst & Young LLP

McLean, Virginia February 14, 2008

# REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Shareholders of Marriott International, Inc.:

We have audited the accompanying consolidated balance sheets of Marriott International, Inc. as of December 28, 2007 and December 29, 2006, and the related consolidated statements of income, cash flows, comprehensive income and shareholders' equity for each of the three fiscal years in the period ended December 28, 2007. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform an audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the consolidated financial position of Marriott International, Inc. as of December 28, 2007 and December 29, 2006, and the consolidated results of its operations and its cash flows for each of the three fiscal years in the period ended December 28, 2007, in conformity with U.S. generally accepted accounting principles.

As discussed in Note 1 to the financial statements, in 2007 the Company adopted the provisions of FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" and FASB Statement No. 156, "Accounting for Servicing of Financial Assets." In 2006 the Company changed its accounting for real estate time-sharing transactions in connection with the adoption of Statement of Position 04-2, "Accounting for Real Estate Time-sharing Transactions," and changed its accounting for stock-based compensation in connection with the adoption of FASB Statement No. 123(R), "Share-Based Payment."

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), Marriott International, Inc.'s internal control over financial reporting as of December 28, 2007, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission and our report dated February 14, 2008, expressed an unqualified opinion thereon.

/s/ Ernst & Young LLP

McLean, Virginia February 14, 2008

# MARRIOTT INTERNATIONAL, INC. CONSOLIDATED STATEMENTS OF INCOME Fiscal Years 2007, 2006, and 2005

(\$ in millions, except per share amounts)

	2007	2006	2005
REVENUES			
Base management fees (1)	\$ 620	\$ 553	\$ 497
Franchise fees (1)	439	390	329
Incentive management fees (1)	369	281	201
Owned, leased, corporate housing, and other revenue (1)	1,240	1,119	944
Timeshare sales and services (including note sale gains of \$81 million for 2007 and \$77 million for 2006)	1,747	1,577	1,487
Cost reimbursements (1)	8,575	8,075	7,671
	12,990	11,995	11,129
OPERATING COSTS AND EXPENSES			
Owned, leased, and corporate housing-direct	1,062	936	778
Timeshare-direct	1,397	1,220	1,228
Reimbursed costs (1)	8,575	8,075	7,671
General, administrative, and other (1)	768	677	753
	11,802	10,908	10,430
OPERATING INCOME	1,188	1,087	699
Gains and other income (1)	97	74	149
Interest expense (1)	(184)	(124)	(106)
Interest income (1)	38	49	79
(Provision for) reversal of provision for loan losses (1)	(17)	3	(28)
Equity in earnings (1)	15	3	36
INCOME FROM CONTINUING OPERATIONS BEFORE INCOME TAXES AND MINORITY INTEREST	1,137	1,092	829
Provision for income taxes	(441)	(380)	(284)
Minority interest	1	_	(2)
INCOME FROM CONTINUING OPERATIONS	697	712	543
Cumulative effect of change in accounting principle, net of tax	_	(109)	_
Discontinued operations, net of tax	(1)	5	126
NET INCOME	\$ 696	\$ 608	\$ 669
EARNINGS PER SHARE-Basic			
Earnings from continuing operations	\$ 1.85	\$ 1.76	\$ 1.26
Losses from cumulative effect of accounting change	<del></del>	(0.27)	_
Earnings from discontinued operations	_	0.01	0.29
Earnings per share	\$ 1.85	\$ 1.50	\$ 1.55
EARNINGS PER SHARE-Diluted			
Earnings from continuing operations	\$ 1.75	\$ 1.65	\$ 1.17
Losses from cumulative effect of accounting change	_	(0.25)	_
Earnings from discontinued operations	_	0.01	0.28
Earnings per share	\$ 1.75	\$ 1.41	\$ 1.45
DIVIDENDS DECLARED PER SHARE	\$0.2875	\$0.2400	\$0.2000
DIVIDENDS DECLARED PER SHARE	\$0.20/5	<b>ΦU.24UU</b>	<b>⊅0.∠000</b>

See Footnote No. 21, "Related Party Transactions," of the Notes to Consolidated Financial Statements for disclosure of related party amounts.

# MARRIOTT INTERNATIONAL, INC. CONSOLIDATED BALANCE SHEETS Fiscal Year-end 2007 and 2006 (\$ in millions)

ASSETS         Current assets         5         2         1         1         4         1         0         0         1         1         0         0         1         0         0         1         0         0         1         0         1         0         0         1         0         1         0         1         0		2007	2006
Cash and equivalents	ASSETS		
Accounts and noise receivable (°)         1,148         1,00           Current deferred taxes, net         1,65         2,00           Assess beld for sale         1,23         4,11           Discontinued operations         1,74         1,80           Other         1,74         1,80           Property and equipment         2,33         3,33           Intangible asses         2,10         5,50           Conobrill         3,5         5,50           Conobrill         3,5         5,50           Equity and cost method investments (°)         3,4         40           Votes receivable         3,1         3,2           Votes receivable         2,1         2,2           Loans to equity method investments (°)         3,0         5,0           Other notes receivable         3,0         5,0           Other notes receivable         3,0         5,0           Contreat against the equity of the pas			
Memotroy			
Assis held for sale			
Possontinued operations			
Other         174         180           Property and equipment         1,329         1,329         1,232           Intangible assers         921         921         921           Contract acquisition costs (1)         665         5.75         5.15         5.75         5.15         5.75         <			
Property and equipment			
Property and equipment         1,23         1,23           Intangible assers         921         921         325           Conford calquisition costs (1)         65         5,55         1,56         <	Otner		
Interplace as a foodwill on costs (a)         361 (b)         361 (b)         363 (b)		3,572	3,319
Goodwill         635         575           Contract acquisition costs (1)         635         575           Equity and cost method investments (2)         340           Notes receivable         321         27           Loans to equity method investees (3)         408         316           Other notes receivable         408         316           Other long-term receivable         176         178           Other long-term receivables         176         178           Other (3)         665         665           Other (4)         688         765           Other (4)         688         765           Other (4)         688         765           Other (5)         688         765           Other (6)         680         768           Other (7)         688         765           Other (7)         688         765           Other (8)         789         868           Other (8)         789         868           Accurate type (4)         889         789           Accurate portion of long-term debt         819         658           Accurate portion of long-term debt         819         658           Accurate	Property and equipment	1,329	1,233
Goodwill         635         575           Contract acquisition costs (1)         635         575           Equity and cost method investments (1)         340         402           Notes receivable         321         27         Loans to equity method investees (1)         408         316         317         316         316         316         316         316         316         316         316         316         316         318         316         316         316         316         316         316         316         316         316         318         316         316         318         316         316         316         316         316         316         316         316         316 <td>Intangible assets</td> <td></td> <td></td>	Intangible assets		
Contract acquisition costs (b)         635         575           Lequity and cost method investments (b)         343         408           Notes receivable         21         27           Loans to equity method investees (b)         21         27           Loans to timeshare owners         408         316         368         356           Other long-term teceivable         176         176         360		921	921
Equity and cost method investments (1)         1,556         1,496           Equity and cost method investments (1)         343         402           Notes receivable         21         27           Loans to timeshare owners         408         316           Other notes receivable         171         217           Defered taxes, net (1)         678         665           Other 10ng-term receivables         176         868           Other (1)         688         735         865           Other (1)         688         735         865           Other (1)         688         735         866           Other (1)         688         735         866           Other (1)         689         735         866           Other (2)         889         868         735         868           Current portion of long-term debt         \$15         \$15         862         658         868			
Equity and cost method investments (1)         34         402           Notes receivable         21         27           Loans to equity method investees (1)         21         27           Loans to timeshare owners         60         500           Other notes receivable         171         217           Other long-term receivables         67         66           Other long-term receivables         67         66           Other (1)         68         73           Deferred taxes, net (1)         68         73           Other (2)         68         73           TABLILITIES AND SHAREHOLDERS' EQUITY         8         78           Current portion of long-term debt         \$175         \$15           Accoural portion of long-term debt         \$175         \$15           Accoural portion of long-term debt         \$1         42           Accoural portion of long-term debt         \$1         42           Accoural portion of long-term debt         \$1         42           Accourage long district of seste held for sale         10         12           Liabilities of assets held for sale         2         10           Liabilities coloratinued operations         12         2			
Notes receivable         21         27           Loans to equity method investees (f)         408         316           Other notes receivable         600         560           Other notes receivable         176         178           Other long-term receivables         678         665           Other long-term receivables         678         665           Other (f)         688         735           Stage (f)         688         735           Current (f)         5 175         5 15           Current portion of long-term debt         5 175         5 15           Accounts payable (f)         789         658           Accrued payroll and benefits         642         614           Liabilities of assets held of sale         -         101         178           Liabilities of assets held for sale         -         102         174           Liabilities of assets held for sale         -         101         178           Liabilities of assets held for sale         -         20           Liabilities of assets held for sale         -         20           Liabilities of assets held for sale         -         10         17           Liabilities prelated to discontinued operations			
Loans to equity method investees (°)         421         27           Loans to timeshare owners         408         316           Other notes receivable         600         560           Other long-term receivables         178         660           Other One descriptions         678         668         735           Other (°)         688         735         \$8.802         \$8.802           Element Liabilities         8.942         \$8.802         \$8.802           LIABILITIES AND SHAREHOLDERS' EQUITY         789         658         8.75         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$15         \$15         \$4.60         \$1         \$10         \$10         \$10 </td <td>Equity and cost method investments (1)</td> <td>343</td> <td>402</td>	Equity and cost method investments (1)	343	402
Loans to timeshare owners         488         316           Other notes receivable         670         500           Other long-term receivables         176         178           Deferred taxes, net (1)         678         665           Other long-term receivables         678         665           Other (1)         688         735           Deferred taxes, net (1)         678         665           Other (1)         5894         5886           Liabilities         878         658           Current portion of long-term debt         789         658           Accounts payable (1)         789         658           Accounts payable (1)         614         614           Account payroll and benefits         614         614           Liabilities of assets held for sale         -         102           Liabilities of assets held for sale         -         101         178           Liabilities related to discontinued operations         13         55           Other payables and accruals (1)         273         516           Liabilities related to discontinued operations         279         1818           Liability for guest loyalty program         97         434			
Other notes receivable         171         217           Chel long-term receivables         176         650           Other long-term receivables         678         665           Deferred taxes, net (1)         688         735           Cher (1)         688         735           Name of the company of			
Other long-term receivables         176         178           Deferred taxes, net (1)         668         655           Other (1)         688         735           Other (2)         688         735           CHABILITIES AND SHAREHOLDERS' EQUITY           Current portion of long-term debt         \$175         \$15           Accounts payable (1)         789         658           Accrued payroll and benefits         642         614           Liabilities of assets held for sale         642         614           Liabilities of assets held for sale         9         102           Timeshare segment deferred revenue         101         178           Liabilities related to discontinued operations         13         55           Other payables and accruals (1)         735         516           Other payables and accruals (1)         13         55           Other payables and accruals (1)         2,970         1,818           Liability for guest loyalty program         9,71         847           Self-insurance reserves         182         484           Other long-term liabilities (1)         5         5           Starting and the payables and accruals (1)         5         5         5			
Other long-term receivables         176         178           Deferred taxes, net (1)         678         665           Other (1)         688         735           158         58,942         \$5,858           LABILITIES AND SHAREHOLDERS' EQUITY           Current liabilities           Current portion of long-term debt         \$175         \$15           Accounts payable (1)         62         614           Accumed payroll and benefits         62         614           Liabilities of assets held for sale         -         102           Timeshare segment deferred revenue         101         178           Liabilities related to discontinued operations         13         55           Other payables and accruals (1)         735         151           Liability for guest loyalty program         13         55           Other payables and accruals (1)         735         151           Liability for guest loyalty program         97         184           Self-insurance reserves         181         36           Other long-term liabilities (1)         5         5         5           Scheinbolders' equity         5         5         5         5           Class A Common St	Other notes receivable		
Deferred taxes, net (i)         678         685         735           Other (i)         688         735<		600	560
Deferred taxes, net (i)         678         685         735           Other (i)         688         735<	Other long-term receivables	176	178
Other (1)         688         735           1 Sayson           LIABILITIES AND SHAREHOLDERS' EQUITY           Current liabilities           Current portion of long-term debt         \$15         \$15           Accruent payable (1)         789         658           Accrued payroll and benefits         612         614           Liability for guest loxylay program         61         12           Liabilities related to discontinued operations         10         178           Chier payables and accrusls (1)         13         55           Other payables and accrusls (1)         279         181           Liability for guest loxylay program         181         36         15           Liability for guest loxylay program         97         25         15           Liability for guest loxylay program         97         84           Liability for guest loxylay program         97         84           Self-insurance reserves         182         184           Other long-term liabilities (1)         5         5           Schrolders' equity         5         5           Class A Common Stock         5         5           Additional paid-in-capital         3,51		678	665
Current liabilities		688	
Current liabilities       Current portion of long-term debt       \$ 15       \$ 15         Accounts payable (1)       789       658         Accrued payroll and benefits       642       614         Liability for guest loyalty program       421       384         Liabilities of assets held for sale       —       102         Timeshare segment deferred revenue       101       178         Liabilities related to discontinued operations       13       55         Other payables and accruals (1)       735       516         Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       59       5         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       3,090         Accumulated other comprehensive income       51       44         4       4,000       1,429       2,618		\$ 8,942	\$ 8,588
Current portion of long-term debt         \$ 15         \$ 15           Accounts payable (1)         789         658           Accrued payroll and benefits         642         614           Liability for guest loyalty program         421         384           Liabilities of assets held for sale         —         102           Timeshare segment deferred revenue         101         178           Liabilities related to discontinued operations         13         55           Other payables and accruals (1)         2,735         516           Long-term debt         2,790         1,818           Liability for guest loyalty program         971         847           Self-insurance reserves         182         184           Other long-term liabilities (1)         694         599           Shareholders' equity         5         5           Class A Common Stock         5         5           Additional paid-in-capital         3,531         3,617           Retained earnings         3,332         2,860           Treasury stock, at cost         (5,490)         3,908           Accumulated other comprehensive income         5         4           4         4         4         4	LIABILITIES AND SHAREHOLDERS' EQUITY	<del></del>	
Accounts payable (1)       789       658         Accrued payroll and benefits       642       614         Liability for guest loyalty program       421       384         Liabilities of assets held for sale       —       102         Timeshare segment deferred revenue       101       178         Liabilities related to discontinued operations       13       55         Other payables and accruals (1)       735       516         Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618	Current liabilities		
Accrued payroll and benefits       642       614         Liability for guest loyalty program       421       384         Liabilities of assets held for sale       —       102         Timeshare segment deferred revenue       101       178         Liabilities related to discontinued operations       13       55         Other payables and accruals (1)       735       516         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618	Current portion of long-term debt	\$ 175	\$ 15
Liability for guest loyalty program       421       384         Liabilities of assets held for sale       —       102         Timeshare segment deferred revenue       101       178         Liabilities related to discontinued operations       13       55         Other payables and accruals (1)       735       516         Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         4,429       2,618	Accounts payable (1)	789	658
Liabilities of assets held for sale       —       102         Timeshare segment deferred revenue       101       178         Liabilities related to discontinued operations       13       55         Other payables and accruals (1)       735       516         Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       2       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         4       1,429       2,618		642	614
Timeshare segment deferred revenue       101       178         Liabilities related to discontinued operations       13       55         Other payables and accruals (1)       735       516         Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618		421	
Liabilities related to discontinued operations       13       55         Other payables and accruals (1)       735       516         Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618		_	
Other payables and accruals (1)       735       516         2,876       2,522         Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         4       1,429       2,618			
Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618			
Long-term debt       2,790       1,818         Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618	Other payables and accruals (1)		
Liability for guest loyalty program       971       847         Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       5       5         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618		2,876	2,522
Self-insurance reserves       182       184         Other long-term liabilities (1)       694       599         Shareholders' equity       Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618		2,790	1,818
Other long-term liabilities (1)       694       599         Shareholders' equity       Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618		971	847
Shareholders' equity         Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618		182	184
Class A Common Stock       5       5         Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618	Other long-term liabilities (1)	694	599
Additional paid-in-capital       3,531       3,617         Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618			
Retained earnings       3,332       2,860         Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618			
Treasury stock, at cost       (5,490)       (3,908)         Accumulated other comprehensive income       51       44         1,429       2,618			
Accumulated other comprehensive income         51         44           1,429         2,618	Retained earnings		
1,429 2,618	Treasury stock, at cost		
	Accumulated other comprehensive income		
\$ 8,942 \$ 8,588		1,429	2,618
		\$ 8,942	\$ 8,588

<sup>(1)</sup> See Footnote No. 21, "Related Party Transactions," of the Notes to Consolidated Financial Statements for disclosure of related party amounts.

# MARRIOTT INTERNATIONAL, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS Fiscal Years 2007, 2006, and 2005 (\$ in millions)

ODED ATTINIC A CITIN METER.	2007	2006	2005
OPERATING ACTIVITIES	ф. coc	Ф. СОО	Ф. 660
Net income	\$ 696	\$ 608	\$ 669
Adjustments to reconcile to cash provided by operating activities:	107	100	104
Depreciation and amortization	197	188	184
Minority interest	(3)	(7)	(44)
Income taxes	(150)	(76)	(86)
Timeshare activity, net	(155)	(104)	(35)
Liability for guest loyalty program	122	113	125
Cumulative effect of change in accounting principle		109	_
Working capital changes and other	71	139	27
Net cash provided by operating activities	778	970	840
INVESTING ACTIVITIES			
Capital expenditures	(671)	(529)	(780)
Dispositions	745	798	298
Loan advances	(31)	(59)	(56)
Loan collections and sales	106	121	706
Equity and cost method investments	(40)	(95)	(216)
Purchase of available-for-sale securities		(27)	(15)
Sale of available-for-sale securities	43	_	
Other	(27)	(90)	(67)
Net cash provided by (used in) investing activities	125	119	(130)
FINANCING ACTIVITIES			
Commercial paper, net	258	(188)	499
Issuance of long-term debt	820	352	356
Repayment of long-term debt	(153)	(17)	(523)
Debt exchange consideration, net	_	_	(29)
Issuance of Class A Common Stock	203	378	125
Dividends paid	(105)	(93)	(84)
Purchase of treasury stock	(1,757)	(1,546)	(1,644)
Other	(28)	15	26
Net cash used in financing activities	(762)	(1,099)	(1,274)
INCREASE (DECREASE) IN CASH AND EQUIVALENTS	141	(10)	(564)
CASH AND EQUIVALENTS, beginning of year	191	201	765
CASH AND EQUIVALENTS, end of year	\$ 332	\$ 191	\$ 201

# MARRIOTT INTERNATIONAL, INC. CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME Fiscal Years 2007, 2006, and 2005 (\$ in millions)

	2007	2006	2005
Net income	\$696	\$608	\$669
Other comprehensive income (loss), net of tax:			
Foreign currency translation adjustments	35	28	(25)
Other derivative instrument adjustments	(2)	_	(2)
Unrealized losses on available-for-sale securities	(8)	27	7
Reclassification of gains upon sale of available-for-sale securities	(18)	_	_
Total other comprehensive income (loss)	7	55	(20)
Comprehensive income	\$703	\$663	\$649

# MARRIOTT INTERNATIONAL, INC. CONSOLIDATED STATEMENTS OF SHAREHOLDERS' EQUITY Fiscal Years 2007, 2006, and 2005

(in millions, except per share amounts)

	Common Shares utstanding		Class A Common Stock	Additional Paid-in- Capital	Deferred Compensation	Retained Earnings	Treasury Stock, at Cost	Accumulated Other Comprehensive Income (Loss)
	451.6	Balance at fiscal year-end 2004	\$ 5	\$ 3,421	\$ (108)	\$ 1,951	\$(1,197)	\$ 9
		Net income			_	669		
	_	Dividends (\$0.2000 per share)	_	_	_	(87)	_	_
	11.6	Employee stock plan issuance and other		141	(29)	(33)	180	(20)
	(51.4)	Purchase of treasury stock		_	_	_	(1,650)	_
	411.8	Balance at fiscal year-end 2005	5	3,562	(137)	2,500	(2,667)	(11)
	_	Net income	_	_	_	608	_	_
	_	Dividends (\$0.2400 per share)	_	_	_	(96)	_	_
	19.2	Employee stock plan issuance and other	_	192	_	(152)	343	55
	_	Impact of adoption of FAS No. 123(R)		(137)	137	_	_	_
	(41.5)	Purchase of treasury stock	_	_	_	_	(1,584)	_
	389.5	Balance at fiscal year-end 2006	5	3,617	_	2,860	(3,908)	44
	_	Impact of adoption of FAS No.156	_	_	_	1	_	_
	_	Impact of adoption of FIN 48	_	(121)	_	(34)	_	_
	389.5	Opening balance fiscal year 2007	5	3,496	_	2,827	(3,908)	44
	_	Net income	_	_	_	696	_	_
	_	Dividends (\$0.2875 per share)	_	_	_	(107)	_	_
	8.6	Employee stock plan issuance and other	_	35	_	(84)	195	7
	(41.0)	Purchase of treasury stock	_	_	_		(1,777)	_
_	357.1	Balance at fiscal year-end 2007	\$ 5	\$ 3,531	\$ —	\$ 3,332	\$(5,490)	\$ 51

# MARRIOTT INTERNATIONAL, INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

# SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Presentation

The consolidated financial statements present the results of operations, financial position, and cash flows of Marriott International, Inc. (together with its subsidiaries, "we," "us," or the "Company").

The preparation of financial statements in conformity with U.S. generally accepted accounting principles ("GAAP") requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities as of the date of the financial statements, the reported amounts of revenues and expenses during the reporting periods, and the disclosures of contingent liabilities. Accordingly, ultimate results could differ from those estimates.

As a result of the discontinuation of our synthetic fuel business on November 3, 2007, the balances and activities of the synthetic fuel reportable segment have been segregated and reported as discontinued operations for all periods presented.

At the beginning of our 2006 fiscal year, we adopted Statement of Position 04-2, "Accounting for Real Estate Time-Sharing Transactions," ("SOP 04-2") as issued by the American Institute of Certified Public Accountants. The American Resort Development Association, a timeshare trade association of which we are a member and the Staff of the Securities and Exchange Commission ("SEC") had communications regarding SOP 04-2 and the income statement presentation of timeshare note securitization gains. As a result of those communications, we now classify Timeshare segment note securitization gains within our "Timeshare sales and services" revenue caption. Accordingly, Timeshare sales and services revenue reflects Timeshare segment note securitization gains of \$81 million and \$77 million for 2007 and 2006, respectively. Gains from the sale of timeshare notes receivable totaling \$69 million for 2005, are classified within the "Gains and other income" caption in our Consolidated Statements of Income.

We have reclassified certain prior year amounts to conform to our 2007 presentation. In our opinion, the accompanying consolidated financial statements reflect all normal and recurring adjustments necessary to present fairly our financial position at fiscal year-end 2007 and fiscal year-end 2006 and the results of our operations and cash flows for fiscal years 2007, 2006, and 2005. We have eliminated all material intercompany transactions and balances between entities consolidated in these financial statements.

#### Fiscal Year

Our fiscal year ends on the Friday nearest to December 31. Unless otherwise specified, each reference to a particular year means the fiscal year ended on the date shown in the following table, rather than the corresponding calendar year:

Fiscal Year	Fiscal Year-end Date	Fiscal Year	Fiscal Year-end Date
2007	December 28, 2007	2003	January 2, 2004
2006	December 29, 2006	2002	January 3, 2003
2005	December 30, 2005	2001	December 28, 2001
2004	December 31, 2004	2000	December 29, 2000

# Revenue Recognition

Our revenues include: (1) base management and incentive management fees; (2) franchise fees; (3) revenues from lodging properties and other businesses owned or leased by us; (4) timeshare sales and services, which also includes resort rental revenue, interest income associated with our "Loans to timeshare owners," and for 2006 and 2007, Timeshare segment note securitization gains, as noted in the "Basis of Presentation" caption earlier; and (5) cost reimbursements. Management fees comprise a base fee, which is a percentage of the revenues of hotels, and an incentive fee, which is generally based on hotel profitability. Franchise fees comprise initial application fees and continuing royalties generated from our franchise programs, which permit the hotel owners and operators to use certain of our brand names. Cost reimbursements include direct and indirect costs that are reimbursed to us by lodging properties that we manage or franchise.

<u>Base Management and Incentive Management Fees</u>: We recognize base management fees as revenue when earned in accordance with the contract. In interim periods and at year-end, we recognize incentive management fees that would be due as if the contract were to terminate at that date, exclusive of any termination fees payable or receivable by us.

*Franchise Fee Revenue*: We recognize franchise fees as revenue in each accounting period as fees are earned from the franchisee.

<u>Owned and Leased Units</u>: We recognize room sales and revenues from other guest services for our owned and leased units when rooms are occupied and services have been rendered.

<u>Timeshare and Fractional Intervals and Condominiums</u>: We recognize sales when: (1) we have received a minimum of 10 percent of the purchase price; (2) the purchaser's period to cancel for a refund has expired; (3) we deem the receivables to be collectible; and (4) we have attained certain minimum sales and construction levels. We defer all revenue using the deposit method for sales that do not meet all four of these criteria. For sales that do not qualify for full revenue recognition as the project has progressed beyond the preliminary stages but has not yet reached completion, all revenue and profit are deferred and recognized in earnings using the percentage of completion method.

<u>Timeshare Residential (Stand-Alone Structures)</u>: We recognize sales under the full accrual method of accounting when we receive our proceeds and transfer title at settlement.

<u>Cost Reimbursements</u>: We recognize cost reimbursements from managed, franchised, and timeshare properties when we incur the related reimbursable costs.

Other Revenue includes third-party licensing fees, other branding fees, land rental income, and other revenue.

# Ground Leases

We are both the lessor and lessee of land under long-term operating leases, which include scheduled increases in minimum rents. We recognize these scheduled rent increases on a straight-line basis over the initial lease term.

#### Real Estate Sales

We account for the sales of real estate in accordance with Financial Accounting Standards ("FAS") No. 66, "Accounting for Sales of Real Estate" ("FAS No. 66"). We reduce gains on sales of real estate by the maximum exposure to loss if we have continuing involvement with the property and do not transfer substantially all of the risks and rewards of ownership. In sales transactions where we retain a management contract, the terms and conditions of the management contract are generally comparable to the terms and conditions of the management contracts obtained directly with third-party owners in competitive bid processes.

# Profit Sharing Plan

We contribute to a profit sharing plan for the benefit of employees meeting certain eligibility requirements and electing participation in the plan. Contributions are determined based on a specified percentage of salary deferrals by participating employees. We recognized compensation costs from profit sharing of \$107 million in 2007, \$86 million in 2006, and \$69 million in 2005.

# Self-Insurance Programs

We are self-insured for certain levels of property, liability, workers' compensation and employee medical coverage. We accrue estimated costs of these self-insurance programs at the present value of projected settlements for known and incurred but not reported claims. We use a discount rate of 4.7 percent to determine the present value of the projected settlements, which we consider to be reasonable given our history of settled claims, including payment patterns and the fixed nature of the individual settlements.

We are subject to a variety of assessments related to our insurance activities, including those by state guaranty funds and workers' compensation second-injury funds. Our liabilities recorded for assessments are reflected within the amounts shown in our Consolidated Balance Sheets on the self-insurance reserves line, are not discounted, and totaled \$5 million for both year-end 2007 and year-end 2006. Our liability of \$5 million as of year-end 2007 for assessments is expected to be paid by the end of 2008.

#### Marriott Rewards

Marriott Rewards is our frequent guest loyalty program. Marriott Rewards members earn points based on their monetary spending at our lodging operations, purchases of timeshare interval, fractional ownership, and residential products and, to a lesser degree, through participation in affiliated partners' programs, such as those offered by airlines and credit card companies. Points, which we track on members' behalf, can be redeemed for stays at most of our lodging operations, airline tickets, airline frequent flyer program miles, rental cars, and a variety of other awards; however, points cannot be redeemed for cash. We provide Marriott Rewards as a marketing program to participating properties. We charge the cost of operating the program, including the estimated cost of award redemption, to properties based on members' qualifying expenditures.

We defer revenue received from managed, franchised, and Marriott-owned/leased hotels and program partners equal to the fair value of our future redemption obligation. We determine the fair value of the future redemption obligation based on statistical formulas that project timing of future point redemption based on historical levels, including an estimate of the "breakage" for points that will never be redeemed, and an estimate of the points that will eventually be redeemed. These judgment factors determine the required liability for outstanding points.

Our management and franchise agreements require that we be reimbursed currently for the costs of operating the program, including marketing, promotion, communication with, and performing member services for the Marriott Rewards members. Due to the requirement that hotels reimburse us for program operating costs as incurred, we receive and recognize the balance of the revenue from properties in connection with the Marriott Rewards program at the time such costs are incurred and expensed. We recognize the component of revenue from program partners that corresponds to program maintenance services over the expected life of the points awarded. Upon the redemption of points, we recognize as revenue the amounts previously deferred and recognize the corresponding expense relating to the costs of the awards redeemed. Our liability for the Marriott Rewards program was \$1,392 million at year-end 2007 and \$1,231 million at year-end 2006.

#### Guarantees

We record a liability for the fair value of a guarantee on the date a guarantee is issued or modified. The offsetting entry depends on the circumstances in which the guarantee was issued. Funding under the guarantee reduces the recorded liability. When no funding is forecasted, the liability is amortized into income on a straight-line basis over the remaining term of the guarantee.

# Rebates and Allowances

We participate in various vendor rebate and allowance arrangements as a manager of hotel properties. There are three types of programs that are common in the hotel industry that are sometimes referred to as "rebates" or "allowances," including unrestricted rebates, marketing (restricted) rebates and sponsorships. The primary business purpose of these arrangements is to secure favorable pricing for our hotel owners for various products and services or enhance resources for promotional campaigns co-sponsored by certain vendors. More specifically, unrestricted rebates are funds returned to the buyer, generally based upon volumes or quantities of goods purchased. Marketing (restricted) allowances are funds allocated by vendor agreements for certain marketing or other joint promotional initiatives. Sponsorships are funds paid by vendors, generally used by the vendor to gain exposure at meetings and events, which are accounted for as a reduction of the cost of the event.

We account for rebates and allowances as adjustments of the prices of the vendors' products and services. We show vendor costs and the reimbursement of those costs as reimbursed costs and cost reimbursements revenue, respectively; therefore, rebates are reflected as a reduction of these line items.

# Cash and Equivalents

We consider all highly liquid investments with an initial maturity of three months or less at date of purchase to be cash equivalents.

# Restricted Cash

Restricted cash, totaling \$153 million and \$117 million at year-end 2007 and year-end 2006, respectively, is recorded in the "Other long-term assets" line in the accompanying Consolidated Balance Sheets. Restricted cash primarily consists of deposits received on timeshare interval, fractional ownership, and residential sales that are held in escrow until the contract is closed.

# Assets Held for Sale

We consider properties (other than Timeshare segment interval, fractional ownership, and residential products, which we classify as inventory) to be assets held for sale when all of the following criteria are met:

- management commits to a plan to sell a property;
- it is unlikely that the disposal plan will be significantly modified or discontinued;
- the property is available for immediate sale in its present condition;
- actions required to complete the sale of the property have been initiated;
- sale of the property is probable and we expect the completed sale will occur within one year; and
- the property is actively being marketed for sale at a price that is reasonable given its current market value.

Upon designation as an asset held for sale, we record the carrying value of each property at the lower of its carrying value or its estimated fair value, less estimated costs to sell, and we cease depreciation.

Assets held for sale totaled \$123 million at year-end 2007 and consisted of property and equipment. The \$123 million total reflected the following segment composition: North American Full-Service-\$17 million; North American Limited-Service-\$17 million; and Luxury-\$89 million. There were no liabilities of assets held for sale at year-end 2007.

Assets held for sale totaled \$411 million at year-end 2006 and consisted of property and equipment of \$391 million, accounts receivable of \$10 million, cash of \$6 million, and other assets of \$4 million. The \$411 million total reflected the following segment composition: International Lodging-\$295 million; Luxury Lodging-\$73 million; and North American Full-Service Lodging-\$43 million. Liabilities of assets held for sale totaled \$102 million at year-end 2006 and consisted of debt totaling \$81 million, accounts payable of \$11 million, accrued payroll and benefits of \$8 million, and other payables and accruals of \$2 million.

In 2007 we reclassified the balances associated with one property, in conformity with other "held and used" properties, as the property no longer satisfied the criteria to be classified as "held for sale." In conjunction with that reclassification, we recorded depreciation expense of \$4 million in 2007 that would have been recognized in 2006 and \$4 million in late 2007 that would have been recognized earlier in 2007 had the asset been continuously classified as "held and used."

# Loan Loss Reserves

# <u>Lodging Senior Loans and Lodging Mezzanine and Other Loans</u>

We measure loan impairment based on the present value of expected future cash flows discounted at the loan's original effective interest rate or the estimated fair value of the collateral. For impaired loans, we establish a specific impairment reserve for the difference between the recorded investment in the loan and the present value of the expected future cash flows or the estimated fair value of the collateral. We apply our loan impairment policy individually to all loans in the portfolio and do not aggregate loans for the purpose of applying such policy. For loans that we have determined to be impaired, we recognize interest income on a cash basis.

# Loans to Timeshare Owners

We record an estimate of expected uncollectibility on notes receivable that we receive from timeshare purchasers as a reduction of revenue at the time we recognize profit on a timeshare sale. We assess uncollectibility based on pools of receivables because we hold large numbers of homogeneous timeshare notes receivable. We estimate uncollectibles based on historical activity for similar timeshare notes receivable over the past three years. We use a technique referred to as static pool analysis, which tracks uncollectibles for each year's sales over the life of those notes.

# Valuation of Goodwill

We evaluate the fair value of goodwill to assess potential impairments on an annual basis, or during the year if an event or other circumstance indicates that we may not be able to recover the carrying amount of the asset. We evaluate the fair value of goodwill at the reporting unit level and make that determination based upon future cash flow projections that assume certain growth projections which may or may not occur. We record an impairment loss for goodwill when the carrying value of the intangible asset is less than its estimated fair value.

#### Investments

We consolidate entities that we control. We account for investments in joint ventures using the equity method of accounting when we exercise significant influence over the venture. If we do not exercise significant influence, we account for the investment using the cost method of accounting. We account for investments in limited partnerships and limited liability companies using the equity method of accounting when we own more than a minimal investment. Our ownership interest in these equity method investments varies generally from 10 percent to 50 percent.

The fair value of our available-for-sale securities totaled \$55 million and \$107 million at year-end 2007 and year-end 2006, respectively. We included net unrealized holding gains on available-for-sale securities of \$9 million at year-end 2007 and \$35 million at year-end 2006 in accumulated other comprehensive income. The amount of net gains reclassified out of accumulated other comprehensive income as a result of the sale of available-for-sale securities totaled \$18 million and zero for 2007 and 2006, respectively. We determined the cost basis of the securities sold using specific identification.

#### Costs Incurred to Sell Real Estate Projects

We charge the majority of sales and marketing costs we incur to sell timeshares to expense when incurred. Selling and marketing costs capitalized were \$6 million at year-end 2007 and \$16 million at year-end 2006 and are included in the accompanying Consolidated Balance Sheets in the "Other" caption within the "Current assets" section. If a contract is canceled, we charge unrecoverable direct selling and marketing costs to expense and record deposits forfeited as income.

## Residual Interests

We periodically sell notes receivable originated by our Timeshare segment in connection with the sale of timeshare interval and fractional products. We continue to service the notes and transfer all proceeds collected to special purpose entities. We retain servicing assets and other interests in the notes and account for these assets and interests as residual interests. The interests are limited to the present value of cash available after paying financing expenses and program fees and absorbing credit losses. Prior to the start of the 2007 fiscal year, we measured servicing assets at the date of sale at their allocated previous carrying amount based on relative fair value, classified those assets as held to maturity under the provisions of FAS No. 115, "Accounting for Certain Investments in Debt and Equity Securities" ("FAS No. 115"), and recorded those assets at amortized cost. On December 30, 2006, the first day of fiscal year 2007, we adopted FAS No. 156, "Accounting for Servicing of Financial Assets-an Amendment of Financial Accounting Standards Board ("FASB") Statement No. 140" ("FAS No. 156"). FAS No. 156 requires that all separately recognized servicing assets and liabilities be initially measured at fair value, if practicable. It also allows an entity to subsequently elect fair value measurement for its servicing assets and liabilities. In conjunction with the adoption of FAS No. 156, we elected to subsequently measure our servicing assets using the fair value method. Under the fair value method, we carry servicing assets on the balance sheet at fair value and report the changes in fair value, primarily due to changes in valuation inputs and assumptions and to the collection or realization of expected cash flows, in earnings in the period in which the change occurs. For additional information regarding the adoption of FAS No. 156, see Footnote No. 11, "Asset Securitizations."

We treat the residual interests, including servicing assets, as "trading" securities under the provisions of FAS No. 115. At the dates of sale and at the end of each reporting period, we estimate the fair value of the residual interests, including servicing assets, using a discounted cash flow model. We report changes in the fair values of these residual interests, including servicing assets, through the accompanying Consolidated Statements of Income.

The rate of prepayment of loans serviced is the most significant estimate involved in the measurement process. Estimates of prepayment rates are based on management's expectations of future prepayment rates, reflecting our historical rate of loan repayments, industry trends, and other considerations. Actual prepayment rates differ from those projected by management due to changes in a variety of economic factors, including prevailing interest rates and the availability of alternative financing sources to borrowers. If actual prepayments of the loans being serviced were to occur more slowly than had been projected, the carrying value of servicing assets could increase, and servicing income would exceed previously projected amounts. Accordingly, the servicing assets actually realized, could differ from the amounts initially recorded.

# Derivative Instruments

We use derivative instruments as part of our overall strategy to manage our exposure to market risks associated with fluctuations in interest rates and foreign currency exchange rates. As a matter of policy, we do not use derivatives for trading or speculative purposes.

We record all derivatives at fair value either as assets or liabilities. We recognize, currently in earnings, changes in fair value of derivatives not designated as hedging instruments and of derivatives designated as fair value hedging instruments. We record changes in the fair value of the hedged item in a fair value hedge as an adjustment to the carrying amount of the hedged item and recognize the change in the fair value of the derivative in earnings in the same income statement line item.

We record the effective portion of changes in fair value of derivatives designated as cash flow hedging instruments as a component of other comprehensive income and report the ineffective portion currently in earnings. We reclassify amounts included in other comprehensive income into earnings in the same period during which the hedged item affects earnings.

#### Foreign Operations

The U.S. dollar is the functional currency of our consolidated and unconsolidated entities operating in the United States. The functional currency for our consolidated and unconsolidated entities operating outside of the United States is generally the currency of the primary economic environment in which the entity primarily generates and expends cash. We translate the financial statements of consolidated entities whose functional currency is not the U.S. dollar, and we do the same, as needed, for unconsolidated entities whose functional currency is not the U.S. dollar. We translate assets and liabilities at the exchange rate in effect as of the financial statement date and translate income statement accounts using the weighted average exchange rate for the period. We include translation adjustments from foreign exchange and the effect of exchange rate changes on intercompany transactions of a long-term investment nature as a separate component of shareholders' equity. We report gains and losses from foreign exchange rate changes related to intercompany receivables and payables that are not of a long-term investment nature, as well as gains and losses from foreign currency transactions, currently in operating costs and expenses, and those amounted to a \$2 million loss in 2007, a \$6 million gain in 2006, and a \$5 million loss in 2005. Gains and other income for 2007 included \$6 million attributable to currency translation adjustment gains, net of losses, from the sale or complete or substantially complete liquidation of investments. There were no similar gains or losses in 2006 or 2005.

#### New Accounting Standards

# FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes"

We adopted FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes," ("FIN 48") on December 30, 2006, the first day of our 2007 fiscal year. FIN 48 is an interpretation of FASB Statement No. 109, "Accounting for Income Taxes," which seeks to standardize practices associated with certain aspects of measurement and recognition in accounting for income taxes. FIN 48 prescribes a recognition threshold and measurement requirement for the financial statement recognition of a tax position taken or expected to be taken on a tax return. Additionally, FIN 48 provides guidance on derecognition, classification, interest and penalties, accounting in interim periods, disclosure, and transition. Under FIN 48, an entity may only recognize or continue to recognize tax positions that meet a "more likely than not" threshold. We recorded the cumulative effect of applying FIN 48 of \$155 million as an adjustment to the opening balance of retained earnings and additional paid-in-capital on December 30, 2006, the first day of our 2007 fiscal year. See Footnote No. 3, "Income Taxes," for additional information.

# Financial Accounting Standards No. 156, "Accounting for Servicing of Financial Assets-an Amendment of FASB Statement No. 140"

We adopted FASB's FAS No. 156 on December 30, 2006, the first day of our 2007 fiscal year. FAS No. 156 requires that all separately recognized servicing assets and liabilities initially be measured at fair value, if practicable. It also allows an entity to subsequently elect fair value measurement for its servicing assets and liabilities. We recorded the cumulative effect of applying FAS No. 156, of \$1 million, net of tax, as an adjustment to the opening balance of retained earnings on December 30, 2006. See Footnote No. 11, "Asset Securitizations," for additional information.

# Future Adoption of Accounting Standards

EITF Issue No. 06-8, "Applicability of the Assessment of a Buyer's Continuing Investment under FASB Statement No. 66, Accounting for Sales of Real Estate, for Sales of Condominiums"

In November 2006, the Emerging Issues Task Force of FASB ("EITF") reached a consensus on EITF Issue No. 06-8, "Applicability of the Assessment of a Buyer's Continuing Investment under FASB Statement No. 66, Accounting for Sales of Real Estate, for Sales of Condominiums" ("EITF 06-8"). EITF 06-8 will require condominium sales to meet the continuing investment criterion in FAS No. 66 in order to recognize profit under the

percentage of completion method. EITF 06-8 will be effective for annual reporting periods beginning after March 15, 2007, which for us begins with our 2008 fiscal year. The cumulative effect of applying EITF 06-8, if any, will be recorded as an adjustment to the opening balance of retained earnings in the year of adoption. We do not expect the impact of adoption of EITF 06-8 to be material.

## Financial Accounting Standards No. 157, "Fair Value Measurements"

In September 2006, the FASB issued FAS No. 157, "Fair Value Measurements" ("FAS No. 157"). This standard defines fair value, establishes a methodology for measuring fair value, and expands the required disclosure for fair value measurements. FAS No. 157 is effective for fiscal years beginning after November 15, 2007, which for us begins with our 2008 fiscal year. Provisions of FAS No. 157 must be applied prospectively as of the beginning of the first fiscal year in which FAS No. 157 is applied. In November 2007, the FASB agreed to partially defer the effective date, for one year, of FAS No. 157, for non-financial assets and liabilities, except those that are recognized or disclosed at fair value in the financial statements on a recurring basis. We are currently evaluating the impact that FAS No. 157 will have on our financial statements.

# Financial Accounting Standards No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities Including an Amendment of FASB Statement No. 115"

In February 2007, the FASB issued FAS No. 159, "The Fair Value Option for Financial Assets and Financial Liabilities Including an Amendment of FASB Statement No. 115" ("FAS No. 159"). This standard permits entities to choose to measure many financial instruments and certain other items at fair value and is effective for the first fiscal year beginning after November 15, 2007, which for us begins with our 2008 fiscal year. We do not expect to elect the fair value measurement option for any financial assets or liabilities at the present time.

# EITF Issue No. 07-6, "Accounting for Sales of Real Estate Subject to the Requirements of FASB Statement No. 66, 'Accounting for Sales of Real Estate,' When the Agreement Includes a Buy-Sell Clause"

In December 2007, the EITF reached a consensus on EITF Issue No. 07-6, "Accounting for Sales of Real Estate Subject to the Requirements of FASB Statement No. 66, 'Accounting for Sales of Real Estate,' When the Agreement Includes a Buy-Sell Clause" ("EITF 07-6"). EITF 07-6 clarifies whether a buy-sell clause is a prohibited form of continuing involvement that would preclude partial sales treatment under FAS No. 66. EITF 07-6 is effective for new arrangements entered into and assessments of existing transactions originally accounted for under the deposit, profit-sharing, leasing, or financing methods for reasons other than the exercise of a buy-sell clause performed in fiscal years beginning after December 15, 2007, which for us begins with our 2008 fiscal year. We do not expect EITF 07-6 to have a material impact on our financial statements.

# Financial Accounting Standards No. 141 (Revised 2007), "Business Combinations"

On December 4, 2007, the FASB issued FAS No. 141 (Revised 2007), "Business Combinations" ("FAS No. 141(R)"). FAS No. 141(R) will significantly change the accounting for business combinations. Under FAS No. 141(R), an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. FAS No. 141(R) also includes a substantial number of new disclosure requirements. FAS No. 141(R) applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008, which for us begins with our 2009 fiscal year. We are currently evaluating the impact that FAS No. 141(R) will have on our financial statements.

# Financial Accounting Standards No. 160, "Non-controlling Interests in Consolidated Financial Statements-an Amendment of ARB No. 51"

On December 4, 2007, the FASB issued FAS No. 160, "Non-controlling Interests in Consolidated Financial Statements-an Amendment of Accounting Research Bulletin ("ARB") No. 51" ("FAS No. 160"). FAS No. 160 establishes new accounting and reporting standards for a non-controlling interest in a subsidiary and for the deconsolidation of a subsidiary. Specifically, this statement requires the recognition of a non-controlling interest (minority interest) as equity in the consolidated financial statements separate from the parent's equity. The amount of net income attributable to the non-controlling interest will be included in consolidated net income on the face of the income statement. FAS No. 160 clarifies that changes in a parent's ownership interest in a subsidiary that do not result in deconsolidation are equity transactions if the parent retains its controlling financial interest. In addition, this statement requires that a parent recognize a gain or loss in net income when a subsidiary is deconsolidated. Such gain or loss will be measured using the fair value of the non-controlling equity investment on the deconsolidation date. FAS No. 160 also includes expanded disclosure requirements regarding the interests of the

parent and its non-controlling interest. FAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008, which for us begins with our 2009 fiscal year. We are currently evaluating the impact that FAS No. 160 will have on our financial statements.

## 2. DISCONTINUED OPERATIONS

Our synthetic fuel operations consisted of four coal-based synthetic fuel production facilities (the "Facilities"). Because tax credits under Section 45K of the Internal Revenue Code are not available for the production and sale of synthetic fuel produced from coal after calendar year-end 2007, and because high oil prices during 2007 will result in the phase-out of a significant portion of the tax credits available for synthetic fuel produced and sold in 2007, on November 3, 2007, we shut down the Facilities and permanently ceased production of synthetic fuel. Accordingly, we now report this business segment as a discontinued operation. The book value of the Facilities was zero at year-end 2007, as the Facilities have been transferred to third parties. Under the site leases for the Facilities, we were required to restore the leased premises to substantially the condition the premises were in when the leases were originally executed. However, we executed agreements with the lessors of the sites pursuant to which we transferred the Facilities to the lessors in exchange for the release of our obligations to restore the leased premises to their original condition. Costs associated with shutting down the synthetic fuel operation and transferring the Facilities to the site lessors were not material.

The following table provides additional income statement and balance sheet information relating to the discontinued operations:

# **Income Statement Summary**

(\$ in millions)	2007	2006	2005
Revenue	\$ 352	\$165	\$ 421
Operating loss	\$(113)	\$ (76)	\$(144)
Gains and other (expense) income	(6)	(15)	32
Interest expense	(8)	(4)	
Loss before income taxes and minority interest	(127)	(95)	(112)
Tax benefit Tax credits	46 80	32 62	23 167
Total tax benefit	126	94	190
(Loss) income before minority interest	(1)	(1)	78
Minority interest		6	47
(Loss) income on discontinued synthetic fuel business, net of tax	(1)	5	125
Income on discontinued distribution services business, net of tax			1
Discontinued operations, net of tax	\$ (1)	\$ 5	\$ 126

# **Balance Sheet Summary**

(\$ in millions)	At Year-end 2007	At Year-end 2006
Property, plant, and equipment	\$ <u> </u>	\$ 5
Other assets	53	86
Liabilities	(13)	(55)

# 3. INCOME TAXES

We adopted the provisions of FIN 48, on December 30, 2006, the first day of fiscal year 2007. As a result of the implementation of FIN 48, we recorded a \$155 million increase in the net liability for unrecognized tax positions, which was recorded as an adjustment to the opening balance of retained earnings and additional paid-in-capital on December 30, 2006. The total amount of unrecognized tax benefits as of year-end 2007 was \$132 million. Included in the balance at year-end 2007 were \$97 million of tax positions that, if recognized, would impact the effective tax rate. As a large taxpayer, we are under continual audit by the Internal Revenue Service ("IRS") and other taxing authorities on several open tax positions, and it is possible that the amount of the liability for unrecognized tax benefits could change during the next 52-week period. While it is possible that one or more of these examinations may be resolved in the next year, it is not anticipated that a significant impact to the unrecognized tax benefit balance will occur.

The unrecognized tax benefit reconciliation from beginning balance to ending balance is as follows:

(\$ in millions)	Amount			
Unrecognized tax benefit at beginning of year (December 30, 2006)	\$ 244			
Change attributable to tax positions taken during a prior period	163			
Change attributable to tax positions taken during the current period	5			
Decrease attributable to settlements with taxing authorities				
Decrease attributable to lapse of statute of limitations	(1)			
Unrecognized tax benefit at end of year (December 28, 2007)	\$ 132			

In accordance with our accounting policy, we recognize accrued interest and penalties related to unrecognized tax benefits as a component of tax expense. This policy did not change as a result of the adoption of FIN 48. Our Consolidated Statements of Income for the year ended December 28, 2007, and our Consolidated Balance Sheets as of that date include interest of \$18 million and \$8 million, respectively.

We file income tax returns, including returns for our subsidiaries, with federal, state, local, and foreign jurisdictions. We are participating in the IRS Compliance Assurance Program (CAP) for the 2006 and 2007 tax years and we intend to participate for 2008. This program accelerates the examination of key transactions with the goal of resolving any issues before the tax return is filed. Our 2005 federal income tax return is currently being examined by the IRS in a traditional audit process. In June 2007, we received IRS Revenue Agents Reports for both the 2000 through 2002 and 2003 through 2004 examination cycles. We have fully resolved all issues and are in the final stages of closing these years. Various state, local, and foreign income tax returns are also under examination by taxing authorities. We do not believe that the outcome of any examination will have a material impact on our financial statements.

As disclosed in our 2006 Form 10-K, the IRS was then auditing the Company's federal tax returns for the 2000, 2001, and 2002 fiscal years. As part of that audit, the IRS reviewed a leveraged employee stock ownership plan ("ESOP") feature of the Company's Employees' Profit Sharing, Retirement and Savings Plan (the "Plan") that was implemented in a transaction (the "ESOP transaction") on June 13, 2000. Principal and interest on the debt related to the transaction was forgiven over a 26-month period as a mechanism for funding Company contributions of elective deferrals and matching contributions to the Plan. We claimed federal income tax deductions for the forgiven principal on the debt in the amount of \$1 billion over that period, along with forgiven interest on the debt. The benefit related to the tax deductions was reflected in equity and did not flow through the provision for income taxes.

On June 7, 2007, we reached a settlement of issues raised during the IRS' and Department of Labor's examination of the ESOP feature of the Plan. The settlement resulted in an after-tax charge in the 2007 second quarter totaling \$54 million and a reduction in shareholders' equity of \$115 million. The \$54 million charge included \$35 million of excise taxes (impacting general, administrative, and other expense), \$13 million of interest expense on those excise taxes, and \$6 million of income tax expense primarily reflecting additional interest. As a result of the settlement, we have made cash payments to the U.S. Treasury and state tax jurisdictions of \$205 million through year-end 2007. The remaining cash payments of approximately \$1 million are expected to be made in 2008. The payments reflect income taxes, excise taxes, and interest charges. No penalties were assessed.

Total deferred tax assets and liabilities as of year-end 2007 and year-end 2006, were as follows:

(\$ in millions)	2007	2006
Deferred tax assets	\$1,013	\$1,042
Deferred tax liabilities	(150)	(177)
Net deferred taxes	\$ 863	\$ 865

The tax effect of each type of temporary difference and carry-forward that gives rise to a significant portion of deferred tax assets and liabilities as of year-end 2007 and year-end 2006, were as follows:

(\$ in millions)	2007_	2006
Self-insurance	\$ 27	\$ 36
Employee benefits	282	250
Deferred income	30	34
Other reserves	42	52
Frequent guest program	101	75
Tax credits	300	383
Net operating loss carry-forwards	132	126
Timeshare financing	(47)	(37)
Property, equipment, and intangible assets	(45)	(32)
Other, net	72	12
Deferred taxes	894	899
Less: valuation allowance	(31)	(34)
Net deferred taxes	\$863	\$865

At year-end 2007, we had approximately \$74 million of tax credits that expire through 2027, \$226 million of tax credits that do not expire, and \$656 million of net operating losses, of which \$360 million expire through 2027. The valuation allowance decreased due to foreign net operating losses that we believe will be realized and the expiration of state net operating losses.

We have made no provision for U.S. income taxes or additional foreign taxes on the cumulative unremitted earnings of non-U.S. subsidiaries (\$738 million as of year-end 2007) because we consider these earnings to be permanently invested. These earnings could become subject to additional taxes if remitted as dividends, loaned to us or a U.S. affiliate or if we sold our interests in the affiliates. We cannot practically estimate the amount of additional taxes that might be payable on the unremitted earnings. We conduct business in countries that grant a "holiday" from income taxes for 10- and 30-year periods. The holidays expire through 2034. The aggregate amount of taxes not incurred due to tax "holidays" and the related earnings per share impacts are \$22 million (\$0.06 per diluted share), \$22 million (\$0.05 per diluted share), and \$11 million (\$0.02 per diluted share) for 2007, 2006, and 2005, respectively.

The (provision for) benefit from income taxes consists of:

(\$ in millions	)	<u>2007</u> <u>200</u>	
Current	- Federal	\$(303) \$(2	(328) \$(328)
	- State	(59)	(61) (56)
	- Foreign	(61)(	$(56) \qquad (35)$
		(423) (4	(419)
Deferred	- Federal	(10)	5 108
	- State	1	1 19
	- Foreign	(9)	18 8
		(18)	24 135
		<u>\$(441)</u> <u>\$(3</u>	\$(284)

The current tax provision does not reflect the benefits attributable to us relating to the exercise of employee stock options of \$115 million in 2007, \$194 million in 2006, and \$87 million in 2005. Included in the preceding table are tax credits of \$4 million in each of 2007 and 2006, and \$5 million in 2005. The taxes applicable to other comprehensive income are not material.

A reconciliation of the U.S. statutory tax rate to our effective income tax rate for continuing operations follows:

	2007	2006	2005
U.S. statutory tax rate	35.0%	35.0%	2005 35.0%
State income taxes, net of U.S. tax benefit	2.8	2.9	3.0
Nondeductible expenses	1.4	0.3	0.3
Change in valuation allowance	_	(1.9)	1.7
Foreign income	(0.9)	(1.8)	(4.4)
Tax credits	(0.3)	(0.4)	(0.6)
Other, net	8.0	0.1	(0.4)
Effective rate	38.8%	34.2%	34.6%

Cash paid for income taxes, net of refunds, was \$350 million in 2007, \$169 million in 2006, and \$182 million in 2005.

#### 4. SHARE-BASED COMPENSATION

Under our 2002 Comprehensive Stock and Cash Incentive Plan ("the Comprehensive Plan"), we award: (1) stock options to purchase our Class A Common Stock ("Stock Option Program"); (2) share appreciation rights for our Class A Common Stock; (3) restricted stock units of our Class A Common Stock; and (4) deferred stock units. We grant awards at exercise prices or strike prices that are equal to the market price of our Class A Common Stock on the date of grant.

We adopted FAS No. 123(R) using the modified prospective transition method at the beginning of our 2006 first quarter. In accordance with the modified prospective transition method, we did not restate our Consolidated Financial Statements for prior periods to reflect the impact of FAS No. 123(R). For all share-based awards granted after the date we adopted FAS No. 123(R) and for the unvested portion of previously granted share-based awards that were outstanding on that date, FAS No. 123(R) requires that we measure compensation costs related to our share-based payment transactions at fair value on the grant date and that we recognize those costs in the financial statements over the vesting period during which the employee provides service in exchange for the award. Previously, under FAS No. 123 and APB Opinion No. 25, we accounted for our share-based employee compensation plans using the intrinsic value method under the recognition and measurement principles of APB Opinion No. 25 and recognized share-based compensation expense for all awards except for our Stock Option Program awards. We recorded share-based compensation expense related to award grants of \$62 million in 2005.

Under FAS No. 123(R), we recorded share-based compensation expense related to award grants of \$104 million in 2007 and \$108 million in 2006. Deferred compensation costs related to unvested awards totaled \$162 million at year-end 2007 and \$168 million at year-end 2006, and the weighted average period over which we expect the deferred compensation costs at year-end 2007 to be recognized is two years.

The following table illustrates the effect on net income and earnings per share as if we had applied the fair value recognition provisions of FAS No. 123 to share-based employee compensation in 2005. We have included the impact of measured but unrecognized compensation costs and excess tax benefits credited to additional paid-in-capital in the calculation of diluted pro forma shares. In addition, we have included the estimated impact of reimbursements from third parties. The reported and pro forma net income and earnings per share figures for 2007 and 2006 in the table are the same because we calculate share-based compensation expense under the provisions of FAS No. 123(R). We have included the 2007 and 2006 amounts in the table below to provide detail for comparative purposes to the 2005 amounts.

(\$ in millions, except per share amounts)	2007	2006	2005
Net income, as reported	\$ 696	\$ 608	\$ 669
Add: Share-based employee compensation expense included in reported net income, net of related tax effects	67	70	40
Deduct: Total share-based employee compensation expense determined under fair value-based method for all			
awards, net of related tax effects and estimated reimbursed costs	(67)	(70)	(61)
Pro forma net income	\$ 696	\$ 608	\$ 648
Earnings per share:			
Basic-as reported	\$1.85	\$1.50	\$1.55
Basic-pro forma	\$1.85	\$1.50	\$1.50
Diluted-as reported	\$1.75	\$1.41	\$1.45
Diluted-pro forma	\$1.75	\$1.41	\$1.40

FAS No. 123(R) requires that share-based compensation expense be recognized over the period from the grant date to the date on which the award is no longer contingent on the employee providing additional service (the "substantive vesting period"). In periods prior to the adoption of FAS No. 123(R), we showed share-based compensation expense in our pro forma disclosure only for option awards to retirement-eligible employees over the awards' stated vesting period. In periods prior to the adoption of FAS No. 123(R), we recorded share-based compensation expense for our other awards to retirement-eligible employees over the awards' stated vesting period. With the adoption of FAS No. 123(R), we continue to follow the stated vesting period for the unvested portion of awards granted prior to adoption of FAS No. 123(R) and follow the substantive vesting period for awards granted after the adoption of FAS No. 123(R).

In accordance with FAS No. 123(R), we present the tax benefits resulting from the exercise of share-based awards as financing cash flows. Prior to the adoption of FAS No. 123(R), we reported the tax benefits resulting from the exercise of share-based awards as operating cash flows. Tax benefits resulting from the exercise of share-based awards totaled \$115 million, \$194 million, and \$87 million for 2007, 2006, and 2005, respectively.

The aggregate amount of cash we received from the exercise of stock options granted under share-based payment arrangements was \$89 million, \$184 million, and \$125 million for 2007, 2006, and 2005, respectively.

We issue restricted stock units under the Comprehensive Plan to certain officers and key employees and those units vest generally over four years in annual installments commencing one year after the date of grant. We recognize compensation expense for the restricted stock units over the service period equal to the fair market value of the stock units on the date of issuance. Upon vesting, restricted stock units convert to shares and are distributed from treasury shares. At year-end 2007 and year-end 2006, we had approximately \$145 million and \$138 million, respectively, in deferred compensation costs related to restricted stock units. Share-based compensation expense associated with restricted stock units was \$82 million, \$77 million, and \$52 million for 2007, 2006, and 2005, respectively. The weighted average remaining term for restricted stock unit grants outstanding at year-end 2007 was two years. Restricted stock units converted and distributed during 2007, 2006, and 2005, had aggregate intrinsic values of \$147 million, \$78 million, and \$52 million, respectively. The weighted average grant-date fair values of restricted stock units granted in 2007, 2006, and 2005 were \$49, \$35, and \$32, respectively.

Changes in our outstanding restricted stock unit grants in 2007 were as follows:

	Number of Restricted Stock Units (in millions)	Average (	ghted Grant-Date Value
Outstanding at year-end 2006	7.6	\$	30
Granted during 2007	2.3		49
Distributed during 2007	(3.0)		26
Forfeited during 2007	(0.2)		37
Outstanding at year-end 2007	6.7		37

Employee stock options may be granted to officers and key employees at exercise prices or strike prices equal to the market price of our Class A Common Stock on the date of grant. Non-qualified options generally expire 10 years after the date of grant, except those issued from 1990 through 2000, which expire 15 years after the date of the grant. Most stock options under the Stock Option Program are exercisable in cumulative installments of one quarter at the end of each of the first four years following the date of grant.

We recognized compensation expense associated with employee stock options of \$12 million in 2007 and \$22 million in 2006. We did not recognize any compensation expense associated with employee stock options in 2005. At year-end 2007 and year-end 2006, there was approximately \$6 million and \$18 million, respectively, in deferred compensation costs related to employee stock options. Upon the exercise of stock options, we issue shares from treasury shares.

Changes in our outstanding Stock Option Program awards in 2007 were as follows:

	Number of Options (in millions)	Weighted Average Exercise Price
Outstanding at year-end 2006	43.9	\$ 17
Granted during 2007	<del>_</del>	49
Exercised during 2007	(6.2)	14
Forfeited during 2007		23
Outstanding at year-end 2007	37.7	17

Stock options issued under the Stock Option Program awards outstanding at year-end 2007, were as follows:

					Outstanding			Exercisable	
		Range of Exercise Prices		Number of Stock Options (in millions)	Weighted Average Remaining Life (in years)	Weighted Average Exercise Price	Number of Stock Options (in millions)	Weighted Average Exercise Price	Weighted Average Remaining Life (in years)
\$	6	to	\$ 8	2.7	2	\$ 7	2.7	\$ 7	2
	9	to	12	1.4	4	10	1.4	10	4
1	13	to	18	21.8	6	15	21.8	15	6
1	19	to	25	10.3	7	22	9.4	22	7
2	26	to	50	1.5	8	33	0.7	32	8
	6	to	50	37.7	6	17	36.0	17	6

The weighted average grant-date fair value of the 33,000 options granted in 2007 was \$19 and the options had a weighted average exercise price of \$49. The weighted average grant-date fair value of the 24,000 options granted in 2006 was \$13 and the options had a weighted average exercise price of \$34. The total intrinsic value of options outstanding at year-end 2007 and year-end 2006 was \$1,279 million and \$1,368 million, respectively, and the total intrinsic value for stock options exercisable as of year-end 2007 and year-end 2006 was \$624 million and \$1,233 million, respectively. The total intrinsic value of stock options exercised during 2007, 2006, and 2005 was approximately \$206 million, \$309 million, and \$173 million, respectively.

Employee share appreciation rights ("Employee SARs") may be granted to officers and key employees at exercise prices or strike prices equal to the market price of our Class A Common Stock on the date of grant. Employee SARs expire 10 years after the date of grant and generally both vest and are exercisable in cumulative installments of one quarter at the end of each of the first four years following the date of grant. Non-employee share appreciation rights ("Non-employee SARs") may be granted to directors at exercise prices or strike prices equal to the market price of

our Class A Common Stock on the date of grant. Non-employee SARs expire 10 years after the date of grant and vest upon grant; however, they are generally not exercisable until one year after grant. We first began issuing share appreciation rights in 2006. On exercise of SARs, employees or Non-employee directors receive the number of shares of Class A Common Stock equal to the number of share appreciation rights that are being exercised multiplied by the quotient of (a) the final value minus the base value, divided by (b) the final value.

We recognized compensation expense associated with Employee SARs and Non-employee SARs of \$5 million in 2007 and \$3 million in 2006. At the end of 2007 and 2006, we had approximately \$7 million and \$4 million, respectively, in deferred compensation costs related to share appreciation rights. Upon the exercise of share appreciation rights, shares are issued from treasury shares. During 2007 and 2006, we granted 0.4 million and 0.5 million, respectively, Employee SARs with a weighted average base value of \$49 and \$34, respectively, and a weighted average grant-date fair value of \$19 and \$13, respectively. During 2007 and 2006, we also granted 4,000 and 8,000, respectively, Non-employee SARs with a weighted average base value of \$46 and \$37, respectively, and a weighted average grant-date fair value of \$20 and \$18, respectively. No SARs have expired or been forfeited in 2007. The total intrinsic value of SARs outstanding at year-end 2007 and year-end 2006 was zero and \$7 million, respectively, and the total intrinsic value of SARs exercised during 2007 was \$100,000. No SARs were exercised in 2006.

We use a binomial method to estimate the fair value of each stock option or SAR granted. The assumptions for stock options for all years and employee SARs for 2007 and 2006 are noted in the following table:

	2007	2006	2005
Annual dividends	\$0.29	\$0.22	\$0.18
Expected volatility	28%	30%	30%
Risk-free interest rate	4.8%	4.5%	4.1%
Expected life (in years)	7	7	8

For Non-employee SARs issued in 2007 and 2006, the only differences in the assumptions versus employee SARs were the use of risk-free interest rates of 4.6 percent and 5.0 percent, respectively, and for each year's issuances, an expected life of 10 years.

Estimated volatilities for 2007 and 2006 were based on the historical share-price volatility for a period equal to the stock option's or share appreciation right's expected lives, ending on the day of grant, and calculated based on weekly data. The weighted average expected stock option or share appreciation right terms for 2007 and 2006 were a product of the lattice-based binomial valuation model that uses suboptimal exercise factors to calculate the expected terms.

We also issue deferred stock units to Non-employee directors. These Non-employee director deferred stock units vest within one year and are distributed upon election. At year-end 2007 and year-end 2006, there was approximately \$227,000 and \$152,000, respectively, in deferred costs related to Non-employee director deferred stock units. We recognized share-based expense associated with Non-employee director deferred stock units of \$666,000, \$492,000, and \$416,000 for 2007, 2006, and 2005, respectively. During 2007 we granted 20,000 Non-employee director deferred stock units with a weighted average grant-date fair value of \$46. For 2006 and 2005, we granted 18,000 and 21,000 Non-employee director deferred stock units with weighted average grant-date fair values of \$37 and \$32, respectively. The aggregate intrinsic value of Non-employee director deferred stock units distributed during 2007, 2006, and 2005, was \$0.3 million, \$1.7 million, and \$0.2 million, respectively. At year-end 2007 and year-end 2006, there were 218,000 and 203,000, respectively, Non-employee deferred stock units outstanding, and the weighted average grant-date fair value of those outstanding deferred stock units was \$24 and \$21, respectively.

Although the Comprehensive Plan also provides for issuance of deferred stock bonus awards, deferred stock awards, and restricted stock awards, our Compensation Policy Committee indefinitely suspended the issuance of deferred bonus stock commencing with our 2001 fiscal year and the issuance of both deferred stock awards and restricted stock awards commencing with the 2003 fiscal year. At year-end 2007 and year-end 2006, there was approximately \$4 million and \$8 million, respectively, in deferred compensation costs related to these suspended award programs, and the weighted average remaining term was two years for such award grants outstanding at year-end 2007. Share-based compensation expense associated with these suspended award programs was \$4 million, \$6 million, and \$10 million for 2007, 2006, and 2005, respectively.

At year-end 2007, 63 million shares were reserved under the Comprehensive Plan including 39 million shares under the Stock Option Program and Share Appreciation Right Program.

## 5. EARNINGS PER SHARE

The table below illustrates the reconciliation of the earnings and number of shares used in the basic and diluted earnings per share calculations.

(in millions, except per share amounts)	2007	2006	2005
Computation of Basic Earnings Per Share			
Income from continuing operations	\$ 697	\$ 712	\$ 543
Weighted average shares outstanding	376.1	404.1	432.7
Basic earnings per share from continuing operations	\$ 1.85	\$ 1.76	\$ 1.26
Computation of Diluted Earnings Per Share		,	
Income from continuing operations	\$ 697	\$ 712	\$ 543
Weighted average shares outstanding	376.1	404.1	432.7
Effect of dilutive securities			
Employee stock option and share appreciation rights plan	16.8	20.2	19.0
Deferred stock incentive plan	1.9	3.4	7.5
Restricted stock units	2.5	2.5	3.1
Shares for diluted earnings per share	397.3	430.2	462.3
Diluted earnings per share from continuing operations	\$ 1.75	\$ 1.65	\$ 1.17

We compute the effect of dilutive securities using the treasury stock method and average market prices during the period. We determine dilution based on earnings from continuing operations.

In accordance with FAS No. 128, "Earnings per Share," we have not included the following stock options and SARs in our calculation of diluted earnings per share because the exercise prices were greater than the average market prices for the applicable periods:

- (a) for the year ended December 28, 2007, 0.4 million options and SARs, with exercise prices ranging from \$45.91 to \$49.03;
- (b) for the year ended December 29, 2006, no options, and
- (c) for the year ended December 30, 2005, no options.

## 6. INVENTORY

Inventory, totaling \$1,557 million and \$1,186 million as of December 28, 2007, and December 29, 2006, respectively, consists primarily of Timeshare segment interval, fractional ownership, and residential products totaling \$1,536 million and \$1,166 million as of December 28, 2007, and December 29, 2006, respectively. Inventory totaling \$21 million and \$20 million as of December 28, 2007, and December 29, 2006, respectively, relates to hotel operating supplies for the limited number of properties we own or lease. We value Timeshare segment interval, fractional ownership, and residential products at the lower of cost or net realizable value and generally value operating supplies at the lower of cost (using the first-in, first-out method) or market. Consistent with recognized industry practice, we classify Timeshare segment interval, fractional ownership, and residential products inventory, which has an operating cycle that exceeds 12 months, as a current asset.

# 7. PROPERTY AND EQUIPMENT

The following table details the composition of our property and equipment balances at year-end 2007 and year-end 2006.

(\$ in millions)		2006
Land	\$ 399	\$ 316
Buildings and leasehold improvements	833	724
Furniture and equipment	900	837
Construction in progress	216	215
	2,348	2,092
Accumulated depreciation	(1,019)	(859)
	\$ 1,329	\$1,233

We record property and equipment at cost, including interest, rent, and real estate taxes incurred during development and construction. Interest capitalized as a cost of property and equipment totaled \$49 million in 2007, \$32 million in 2006, and \$30 million in 2005. We capitalize the cost of improvements that extend the useful life of property and equipment when incurred. These capitalized costs may include structural costs, equipment, fixtures, floor, and wall coverings. We expense all repair and maintenance costs as incurred. We compute depreciation using the straight-line method over the estimated useful lives of the assets (three to 40 years), and we amortize leasehold improvements over the shorter of the asset life or lease term. Depreciation expense totaled \$162 million in 2007, \$155 million in 2006, and \$156 million in 2005.

## 8. ACQUISITIONS AND DISPOSITIONS

## 2007 Acquisitions

During 2007, we acquired one full-service property, one limited-service property, and one extended-stay property for cash consideration of \$199 million. These three properties were acquired in conjunction with a land assemblage for a large hotel complex that is still in the formative development stage. In addition, we acquired certain land parcels in 2007 for cash consideration of \$52 million. Also during 2007, we acquired the fee simple interest in the improvements of three properties and the leasehold interest in the ground underlying the three properties for cash consideration of \$58 million. The purchase included one full-service property and two limited-service properties, which were each sold later in the same year.

During the first half of 2007, we were party to a venture that developed and marketed fractional ownership and residential interests. In the second half of 2007, we purchased our partner's interest in the joint venture for \$6 million. In conjunction with that transaction, we acquired assets and liabilities totaling \$90 million and \$84 million, respectively, on the date of the purchase. During the first half of 2007, we were party to another venture that was established to develop and market timeshare and residential interests. In the second half of 2007, we purchased our partner's interest in that joint venture, and concurrent with this transaction, we purchased additional land and assets from our partner as well. Aggregate cash and notes payable issued for these transactions totaled \$106 million, and we acquired assets and liabilities totaling \$182 million and \$76 million, respectively, on the date of purchase.

At year-end 2007, we were party to a venture that developed and marketed fractional and whole ownership interests. Subsequent to year-end 2007, we purchased our partner's interest in that joint venture and concurrent with this transaction, we purchased additional land from our partner as well. Cash consideration for this transaction totaled \$37 million and we acquired assets and liabilities totaling \$74 million and \$37 million, respectively, on the date of purchase.

# 2007 Dispositions

In 2007, we sold nine properties for cash proceeds of \$601 million and recognized gains totaling \$24 million. We continue to operate eight of the nine properties under long-term agreements. We sold two parcels of land for \$55 million in cash proceeds that were under development and recognized a gain of \$2 million. We also sold the fee simple interest in the improvements of three properties and the leasehold interest in the ground underlying the three properties, initially acquired in early 2007, for book value and received \$58 million in cash proceeds. We continue to manage the properties under long-term agreements. Each of the aforementioned sales was accounted for under the full accrual method in accordance with FAS No. 66. During the year, we also sold our interests in five joint ventures for cash proceeds of \$30 million and recognized gains totaling \$13 million. We had other asset sales during the year, which generated proceeds totaling \$1 million. Cash flows totaling \$745 million for all the preceding dispositions in 2007 are reflected in the "Dispositions" line in our Consolidated Statements of Cash Flows.

In 2007, we also sold land that was under development. Due to a contingency in the sales contract, this sale was accounted for under the deposit method of FAS No. 66. Accordingly, the cash proceeds, totaling \$90 million, were reflected in "Other investing activities" in our Consolidated Statements of Cash Flows.

## 2006 Acquisitions

During 2006, we acquired one full-service property for \$130 million including aggregate cash consideration of \$46 million plus the assumption of debt. In addition we acquired three other full-service properties for aggregate cash consideration of \$134 million. We sold each of the four properties to third-party owners during the 2007 fiscal year.

# 2006 Dispositions

In 2006 we sold our interest in the 50/50 joint venture with Whitbread PLC ("Whitbread"), which held 46 hotels consisting of more than 8,000 rooms and we received approximately \$164 million in cash, net of transaction costs, which was approximately equal to the investment's book value. We continue to manage the hotels under the Marriott Hotels & Resorts and Renaissance Hotels & Resorts brands. We also sold our minority interest in five other joint ventures during 2006 for cash proceeds of \$64 million and recognized gains of \$43 million. Additionally, one cost method investee redeemed the preferred stock we held for \$81 million in cash consideration and we recognized income of \$25 million on the redemption.

During 2006 we also sold 10 full-service properties for cash proceeds of \$487 million and recognized gains totaling \$14 million. We accounted for each of the sales under the full accrual method in accordance with FAS No. 66 and will continue to operate eight of the properties under long-term management agreements. The sold properties included eight properties purchased in 2005 from CTF Holdings Ltd. and certain of its affiliates (collectively "CTF"). For additional information regarding the CTF transaction, see the "2005 Acquisitions" caption later in this footnote. Prior to the sale of one property, balances associated with that property were reclassified in conformity with other "held and used" properties, in the first half of 2006 as the property was not expected to be sold, within one year of its classification as "held for sale." In conjunction with that reclassification, we recorded depreciation expense of \$2 million in the first half of 2006 that would have been recognized had the asset been continuously classified as "held and used." Cash proceeds of \$26 million for one hotel sold in 2006 are not reflected in the disposition proceeds of \$487 million as the proceeds were initially recorded as a deposit because of a contingency and impacted the "Other investing activities" section of our Consolidated Statements of Cash Flows rather than "Dispositions." The contingency was subsequently resolved and sale accounting was achieved in 2006. Other asset sales generated cash proceeds of \$2 million.

Late in 2006, we sold a 75 percent interest in a joint venture to a third party for its book value of \$14 million. At the time of the sale, the joint venture's only asset was a parcel of land. In conjunction with the sale, we made a \$25 million bridge loan to the joint venture, which matured in early 2007. Following the guidance found in EITF 98-8, "Accounting for Transfers of Investments That Are in Substance Real Estate," and FAS No. 66 due to our continuing involvement with the joint venture, we consolidated the joint venture for the period of time that the bridge loan was outstanding. Subsequent to the bridge loan's repayment, we account for our remaining interest in the joint venture under the equity method as required by APB Opinion No. 18, "The Equity Method of Accounting for Investments in Common Stock."

In 1988, the Company as landlord, entered into a 59-year ground lease with a lessee for land that was improved with a hotel that is owned by the lessee. The hotel was previously branded a Marriott property. The lease contained contractual rental increases over the term of the lease and annual ground rent on the land totaled approximately \$5 million in 2006. The lease also contained a provision that permitted the lessee, under certain circumstances, to purchase the land for a fixed price. We and the lessee had various discussions in 2006 concerning the land as well as the hotel. During the 2006 second quarter, it became probable that none of the proposed transactions were acceptable to both parties and the lessee indicated its intent to exercise its option to purchase the land. Accordingly, in the 2006 second quarter, we reclassified the land from the "Property and equipment" caption in our Consolidated Balance Sheets to the "Assets held for sale" caption and recorded a \$37 million non-cash charge to adjust the carrying amount to net realizable value. We completed the sale of the land, at book value, to the lessee in 2007 and this transaction is reflected in the figures noted earlier for dispositions in 2007.

# 2005 Acquisitions

During the third quarter of 2005, we purchased from CTF and certain of its affiliates 13 properties (in each case through a purchase of real estate, a purchase of the entity that owned the hotel, or an assignment of CTF's leasehold rights) and certain joint venture interests from CTF for an aggregate price of \$381 million. Prior to the sale, all of the properties were operated by us or our subsidiaries.

At the purchase date, we planned to sell eight of the properties to third-party owners and the balances related to these full-service properties were classified within the "Assets held for sale" and "Liabilities of assets held for sale" captions in our Consolidated Balance Sheets at year-end 2005. All eight properties were sold in 2006. One operating lease terminated in 2005. At the purchase date we operated four remaining properties under leases, three of which expire by 2012. Under the purchase and sale agreement we signed with CTF in the second quarter of 2005, we were obligated to purchase two additional properties for \$17 million, the acquisition of which was postponed pending receipt of certain third-party consents. We did not purchase these additional two properties and the obligation expired.

On the closing date we and CTF also modified management agreements on 29 other CTF-leased hotels, 28 located in Europe and one located in the United States. We became secondarily liable for annual rent payments for certain of these hotels when we acquired the Renaissance Hotel Group N.V. in 1997. At the closing date, we continued to manage 16 of these hotels under new long-term management agreements; however, due to certain provisions in the management agreements, we account for these contracts as operating leases. CTF placed approximately \$89 million in trust accounts to cover possible shortfalls in cash flow necessary to meet rent payments under these leases. In turn, we released CTF from its guarantees in connection with these leases. In 2007, the lease agreement associated with one of these properties was terminated. In 2007, we also entered into a transaction whereby the landlord allowed us to assume the lease agreement for another of the properties for which we then became the primary obligor. In conjunction with becoming the primary obligor, we received a \$16 million distribution from the trust, and the balance of the funds was distributed to the landlord. We accounted for our receipt of trust funds as a lease incentive, the reduction of which will be recorded on a straight-line basis as an adjustment to lease expense over the term expiring in 2033. At year-end 2007, approximately \$38 million remained in these trust accounts for the 14 properties we still manage and account for as leases. Minimum lease payments relating to these leases are as follows: \$30 million in each of 2008, 2009, and 2010; \$26 million in 2011; \$17 million in 2012; and \$33 million thereafter, for a total of \$166 million.

For 13 European leased hotels, of the 29 properties mentioned in the preceding paragraph, CTF may terminate management agreements with us if and when CTF obtains releases from landlords of our back-up guarantees. Pending completion of the CTF-landlord agreements, we continue to manage these hotels under modified management agreements and remain secondarily liable under certain of these leases. We are also secondarily obligated for real estate taxes and other charges associated with the leases. Third parties have severally indemnified us for all payments we may be required to make in connection with these obligations. Since we assumed these guarantees, we have not funded any amounts and we do not expect to fund any amounts under these guarantees in the future. CTF had originally made available €35 million in cash collateral in the event that we were required to fund under such guarantees. At year-end 2007, we still managed five of the original 13 properties. Approximately €7 million (\$11 million) of cash collateral remained available at year-end 2007. Our contingent liability exposure at year-end 2007 associated with the five remaining properties totaled \$77 million as also noted in Footnote No. 17, "Contingencies." As CTF obtains releases from the landlords and these remaining hotels exit the system, our contingent liability exposure will decline.

At the closing date, we continued to manage three other hotels in the United Kingdom under amended management agreements with CTF and continued to manage 14 other properties in Asia on behalf of New World Development Company Limited and its affiliates. CTF's principals are officers, directors, and stockholders of New World Development Company Limited. At the closing date, the owners of the United Kingdom and Asian hotels agreed to invest \$17 million to renovate those properties.

We and CTF also exchanged legal releases effective as of the closing date and litigation and arbitration that was outstanding between the two companies and their affiliates was dismissed.

Simultaneously with the closing on the foregoing transactions, CTF also sold five properties and one minority joint venture interest to Sunstone Hotel Investors, Inc. ("Sunstone") for \$419 million, eight properties to Walton Street Capital, L.L.C. ("Walton Street") for \$578 million, and two properties to Tarsadia Hotels ("Tarsadia") for \$29 million, in each case as substitute purchasers under our purchase and sale agreement with CTF. Prior to consummation of the sales, we also operated all of these properties. At closing, Walton Street and Sunstone entered into new long-term management agreements with us and agreed to invest a combined \$68 million to further upgrade the 13 properties they acquired. At the closing date, the two properties purchased by Tarsadia were operated under short-term management and franchise agreements.

When we signed the purchase and sale agreement for the foregoing transactions in the 2005 second quarter, we recorded a \$94 million pretax charge primarily due to the non-cash write-off of deferred contract acquisition costs associated with the termination of the existing management agreements for properties involved in these transactions. As described above, we entered into new long-term management agreements with CTF, Walton Street, and Sunstone at the closing of the transactions.

In 2005 we also purchased two full-service properties, one in Paris, France and the other in Munich, Germany, for aggregate cash consideration of \$146 million. We planned to sell these two full-service properties to third-party owners and the balances related to these properties were classified within the "Assets held for sale" and "Liabilities of assets held for sale" line items on our Consolidated Balance Sheets at year-end 2005. The property in Paris, France, was sold in 2006 and the property located in Munich, Germany, was sold in 2007. The balances associated with the Munich, Germany, property were reclassified, in conformity with other "held and used" properties in the 2006 fourth quarter as the property was not sold at that time, as expected, within one year of its classification as "held for sale." In conjunction with the 2006 reclassification, we recorded depreciation expense of \$5 million in the 2006 fourth quarter that would have been recognized had the asset been continuously classified as "held and used."

#### 2005 Dispositions

Late in 2005, we contributed land underlying an additional nine Courtyard hotels, worth approximately \$40 million, to CBM Land Joint Venture limited partnership ("CBM Land JV"), a joint venture the majority of which was owned, at the time of the transaction, by a third party on behalf of an institutional investor, thereby obtaining a 23 percent equity stake in CBM Land JV. At the same time we completed the sale of a portfolio of land underlying 75 Courtyard by Marriott hotels for approximately \$246 million in cash to CBM Land JV. In conjunction with this transaction, we recognized a gain of \$17 million in 2005, we deferred recognition of a \$5 million gain due to our minority interest in the joint venture, and we also deferred recognition of a \$40 million gain due to contingencies in the transaction documents. As those contingencies expire in subsequent years, we will recognize additional gains. In each of 2007 and 2006, we recognized gains of \$4 million, and at year-end 2007, the aggregate remaining deferred gains totaled \$37 million.

We also sold a number of other land parcels in 2005 for \$38 million in cash, net of transaction costs and recognized gains totaling \$6 million, and we sold two minority interests in joint ventures for \$14 million in cash and recognized gains totaling \$7 million.

## 9. GOODWILL AND INTANGIBLE ASSETS

The following table details the composition of our goodwill and other intangible assets at year-end 2007 and year-end 2006.

(\$ in millions)	2007	2006
Contract acquisition costs and other	\$ 899	\$ 809
Accumulated amortization	(264)	(234)
	\$ 635	\$ 575
Goodwill	\$1,049	\$1,049
Accumulated amortization	(128)	(128)
	\$ 921	\$ 921

We capitalize costs incurred to acquire management, franchise, and license agreements that are both direct and incremental. We amortize these costs on a straight-line basis over the initial term of the agreements, ranging from 15 to 30 years. We evaluate the carrying values of intangible assets for impairment under the provisions of FAS No. 142, "Goodwill and Other Intangible Assets." Amortization expense totaled \$35 million in 2007, \$33 million in 2006, and \$28 million in 2005. Estimated aggregate amortization expense for each of the next five fiscal years is as follows: \$29 million for 2008; \$26 million for 2010; \$25 million for 2011; and \$25 million for 2012.

## 10. NOTES RECEIVABLE

The following table details the composition of our notes receivable balances at year-end 2007 and year-end 2006.

(\$ in millions)	2007	2006
Loans to timeshare owners	\$476	\$ 386
Lodging senior loans	7	9
Lodging mezzanine and other loans	206	268
	689	663
Less current portion	(89)	(103)
	<u>\$600</u>	\$ 560

We classify notes receivable due within one year as current assets in the caption "Accounts and notes receivable" in the accompanying Consolidated Balance Sheets, including \$68 million and \$70 million, as of year-end 2007 and year-end 2006, respectively, related to "Loans to timeshare owners."

Our notes receivable are due as follows: 2008-\$89 million; 2009-\$51 million; 2010-\$134 million; 2011-\$58 million; 2012-\$57 million; and \$300 million thereafter. The 2007 notes receivable balance is net of unamortized discounts totaling \$25 million and the 2006 notes receivable balance is net of unamortized discounts totaling \$28 million. Gains from the sale of notes receivable totaled approximately \$82 million, \$79 million, and \$94 million during 2007, 2006, and 2005, respectively.

Lodging Senior Loans and Lodging Mezzanine and Other Loans

Interest income associated with "Lodging senior loans" and "Lodging mezzanine and other loans" is reflected in the accompanying Consolidated Statements of Income in the "Interest income" caption. We do not accrue interest on "Lodging senior loans" and "Lodging mezzanine and other loans" that are impaired. At year-end 2007, our recorded investment in impaired "Lodging senior loans" and "Lodging mezzanine and other loans" was \$112 million and we had a \$92 million allowance for credit losses, leaving \$20 million of our investment in impaired loans for which there was no related allowance for credit losses. At year-end 2006, our recorded investment in impaired "Lodging senior loans" and "Lodging mezzanine and other loans" was \$92 million and we had a \$70 million allowance for credit losses, leaving \$22 million of our investment in impaired loans for which there was no related allowance for credit losses. During 2007 and 2006, our average investment in impaired loans totaled \$102 million and \$138 million, respectively.

The following table summarizes the activity related to our "Lodging senior loans" and "Lodging mezzanine and other loans" notes receivable reserve for 2005, 2006, and 2007:

(\$ in millions)	Notes Receivable Reserve
Year-end 2004 balance	\$ 92
Additions	11
Reversals	_
Write-offs	(9)
Transfers and other	9
Year-end 2005 balance	103
Additions	_
Reversals	(5)
Write-offs	(38)
Transfers and other	10
Year-end 2006 balance	70
Additions	11
Reversals	_
Write-offs	_
Transfers and other	11
Year-end 2007 balance	\$ 92

#### Loans to Timeshare Owners

At year-end 2007, the weighted average interest rate for our "Loans to timeshare owners" was 12.7 percent and the stated interest rates associated with these loans ranged from zero to 19.9 percent. We reflect interest income associated with "Loans to timeshare owners" of \$50 million, \$41 million, and \$38 million for 2007, 2006, and 2005, respectively, in the accompanying Consolidated Statements of Income in the "Timeshare sales and services" revenue caption. We do not accrue interest on "Loans to timeshare owners" that are over 90 days past due. Our recorded investment in "Loans to timeshare owners" on nonaccrual status at year-end 2007 and year-end 2006 totaled \$59 million and \$67 million, respectively. We established the reserve for "Loans to timeshare owners" notes receivable in 2006 in conjunction with the adoption of SOP 04-2.

The following table summarizes the activity related to our "Loans to timeshare owners" notes receivable reserve for 2006 and 2007.

(\$ in millions)	Rec	lotes eivable eserve
Year-end 2005 balance	\$	_
Establishment of reserve		25
Additions for current year sales		20
Write-offs		(16)
Year-end 2006 balance		29
Additions for current year sales		29
Write-offs		(24)
Other		(15)
Year-end 2007 balance	\$	19

## 11. ASSET SECURITIZATIONS

We periodically sell, without recourse, through special purpose entities, notes receivable originated by our Timeshare segment in connection with the sale of timeshare interval and fractional products. We continue to service the notes and transfer all proceeds collected to special purpose entities. We retain servicing assets and other interests in the notes and account for these assets and interests as residual interests. The interests are limited to the present value of cash available after paying financing expenses and program fees and absorbing credit losses. Prior to the start of the 2007 fiscal year, we measured servicing assets at the date of sale at their allocated previous carrying amount based on relative fair value, classified those assets as held to maturity under the provisions of FAS No. 115 and recorded those assets at amortized cost.

On December 30, 2006, the first day of fiscal year 2007, we adopted FAS No. 156. In conjunction with the adoption of FAS No. 156, we elected to subsequently measure our servicing assets using the fair value method. Under the fair value method, we carry servicing assets on the balance sheet at fair value, and report the changes in fair value, primarily due to changes in valuation inputs and assumptions and to the collection or realization of expected cash flows, in earnings in the period in which the change occurs.

To determine the fair value of servicing assets, we use a valuation model that calculates the present value of estimated future net servicing income, which is based on the monthly fee we receive for servicing the securitized notes. We use market assumptions in the valuation model, including estimates of prepayment speeds, default rates, and discount rates. We have inherent risk for changes in the fair value of the servicing asset but do not deem the risk significant and therefore, do not use other financial instruments to mitigate this risk.

Effective December 30, 2006, upon the remeasurement of our servicing assets at fair value, we recorded a cumulative-effect adjustment to the 2007 beginning balance of retained earnings of \$1 million after-tax (\$2 million pretax) in our Consolidated Statements of Shareholders' Equity. Accordingly, servicing assets totaled \$11 million at year-end 2006 and \$13 million on the first day of fiscal year 2007. At year-end 2007, servicing assets totaled \$15 million.

The table below reconciles the servicing assets balance at year-end 2006, to the beginning balance on December 30, 2006.

(\$ in millions)	Servi Ass	
Balance at year-end 2006 (December 29, 2006)	\$	11
Remeasurement upon adoption of FAS No. 156		2
Beginning balance at December 30, 2006	\$	13

The changes in servicing assets, measured using the fair value method, were:

(\$ in millions)	ssets
Fair value, beginning of period (December 30, 2006)	\$ 13
Servicing from securitizations	6
Changes in fair value (1)	(4)
Fair value, end of period (December 28, 2007)	\$ 15

Principally represents changes due to collection/realization of expected cash flows over time and changes in fair value due to changes in key variables listed below.

Contractually specified servicing fees, late fees, and ancillary fees earned for 2007, 2006, and 2005 totaled \$6 million, \$5 million, and \$4 million, respectively, and were reflected within the changes in fair value to the servicing assets noted above.

We have included gains from the sales of timeshare notes receivable totaling \$81 million in 2007 and \$77 million in 2006 within the "Timeshare sales and services" revenue caption in our Consolidated Statements of Income. Gains from the sale of timeshare notes receivable of \$69 million in 2005 are in the "Gains and other income" caption in the accompanying Consolidated Statements of Income. For additional information regarding the classification of gains from the sale of timeshare notes receivable, see the "Basis of Presentation" caption in Footnote No. 1, "Summary of Significant Accounting Policies." In addition, in September 2006, we repurchased notes receivable with a principal balance of \$31 million and in November 2006, sold those notes, along with \$249 million of additional notes in a \$280 million note securitization. The gain on the sale of these notes is included in the \$77 million gain noted earlier in this paragraph.

We had residual interests of \$238 million and \$221 million, respectively, at year-end 2007 and year-end 2006, which are recorded in the accompanying Consolidated Balance Sheets as other long-term receivables of \$157 million and \$137 million, respectively, and other current assets of \$81 million and \$84 million, respectively.

At the dates of sale and at the end of each reporting period, we estimate the fair value of residual interests, including servicing assets, using a discounted cash flow model. These transactions may utilize interest rate swaps to protect the net interest margin associated with the beneficial interest. We report in income, changes in the fair value of residual interests, including servicing assets, as they are considered trading securities under the provisions of FAS No. 115. During 2007, 2006, and 2005, we recorded trading gains of \$30 million, \$19 million, and \$2 million, respectively. We used the following key assumptions to measure the fair value of the residual interests, including servicing assets, at the date of sale during 2007, 2006, and 2005: average discount rate of 9.02 percent, 9.22 percent, and 8.56 percent, respectively; average expected annual prepayments, including defaults, of 25.02 percent, and 23.56 percent, respectively; expected weighted average life of prepayable notes receivable, excluding prepayments and defaults, of 75 months, 70 months, and 79 months, respectively; and expected weighted average life of prepayable notes receivable, including prepayments and defaults of 34 months, 32 months, respectively. Our key assumptions are based on experience.

We used the following key assumptions in measuring the fair value of the residual interests, including servicing assets, in our 11 outstanding note sales at year-end 2007: an average discount rate of 7.96 percent; an average expected annual prepayment rate, including defaults, of 19.58 percent; an expected weighted average life of prepayable notes receivable, excluding prepayments and defaults, of 62 months; and an expected weighted average life of prepayable notes receivable, including prepayments and defaults of 35 months.

Cash flows between us and third-party purchasers during 2007, 2006, and 2005, were as follows: net proceeds to us from new timeshare note sales of \$515 million, \$508 million, and \$399 million, respectively; repurchases by us of defaulted loans (over 150 days overdue) of \$30 million, \$24 million, and \$23 million, respectively; repurchases by us of other loans in 2006 of \$31 million; servicing fees received by us of \$6 million, \$5 million, and \$4 million, respectively; and cash flows received from our retained interests of \$100 million, \$91 million, and \$86 million, respectively.

At year-end 2007, \$1,263 million of principal remained outstanding in all sales in which we had a retained residual interest. Delinquencies of more than 90 days at year-end 2007 amounted to \$10 million. Existing reserves were adequate for defaulted loans that were resolved during 2007. We have been able to resell timeshare units underlying defaulted loans without incurring material losses.

We completed a stress test on the fair value of the residual interests as of year-end 2007 with the objective of measuring the change in value associated with independent changes in individual key variables. The methodology used applied unfavorable changes that would be considered statistically significant for the key variables of prepayment rate, discount rate, and weighted average remaining term. The fair value of the residual interests was \$238 million at year-end 2007, before we applied any stress test changes. An increase of 100 basis points in the prepayment rate would decrease the year-end valuation by \$4 million, or 1.9 percent, and an increase of 200 basis points in the prepayment rate would decrease the year-end valuation by \$9 million, or 3.7 percent. An increase of 100 basis points in the discount rate would decrease the year-end valuation by \$11 million, or 4.4 percent. A decline of two months in the weighted average remaining term would decrease the year-end valuation by \$3 million, or 1.1 percent, and a decline of four months in the weighted average remaining term would decrease the year-end valuation by \$5 million, or 2.3 percent.

## 12. LONG-TERM DEBT

Our long-term debt at year-end 2007 and year-end 2006 consisted of the following:

(\$ in millions)	2007	2006
Senior Notes:		
Series C, interest rate of 7.875%, face amount of \$76, maturing September 15, 2009 (effective interest rate of		
8.018%)	\$ 76	\$ 76
Series E, interest rate of 7.000%, face amount of \$91, maturing January 15, 2008 (effective interest rate of 7.194%)	91	91
Series F, interest rate of 4.625%, face amount of \$350, maturing June 15, 2012 (effective interest rate of 4.796%)	349	349
Series G, interest rate of 5.810%, face amount of \$427, maturing November 10, 2015 (effective interest rate of		
6.571%)	402	399
Series H, interest rate of 6.200%, face amount of \$350, maturing June 15, 2016 (effective interest rate of 6.294%)	349	349
Series I, interest rate of 6.375%, face amount of \$350, maturing June 15, 2017 (effective interest rate of 6.439%)	346	_
Series J, interest rate of 5.625%, face amount of \$400, maturing February 15, 2013 (effective interest rate of 5.661%)	397	_
Commercial paper, average interest rate of 5.4% at year-end 2007	585	315
Mortgage debt, average interest rate of 7.2% at year-end 2007, maturing through May 1, 2025	196	167
Other	174	87
	2,965	1,833
Less current portion	(175)	(15)
	\$2,790	\$1,818

As of year-end 2007, all debt, other than mortgage debt and \$1 million of other debt, is unsecured.

During 2007, we amended and restated our multicurrency revolving credit agreement, originally entered into in 2005, to increase the aggregate borrowings and letters of credit available under the facility from \$2 billion to \$2.5 billion and to extend the expiration of the facility from 2011 to 2012. The availability of revolving credit borrowings supports our commercial paper program. Borrowings under the facility bear interest at the London Interbank Offered Rate (LIBOR) plus a spread, based on our public debt rating. Additionally, we pay quarterly fees on the facility at a rate also based on our public debt rating.

In 2005 we began issuing short-term commercial paper in Europe in addition to our long-standing commercial paper program in the United States. Our United States and European commercial paper issuances are subject to the availability of the commercial paper market, as we have no commitment from buyers to purchase our commercial

paper. We reserve unused capacity under our credit facility to repay outstanding commercial paper borrowings in the event that the commercial paper market is not available to us for any reason when outstanding borrowings mature. We classify commercial paper as long-term debt based on our ability and intent to refinance it on a long-term basis.

During 2007, we issued \$350 million of aggregate principal amount of 6.375 percent Series I Senior Notes due 2017. The offering of the Notes closed on June 25, 2007. We received net proceeds before expenses of approximately \$346 million from this offering, after deducting the underwriting discount and estimated expenses of the offering. We used these proceeds for general corporate purposes, including the repayment of commercial paper borrowings. Interest on these notes is paid on June 15 and December 15 of each year, and commenced on December 15, 2007. The notes will mature on June 15, 2017, and are redeemable, in whole or in part, at any time and from time to time under the terms provided in the form of note.

Also in 2007, we issued \$400 million of aggregate principal amount of 5.625 percent Series J Senior Notes due 2013. The offering of the notes closed on October 19, 2007. We received net proceeds before expenses of approximately \$396 million from this offering, after deducting the underwriting discount and estimated expenses of the offering. We used these proceeds for general corporate purposes, including working capital, capital expenditures, acquisitions, stock repurchases, and the repayment of commercial paper borrowings. Interest on these notes is paid on February 15 and August 15 of each year, and commenced on February 15, 2008. The notes will mature on February 15, 2013, and are redeemable, in whole or in part, at any time and from time to time under the terms provided in the form of note.

Both the Series I Senior Notes and the Series J Senior Notes were issued under an indenture with The Bank of New York, successor to JPMorgan Chase Bank, N.A. (formerly known as The Chase Manhattan Bank), as trustee, dated as of November 16, 1998.

We are in compliance with covenants in our loan agreements, which require the maintenance of certain financial ratios and minimum shareholders' equity and also include, among other things, limitations on additional indebtedness and the pledging of assets.

Aggregate debt maturities are: 2008-\$175 million; 2009-\$118 million; 2010-\$51 million; 2011-\$19 million; 2012-\$360 million; and \$2,242 million thereafter.

Cash paid for interest, net of amounts capitalized, was \$115 million in 2007, \$73 million in 2006, and \$87 million in 2005.

Subsequent to year-end 2007, on January 15, 2008, we made a \$94 million cash payment of principal and interest to retire, at maturity, all of our outstanding Series E Senior Notes.

## 13. SELF-INSURANCE RESERVE FOR LOSSES AND LOSS ADJUSTMENT EXPENSES

The activity in the reserve for losses and loss adjustment expenses is summarized as follows:

(\$ in millions)	2007	2006
Balance at beginning of year	\$271	\$264
Less: reinsurance recoverable	(16)	(24)
Net balance at beginning of year	(16) 255	240
Incurred related to:		
Current Year	120	122
Prior year	(34)	(20)
Total incurred	86	102
Paid related to:		
Current Year	(31)	(37)
Prior year	(50)	(50)
Total paid	(81)	(87)
Net balance at end of year	260	255
Add: reinsurance recoverable	14	16
Balance at end of year	\$274	\$271

The provision for unpaid loss and loss adjustment expenses decreased by \$34 million and \$20 million in 2007 and 2006, respectively, as a result of changes in estimates from insured events of the prior years due to changes in underwriting experience and frequency and severity trends. The year-end 2007 self-insurance reserve of \$274 million is comprised of a current portion of \$92 million and long-term portion of \$182 million. The year-end 2006 self-insurance reserve of \$271 million is comprised of a current portion of \$87 million and a long-term portion of \$184 million.

## 14. SHAREHOLDERS' EQUITY

Eight hundred million shares of our Class A Common Stock, with a par value of \$.01 per share, are authorized, and 10 million shares of preferred stock, without par value, are authorized. As of the 2007 fiscal year-end, there were 357.1 million shares of our Class A Common Stock outstanding and no shares of our preferred stock were outstanding.

On March 27, 1998, our Board of Directors adopted a shareholder rights plan under which one preferred stock purchase right was distributed for each share of our Class A Common Stock. Each right entitles the holder to buy 1/1000th of a share of a newly issued series of junior participating preferred stock of the Company at an exercise price of \$175. The rights may not presently be exercised, but will be exercisable 10 days after a person or group acquires beneficial ownership of 20 percent or more of our Class A Common Stock or begins a tender or exchange for 30 percent or more of our Class A Common Stock. Shares owned by a person or group on March 27, 1998, and held continuously thereafter, are exempt for purposes of determining beneficial ownership under the rights plan. The rights are nonvoting and will expire on March 27, 2008, unless previously exercised or redeemed by us for \$.01 each. If we are involved in a merger or certain other business combinations not approved by the Board of Directors prior to the expiration of the rights, each right entitles its holder, other than the acquiring person or group, to purchase common stock of either the Company or the acquirer having a value of twice the exercise price of the right. The Company does not plan to extend the shareholder rights plan or the rights beyond their March 27, 2008 expiration.

Accumulated other comprehensive income of \$51 million at year-end 2007 primarily consisted of gains totaling \$9 million associated with available-for-sale securities and gains totaling \$46 million associated with foreign currency translation adjustments, partially offset by net losses of \$5 million associated with interest rate swap agreement cash flow hedges. Accumulated other comprehensive income of \$44 million at year-end 2006 primarily consisted of gains totaling \$35 million associated with available-for-sale securities and gains totaling \$12 million associated with foreign currency translation adjustments.

# 15. FAIR VALUE OF FINANCIAL INSTRUMENTS

We believe that the fair values of current assets and current liabilities approximate their reported carrying amounts. The fair values of non-current financial assets, liabilities, and derivatives are shown in the following table.

(\$ in millions)	At Year-end 2007 At Ye		At Year-o	At Year-end 2006	
	Carrying Amount	Fair Value	Carrying Amount	Fair Value	
Notes and other long-term assets	\$ 996	\$ 1,002	\$ 993	\$ 996	
Long-term debt and other long-term liabilities	\$(2,967)	\$(3,050)	\$(1,816)	\$(1,847)	
Derivative instruments	<b>\$</b> (7)	<b>\$</b> (7)	\$ 6	\$ 6	

We value notes and other receivables based on the expected future cash flows discounted at risk-adjusted rates. We determine valuations for long-term debt and other long-term liabilities based on quoted market prices or expected future payments discounted at risk-adjusted rates.

# 16. DERIVATIVE INSTRUMENTS

During 2007, we entered into interest rate swap agreements to manage the volatility of the U.S. Treasury component of the interest rate risk associated with the forecasted issuance of our Series I and Series J Senior Notes. During 2006, we entered into an interest rate swap agreement to manage the volatility of the U.S. Treasury component of the interest rate risk associated with the forecasted issuance of our Series H Notes. During 2005, we entered into two similar instruments in conjunction with the forecasted issuance of our Series F Notes and the exchange of our Series C and Series E Senior Notes for new Series G Notes. All five swaps were designated as cash

flow hedges under FAS No. 133, "Accounting for Derivative Instruments and Hedging Activities" ("FAS No. 133") and were terminated upon pricing of the notes. All five swaps were highly effective in offsetting fluctuations in the U.S. Treasury component. Thus, there was no net gain or loss reported in earnings during 2007, 2006, or 2005. The net losses for these swaps were \$4 million in 2007 associated with the Series I and Series J Senior Notes, zero in 2006 associated with the Series H Senior Notes, and \$2 million in 2005 associated with the Series G Senior Notes were recorded in other comprehensive income and are being amortized to interest expense using the interest method over the life of the notes.

At year-end 2007, we had four outstanding interest rate swap agreements to manage interest rate risk associated with the residual interests we retain in conjunction with our timeshare note sales. Historically, we were required by purchasers and/or rating agencies to utilize interest rate swaps to protect the excess spread within our sold note pools. The aggregate notional amount of the swaps is \$157 million, and they expire through 2022. These swaps are not accounted for as hedges under FAS No. 133. The fair value of the swaps was a net asset of \$2 million at year-end 2007 and \$5 million at year-end 2006. For the outstanding interest rate swaps, we recorded a \$3 million net loss during 2007, a \$1 million net loss during 2006 and a \$2 million net gain during 2005.

During 2007, 2006, and 2005, we entered into interest rate swaps to manage interest rate risk associated with forecasted timeshare note sales. During 2007, five swaps were designated as cash flow hedges under FAS No. 133 and were highly effective in offsetting interest rate fluctuations. The amount of the ineffectiveness was immaterial. In 2007, we terminated two of those five swaps and recognized no gain or loss from the sales of Timeshare segment notes receivable. The aggregate notional amount of the remaining three swaps is \$238 million at year-end 2007, and they expire through 2012. The fair value of the remaining three swaps was a net liability of \$7 million at year-end 2007. For the outstanding interest rate swaps, we recorded in 2007 a \$7 million net loss in other comprehensive income. During 2006, two swaps (the "2006 swaps") were designated as cash flow hedges under FAS No. 133 and were highly effective in offsetting interest rate fluctuations. In 2006, we terminated the 2006 swaps and recognized a net gain of \$1 million in the gains from the sales of Timeshare segment notes receivable. During 2005, one swap was designated as a cash flow hedge under FAS No. 133 and was highly effective in offsetting interest rate fluctuations. The amount of the ineffectiveness was immaterial and upon termination we recognized a net gain of \$2 million in gains from the sales of Timeshare segment notes receivable. The second swap entered into in 2005 did not qualify for hedge accounting. The non-qualifying swap resulted in a loss of \$3 million during 2005, which was also recognized in gains from the sales of Timeshare segment notes receivable.

During 2007, 2006, and 2005, we entered into forward foreign exchange contracts to manage the foreign currency exposure related to certain monetary assets. The aggregate dollar equivalent of the notional amount of the contracts was \$133 million at year-end 2007. The forward exchange contracts do not qualify as hedges in accordance with FAS No. 133. The fair value of the forward contracts was an asset of \$2 million at year-end 2007 and zero at year-end 2006. We recorded a \$10 million loss during 2007, a \$34 million loss during 2006, and a \$26 million gain in 2005, relating to these forward foreign exchange contracts. The net gains and losses for all years were recorded as general, administrative, and other expense and were offset by income and losses recorded from translating the related monetary assets denominated in foreign currencies into U.S. dollars.

During 2007, 2006, and 2005, we entered into foreign exchange option contracts to hedge the potential volatility of earnings and cash flows associated with variations in foreign exchange rates. The aggregate dollar equivalent of the notional amounts of the contracts was \$86 million at year-end 2007. These contracts have terms of less than one year and are classified as cash flow hedges. Changes in their fair values are recorded as a component of other comprehensive income. The fair value of the option contracts was approximately \$1 million at year-end 2007 and zero at year-end 2006. We recorded an immaterial amount in general, administrative, and other expense due to changes in the time value of these contracts, which is excluded from the assessment of our hedge effectiveness. The hedges were highly effective and there was no net gain or loss reported in earnings for 2007, 2006, and 2005. As of year-end 2007, the deferred gains or losses on existing contracts accumulated in other comprehensive income that we expect to reclassify into earnings over the next year were zero.

During 2007, we entered into foreign exchange forward contracts to hedge forecasted transactions for contracts denominated in foreign currencies. The aggregate dollar equivalent of the notional amounts of the contracts was \$58 million at year-end 2007. These contracts have terms of less than three years and are classified as cash flow hedges. Changes in their fair value are recorded as a component of other comprehensive income. The fair value of the forward contracts was zero at year-end 2007. We recorded an immaterial amount in general, administrative, and other expense due to changes in the time value of these contracts, which is excluded from the assessment of our hedge effectiveness. The hedges were highly effective and there was no gain or loss reported in 2007. As of year-end 2007, the deferred gains or losses on existing contracts accumulated in other comprehensive income that we expect to reclassify into earnings over the next three years were \$1 million.

During 2007, 2006, and 2005, we entered into forward foreign exchange contracts to manage currency exchange rate volatility associated with certain investments in foreign operations. The contracts offset the gains and losses associated with translation adjustments for various investments in foreign operations. One contract was designated as a hedge in the net investment of a foreign operation under FAS No. 133 at year-end 2006. The hedge was highly effective and resulted in a \$1 million net loss for 2007, no net gain or loss for 2006, and a \$1 million net loss for 2005 in the foreign currency translation adjustment section of other comprehensive income. Certain contracts did not qualify as hedges under FAS No. 133 and resulted in no gain or loss for 2007, a loss of \$17 million for 2006, and a gain of \$3 million for 2005, impacting our general, administrative, and other expenses. No contracts remained at year-end 2007. One contract remained at year-end 2006, which had an aggregate dollar equivalent of the notional amount of \$43 million and a fair value of zero. Contracts remaining at year-end 2005 had an aggregate dollar equivalent of the notional amount of \$229 million and a fair value of approximately zero.

During 2007 and 2006, in response to high oil prices and uncertainty surrounding the potential phase out of tax credits, we entered into oil price hedges in conjunction with our synthetic fuel operation. These hedges do not qualify as cash flow hedges under FAS No. 133. Therefore, changes in the fair values of these instruments are marked-to-market through interest income at each reporting period. Correspondingly, we recorded a net loss of \$8 million during 2007 and a net loss of \$2 million during 2006. At year-end 2007, hedges with a total fair value of \$4 million remained outstanding. At year-end 2006, hedges with a total fair value of \$5 million remained outstanding.

During 2003, we entered into an interest rate swap agreement under which we receive a floating rate of interest and pay a fixed rate of interest. The swap modifies our interest rate exposure by effectively converting a note receivable with a fixed rate to a floating rate. The aggregate notional amount of the swap is \$92 million and it matures in 2010. The swap is classified as a fair value hedge under FAS No. 133 and the change in the fair value of the swap, as well as the change in the fair value of the underlying note receivable, is recognized in interest income. The fair value of the swap was a \$2 million liability at year-end 2007, and a \$1 million asset at year-end 2006 and 2005. The hedge is highly effective and, therefore, no net gain or loss was reported during 2007, 2006, and 2005.

# 17. CONTINGENCIES

# Guarantees

We issue guarantees to certain lenders and hotel owners primarily to obtain long-term management contracts. The guarantees generally have a stated maximum amount of funding and a term of three to 10 years. The terms of guarantees to lenders generally require us to fund if cash flows from hotel operations are inadequate to cover annual debt service or to repay the loan at the end of the term. The terms of the guarantees to hotel owners generally require us to fund if the hotels do not attain specified levels of operating profit. Guarantee fundings to lenders and hotel owners are generally recoverable as loans repayable to us out of future hotel cash flows and/or proceeds from the sale of hotels. We also enter into project completion guarantees with certain lenders in conjunction with hotels and Timeshare segment properties that we or our joint venture partners are building.

The maximum potential amount of future fundings for guarantees where we are the primary obligor and the carrying amount of the liability for expected future fundings at year-end 2007 are as follows:

(\$ in millions)  Guarantee Type	Maximum Potential Amount of Future Fundings	Liability for Future Fundings at Year-end 2007		
Debt service	\$ 35	\$ 1		
Operating profit	204	31		
Other	76	7		
Total guarantees where we are the primary obligor	\$ 315	\$ 39		

Our guarantees of \$315 million listed in the preceding table include \$41 million of operating profit guarantees that will not be in effect until the underlying properties open and we begin to operate the properties.

The guarantees of \$315 million in the preceding table do not include \$245 million of guarantees that expire in the years 2011 through 2013, related to Senior Living Services lease obligations and lifecare bonds for which we are secondarily liable. Sunrise Senior Living, Inc. ("Sunrise") is the primary obligor of the leases and a portion of the lifecare bonds, and CNL Retirement Properties, Inc. ("CNL"), which subsequently merged with Health Care Property Investors, Inc., is the primary obligor of the remainder of the lifecare bonds. Prior to our sale of the Senior Living Services business in 2003, these preexisting guarantees were guarantees by us of obligations of consolidated Senior Living Services subsidiaries. Sunrise and CNL have indemnified us for any guarantee fundings we may be called on to make in connection with these lease obligations and lifecare bonds. We do not expect to fund under the guarantees.

The table also does not include lease obligations for which we became secondarily liable when we acquired the Renaissance Hotel Group N.V. in 1997, consisting of annual rent payments of approximately \$7 million and total remaining rent payments through the initial term of approximately \$77 million. Most of these obligations expire at the end of the 2023 calendar year. CTF had originally made available €35 million in cash collateral in the event that we are required to fund under such guarantees (approximately €7 million [\$11 million] remained at year-end 2007). As CTF obtains releases from the landlords and these hotels exit the system, our contingent liability exposure of approximately \$77 million will decline. Since the time we assumed these guarantees, we have not funded any amounts and we do not expect to fund any amounts under these guarantees in the future.

Furthermore, in addition to the guarantees noted in the preceding table, we have provided a project completion guarantee to a lender for a project with an estimated aggregate total cost of \$586 million. Payments for cost overruns for this project will be satisfied by the joint venture via contributions from the partners, and we are liable on a several basis with our partners in an amount equal to our pro rata ownership in the joint venture, which is 34 percent. We do not expect to fund under the guarantee. We have also provided a project completion guarantee to another lender for a project with an estimated aggregate total cost of \$80 million. Payments for cost overruns for this project will be satisfied by the joint venture via contributions from the partners, and we are liable on a several basis with our partners in an amount equal to our pro rata ownership in the joint venture, which is 25 percent. We do not expect to fund under this guarantee. The carrying value of the liabilities associated with these two project completion guarantees is \$7 million.

In addition to the guarantees described in the preceding paragraphs, in conjunction with financing obtained for specific projects or properties owned by joint ventures in which we are a party, we may provide industry standard indemnifications to the lender for loss, liability, or damage occurring as a result of the actions of the other joint venture owner or our own actions.

# Commitments and Letters of Credit

In addition to the guarantees noted previously, as of year-end 2007, we had extended approximately \$4 million of loan commitments to owners of lodging properties, under which we expect to fund approximately \$2 million within one year. We do not expect to fund the remaining \$2 million of commitments, which expire after five years.

At year-end 2007, we also have commitments to invest up to \$44 million of equity for minority interests in partnerships that plan to purchase North American full-service and limited-service properties or purchase or develop hotel anchored mixed-use real estate projects, which expire as follows: \$14 million in one to two years; and \$30 million in three to five years. As of year-end 2007, we also have a commitment to invest up to \$25 million in a joint venture of which we have funded \$12 million and have \$13 million remaining that we expect to fund within one year. As of year-end 2007, we also had a commitment to invest up to \$29 million (€20 million) in a joint venture in which we are an investor. We currently do not expect to fund under this commitment.

At year-end 2007, we had \$96 million of letters of credit outstanding on our behalf, the majority of which related to our self-insurance programs. Surety bonds issued on our behalf at year-end 2007, totaled \$468 million, the majority of which were requested by federal, state or local governments related to our lodging operations, including our Timeshare segment and self-insurance programs.

# Synthetic Fuel

The tax credits available under the Internal Revenue Code for the production and sale of synthetic fuels were established by Congress to encourage the development of alternative domestic energy sources. Congress deemed that the incentives provided by the tax credits would not be necessary if the price of oil increased beyond certain thresholds as prices would then provide a more natural market for these alternative fuels. As a result, the tax credits available under the Internal Revenue Code for the production and sale of synthetic fuel in any given calendar year are phased out if the Reference Price of a barrel of oil for that year falls within a specified range. The Reference Price of a barrel of oil is an estimate of the annual average wellhead price per barrel of domestic crude oil and is determined for each calendar year by the Secretary of the Treasury by April 1 of the following year. In 2005 and 2006, the Reference Price was roughly equal to 89 percent and 90 percent, respectively, of the average price in those years of the benchmark NYMEX futures contract for a barrel of light, sweet crude oil. The price range within which the credit is phased out was set in 1980 and is adjusted annually for inflation. In 2006, the Reference Price phase-out range was \$55.06 to \$69.12. Because the Reference Price of a barrel of oil for 2006 was within that range, at \$59.68, there was a 33 percent reduction of the tax credits available for synthetic fuel produced and sold in 2006.

Assuming a 2.5 percent inflation adjustment factor for 2007, and assuming the ratio of the Reference Price to the average price of the benchmark NYMEX futures contract remains the same for November and December 2007 as it has been in the five preceding months, we currently estimate that the tax credits available for production and sale of synthetic fuel in 2007 would begin to be phased out if the average price of the benchmark NYMEX futures contract in 2007 exceeds approximately \$62 and would be fully phased out if the average price of the benchmark NYMEX futures contract in 2007 exceeds approximately \$78. For the year ended December 28, 2007, our results reflect a provision for an estimated 70.71 percent tax credit phase-out as a result of high oil prices. The average price of the benchmark NYMEX futures contract for 2007, through December 31, 2007, was approximately \$72.41.

Late in 2006 and early in 2007, we entered into hedge agreements to minimize operating losses that could occur if more than a majority of the tax credits is phased out in 2007.

See Footnote No. 2, "Discontinued Operations," earlier in this report, for additional information related to the synthetic fuel operations, including information related to the cessation of synthetic fuel production permanently in November 2007.

# Investment in Leveraged Lease

Historically, we had a \$23 million investment in an aircraft leveraged lease with Delta Air Lines, Inc. ("Delta") which we acquired in 1994. The gross investment was comprised of rentals receivable and the residual value of the aircraft offset by unearned income. On September 14, 2005, Delta filed for bankruptcy protection under Chapter 11 of the U.S. Bankruptcy Code and informed us that it wished to restructure the lease. As a result, our investment was impaired and we had recorded pretax charges of approximately \$18 million through 2006. We recorded an additional \$5 million loss related to this investment in fiscal 2007. We have no remaining exposure related to this historical investment.

#### 18. LEASES

We have summarized our future obligations under operating leases at year-end 2007, below:

#### (\$ in millions)

Fiscal Year	Minimum Le Payments	
2008	\$	142
2009		147
2010		139
2011		126
2012		123
Thereafter		1,156
Total minimum lease payments where we are the primary obligor	\$	1,833

Most leases have initial terms of up to 20 years and contain one or more renewal options, generally for five- or 10-year periods. These leases provide for minimum rentals and additional rentals based on our operations of the leased property. The total minimum lease payments above include \$439 million, representing obligations of consolidated subsidiaries that are non-recourse to Marriott International, Inc.

As discussed in Footnote No. 8, "Acquisitions and Dispositions," we became secondarily liable for annual rent payments for certain hotels when we acquired the Renaissance Hotel Group N.V. in 1997. At year-end 2007, we continued to manage 14 of these hotels under long-term management agreements; however, due to certain provisions in the management agreements, we account for these contracts as operating leases. CTF placed funds into trust accounts to cover possible shortfalls in cash flow necessary to meet rent payments under these leases. In turn, we released CTF affiliates from their guarantees in connection with these leases. Approximately \$38 million remained in these trust accounts at year-end 2007. Minimum lease payments relating to these leases, which are not reflected in the \$1,833 million in the preceding table are as follows: \$30 million in each of 2008, 2009, and 2010; \$26 million in 2011; \$17 million in 2012; and \$33 million thereafter, for a total of \$166 million.

The composition of rent expense for the last three fiscal years associated with operating leases is detailed in the following table:

(\$ in millions)	2007	2006	2005
Minimum rentals	\$272	\$253	\$211
Additional rentals	100	109	109
	\$372	\$362	\$320

We have summarized our future obligations under capital leases at year-end 2007, in the following table:

(\$ in millions)	
Fiscal Year Mi	inimum Lease Payments
2008 \$	1
2009	1
2010	1
2011	1
2012	1
Thereafter	7
Total minimum lease payments	12
Less: Amount representing interest	(5)
Present value of net minimum lease payments	7

The accompanying Consolidated Balance Sheets for year-end 2007, included \$7 million in the "Long-term debt" caption that represented the present value of net minimum lease payments associated with capital leases.

## 19. BUSINESS SEGMENTS

We are a diversified hospitality company with operations in five business segments:

- North American Full-Service Lodging, which includes Marriott Hotels & Resorts, Marriott Conference Centers, JW Marriott Hotels & Resorts, Marriott Conference Centers, Renaissance Hotels & Resorts, and Renaissance ClubSport properties located in the continental United States and Canada;
- North American Limited-Service Lodging, which includes Courtyard, Fairfield Inn, SpringHill Suites, Residence Inn, TownePlace Suites, and Marriott ExecuStay properties located in the continental United States and Canada;
- International Lodging, which includes Marriott Hotels & Resorts, JW Marriott Hotels & Resorts, Renaissance Hotels & Resorts, Courtyard, Fairfield Inn, Residence Inn, Ramada International, and Marriott Executive Apartments properties located outside the continental United States and Canada;
- Luxury Lodging, which includes The Ritz-Carlton and Bulgari Hotels & Resorts properties worldwide; and
- Timeshare, which includes the development, marketing, operation, and sale of timeshare, fractional ownership, and residential properties worldwide under Marriott Vacation Club, The Ritz-Carlton Club, Grand Residences by Marriott, and Horizons by Marriott Vacation Club.

In addition to the segments above, in 2007 we exited the synthetic fuel business, which was formerly a separate segment but which we now report under discontinued operations.

In addition to the brands noted above, in 2007 we announced our new brand of family-friendly resorts and spas, "Nickelodeon Resorts by Marriott" and a new brand of lifestyle boutique hotels, "Edition." At year-end 2007, no properties were yet open under either brand.

In 2006, we analyzed our internal reporting process and implemented changes in the fourth quarter that were designed to improve efficiency and, as part of this process, we evaluated the impact on segment reporting. Accordingly, we now report five operating segments, and no longer allocate indirect administrative expenses to our segments. We reflect this revised segment reporting throughout this report for all periods presented, and present historical figures in a manner that is consistent with the revised segment reporting. See also the Form 8-K that we filed on March 19, 2007, furnishing quarterly Revenues and Income from Continuing Operations for each of 2006 and 2005 in the new segment format.

We evaluate the performance of our segments based primarily on the results of the segment without allocating corporate expenses, interest expense, or indirect general, administrative, and other expenses. With the exception of the Timeshare segment, we do not allocate interest income to our segments. Because note sales are an integral part of the Timeshare segment, we include note sale gains in our Timeshare segment results. We include interest income associated with Timeshare segment notes in our Timeshare segment results because financing sales are an integral part of that segment's business. We also allocate other gains or losses as well as equity in earnings or losses from our joint ventures and divisional general, administrative, and other expenses to each of our segments. "Other unallocated corporate" represents that portion of our revenues, general, administrative, and other expenses, equity in earnings or losses, and other gains or losses that are not allocable to our segments.

We aggregate the brands presented within our North American Full-Service, North American Limited-Service, International, Luxury, and Timeshare segments considering their similar economic characteristics, types of customers, distribution channels, the regulatory business environment of the brands and operations within each segment and our organizational and management reporting structure.

## Revenues

(\$ in millions)	2007	2006	2005
North American Full-Service Segment	\$ 5,476	\$ 5,196	\$ 5,116
North American Limited-Service Segment	2,198	2,060	1,886
International Segment	1,594	1,411	1,017
Luxury Segment	1,576	1,423	1,333
Timeshare Segment	2,065	1,840	1,721
Total segment revenues	12,909	11,930	11,073
Other unallocated corporate	81	65	56
	\$12,990	\$11,995	\$11,129

# **Income from Continuing Operations**

(\$ in millions)	2007	2006	2005
North American Full-Service Segment	\$ 478	\$ 455	\$ 349
North American Limited-Service Segment	461	380	303
International Segment	271	237	133
Luxury Segment	72	63	45
Timeshare Segment	306	280	271
Total segment financial results	1,588	1,415	1,101
Other unallocated corporate	(287)	(251)	(219)
Interest income, provision for loan losses, and interest expense	(163)	(72)	(55)
Income taxes	(441)	(380)	(284)
	\$ 697	\$ 712	\$ 543

# **Equity in Earnings (Losses) of Equity Method Investees**

(\$ in millions)	2007	2006	2005		
North American Full-Service Segment	\$ 3	\$ 2	\$ 21		
North American Limited-Service Segment	2	_	(6)		
International Segment	3	_	20		
Luxury Segment	(4)	(2)	(1)		
Timeshare Segment	10	(2)	1		
Total segment equity in earnings (losses)	14	(2)	35		
Other unallocated corporate	1	5	1		
	\$ 15	\$ 3	\$ 36		

# **Depreciation and Amortization**

(\$ in millions)	2	2007	2	2006	2005		
North American Full-Service Segment	\$	25	\$	24	\$	3	19
North American Limited-Service Segment		23		24			24
International Segment		23		23			18
Luxury Segment		17		7			10
Timeshare Segment		39		39			46
Total segment depreciation and amortization		127		117	_		117
Other unallocated corporate		62		61			57
Discontinued operations		8		10			10
	\$	197	\$	188	\$	3	184

# Assets

(\$ in millions)	At Year-end 2007		At Year-end 2007								At	Year-end 2006	A	t Year-end 2005
North American Full-Service Segment	\$	1,322	\$	1,104	\$	1,309								
North American Limited-Service Segment		486		565		613								
International Segment		855		1,225		1,333								
Luxury Segment		748		755		656								
Timeshare Segment		3,142		2,560		2,454								
Total segment assets		6,553		6,209		6,365								
Other unallocated corporate		2,336		2,288		2,062								
Discontinued operations		53		91		103								
	\$	8,942	\$	8,588	\$	8,530								

# **Equity Method Investments**

(\$ in millions)		At Year-end 2007		At Year-end 2006		ear-end 2005
North American Full-Service Segment	\$	16	\$	18	\$	21
North American Limited-Service Segment		75		35		50
International Segment		62		73		119
Luxury Segment		41		17		18
Timeshare Segment		99		168		115
Total segment equity method investments	· ·	293	· <u> </u>	311		323
Other unallocated corporate		23		21		26
	\$	316	\$	332	\$	349

#### Goodwill

(\$ in millions)	Year-end At Year-end 2007 2006			At Year-end 2005		
North American Full-Service Segment	\$ 406	\$	406	\$	407	
North American Limited-Service Segment	72		72		72	
International Segment	273		273		273	
Luxury Segment	170		170		172	
Total segment goodwill	\$ 921	\$	921	\$	924	

# **Capital Expenditures**

(\$ in millions)	<u>:</u>	2007		2007 2006		2006		2005
North American Full-Service Segment	\$	446	\$	75	\$	197		
North American Limited-Service Segment		36		38		10		
International Segment		44		215		376		
Luxury Segment		35		104		84		
Timeshare Segment		56		28		27		
Total segment capital expenditures		617		460		694		
Other unallocated corporate		54		69		86		
	\$	671	\$	529	\$	780		

Segment expenses include selling expenses directly related to the operations of the businesses, aggregating \$616 million in 2007, \$600 million in 2006, and \$609 million in 2005. Approximately 89 percent for 2007 and 90 percent for each of 2006 and 2005 of the selling expenses are related to our Timeshare segment.

The consolidated financial statements include the following related to operations located outside the United States (which are predominately related to our International lodging segment): Revenues of \$2,287 million in 2007, \$1,869 million in 2006, and \$1,388 million in 2005; segment financial results of \$341 million in 2007 (32 percent from Europe, 30 percent from the Americas excluding the United States, 17 percent from Asia, 11 percent from the United Kingdom, 8 percent from the Middle East and Africa, and 2 percent from Australia), \$298 million in 2006, and \$178 million in 2005; and fixed assets of \$257 million in 2007 and \$684 million in 2006. At year-end 2007, fixed assets totaling \$257 million located outside the United States are included within the "Property and equipment" caption in our Consolidated Balance Sheets. At year-end 2006, fixed assets totaling \$684 million of fixed assets which are included within the "Assets held for sale" caption in our Consolidated Balance Sheets. No individual country, other than the United States, constitutes a material portion of our revenues, financial results or fixed assets.

# 20. VARIABLE INTEREST ENTITIES

We currently consolidate a holding company that holds 100 percent interest in four entities that are variable interest entities under FIN 46, "Consolidation of Variable Interest Entities-revised" ("FIN 46(R)"). At year-end 2007, the combined capital in the four variable interest entities is less than \$1 million, which is used primarily to fund hotel working capital. Our equity at risk was \$3 million and we held 55 percent of the common equity shares of the holding company. The creditors of the holding company do not have general recourse to our credit.

We are party to a venture that develops and markets fractional ownership and residential interests. During the 2007 second quarter, we issued a guarantee to the senior lender of the venture in support of the senior loan facility and reevaluated our variable interests in the venture under FIN 46(R). At that time, we determined that we were the primary beneficiary and as such, we also consolidated this venture. During the 2007 third quarter, the guarantee was replaced with the issuance of a loan facility for \$40 million, of which \$36 million is receivable and outstanding at year-end 2007. Our issuance of the loan facility was a reconsideration event under FIN 46(R); we again determined we were the primary beneficiary and continue to consolidate the joint venture. At year-end 2007, the carrying amount of consolidated assets that are collateral for the variable interest entity's obligations totaled \$141 million and comprised \$24 million of accounts receivable, \$106 million of real estate held for development, property, equipment, and other long-term assets, and \$11 million of cash. The creditors of the variable interest entity do not have general recourse to our credit.

In conjunction with the transaction with CTF described more fully in Footnote No. 8, "Acquisitions and Dispositions," under the caption "2005 Acquisitions" we manage certain hotels on behalf of four tenant entities 100 percent owned by CTF, which lease the hotels from third-party owners. At year-end 2007, the number of hotels totaled 14. The entities have minimal equity and minimal assets comprised of hotel working capital. CTF placed money in a trust account to cover cash flow shortfalls and to meet rent payments. The terms of the trust require that the cash flows for the four tenant entities be pooled for purposes of making rent payments and determining cash flow shortfalls. At year-end 2007, the trust account held approximately \$38 million. The entities are variable interest entities under FIN 46(R). We do not consolidate the entities because we do not bear the majority of the expected losses. We are secondarily liable for rent payments for eight of the 14 hotels in the event that there are cash flow shortfalls and there is no money left in the trust. Future minimum lease payments through the end of the lease term for these eight hotels total approximately \$122 million. In addition, we are also secondarily liable for rent payments of up to an aggregate cap of \$44 million for the six other hotels in the event that there are cash flow shortfalls.

## 21. RELATED PARTY TRANSACTIONS

## **Equity Method Investments**

We have equity method investments in entities that own properties for which we provide management and/or franchise services and receive fees. In addition, in some cases we provide loans, preferred equity or guarantees to these entities. Our ownership interest in these equity method investments generally varies from 10 to 50 percent. The amount of our consolidated retained earnings that represents undistributed earnings attributable to our equity investments totaled \$9 million at year-end 2007.

The following tables present financial data resulting from transactions with these related parties:

## **Income Statement Data**

(\$ in millions)	2007	2006	2005
Base management fees	\$ 56	\$ 62	\$ 83
Franchise fees	1	2	2
Incentive management fees	26	22	14
Cost reimbursements	510	649	936
Owned, leased, corporate housing, and other revenue	_	<del>_</del>	19
Total revenue	\$ 593	\$ 735	\$1,054
General, administrative, and other	\$ (4)	\$ (1)	\$ (19)
Reimbursed costs	(510)	(649)	(936)
Gains and other income	25	28	54
Interest expense	(1)	(1)	_
Interest income	4	4	31
(Provision for) reversal of provision for loan losses	(12)	1	_
Equity in earnings	15	3	36

# **Balance Sheet Data**

(\$ in millions)	At Year-end 2007	At Year-end 2006	
Current assets-accounts and notes receivable	\$ 42	\$ 76	
Deferred development	2	_	
Contract acquisition costs	33	34	
Equity and cost method investments	316	377	
Loans to equity method investees	21	27	
Long-term deferred tax assets, net	1	4	
Current liabilities:			
Other payables and accruals	(2)	(2)	
Other long-term liabilities	(16)	(13)	

Summarized information for the entities in which we have equity method investments is as follows:

# **Income Statement Summary**

(\$ in millions)		2006	2005
Sales	\$1,622	\$1,479	\$1,975
Net income	\$ 134	\$ 170	\$ 259

# **Balance Sheet Summary**

(\$ in millions)	At Year-end 2007	At Year-end 2006	
Assets (primarily comprised of hotel real estate managed by us)	\$ 3,856	\$ 4,325	
Liabilities	\$ 2,536	\$ 2,830	

## 22. RELATIONSHIP WITH MAJOR CUSTOMER

As of year-end 2007, Host Hotels & Resorts, Inc. ("Host"), known as Host Marriott Corporation prior to April 18, 2006, owned or leased 149 lodging properties operated by us under long-term agreements. We recognized revenues from lodging properties owned or leased by Host (which are included in our North American Full-Service, North American Limited-Service, and International segments) for the last three fiscal years as shown in the following table:

(\$ in millions)	2007	2006	2005
Revenues	\$2,580	\$2,475	\$2,427

Additionally, Host is a partner in several unconsolidated partnerships that own lodging properties operated by us under long-term agreements. As of year-end 2007, Host was affiliated with 124 such properties operated by us. The revenues associated with those properties (which are included in our North American Full-Service, North American Limited-Service, and International segments) that were recognized by the Company for the last three fiscal years are shown in the following table:

(\$ in millions)	2007	2006	2005
Revenues	\$ 350	\$ 353	\$ 352

# QUARTERLY FINANCIAL DATA – UNAUDITED

(\$ in millions, except per share data)	Fiscal Year 2007 <sup>(1), (2)</sup>				
	Fir+st Quarter	Second Quarter	Third Quarter	Fourth Quarter	Fiscal Year
Revenues (3)	\$2,836	\$3,122	\$2,943	\$4,089	\$ 12,990
Operating income (3)	\$ 237	\$ 335	\$ 210	\$ 406	\$ 1,188
Income from continuing operations	\$ 164	\$ 175	\$ 122	\$ 236	\$ 697
Discontinued operations, net of tax	18	32	9	(60)	(1)
Net income	\$ 182	\$ 207	\$ 131	\$ 176	\$ 696
Diluted earnings per share from continuing operations	\$ 0.40	\$ 0.43	\$ 0.31	\$ 0.62	\$ 1.75
Diluted earnings per share from discontinued operations	0.04	0.08	0.02	(0.16)	_
Diluted earnings per share	\$ 0.44	\$ 0.51	\$ 0.33	\$ 0.46	\$ 1.75
(\$ in millions, except per share data)	Fiscal Year 2006 <sup>(1), (2)</sup>				
	First Quarter	Second Quarter	Third Quarter	Fourth Quarter	Fiscal Year
Revenues (3)	\$2,648	\$2,852	\$2,697	\$3,798	\$ 11,995
Operating income (3)	\$ 230	\$ 292	\$ 227	\$ 338	\$ 1,087
Income from continuing operations	\$ 167	\$ 182	\$ 144	\$ 219	\$ 712
Discontinued operations, after-tax	3	4	(3)	1	5
Cumulative effect of change in accounting principle, net of tax	(109)				(109)
Net income	\$ 61	\$ 186	\$ 141	\$ 220	\$ 608
Diluted earnings per share from continuing operations	\$ 0.38	\$ 0.42	\$ 0.34	\$ 0.52	\$ 1.65
Diluted losses per share from cumulative effect of accounting change	(0.25)	_			(0.25)
Diluted earnings per share from discontinued operations	0.01	0.01	(0.01)		0.01
Diluted earnings per share	\$ 0.14	\$ 0.43	\$ 0.33	\$ 0.52	\$ 1.41

The quarters consisted of 12 weeks, except for the fourth quarters, which consisted of 16 weeks.

All share and per share amounts reflect the June 2006 stock split. The sum of the earnings per share for the four quarters differs from annual earnings per share due to the required method of computing the weighted average shares in interim periods.

Balances do not reflect the impact of the synthetic fuel business as the impact of that business is now reflected in discontinued operations.

## Item 9. Changes in and Disagreements With Accountants on Accounting and Financial Disclosure.

None.

#### Item 9A. Controls and Procedures.

## Disclosure Controls and Procedures

As of the end of the period covered by this annual report, we carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) of the Securities Exchange Act of 1934 (the "Exchange Act")), and management necessarily applied its judgment in assessing the costs and benefits of such controls and procedures, which, by their nature, can provide only reasonable assurance regarding management's control objectives. You should note that the design of any system of controls is based in part upon certain assumptions about the likelihood of future events, and we cannot assure you that any design will succeed in achieving its stated goals under all potential future conditions, regardless of how remote. Based upon the foregoing evaluation, our Chief Executive Officer and the Chief Financial Officer concluded that our disclosure controls and procedures were effective to provide reasonable assurance that information required to be disclosed by us in the reports that we file or submit under the Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the rules and forms of the SEC, and to provide reasonable assurance that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

## Internal Control Over Financial Reporting

There were no changes in internal control over financial reporting that occurred during the fourth quarter that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

## Item 9B. Other Information.

None.

## PART III

# Items 10, 11, 12, 13, 14.

As described below, we incorporate certain information appearing in the Proxy Statement we will furnish to our shareholders in connection with the 2008 Annual Meeting of Shareholders by reference in this Annual Report on Form 10-K.

- ITEM 10. We incorporate this information by reference to the "Directors Standing For Election," "Directors Continuing In Office," "Section 16(a) Beneficial Ownership Reporting Compliance," "Audit Committee," and "Selection of Director Nominees" sections of our Proxy Statement. We have included information regarding our executive officers and our Code of Ethics below.
- ITEM 11. We incorporate this information by reference to the "Executive Compensation" section of our Proxy Statement.
- ITEM 12. We incorporate this information by reference to the "Securities Authorized for Issuance Under Equity Compensation Plans" and the "Stock Ownership" sections of our Proxy Statement.
- ITEM 13. We incorporate this information by reference to the "Certain Relationships and Related Transactions, and Director Independence" section of our Proxy Statement.
- ITEM 14. We incorporate this information by reference to the "Principal Independent Auditor Fee Disclosure" and the "Pre-Approval of Independent Auditor Fees and Services Policy" sections of our Proxy Statement.

## EXECUTIVE OFFICERS OF THE REGISTRANT

Set forth below is certain information with respect to our executive officers. The information set forth below is as of February 1, 2008.

the Company's Board of Directors.

Age

J.W. Marriott, Jr.	75	J.W. Marriott, Jr. has been Chief Executive Officer of Marriott International, Inc. (the Company) and its
Chairman of the Board and		predecessors since 1972, and Chairman of the Board since 1985. He joined Marriott Corporation (now
Chief Executive Officer		known as Host Hotels & Resorts, Inc.) in 1956, became President and a director in 1964. Mr. Marriott
		serves as a director of the United States Naval Academy Foundation. He also serves on the Board of
		Trustees of the National Geographic Society, The J. Willard & Alice S. Marriott Foundation and the
		National Business Council. He is a member of the Executive Committee of the World Travel & Tourism
		Council and the chairman of the President's Export Council, a presidential advisory committee on export
		trade. Mr. Marriott is also on the U.S. Travel and Tourism Advisory Board (TTAB) and the Secure Borders
		Open Doors Advisory Committee (SBODAC). In addition, he serves as chairman of the Leadership
		Council of the Laura Bush Foundation for America's Libraries. Mr. Marriott has served as a director of the
		Company since March 1998. He is the father of John W. Marriott III, the non-employee Vice Chairman of

- Carl T. Berquist Executive Vice President Financial Information and Enterprise Risk Management
- Simon F. Cooper President and Chief Operating Officer The Ritz-Carlton Hotel Company, L.L.C.

Name and Title

56 Carl T. Berquist joined Marriott International in December 2002 as Executive Vice President of Financial Information and Enterprise Risk Management. He also serves as Principal Accounting Officer of the Company. Prior to joining Marriott, Mr. Berquist was a partner at Arthur Andersen LLP, most recently as managing partner of the mid-Atlantic region, which included five offices from Philadelphia, PA, to Richmond, VA. Mr. Berquist is a member of Penn State's Smeal Business School's Board of Visitors and is a member of the board of Hertz Global Holdings, Inc., and several private companies.

**Business Experience** 

62 Simon F. Cooper joined Marriott International in 1998 as President of Marriott Lodging Canada and Senior Vice President of Marriott Lodging International. In 2000, the Company added the New England Region to his Canadian responsibilities. Prior to joining Marriott, Mr. Cooper was President and Chief Operating Officer of Delta Hotels and Resorts. He is a fellow of the Board of Trustees for the Educational Institute of the American Hotel and Motel Association. Mr. Cooper was appointed to his current position as President and COO, The Ritz-Carlton Hotel Company, in February 2001.

Name and Title Age 62 **Business Experience** Edwin D. Fuller President and Managing Director International Lodging

Robert J. McCarthy **Executive Vice President** North American Lodging Operations and Global Brand Management

David A. Rodriguez **Executive Vice President** Global Human Resources

Edward A. Ryan Executive Vice President and General Counsel

William J. Shaw Director, President and Chief Operating Officer

- Edwin D. Fuller joined Marriott in 1972 and held several sales positions before being appointed Vice President of Marketing in 1979. He became Regional Vice President of the Midwest Region in 1985, Regional Vice President of the Western Region in 1988, and in 1990 was promoted to Senior Vice President and Managing Director of International Lodging, with a focus on developing the international group of hotels. He was named Executive Vice President and Managing Director of International Lodging in 1994 and was promoted to his current position in 1997.
- 54 Robert J. "Bob" McCarthy has over 30 years experience with Marriott. Prior to his current position, he served as Executive Vice President of North American Lodging Operations for Marriott International; Senior Vice President of the Northeast region for Marriott Lodging; and Vice President of Operations for Courtyard and Fairfield Inn. He is also Chairman of the Board of Managers of Avendra, LLC, a procurement company serving the hospitality industry.
- 49 David A. Rodriguez joined Marriott International and assumed the role of Senior Vice President-Staffing & Development in 1998. In 2003, he was appointed Executive Vice President-Human Resources for Marriott Lodging and in 2006 he assumed the role of Executive Vice President-Global Human Resources for Marriott International. Prior to joining Marriott International, he held several senior roles in human resources at Citicorp (now Citigroup) from 1989-1998.
- 54 Edward A. Ryan was named Executive Vice President and General Counsel in November 2006. He joined Marriott in 1996 as Assistant General Counsel, was promoted to Senior Vice President and Associate General Counsel in 1999, when he had responsibility for all new management agreements and real estate development worldwide for full-service and limited-service hotels; in 2005 he also assumed responsibility for all corporate transactions and corporate governance. Prior to joining Marriott, Mr. Ryan was a partner at the law firm of Hogan & Hartson in Washington, D.C.
- 62 William J. Shaw has served as President and Chief Operating Officer of the Company since 1997 (including service in the same capacity with Old MI until March 1998). He joined Marriott Corporation in 1974, was elected Corporate Controller in 1979 and a Corporate Vice President in 1982. In 1986, Mr. Shaw was elected Senior Vice President-Finance and Treasurer of Marriott Corporation. He was elected Chief Financial Officer and Executive Vice President of Marriott Corporation in 1988. In 1992, he was elected President of the Marriott Service Group. He also serves on the Board of Trustees of the University of Notre Dame and on the NCAA Leadership Advisory Board. Mr. Shaw served as a director of Old MI from March 1998 through June 2001. He has served as a director of the Company since March 1997.

Name and Title	Age	Business Experience
Arne M. Sorenson Executive Vice President, Chief Financial Officer and President, Continental European Lodging	49	Arne M. Sorenson joined Marriott in 1996 as Senior Vice President of Business Development. He was instrumental in our acquisition of the Renaissance Hotel Group in 1997. Prior to joining Marriott, he was a partner in the law firm of Latham & Watkins in Washington, D.C., where he played a key role in 1992 and 1993 in the distribution of Old MI by Marriott Corporation. Mr. Sorenson was appointed Executive Vice President and Chief Financial Officer in 1998 and assumed the additional title of President, Continental European Lodging, in January 2003. Mr. Sorenson serves on the Board of Regents of Luther College in Decorah, Iowa. He also serves on the Board of Directors of The Luther Institute.
James M. Sullivan Executive Vice President Lodging Development	64	James M. Sullivan joined Marriott Corporation in 1980, departed in 1983 to acquire, manage, expand, and subsequently sell a successful restaurant chain, and returned to Marriott Corporation in 1986 as Vice President of Mergers and Acquisitions. Mr. Sullivan became Senior Vice President, Finance-Lodging in 1989, Senior Vice President-Lodging Development in 1990 and was appointed to his current position in 1995. He serves as a director of IntegraLifeSciences, Holding Corporation.
Stephen P. Weisz President Marriott Vacation Club	57	Stephen P. Weisz joined Marriott Corporation in 1972 and was named Regional Vice President of the Mid-Atlantic Region in 1991. Mr. Weisz had previously served as Senior Vice President of Rooms Operations before being appointed Vice President of the Revenue Management Group. Mr. Weisz became Senior Vice President of Sales and Marketing for Marriott Hotels, Resorts & Suites in 1992 and Executive Vice President-Lodging Brands in 1994. Mr. Weisz was appointed to his current position in 1996. He serves as a director of Mindshare Technologies, Inc.

## **Code of Ethics**

The Company has long maintained and enforced an Ethical Conduct Policy that applies to *all* Marriott associates, including our Chief Executive Officer, Chief Financial Officer, and Principal Accounting Officer, and to each member of our Board of Directors. We have posted our Code of Ethics ("Ethical Conduct Policy") in the "Corporate Governance" section of our Investor Relations Web site, <u>www.Marriott.com/investor</u>. Any future changes or amendments to our Ethical Conduct Policy and any waiver of our Ethical Conduct Policy that applies to our Chief Executive Officer, Chief Financial Officer, Principal Accounting Officer, or member of the Board of Directors, will be posted to <u>www.Marriott.com/investor</u>.

## **PART IV**

## Item 15. Exhibits and Financial Statement Schedules.

## LIST OF DOCUMENTS FILED AS PART OF THIS REPORT

## (1) FINANCIAL STATEMENTS

The response to this portion of Item 15 is submitted under Item 8 of this Report on Form 10-K.

## (2) FINANCIAL STATEMENT SCHEDULES

Information relating to schedules for which provision is made in the applicable accounting regulations of the SEC is included in the notes to the financial statements and is incorporated herein by reference.

## (3) EXHIBITS

Any shareholder who wants a copy of the following Exhibits may obtain one from us upon request at a charge that reflects the reproduction cost of such Exhibits. Requests should be made to the Secretary, Marriott International, Inc., Marriott Drive, Department 52/862, Washington, D.C. 20058.

Incorporation by Reference

Exhibit No.	Description	(where a report is indicated below, that document has been previously filed with the SEC and the applicable exhibit is incorporated by reference thereto)
3.(i).1	Restated Certificate of Incorporation of the Company.	Exhibit No. 3.(i) to our Form 8-K filed August 22, 2006 (File No. 001-13881).
3.(i).2	Certificate of Elimination for the Capped Convertible Preferred Stock and ESOP Convertible Preferred Stock (nullifying and eliminating supplementary Certificates of Designation previously filed as Exhibits No. 3.1 and 3.2 to our Form 10-Q for the fiscal quarter ended June 16, 2000, filed July 28, 2000).	Exhibit 3(i) to our Form 8-K filed February 7, 2006 (File No. 001-13881).
3.(ii)	Amended and Restated Bylaws.	Exhibit No. 3.(ii) to our Form 8-K filed August 22, 2006 (File No. 001-13881).
4.1	Form of Common Stock Certificate.	Exhibit No. 4.5 to our Form S-3ASR filed December 8, 2005 (File No. 333-130212).
4.2	Amended and Restated Rights Agreement dated as of August 9, 1999, with The Bank of New York, as Rights Agent.	Exhibit No. 4.1 to our Form 10-Q for the fiscal quarter ended September 10, 1999 (File No. 001-13881).
4.3	Form of Rights Certificate.	Exhibit No. 99.4 to our Form 8-A/A filed April 3, 1998 (File No. 001-13881).
4.4	Indenture dated November 16, 1998, with JPMorgan Chase Bank, N.A., as Trustee, formerly known as The Chase Manhattan Bank.	Exhibit No. $4.1$ to our Form $10$ -K for the fiscal year ended January 1, 1999 (File No. $001$ - $13881$ ).
4.5	Form of 7.875% Series C Note due 2009.	Exhibit No. 4.1 to our Form 8-K filed on September 21, 1999 (File No. 001-13881).
4.6	Form of 7.000% Series E Note due 2008.	Exhibit No. 4.1(f) to our Form S-3 filed January 17, 2001 (File No. 333-53860).

#### Incorporation by Reference (where a report is indicated below, that document has been previously filed with the SEC and the applicable exhibit is incorporated by reference thereto)

Exhibit No.	Description	reference thereto)
4.7	Form of 4.625% Series F Note due 2012.	Exhibit No. 4.2 to our Form 8-K filed June 14, 2005 (File No. 001-13881).
4.8	Form of 5.810% Series G Note due 2015.	Exhibit No. 4.1 to our Form 10-Q for the fiscal quarter ended June 16, 2006 (File No. 001-13881).
4.9	Form of 6.200% Series H Note due 2016.	Exhibit No. 4.2 to our Form 8-K filed June 14, 2006 (File No. 001-13881).
4.10	Form of 6.375% Series I Note due 2017.	Exhibit No. 4.2 to our Form 8-K filed June 25, 2007 (File No. 001-13881).
4.11	Form of 5.625% Series J Note due 2013.	Exhibit No. 4.2 to our Form 8-K filed October 19, 2007 (File No. 001-13881).
10.1	U.S. \$2.5 billion Amended and Restated Credit Agreement dated as of May 14, 2007, with Citibank, N.A., as Administrative Agent and certain banks.	Exhibit No. 10 to our Form 8-K filed May 16, 2007 (File No. 001-13881).
*10.2	Marriott International, Inc. Stock and Cash Incentive Plan-Amended and Restated as of January 1, 2008.	Filed with this report.
*10.3	Marriott International, Inc. Executive Deferred Compensation Plan- Amended and Restated as of January 1, 2008.	Filed with this report.
*10.4	Form of Employee Non-Qualified Stock Option Agreement for the Marriott International, Inc. Stock and Cash Incentive Plan.	Filed with this report.
*10.5	Form of Executive Restricted Stock Unit Agreement for the Marriott International, Inc. Stock and Cash Incentive Plan.	Filed with this report.
*10.6	Form of MI Shares Agreement for the Marriott International, Inc. Stock and Cash Incentive Plan.	Filed with this report.
*10.7	Form of Stock Appreciation Right Agreement for the Marriott International, Inc. Stock and Cash Incentive Plan.	Filed with this report.
*10.8	Form of Stock Appreciation Right Agreement for the Marriott International, Inc. Stock and Cash Incentive Plan (For Non-Employee Directors).	Filed with this report.
*10.9	Summary of Marriott International, Inc. Director Compensation.	Filed with this report.
*10.10	Marriott International, Inc. Executive Officer Incentive Plan and Executive Officer Individual Performance Plan.	Filed with this report.

Incorporation by Reference (where a report is indicated below, that document has been previously filed with the SEC and the applicable exhibit is incorporated by

Exhibit No.	Description	reference thereto)
12	Statement of Computation of Ratio of Earnings to Fixed Charges.	Filed with this report.
21	Subsidiaries of Marriott International, Inc.	Filed with this report.
23	Consent of Ernst & Young LLP.	Filed with this report.
31.1	Certification of Chief Executive Officer Pursuant to Rule 13a-14(a).	Filed with this report.
31.2	Certification of Chief Financial Officer Pursuant to Rule 13a-14(a).	Filed with this report.
32	Section 1350 Certifications.	Furnished with this report.

<sup>\*</sup> Denotes management contract or compensatory plan.

MARRIOTT INTERNATIONAL, INC.

# **SIGNATURES**

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, we have duly caused this Form 10-K to be signed on our behalf by the undersigned, thereunto duly authorized, on this 15<sup>th</sup> day of February 2008.

By: /s/ J.W. Marriott, Jr.	
J.W. Marriott, Jr.	
Chief Executive Officer	
Pursuant to the requirements of the Securities Exchange Act of 1934 capacities indicated and on the date indicated above.	I, this Form 10-K has been signed by the following persons on our behalf in the
PRINCIPAL EXECUTIVE OFFICER:	
/s/ J.W. Marriott, Jr.	
J.W. Marriott, Jr.	Chairman of the Board, Chief Executive Officer and Director
PRINCIPAL FINANCIAL OFFICER:	
/s/ Arne M. Sorenson	
Arne M. Sorenson	Executive Vice President and Chief Financial Officer
PRINCIPAL ACCOUNTING OFFICER:	
/s/ Carl T. Berquist	
Carl T. Berquist	Executive Vice President, Financial Information and Enterprise Risk Management
DIRECTORS:	
/s/ John W. Marriott III	/s/ Harry J. Pearce
John W. Marriott III, Vice Chairman of the Board	Harry J. Pearce, Director
/s/ Lawrence W. Kellner	/s/ Steven S Reinemund
Lawrence W. Kellner, Director	Steven S Reinemund, Director
/s/ Debra L. Lee	/s/ William J. Shaw
Debra L. Lee, Director	William J. Shaw, Director
/s/ Floretta Dukes McKenzie	/s/ Lawrence M. Small
Floretta Dukes McKenzie, Director	Lawrence M. Small, Director
/s/ George Muñoz	
George Muñoz, Director	

Marriott International, Inc.

**Stock And Cash Incentive Plan** 

As Amended and Restated Effective January 1, 2008

## **Contents**

		Page
Article 1.	Establishment, Objectives, and Duration	1
Article 2.	Definitions	1
Article 3.	Administration	6
Article 4.	Shares Subject to the Plan and Maximum Awards	7
Article 5.	Eligibility and Participation	7
Article 6.	SARs and Stock Options	8
Article 7.	Restricted Stock	11
Article 8.	Deferred Stock	12
Article 9.	Special Recognition Stock Awards	16
Article 9A.	MI Shares	16
Article 10.	Other Awards	18
Article 11.	Performance Measures for Awards	19
Article 12.	Directors' Share Awards, Fee Deferral Elections, and Director SARs and Options	20
Article 13.	1998 Conversion Awards	22
Article 14.	Beneficiary Designation	22
Article 15.	Reserved	23
Article 16.	Rights of Participants	23
Article 17.	Amendment, Modification, and Termination	23
Article 18.	Withholding	24
Article 19.	Indemnification	24
Article 20.	Successors	24
Article 21.	Legal Construction	25

# MARRIOTT INTERNATIONAL, INC. STOCK AND CASH INCENTIVE PLAN

### Article 1. Establishment, Objectives, and Duration

**1.1 Establishment of the Plan.** Marriott International, Inc., a Delaware corporation (the "Company"), hereby establishes an incentive compensation plan to be known as the Marriott International, Inc. Stock and Cash Incentive Plan (hereinafter referred to as the "Plan"), as set forth in this document.

The Plan shall become effective as of the Effective Date, as defined below, and shall remain in effect as provided in Article 1.3 hereof.

**1.2 Purpose of the Plan.** The purpose of the Plan is to promote and enhance the long-term growth of the Company by aligning the personal interests of Employees and Non-Employee Directors to those of Company shareholders and allowing such Employees and Non-Employee Directors to participate in the growth, development and financial success of the Company.

The Plan is further intended to provide flexibility to the Company in its ability to motivate, attract, and retain the services of key individuals.

**1.3 Duration of the Plan.** The Plan shall commence on the Effective Date, as described in Article 1.1 hereof, and shall remain in effect, subject to the right of the Board of Directors to amend or terminate the Plan at any time pursuant to Article 17 hereof, until all Shares subject to it shall have been purchased or acquired according to the Plan's provisions.

### **Article 2. Definitions**

Whenever used in the Plan, the following terms shall have the meanings set forth below, and when the meaning is intended, the initial letter of the word shall be capitalized:

- **2.1 "Allocation Agreement"** means the Employee Benefits and Other Employment Matters Allocation Agreement by and between Marriott International, Inc. (To Be Renamed Sodexho Marriott Services, Inc.) and New Marriott MI, Inc. (To Be Renamed Marriott International, Inc.) dated as of September 30, 1997.
  - 2.2 "Annual Meeting" means the annual meeting of the stockholders of the Company at which Directors are elected.
- **2.3 "Award"** means, individually or collectively, a grant under this Plan of MI Shares, SARs, Nonqualified Stock Options, Incentive Stock Options, Restricted Stock, Deferred Stock Bonus Awards, Deferred Stock Agreements, Special Recognition Stock Awards, 1998 Conversion Awards, Other Share-Based Awards, Other Cash Performance-Based Awards, Non-Employee Director Share Awards, Stock Units, and Director SARs and Options.

- **2.4 "Award Agreement"** means an agreement entered into by the Company and each Participant setting forth the terms and provisions applicable to an Award granted under this Plan.
- **2.5 "Beneficial Owner"** or **"Beneficial Ownership"** shall have the meaning ascribed to such term in Rule 13d-3 of the General Rules and Regulations under the Exchange Act.
  - **2.6 "Beneficiary**" means the person or persons designated pursuant to Article 14 hereof.
  - 2.7 "Board" or "Board of Directors" means the Board of Directors of the Company.
  - **2.8 "Code"** means the Internal Revenue Code of 1986, as amended from time to time.
- **2.9 "Committee"** means the Compensation Policy Committee of the Board, as specified in Article 3 herein, or such other Committee appointed by the Board to administer the Plan with respect to grants of Awards.
  - 2.10 "Company" means Marriott International, Inc., together with any and all Subsidiaries, and any successor thereto as provided in Article 20 herein.
  - 2.11 "Current Award" means a Deferred Stock Bonus Award granted under the terms and conditions described in Article 8.2(c) hereof.
- **2.12 "Covered Employee"** means a Participant who, as of the date of grant, vesting and/or payout of an Award, as applicable, is one of the group of "covered employees," as defined in the regulations promulgated under Code Section 162(m), or any successor statute.
  - 2.13 "Deferred Award" means a Deferred Stock Bonus Award granted under the terms and conditions described in Article 8.2(b) hereof.
  - **2.14 "Deferred Stock"** means an Award granted to a Participant as described in Article 8 herein.
  - 2.15 "Deferred Stock Bonus Award" means a grant of a right to receive Shares on a deferred basis, pursuant to Article 8.2 hereof.
  - 2.16 "Deferred Stock Agreement" means an Award granted to a Participant as described in Article 8.3 herein.
  - **2.17 "Director"** means any member of the Board.
  - 2.18 "Director SAR" and "Director Option" mean, respectively, a SAR and a Nonqualified Stock Option as described in Article 12 herein.

- **2.19 "Disability"** means a permanent and total disability, within the meaning of Code Section 22(e)(3), as determined by the Committee in good faith, upon receipt of sufficient competent medical advice from one or more individuals, selected by or satisfactory to the Committee, who are qualified to give professional medical advice.
  - 2.20 "Distribution" means the distribution of all the outstanding shares of capital stock of the Company as provided in the Distribution Agreement.
- **2.21 "Distribution Agreement"** means the Distribution Agreement between Marriott International, Inc. (To Be Renamed Sodexho Marriott Services, Inc.) and the Company dated as of September 30, 1997.
  - 2.22 "Distribution Date" means the date on which the Distribution shall be effected pursuant to the Distribution Agreement.
  - 2.23 "Effective Date" means January 1, 2008, except as otherwise indicated herein.
- **2.24 "Employee"** means any individual who is, or will become, a full-time, active, non-union employee of the Company. Any Employee who, at the request of the Company, and on the written assignment of the Company specifically referencing this provision of the Plan, becomes an employee of another employer shall continue to be treated as an Employee for all purposes hereunder during the period of such assignment. Directors who are not employed by the Company shall not be considered Employees under this Plan.
- **2.25 "Engaging in Competition"** means (i) engaging, individually or as an employee, consultant, owner (more than five percent (5%)) or agent of any entity, in or on behalf of any business engaged in significant competition (or that transacts or cooperates with another business in activities of significant competition) with any business operated by the Company or with interests adverse to those of the Company; (ii) soliciting and hiring a key employee of the Company in another business, whether or not in significant competition with any business operated by the Company; or (iii) using or disclosing confidential or proprietary information, in each case, without the approval of the Company.
  - **2.26** "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, or any successor act thereto.
- **2.27 "Exercise Price"** means the price at which a Share may be purchased by a Participant pursuant to an Option or the base price from which appreciation in Shares is measured under a SAR.
- **2.28 "Fair Market Value"** means the average of the highest and lowest quoted selling prices for the Shares on the relevant date, or (if there were no sales on such date) the average so computed on the nearest day before or the nearest day after the relevant date, as reported in *The Wall Street Journal* or a similar publication selected by the Committee.

- **2.29** "Fee Deferral Election" means an election made by a Non-Employee Director to defer the receipt of Fees, as described in Article 12.3 hereof.
- 2.30 "Fees" means all or part of any retainer and/or fees payable to a Non-Employee Director in his or her capacity as such.
- **2.31 "Incentive Stock Option"** or **"ISO"** means an option to purchase Shares granted under Article 6 herein, which is designated as an Incentive Stock Option and which is intended to meet the requirements of Code Section 422.
- **2.32 "Insider"** shall mean an individual who is, on the relevant date, an officer, Director or more than ten percent (10%) beneficial owner of any class of the Company's equity securities that is registered pursuant to Section 12 of the Exchange Act, all as defined under Section 16 of the Exchange Act.
  - **2.33 "MI Share"** means an Award granted to a Participant pursuant to Article 9A herein.
- **2.34 "1998 Conversion Award"** means an Award made pursuant to Article 13 to reflect the effect of the Distribution on outstanding awards which were made under the Predecessor Plans and which were held by the grantee immediately before the Distribution.
  - 2.35 "Non-Employee Director" means a Director who is not an Employee of the Company.
  - **2.36 "Non-Employee Director Share Award"** shall mean an award of Shares to a Non-Employee Director, as described in Article 12.2 herein.
- **2.37 "Nonqualified Stock Option"** or **"NQSO"** means an option to purchase Shares granted under Article 6 herein and which is not intended to meet the requirements of Code Section 422.
- **2.38 "Option"** means an Incentive Stock Option or a Nonqualified Stock Option, as described in Article 6 herein, or a Director Option as described in Article 12 herein.
  - 2.39 "Other Cash Performance-Based Awards" means an Other Cash Performance-Based Award, as described in Article 10 herein.
  - 2.40 "Other Share-Based Award" means an Other Share-Based Award, as described in Article 10 herein.
  - 2.41 "Participant" means an individual who has an outstanding Award granted under the Plan.
  - 2.42 "Performance-Based Exception" means the performance-based exception from the tax deductibility limitations of Code Section 162(m).

- **2.43 "Period of Restriction"** means the period during which the transfer of Shares of Restricted Stock is limited in some way (based on the passage of time, the achievement of performance objectives, or upon the occurrence of other events as determined by the Committee, in its discretion), and the Shares are subject to a substantial risk of forfeiture, as provided in Article 7 herein.
- **2.44 "Person"** shall have the meaning ascribed to such term in Section 3(a)(9) of the Exchange Act and used in Sections 13(d) and 14(d) thereof, including a "group" as defined in Section 13(d) thereof.
- **2.45 "Predecessor Plans"** means the Marriott International, Inc. 1993 Comprehensive Stock Incentive Plan, the Marriott International, Inc. 1996 Comprehensive Stock Incentive Plan and the Marriott International, Inc. 1995 Non-Employee Directors' Deferred Stock Compensation Plan.
  - **2.46 "Restricted Stock"** means an Award granted to a Participant pursuant to Article 7 herein.
  - 2.47 "Shares" means shares of Class A Common Stock of the Company or of any successor company adopting this Plan.
  - 2.48 "Special Recognition Stock Award" means an Award granted to a Participant pursuant to Article 9 herein.
  - 2.49 "SAR" means a stock appreciation right Award granted to a Participant pursuant to Article 6 herein which shall be settled in Shares.
- **2.50 "Stock Units"** means the credits to a Non-Employee Director's Stock Unit Account, each of which represents the right to receive one Share upon settlement of the Stock Unit Account.
  - **2.51 "Stock Unit Account"** means the bookkeeping account established by the Company pursuant to Article 12.3.
- **2.52 "Subsidiary"** means any corporation, partnership, joint venture or other entity in which the Company has a controlling interest as defined in Treasury Regulation Section 1.414(c)-2(b)(2), except that the threshold interest shall be "more than fifty percent (50%)" instead of "at least eighty percent (80%)."
  - **2.53 "Termination of Service"** means termination of service as a Non-Employee Director in any of the following circumstances:
    - (a) Where the Non-Employee Director voluntarily resigns or retires;
  - (b) Where the Non-Employee Director is not re-elected (or elected in the case of an appointed Non-Employee Director) to the Board by the shareholders; or

(c) Where the Non-Employee Director dies.

With respect to any Awards that are or become subject to Section 409A of the Code, Termination of Service shall not include any event that is not within the meaning of "separation from service" as set forth in Treasury Regulation Section 1.409A-1(h).

**2.54 "Year of Service"** means a period of twelve (12) consecutive calendar months during which an Employee was paid for twelve hundred (1200) or more hours of work for the Company.

#### **Article 3. Administration**

- **3.1 The Committee.** The Plan shall be administered by the Compensation Policy Committee of the Board, or by any other committee appointed by the Board, the members of which shall be "Non-Employee Directors" within the meaning of Rule 16b-3 under the Exchange Act, or any successor provision. The members of the Committee shall be appointed from time to time by, and shall serve at the discretion of, the Board of Directors.
- 3.2 Authority of the Committee. Except as limited by law or by the Articles of Incorporation or Bylaws of the Company, and subject to the provisions herein, the Committee shall have full power to select Employees and Directors who shall participate in the Plan; determine the sizes and types of Awards; determine the terms and conditions of Awards in a manner consistent with the Plan; construe and interpret the Plan and any agreement or instrument entered into under the Plan; establish, amend, or waive rules and regulations for the Plan's administration; and (subject to the provisions of Article 17 herein) amend the terms and conditions of any outstanding Award to the extent such terms and conditions are within the discretion of the Committee as provided in the Plan. Further, the Committee shall make all other determinations that may be necessary or advisable for the administration of the Plan. The Committee's determinations under the Plan (including without limitation, determinations of the persons to receive Awards, the form, amount and timing of such Awards, the terms and provisions of such Awards and the Award Agreements evidencing such Awards) need not be uniform and may be made by the Committee selectively among persons who receive, or are eligible to receive, Awards under the Plan, whether or not such persons are similarly situated. As permitted by law, the Committee may delegate its authority under the Plan to a Director or Employee.
- **3.3 Decisions Binding.** All determinations and decisions made by the Committee or its designee pursuant to the provisions of the Plan and all related orders and resolutions of the Board shall be final, conclusive and binding on all parties.
- **3.4 Unanimous Consent in Lieu of Meeting.** A memorandum signed by all members of the Committee shall constitute the act of the Committee without the necessity in such event to hold a meeting.
- **3.5 Serious Misconduct.** Notwithstanding anything to the contrary in the Plan or any Award Agreement, if a Participant terminates employment for serious misconduct, the

Committee may, in its sole discretion, refuse or revoke Approved Retiree status or other retirement approval for such Participant, or otherwise determine that such Participant may not receive, vest in or exercise any Awards or otherwise receive Shares thereunder to the extent the Awards are not granted, vested or fully exercised, or Shares are not received, as of such determination.

### Article 4. Shares Subject to the Plan and Maximum Awards

- **4.1 Number of Shares.** Subject to Articles 4.2 and 4.3 herein, (a) no more than 85 million shares of Class A Common Stock of the Company may be issued pursuant to Awards granted under the Plan, and (b) the maximum aggregate number of Shares that may be subject to any Awards (other than 1998 Conversion Awards) granted in any one fiscal year to any single Employee shall be 750,000.
- **4.2 Lapsed Awards.** If any Award granted under the Plan is canceled, terminates, expires, or lapses for any reason, any Shares subject to such Award shall again be available for the grant of an Award under the Plan.
- 4.3 Adjustments in Authorized Shares and Awards. In the event of any change in corporate capitalization, such as a stock split, or a corporate transaction, such as any merger, consolidation, separation, including a spin-off, or other distribution of stock or property of the Company, any reorganization (whether or not such reorganization comes within the definition of such term in Code Section 368) or any partial or complete liquidation of the Company, (a) such adjustment shall be made in the number and class of Shares which may be delivered under Article 4.1 and the Award limits set forth in Article 4.1 as may be determined to be appropriate and equitable by the Committee, in its sole discretion, to prevent dilution or enlargement of rights; and (b) the Committee or the board of directors, compensation committee or similar body of any other legal entity assuming the obligations of the Company hereunder, shall either (i) except for Awards of Deferred Stock held by former associates, make appropriate provision for the protection of outstanding Awards by the substitution on an equitable basis of appropriate equity interests or awards similar to the Awards (or, in the event no such similar equity interests may be identified, a nonqualified deferred compensation account allocation of equivalent value), provided that the substitution neither enlarges nor diminishes the value and rights under the Awards; or (ii) except for Awards of Deferred Stock held by active associates, upon written notice to the Participants, provide that Awards will be exercised, distributed, canceled or exchanged for value pursuant to such terms and conditions (including the waiver of any existing terms or conditions) as shall be specified in the notice. Any adjustment of an ISO under this paragraph shall be made in such a manner so as not to constitute a "modification" within the meaning of Section 424(h)(3) of the Code.

## Article 5. Eligibility and Participation

**5.1 Eligibility.** Employees shall be eligible to participate in this Plan with respect to Awards specified in Articles 6 through 10. Non-Employee Directors shall be eligible to participate in the Plan with respect to Awards specified in Article 12. Persons eligible to receive 1998 Conversion Awards under the Allocation Agreement shall be eligible to participate in the Plan with respect to Awards specified in Article 13.

**5.2 Actual Participation by Employees.** Subject to the provisions of the Plan, the Committee may, from time to time, select from all eligible Employees, those to whom Awards shall be granted and shall determine the nature and amount of each Award.

### **Article 6. SARs and Stock Options**

- **6.1 Grant of SARs and Options.** Subject to the terms and provisions of the Plan, SARs and/or Options may be granted to Employees in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee. SARs and Options may include provisions for reload of SARs and Options, respectively, exercised (in the case of Options) by the tender of Shares or the withholding of Shares with respect to the exercise of the SARs and Options. A SAR or an Option, once granted, may not thereafter be amended to change the Exercise Price.
- **6.2 Award Agreement.** Each SAR and Option grant shall be evidenced by an Award Agreement that shall specify the Exercise Price, the duration of the Award, the number of Shares to which the Award pertains, and such other provisions as the Committee shall determine. The Award Agreement, if pertaining to an Option, also shall specify whether the Option is intended to be an ISO within the meaning of Code Section 422, or an NQSO whose grant is intended not to fall under the provisions of Code Section 422.
- **6.3** Exercise Price. The Exercise Price for each grant of a SAR or an Option under this Article 6 shall be at least equal to one hundred percent (100%) of the Fair Market Value of a Share on the date the SAR or Option is granted.
- **6.4 Duration of SARs and Options.** Each SAR and Option granted under this Article 6 shall expire at such time as the Committee shall determine at the time of grant; provided, however, that no SAR or Option shall be exercisable later than the fifteenth (15<sup>th</sup>) anniversary date of its grant.
- **6.5** Exercise of SARs and Options. SARS and Options granted under this Article 6 shall be exercisable at such times and be subject to such restrictions and conditions as the Committee shall in each instance approve, which need not be the same for each grant or for each Participant.

The ability of a Participant to exercise a SAR or an Option is conditioned upon the Participant not committing any criminal offense or malicious tort relating to or against the Company, or, as determined by the Committee in its sole discretion, engaging in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation.

**6.6 Notice and Payment.** SARs and Options granted under this Article 6 shall be exercised by the delivery of notice of exercise to the Company by such means as the Committee shall approve from time to time, setting forth the number of Shares with respect to which the SAR or Option is to be exercised, accompanied, in the case of Options, by full payment for the Shares.

The Exercise Price upon exercise of any Option shall be payable to the Company in full either: (a) in cash or its equivalent, or (b) if permitted in the governing Award Agreement, by tendering previously acquired Shares having an aggregate Fair Market Value at the time of exercise equal to the total Exercise Price (provided that the Shares which are tendered must have been held by the Participant for at least six (6) months prior to their tender to satisfy the Exercise Price), or (c) if permitted in the governing Award Agreement, by a combination of (a) and (b).

The Committee also may allow cashless exercise as permitted under the Federal Reserve Board's Regulation T, subject to applicable securities law restrictions, or by any other means which the Committee determines to be consistent with the Plan's purpose and applicable law.

**6.8 Restrictions on Share Transferability.** The Committee may impose such restrictions on any Shares acquired pursuant to the exercise of a SAR or an Option granted under this Article 6 as it may deem advisable, including, without limitation, restrictions under applicable Federal securities laws, under the requirements of any stock exchange or market upon which such Shares are then listed or traded, and under any blue sky or state securities laws applicable to such Shares.

**6.9 Termination of Employment or Leave of Absence.** In the event that a Participant who is an Employee, during his or her lifetime has been on leave of absence for a period of greater than twelve (12) months (except a leave of absence approved by the Board or the Committee, as the case may be), or ceases to be an Employee of the Company or of any Subsidiary for any reason, including retirement, the portion of any SAR or Option which is not exercisable on the date on which the Participant ceased to be an Employee or has been on leave for over twelve (12) months (except a leave of absence approved by the Board or the Committee, as the case may be) shall expire on such date and any unexercised portion thereof which was otherwise exercisable on such date shall expire unless exercised within a period of three (3) months from such date, but in no event after the expiration of the term for which the SAR or Option was granted; provided, however, that in the case of an awardee of a SAR or a NQSO who is an "Approved Retiree" (as hereinafter defined), the SAR or NQSO shall continue to vest for up to five years from the date of retirement and said awardee may exercise such SAR or NQSO, as applicable, until the soonest to occur of (i) the expiration of such SAR or NQSO in accordance with its original term; (ii) the expiration of five (5) years from the date of retirement; or (iii) with respect to SARs or Options granted after 2005 and less than one year before the date the Approved Retiree retires, expiration of the SAR or Option on such retirement date, except not with respect to that portion of the SARs or Options equal to such number of shares multiplied by the ratio of (I) the number of days between the grant date and the retirement date inclusive, over (II) the number of days in the twelve (12) month period following the grant date. For purposes of the proviso to the preceding sentence:

- (a) An "Approved Retiree" is any awardee of a SAR or an Option who (i) terminates employment by reason of a Disability, or (ii) (A) retires from employment with the Company with the specific approval of the Committee on or after such date on which the awardee has attained age fifty-five (55) and completed ten (10) Years of Service or, with respect to Options granted prior to 2006, has completed twenty (20) Years of Service, and (B) has entered into and has not breached an agreement to refrain from Engaging in Competition in form and substance satisfactory to the Committee; and
- (b) If the Committee subsequently determines, in its sole discretion, that an Approved Retiree has violated the provisions of the agreement to refrain from Engaging in Competition referred to in clause (a)(ii)(B) of this Article, or has engaged in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation, such Approved Retiree shall have ninety (90) days from the date of such finding within which to exercise any SARs or Options or portions thereof which are exercisable on such date, and any Options or portions thereof which are not exercised within such ninety- (90-) day period shall expire, and any SARs or Options or portion thereof which are not exercisable on such date shall be cancelled on such date.

In the event of the death of an awardee during the three (3)-month period described above for exercise of a SAR or an Option by a terminated awardee or one on leave for over twelve (12) months (except a leave of absence approved by the Board or the Committee, as the case may be), the Option shall be exercisable by the awardee's personal representatives, heirs or legatees to the same extent and during the same period that the awardee could have exercised the SAR or Option if the awardee had not died.

Notwithstanding anything in Article 6.5 to the contrary, in the event of the death of an awardee while an Employee or Approved Retiree of the Company or any Subsidiary, an outstanding SAR or Option held by such awardee upon death shall become fully vested upon death and shall be exercisable by the awardee's personal representatives, heirs or legatees at any time prior to the expiration of one (1) year from the date of death of the awardee, but in no event after the expiration of the term for which the SAR or Option was granted.

## 6.10 Nontransferability of SARs and Options.

- (a) **Incentive Stock Options.** No ISO granted under the Plan may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution. Further, all ISOs granted to a Participant under the Plan shall be exercisable during his or her lifetime only by such Participant.
- (b) **SARs and Nonqualified Stock Options.** Except as otherwise provided in a Participant's Award Agreement or pursuant to policies adopted by the Committee, no SAR or NQSO granted under this Article 6 may be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution. Further, except as otherwise provided in the Plan or a Participant's Award Agreement, all SARs and NQSOs granted to a Participant under this Article 6 shall be exercisable during his or her lifetime only by such Participant.

#### **Article 7. Restricted Stock**

- **7.1 Grant of Restricted Stock.** Subject to the terms and provisions of the Plan, the Committee, at any time and from time to time, may grant Shares of Restricted Stock to Employees in such amounts as the Committee shall determine.
- **7.2 Restricted Stock Agreement.** Each Restricted Stock grant shall be evidenced by a Restricted Stock Award Agreement that shall specify the Period(s) of Restriction, the number of Shares of Restricted Stock granted, and such other provisions as the Committee shall determine.
- **7.3 Transferability.** Except as provided in this Article 7, the Shares of Restricted Stock granted herein may not be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated until the end of the applicable Period of Restriction established by the Committee and specified in the Restricted Stock Award Agreement, or upon earlier satisfaction of any other conditions, as specified by the Committee in its sole discretion and set forth in the Restricted Stock Award Agreement. All rights with respect to the Restricted Stock granted to a Participant under the Plan shall be available during his or her lifetime only to such Participant.
- **7.4 Other Restrictions.** The Committee shall impose such conditions and/or restrictions on any Shares of Restricted Stock granted pursuant to the Plan as it may deem advisable including, without limitation, a requirement that Participants pay a stipulated purchase price for each Share of Restricted Stock, restrictions based upon the achievement of specific performance objectives (Company-wide, business unit, and/or individual), time-based restrictions on vesting following the attainment of the performance objectives, and/or restrictions under applicable Federal or state securities laws.

The Company shall retain the certificates representing Shares of Restricted Stock in the Company's possession until such time as all conditions and/or restrictions applicable to such Shares have been satisfied.

Except as otherwise provided in this Article 7, Shares of Restricted Stock covered by each Restricted Stock grant made under the Plan shall become freely transferable by the Participant after the last day of the applicable Period of Restriction.

Distribution of Shares of Restricted Stock is conditioned upon the Participant not committing any criminal offense or malicious tort relating to or against the Company or, as determined by the Committee in its sole discretion, engaging in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the company's operations, financial condition or business reputation.

**7.5 Voting Rights.** During the Period of Restriction, Participants holding Shares of Restricted Stock granted hereunder may exercise full voting rights with respect to those Shares.

**7.6 Dividends and Other Distributions.** During the Period of Restriction, Participants holding Shares of Restricted Stock granted hereunder may be credited with regular cash dividends paid with respect to the underlying Shares while they are so held. Such dividends may be paid currently or converted into additional shares of Restricted Stock, upon such terms as the Committee establishes.

The Committee may apply any restrictions to the dividends that the Committee deems appropriate. Without limiting the generality of the preceding sentence, if the grant or vesting of Restricted Stock granted to a Covered Employee is designed to comply with the requirements of the Performance-Based Exception, the Committee may apply any restrictions it deems appropriate to the payment of dividends declared with respect to such Restricted Stock, such that the dividends and/or the Restricted Stock maintain eligibility for the Performance-Based Exception.

**7.7 Termination of Employment.** In the event a Participant's employment with the Company is terminated because of the Participant's Disability or death during the Period of Restriction, the Period of Restriction shall end and the Participant's rights thereunder shall inure to the benefit of his or her Beneficiary.

In the event that a Participant's employment with the Company is terminated during the Period of Restriction because of either the Participant's:

(a) retirement with specific approval from the Committee following attainment of age fifty-five (55) and with ten (10) Years of Service or (b) with respect to Awards granted before January 1, 2006, retirement with specific approval from the Committee and with twenty (20) Years of Service, the Committee shall have complete discretion in determining the percentage, if any, of a Participant's outstanding Restricted Shares as to which the Period of Restriction shall end. In the event that a Participant's employment with the Company is terminated for any other reason during the Period of Restriction, such Participant's outstanding Restricted Shares shall be forfeited to the Company without payment.

#### **Article 8. Deferred Stock**

- **8.1 Award of Deferred Stock.** Subject to the terms and provisions of the Plan, Deferred Stock Bonus Awards or Deferred Stock Agreements may be granted to Employees at any time and from time to time as shall be determined by the Committee. The Committee shall have complete discretion in determining the amount of Deferred Stock granted to each Employee (subject to Article 4 herein) and, consistent with the provisions of the Plan, in determining the terms and conditions pertaining to such Awards of Deferred Stock.
- **8.2 Deferred Stock Bonus Awards.** Deferred Stock Bonus Awards may be granted as part of a management incentive program under which part of the annual performance bonus awarded to managers and other key Employees is made in Deferred Stock. Subject to the terms of the Plan, Deferred Stock Bonus Awards shall have such terms and conditions as determined by the Committee. As determined by the Committee and subject to the terms of the Plan, Participants selected by the Committee in its discretion may elect to receive their Deferred Stock Bonus Award in the form of either a Current Award or a Deferred Award.

(a) **Method of Election.** Each Participant who is granted a Deferred Stock Bonus Award and selected by the Committee in its discretion may elect, in writing, on a form to be furnished by the Company, to receive a Current Award or a Deferred Award. Notwithstanding the foregoing, any eligible Participant who does not elect to receive a Deferred Award within the time designated by the Company shall be granted a Current Award.

### (b) Deferred Award.

- (i) *Vesting.* Deferred Stock granted in connection with a Deferred Award shall contingently vest, pro rata, in annual installments commencing one year after the date of the Deferred Stock Bonus Award and continuing on each January 2 thereafter until the expiration of a ten (10)-year period from such commencement date. Notwithstanding the foregoing, all unvested Deferred Stock subject to a Deferred Award shall vest upon the Participant's: (1) termination of employment with retirement approval from the Committee following attainment of age fifty-five (55) with ten (10) Years of Service; (2) termination of employment with retirement approval from the Committee and with twenty (20) Years of Service; (3) Disability, or (4) death. Subject to Article 4.3 herein, unvested Deferred Stock shall not vest following termination of employment for any other reason.
- (ii) *Distribution of Shares*. Vested Shares will be distributed to the Participant in two (2) to ten (10) approximately equal annual installments, as elected by the Participant. Such distribution shall commence in the month of January following the date the Participant terminates employment; provided, however, that the Participant may elect, at the time of grant and prior to vesting in any shares of Deferred Stock subject to an Award, to receive his or her vested Shares in a single distribution which shall take place in the month of January following the year during which his or her termination of employment occurs.

All such elections made pursuant to this Article 8.2(b)(ii), shall be made at the time the Deferred Stock Bonus Award is granted, and shall be made, in writing, on a form prescribed by the Committee. Upon a Participant's death, all undistributed vested Deferred Stock will be distributed in one distribution as provided in Article 10 herein.

### (c) Current Award.

(i) *Distribution of Shares*. Shares subject to a Current Award will be distributed in ten (10) consecutive, approximately equal, annual installments, commencing in the first calendar quarter of the year following the year in which the Deferred Stock Bonus Award is granted. If a Participant dies prior to distribution of all Shares to which he or she is entitled, the remaining Shares will be distributed in one distribution as provided in Article 10 herein.

- (ii) *Forfeiture of Shares.* Any undistributed Shares subject to a Current Award will be forfeited and the Deferred Stock Bonus Award relating thereto terminated, without payment, if the Participant's employment with the Company is terminated for any reason other than the Participant's:
- (1) termination of employment with retirement approval from the Committee at or beyond age fifty-five (55) with ten (10) Years of Service,
- (2) retirement after twenty (20) Years of Service with approval from the Committee, (3) Disability, or (4) death. Any undistributed Shares not subject to forfeiture shall continue to be distributed to the Participant under the distribution schedule which would have applied to those Shares if the Participant had not terminated employment; or over such shorter period as may be determined by the Committee.
- (d) **Conditions.** Notwithstanding anything to the contrary in the Plan, distribution of Shares under Current Awards and Deferred Awards is conditioned upon:
  - (i) the Participant not committing any criminal offense or malicious tort relating to or against the Company, or, as determined by the Committee in its sole discretion, engaging in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation;
    - (ii) the Participant not Engaging in Competition; and
    - (iii) the Participant having provided the Committee with a current address where the Deferred Stock Bonus Award may be distributed.

If said conditions are not met, all undistributed Shares will be forfeited and the Deferred Stock Bonus Award terminated, without payment.

- (e) **Lump Sum Payments.** Notwithstanding anything in the Plan to the contrary, any Participant entitled upon termination of employment to receive a distribution pursuant to this Article 8 which has a total Fair Market Value at the time of such termination of \$5,000 or less shall receive such distribution in one lump sum as soon as practicable following termination of employment.
- 8.3 Deferred Stock Agreements. Deferred Stock Agreements represent Deferred Stock granted to a Participant subject to the following conditions:
- (a) **Vesting.** Deferred Stock granted pursuant to this Article 8.3 shall contingently vest over a specified number of years, as determined by the Committee. Notwithstanding the foregoing, the Committee shall have complete discretion in determining the vested percentage, if any, of all unvested Deferred Stock subject to a Deferred Stock Agreement upon either the Participant's (1) termination of employment

with retirement approval from the Committee following attainment of age fifty-five (55) and with ten (10) Years of Service or (2) termination of employment with retirement approval from the Committee and with twenty (20) Years of Service. All unvested Deferred Stock subject to a Deferred Stock Agreement shall immediately vest upon the Participant's termination of employment as a result of the Participant's Disability or death. Subject to Article 4.3 herein, unless otherwise provided in the Deferred Stock Agreement, if the Participant's employment with the Company is terminated for any other reason, all Deferred Stock that is not vested before such termination of employment shall be forfeited and the Deferred Stock Agreement terminated without payment.

- (b) **Distribution of Shares.** Vested Deferred Stock granted pursuant to this Article 8.3 shall be distributed to the Participant in the form of Shares in the manner specified in the Deferred Stock Agreement. Such distribution shall commence in accordance with the terms of the Deferred Stock Agreement; provided that upon the Participant's death, all unpaid vested Deferred Stock shall be distributed in the form of Shares, in one distribution, as provided in Article 14 hereof.
- (c) **Conditions.** Notwithstanding anything to the contrary in the Plan, distribution of Shares subject to Deferred Stock Agreements is conditioned upon:
  - (i) the Participant not Engaging in Competition,
  - (ii) the Participant not committing any criminal offense or malicious tort relating to or against the Company, or, as determined by the Committee in its sole discretion, engaging in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation; and
  - (iii) the Participant having provided the Committee with a current address where the Deferred Stock may be distributed.

    If said conditions are not met, all undistributed Deferred Stock will be forfeited and the Deferred Stock Agreement terminated without payment.
- **8.4 Assignment.** A Participant's rights under a Deferred Stock Agreement or Deferred Stock Bonus Award may not, without the Company's written consent, be assigned or otherwise transferred, nor shall they be subject to any right or claim of a Participant's creditors, provided that the Company may offset any amounts owing to or guaranteed by the Company, or owing to any credit union related to the Company against the value of Deferred Stock and underlying Shares to be distributed under Deferred Stock Agreements and Deferred Stock Bonus Awards.
- **8.5** Change in Distribution Schedule. Effective August 4, 2005, Participants with Deferred Awards or Deferred Stock Agreements may elect to change the schedule under which they receive Share distributions under such Awards, provided that (i) the alternative distribution schedule shall be a lump sum distribution occurring on March 22, 2006, for all such Shares that

are vested as of that date, and for all unvested Shares the alternative schedule of distributions shall be the sooner to occur of the fixed date(s) on which the Shares are scheduled to vest or the Participant's termination of employment; and (ii) the Participant elections shall be made no later than October 14, 2005, pursuant to procedures established by the Committee or its designee. This Article shall also apply to all Conversion Awards that are substantially similar in form to Deferred Awards and Deferred Stock Agreements.

**8.6 Key Employees.** Notwithstanding Articles 8.2(b)(ii) and 8.3, for Awards that are subject to Section 409A of the Code, distributions to a Participant who is a Key Employee (as defined below) on account of a Termination of Service shall be made or commence not before the date which is six (6) months following the Termination of Service, except in the event of the Participant's death. For this purpose, a Key Employee is a person described in Treasury Regulation Section 1.409A-1(i), applying the default rules thereunder.

### **Article 9. Special Recognition Stock Awards**

Subject to the terms and provisions of the Plan, the Committee or its designee, at any time and from time to time, may grant Special Recognition Stock Awards to Employees in such amounts and upon such conditions as the Committee or its designee shall determine.

#### Article 9A. MI Shares

**9A.1 MI Shares.** Subject to the terms and conditions of the Plan, the Committee, at any time and from time to time, may grant Awards of MI Shares to eligible Employees in such amounts as the Committee shall determine.

**9A.2 MI Share and Common Share Rights.** MI Shares shall represent an Employee's unsecured right to receive from the Company the transfer of title to Shares in accordance with the schedule of vesting dates set forth in Article 9A.3 below, provided that the Employee has satisfied the conditions of transfer set forth in Article 9A.4 below, and subject to the satisfaction of the provision on withholding taxes set forth in Article 9A.6 below. On each such vesting date, if it occurs, the Company shall transfer a corresponding number of Shares (which may be reduced by the number of Shares withheld to satisfy withholding taxes as set forth in Article 9A.6 below, if share reduction is the method utilized for satisfying the tax withholding obligation) to an individual brokerage account established and maintained in the Employee's name. The Employee shall have all the rights of a stockholder with respect to such Shares transferred to the brokerage account, including but not limited to the right to vote the Shares, to sell, transfer, liquidate or otherwise dispose of the Shares, and to receive all dividends or other distributions paid or made with respect to the Shares from the time they are deposited in the account. The Employee shall have no voting, transfer, liquidation, dividend or other rights of a Share stockholder with respect to MI Shares prior to such time that the corresponding Shares are transferred, if at all, to the Employee's brokerage account.

**9A.3 Vesting in MI Shares.** If an MI Share Award is granted on the fifteenth (15<sup>th)</sup> or preceding day of any month, the Award shall vest pro rata with respect to an additional twenty-five percent (25%) of the MI Shares granted hereunder on the fifteenth (15<sup>th)</sup> day of the month in

which occurs the first (1<sup>st</sup>), second (2<sup>nd</sup>), third (3<sup>rd</sup>) and fourth (4<sup>th</sup>) twelve- (12-) month anniversaries of the grant date, respectively. If an MI Share Award is granted on the sixteenth (16<sup>th</sup>) or succeeding day of any month, the Award shall vest pro rata with respect to twenty-five (25%) of the MI Shares granted hereunder on the fifteenth (15<sup>th</sup>) day of the month following the first (1st), second (2nd), third (3rd) or fourth (4th) twelve- (12-) month anniversaries of the grant date, respectively. Notwithstanding the foregoing, in the event that any such fifteenth (15<sup>th</sup>) day of the month is a Saturday, Sunday or other day on which stock of the Company is not traded on the New York Stock Exchange or another national exchange, then the vesting date shall be the next following day on which the stock of the Company is traded on the New York Stock Exchange or another national exchange.

- **9A.4 Conditions of Transfer.** A transfer of Shares in accordance with paragraph 9A.2 above shall be conditioned upon the Employee meeting all of the following conditions during the entire period from the grant date through the vesting date(s) relating to such MI Shares:
  - (a) The Employee must continue to be an active employee of the Company or one of its Subsidiaries;
  - (b) The Employee must refrain from Engaging in Competition; and
  - (c) The Employee must refrain from committing any criminal offense or malicious tort relating to or against the Company or, as determined by the Committee in its discretion, engaging in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation.

If the Employee fails to meet the requirements of Article 9A.4(a) through (c), then the Employee shall forfeit the right to vest in any MI Shares that have not already vested as of the time such failure is determined, and the Employee shall accordingly forfeit the right to receive the transfer of title to any corresponding Shares. The forfeiture of rights with respect to unvested MI Shares (and corresponding Shares) shall not affect the rights of the Employee with respect to any MI Shares that already have vested nor with respect to any Shares the title of which has already been transferred to the Employee's brokerage account.

## **9A.5** Effect of Termination of Employment. Notwithstanding the contrary in Articles 9A.3 and 9A.4:

(a) In the event the Employee's employment is terminated prior to the relevant vesting date on account of death, and if the Employee had otherwise met the requirements of Article 9A.4(a) through (c) from the grant date through the date of such death, then the Employee's unvested MI Shares shall immediately vest in full upon death and the Employee's rights hereunder with respect to any such MI Shares shall inure to the benefit of the Employee's executors, administrators, personal representatives and assigns.

(b) In the event Employee's employment is terminated prior to the relevant vesting date on account of the Employee's Disability or Retirement (as defined below), and if the Employee had otherwise met the requirements of Article 9A.4(a) through (c) from the grant date through the date of such Disability or Retirement, and provided that the Employee continues to meet the requirements of Article 9A.4(b) and (c), then the Employee's rights hereunder with respect to any outstanding, unvested MI Shares shall continue in the same manner as if the Employee continued to meet the continuous employment requirement of Article 9A.4(a) through the vesting dates related to the Award, except not for that portion of MI Shares granted less than one (1) year prior to the Employee's termination equal to such number of shares multiplied by the ratio of (I) the number of days after the termination date and before the first (1<sup>st</sup>) anniversary of the grant date, over (II) the number of days on and after the grant date and before the first (1<sup>st</sup>) anniversary of the grant date. For purposes of this Article 9A.5(b), "Retirement" shall mean termination of employment by retiring with special approval of the Committee following age fifty-five (55) with ten (10) years of service.

**9A.6. Taxes.** The transfer of Shares upon each vesting date, pursuant to Articles 9A.2 and 9A.4 above, shall be subject to the further condition that the Company shall provide for the withholding of any taxes required by federal, state, or local law in respect of that vesting date by reducing the number of Shares to be transferred to the Employee's brokerage account or by such other manner as the Committee shall determine in its discretion.

### **Article 10. Other Awards**

- **10.1 Grant of Other Share-Based Awards.** The Committee may grant Other Share-Based Awards to Participants in such number, and upon such terms, and at any time and from time to time, as shall be determined by the Committee.
- **10.2 Terms of Other Share-Based Awards.** Other Share-Based Awards shall contain such terms and conditions as the Committee may from time to time specify and may be denominated in cash, in Shares, in Share-equivalent units, in Share appreciation units, in securities or debentures convertible into Shares or in a combination of the foregoing and may be paid in cash or in Shares, all as determined by the Committee. Other Share-Based Awards may be issued alone or in tandem with other Awards granted to Employees.
- **10.3 Other Share-Based Award Agreement.** Each Other Share-Based Award shall be evidenced by an Award Agreement that shall specify such terms and conditions as the Committee shall determine.
- **10.4 Other Cash Performance-Based Awards.** The Committee may grant Other Cash Performance-Based Awards based on performance measures set forth in Article 11 not based on Shares upon such terms and at any time and from time to time as shall be determined by the Committee. Each such Other Cash Performance-Based Award shall be evidenced by an award agreement that shall specify such terms and conditions as the Committee shall determine. An Other Cash Performance-Based Award not based upon Shares shall not decrease the number

of Shares under Article 4 that may be issued pursuant to other Awards. No individual shall be eligible to receive a payment with respect to cash performance-based awards in excess of \$4 million in any calendar year. Other Cash Performance-Based Awards may relate to annual bonus or long-term performance awards.

### **Article 11. Performance Measures for Awards**

- 11.1 Performance Measures. Unless and until the Committee proposes for shareholder vote and shareholders approve a change in the general performance measures set forth in this Article 11, the attainment of which may determine the degree of payout and/or vesting with respect to Awards granted to Covered Employees which are designed to qualify for the Performance-Based Exception, the performance measure(s) to be used for purposes of such Awards shall be chosen from among the following alternatives:
  - (a) Consolidated cash flows,
  - (b) Consolidated financial reported earnings,
  - (c) Consolidated economic earnings,
  - (d) Earnings per share,
  - (e) Business unit financial reported earnings,
  - (f) Business unit economic earnings,
  - (g) Business unit cash flows,
  - (h) Appreciation in the Fair Market Value of Shares either alone or as measured against the performance of the stocks of a group of companies approved by the Committee,
    - (i) Return on invested capital,
    - (j) Consolidated earnings before interest, taxes, depreciation and amortization ("EBITDA"), and
    - (k) Business unit EBITDA.
- **11.2 Adjustments.** The Committee shall have the discretion to adjust the determinations of the degree of attainment of the preestablished performance objectives; provided, however, that Awards that are designed to qualify for the Performance-Based Exception and that are held by Covered Employees may not be adjusted upward (the Committee shall retain the discretion to adjust such Awards downward).
- **11.3 Committee Discretion.** In the event that applicable tax and/or securities laws change to permit Committee discretion to alter the governing performance measures without

obtaining shareholder approval of such changes, the Committee shall have sole discretion to make such changes without obtaining shareholder approval. In addition, in the event that the Committee determines that it is advisable to grant Awards that do not qualify for the Performance-Based Exception; the Committee may make such grants without satisfying the requirements of Code Section 162(m).

### Article 12. Directors' Share Awards, Fee Deferral Elections, and Director SARs and Options.

- **12.1 Eligibility.** Only Non-Employee Directors shall be eligible to receive Non-Employee Director Share Awards and Director SARs and Options and to make Fee Deferral Elections.
- **12.2 Non-Employee Director Share Awards.** On the first (1<sup>st</sup>) full trading day immediately following each Annual Meeting, each Non-Employee Director designated by the Board shall receive a Non-Employee Director Share Award of a number of Shares determined by the Board before such Annual Meeting. Each Non-Employee Director Share Award shall be fully vested and nonforfeitable when granted.

#### 12.3 Fee Deferral Elections.

- (a) *Elections to Defer Payment of Fees.* Payment of all or any part of any Fees payable to a Non-Employee Director may be deferred by election of the Non-Employee Director. Each such election must be made in writing on a form prescribed by the Committee and irrevocably delivered to the Company in the year preceding the year which commences with the next Annual Meeting (the "Election Year") and must be irrevocable for such Election Year. No election may be made under this Article 12.3(a) with respect to Fees for which an election is made under Article 12.5.
- (b) *Crediting Stock Units to Accounts.* Amounts deferred pursuant to a Fee Deferral Election shall be credited as of the date of the deferral to a Stock Unit Account in Stock Units. The number of Stock Units credited to a Stock Unit Account with respect to any Non-Employee Director shall equal (i) the amount deferred pursuant to the Fee Deferral Election divided by (ii) the Fair Market Value of a Share on the date on which the Fees subject to the Fee Deferral Election would have been paid but for the Fee Deferral Election, with fractional units calculated to at least three (3) decimal places.
- (c) *Fully Vested Stock Units*. All Stock Units credited to a Non-Employee Director's Stock Unit Account pursuant to this Article 12.3 shall be at all times fully vested and nonforfeitable.
- (d) *Credit of Dividend Equivalents.* As of each dividend payment date with respect to Shares, each Non-Employee Director shall have credited to his or her Stock Unit Account an additional number of Stock Units equal to the product of (i) the per-share cash dividend payable with respect to a Share on such dividend payment date multiplied by the number of Stock Units credited to his or her Stock Unit Account as of

the close of business on the record date for such dividend, divided by (ii) the Fair Market Value of a Share on such dividend payment date. If dividends are paid on Shares in a form other than cash, then such dividends shall be notionally converted to cash, if their value is readily determinable, and credited in a manner consistent with the foregoing and, if their value is not readily determinable, shall be credited "in kind" to the Non-Employee Director's Stock Unit Account.

- (e) *Payment of Stock Units*. Upon Termination of Service, the Stock Units credited to a Non-Employee Director's Stock Unit Account shall be paid to the Non-Employee Director in an equal number of shares of Stock in a single lump sum or in substantially equal annual installments over a period not to exceed ten (10) years, as irrevocably elected in writing by the Non-Employee Director at the time of the Non-Employee Director's election to defer Fees under Article 12.3(a), pursuant to rules established from time to time by the Committee.
- **12.4 Unfunded Status.** The interest of each Non-Employee Director in any Fees deferred under this Article 12 (and any Stock Units or Stock Unit Account relating thereto) or in any Director Stock Award shall be that of a general creditor of the Company. Stock Unit Accounts and Stock Units (and, if any, "in kind" dividends) credited thereto shall at all times be maintained by the Company as bookkeeping entries evidencing unfunded and unsecured general obligations of the Company.

## 12.5 Director SARs and Options.

- (a) *Elections to Receive Payment of Fees in the Form of SARs or Options.* A Non-Employee Director may elect to receive payment of all or any part of his or her cash retainer in the form of Director SARs or Options, as determined by the Committee, in lieu of cash. Each such election must be made in writing on a form prescribed by the Committee and delivered to the Company in the calendar year preceding the calendar year in which occurs the Annual Meeting that marks the commencement of the annual period of service during which such Fees are earned. Each election is irrevocable for that annual period. Elections under this Article 12.5 may not be made with respect to Fees deferred under Article 12.3.
- (b) *Grant of Director SARs and Options.* On the first (1<sup>st</sup>) full trading day immediately following each Annual Meeting, each Non-Employee Director who has filed an election under Article 12.5(a) for the annual period of service that commences with such Annual Meeting shall be granted Director SARs or Options that have a value on the date of grant substantially equal to the amount of Fees otherwise payable to the Director in cash but for the election to receive Director SARs or Options. The value of Director SARs or Options shall be determined by the Committee in its sole discretion, at a meeting held prior to the Annual Meeting, based on a Black-Scholes option pricing model or other valuation model that the Committee determines to be appropriate in its sole discretion.
- (c) *Terms of Director SARs and Options*. Each Director SAR and Option shall be evidenced by an Award Agreement that shall specify the Exercise Price, the

duration of the SAR or Option, and the number of Shares to which the SAR or Option pertains. Each Director SAR and Option shall (i) have an Exercise Price equal to the Fair Market Value of a Share on the date the Award is granted; (ii) become one hundred percent (100%) vested and first exercisable on the last business day immediately preceding the Annual Meeting next following the date the SAR or Option is granted or, if earlier, upon the Director's Termination of Service due to death or Disability; (iii) expire on the tenth (10<sup>th</sup>) anniversary of the date of its grant; and (iv) be nontransferable unless otherwise specified by the Committee.

(d) *Payment.* Director SARs and Options granted under this Article 12 shall be exercised by the delivery of notice of exercise to the Company in such manner as the Committee shall determine, setting forth the number of Shares with respect to which the SAR or Option is to be exercised, accompanied by full payment for the Shares. The Exercise Price upon exercise of any Director SAR or Option shall be payable to the Company in full either: (i) in cash or its equivalent, (ii) by tendering previously acquired Shares having an aggregate Fair Market Value at the time of exercise equal to the total Exercise Price (provided that the Shares which are tendered must have been held by the Director for at least six (6) months prior to their tender to satisfy the Exercise Price), or (iii) by a combination of (i) and (ii). The Committee also may allow cashless exercise as permitted under the Federal Reserve Board's Regulation T, subject to applicable securities law restrictions, or by any other means which the Committee determines to be consistent with the Plan's purpose and applicable law.

### **Article 13. 1998 Conversion Awards**

All 1998 Conversion Awards which, under the Allocation Agreement, are to be denominated in equal numbers of shares of Class A Common Stock of the Company shall be issued under the Plan as provided in the Allocation Agreement. The Committee shall administer all such 1998 Conversion Awards under this Plan, giving service credit to the grantee of each such 1998 Conversion Award to the extent required under the Allocation Agreement. All 1998 Conversion Awards shall be subject to substantially similar terms and conditions as provided in the holder's corresponding awards under the Predecessor Plan.

### **Article 14. Beneficiary Designation**

Each Participant under the Plan may, from time to time, name any beneficiary or beneficiaries (who may be named contingently or successively) to whom any benefit under the Plan is to be paid in case of the Participant's death before the Participant has received any or all of such benefit. Each such designation shall revoke all prior designations by the same Participant, shall be in a form prescribed by the Company, and will be effective only when filed by the Participant in writing with the Company during the Participant's lifetime. In the absence of any such designation, benefits remaining unpaid at the Participant's death shall be paid to the Participant's estate.

### Article 15. [Reserved]

## **Article 16. Rights of Participants**

- **16.1 Employment or Service.** Nothing in the Plan shall interfere with or limit in any way the right of the Company to terminate any Participant's employment or service at any time, nor confer upon any Participant any right to continue in the employ or service of the Company.
- **16.2 Participation.** No Employee shall have the right to be selected to receive an Award under this Plan, or, having been so selected, to be selected to receive a future Award.

#### Article 17. Amendment, Modification, and Termination

- **17.1 Amendment, Modification, and Termination.** The Board may at any time and from time to time, alter, amend, suspend or terminate the Plan in whole or in part; provided, however, that the Board may, in its sole discretion, condition the adoption of any amendment of the Plan on the approval thereof by the requisite vote of the shareholders of the Company entitled to vote thereon.
- 17.2 Adjustment of Awards upon the Occurrence of Certain Unusual or Nonrecurring Events. Subject to the restriction set forth in Article 11 herein on the exercise of upward discretion with respect to Awards which have been designed to comply with the Performance-Based Exception, the Committee may make adjustments in the terms and conditions of, and the criteria included in, Awards in recognition of unusual or nonrecurring events (including, without limitation, the events described in Article 4.3 hereof) affecting the Company or the financial statements of the Company or of changes in applicable laws, regulations, or accounting principles, whenever the Committee determines that such adjustments are appropriate in order to prevent dilution or enlargement of the benefits or potential benefits intended to be made available under the Plan.
- **17.3 Awards Previously Granted.** No termination, amendment, or modification of the Plan or any Award shall adversely affect in any material way any Award previously granted under the Plan, without the written consent of the Participant holding such Award.
- **17.4 Compliance with Code Section 162(m).** At all times when Code Section 162(m) is applicable, all Awards granted under this Plan shall comply with the requirements of Code Section 162(m); provided, however, that in the event the Committee determines that such compliance is not desired with respect to any Award or Awards available for grant under the Plan, then compliance with Code Section 162(m) will not be required. In addition, in the event that changes are made to Code Section 162(m) to permit greater flexibility with respect to any Award or Awards available under the Plan, the Committee may, subject to this Article 17, make any adjustments it deems appropriate.
- **17.5 Substitution of Awards in Mergers and Acquisitions.** Awards may be granted under the Plan from time to time in substitution for awards held by employees or directors of

entities who become or are about to become employees or directors of the Company or a Subsidiary as the result of a merger, consolidation or other acquisition of the employing entity or the acquisition by the Company or a Subsidiary of the assets or stock of the employing entity. The terms and conditions of any substitute awards so granted may vary from the terms and conditions set forth herein to the extent that the Committee deems appropriate at the time of grant to conform the substitute awards to the provisions of the awards for which they are substituted.

#### Article 18. Withholding

**18.1 Tax Withholding.** The Company shall have the power and the right to deduct from any amount otherwise due to the Participant, or withhold, or require a Participant to remit to the Company, an amount sufficient to satisfy Federal, state, and local income, employment or other related taxes, domestic or foreign, required by law or regulation to be withheld with respect to any taxable event arising as a result of this Plan.

**18.2 Share Withholding.** With respect to withholding required in connection with any Award, the Company may require, or the Committee may permit a Participant to elect, that the withholding requirement be satisfied, in whole or in part, by having the Company withhold Shares having a Fair Market Value on the date the tax is to be determined equal to the minimum statutory total tax which could be withheld on the transaction. Any election by a Participant shall be irrevocable, made in writing, signed by the Participant, and shall be subject to any restrictions or limitations that the Committee, in its sole discretion, deems appropriate.

### **Article 19. Indemnification**

Each person who is or shall have been a member of the Committee, or of the Board, shall be indemnified and held harmless by the Company against and from any loss, cost, liability or expense that may be imposed upon or reasonably incurred by him or her in connection with or resulting from any claim, action, suit or proceeding to which he or she may be a party or in which he or she may be involved by reason of any action taken or failure to act under the Plan and against and from any and all amounts paid by him or her in settlement thereof, with the Company's approval, or paid by him or her in satisfaction of any judgment in any such action, suit or proceeding against him or her, provided he or she shall give the Company an opportunity, at its own expense, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such persons may be entitled under the Company's Articles of Incorporation or Bylaws, as a matter of law, or otherwise, or any power that the Company may have to indemnify them or hold them harmless.

### **Article 20. Successors**

All obligations of the Company under the Plan with respect to Awards granted hereunder shall be binding on any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, of all or substantially all of the business and/or assets of the Company, or a merger, consolidation or otherwise.

### **Article 21. Legal Construction**

- **21.1 Gender and Number.** Except where otherwise indicated by the context, any masculine term used herein also shall include the feminine, the plural shall include the singular and the singular shall include the plural.
- **21.2 Severability.** In the event any provision of the Plan shall be held illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining provisions of the Plan, and the Plan shall be construed and enforced as if the illegal or invalid provision had not been included.
- **21.3 Requirements of Law.** The granting of Awards and the issuance of Shares under the Plan shall be subject to all applicable laws, rules, and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.
- **21.4 Securities Law Compliance.** With respect to Insiders, transactions under this Plan are intended to comply with all applicable conditions of Rule 16b-3 or its successors under the Exchange Act. To the extent any provision of the plan or action by the Committee fails to so comply, it shall be deemed null and void, to the extent permitted by law and deemed advisable by the Committee.
- **21.5 Governing Law.** To the extent not preempted by Federal law, the Plan, and all agreements hereunder, shall be construed in accordance with and governed by the laws of the State of Maryland.

# MARRIOTT INTERNATIONAL, INC.

# EXECUTIVE DEFERRED COMPENSATION PLAN

Amended and Restated as of January 1, 2008

## TABLE OF CONTENTS

PREAM	MBLE	1
ARTICI	LE I - DEFINITIONS	2
1.1	ACCOUNT	2
1.2	ADMINISTRATOR	2
1.3	CODE	2
1.4	COMMITTEE	2
1.5	COMPANY	2
1.6	COMPANY ACCRUALS	2
1.7	COMPENSATION	2
1.8	Defferal Percentage	2
1.9	DEFERRED COMPENSATION	2
1.10	DEFERRED COMPENSATION RESERVE	2
1.11	EFFECTIVE DATE	3
1.12	ELECTION	3
1.13	ELECTION YEAR	3
1.14	EMPLOYEE	3
1.15	FISCAL YEAR	3
1.16	HR OFFICER	3
1.17	IN-SERVICE WITHDRAWAL	J n
1.18	LTCI COMPENSATION  NON EMPLOYER DEPOSTOR	
1.19 1.20	Non-Employee Director Participant	ა ე
1.21	PERMANENT DISABILITY	3
1.22	PLAN	4
1.23	REINSTATEMENT OR REINSTATED	5
1.24	RETIRE OR RETIREMENT	5
1.25	RETIREMENT SAVINGS PLAN	5
1.26	SEPARATION FROM SERVICE	5
1.27	SUBSIDIARY	5
1.28	VESTED PORTION	5
1.29	YEAR OF SERVICE	6
ARTICI	LE II - PARTICIPANT ELECTIONS	7
2.1	DEFERRED COMPENSATION RESERVE	7
2.2	ELECTIONS	7
2.3	FORM OF ELECTION	8
ARTICI	LE III - PARTICIPANT ACCOUNTS	9
3.1	INDIVIDUAL ACCOUNTS	q
3.2	COMPANY ACCRUALS	g
3.3	VESTING	10
3.4	FORFEITURES	11
3.5	CREDITING OF EARNINGS	11
3.6	ACCOUNTS DO NOT RESULT IN PROPERTY RIGHTS	11
3.7	NO ASSIGNMENT OF INTERESTS	12
3.8	FEDERAL AND STATE TAXES	12
ARTICI	LE IV - DISTRIBUTIONS	13
4.1	ELECTION OF DISTRIBUTION	13
4.2	FORM AND TIMING OF DISTRIBUTION	13
4.3	TAX IMPACT	16
4.4	CHANGES IN DISTRIBUTION ELECTION	16
4.5	BENEFICIARIES	17
4.6	DISCHARGE OF OBLIGATION FOR PAYMENT	17

ARTICLE V - ADMINISTRATION		18
5.1 5.2	ADMINISTRATOR EXPENSES	18 18
ARTICLE VI - CLAIMS PROCEDURE		19
6.1 6.2	INITIAL CLAIMS APPEALS	19 19
ARTIC	CLE VII - MISCELLANEOUS	20
7.1 7.2 7.3 7.4 7.5 7.6	PLAN NOT AN EMPLOYMENT CONTRACT NO TRUST CREATED AMENDMENT OR TERMINATION OF PLAN EFFECT OF PLAN SEVERABILITY APPLICABLE LAW	20 20 20 20 20 20 21
APPEN	NDIX A	22
BENC	CHMARK FUNDS	22

## MARRIOTT INTERNATIONAL, INC. EXECUTIVE DEFERRED COMPENSATION PLAN

## **PREAMBLE**

WHEREAS, as of March 27, 1998, the Company established an unfunded deferred compensation arrangement known as the Marriott International, Inc. Executive Deferred Compensation Plan (the "Plan") for the benefit of a select group of management and highly compensated employees of the Company and its subsidiaries; and

WHEREAS, effective January 1, 2001, the Plan was amended and restated to reflect amendments made to the Plan following March 27, 1998; and

WHEREAS, effective October 1, 2004, the Plan was amended and restated to reflect amendments made to the Plan following January 1, 2001; and

WHEREAS, the Company wishes to amend and restate the Plan to reflect amendments that have been made following the October 1, 2004 restatement and certain administrative changes, and to comply with section 409A of the Internal Revenue Code.

NOW THEREFORE, the Plan, as herein amended and restated, shall be effective as of January 1, 2008.

## ARTICLE I

## **DEFINITIONS**

For purposes of this Plan, unless the context requires otherwise, the following words and phrases, when used herein with initial capital letters, shall have the meanings indicated:

- 1.1 "Account" shall mean, with respect to each Participant, the amount of Company Accruals, Deferred Compensation and earnings credited to a Participant under the Deferred Compensation Reserve.
  - 1.2 "Administrator" means the Company's Senior Vice President for Executive Compensation.
  - 1.3 "Code" means the Internal Revenue Code of 1986, as amended, or any successor statute, including the regulations issued thereunder.
  - 1.4 "Committee" means the Compensation Policy Committee appointed by the Board of Directors of Marriott International, Inc.
  - 1.5 "Company" means Marriott International, Inc. and any Subsidiary that (a) elects to join the Plan, and (b) obtains the consent of the Committee to do so.
  - 1.6 "Company Accruals" means the amounts credited to the Deferred Compensation Reserve pursuant to Section 3.2.
- 1.7 "Compensation" means (a) with respect to Employees, Compensation as defined for purposes of computing contributions under the Retirement Savings Plan, determined, however, by including LTCI Compensation and without regard to any Elections made by the Employee to defer any compensation under this Plan; and (b) with respect to Non-Employee Directors, fees payable by the Company during the Election Year. Notwithstanding the preceding sentence, effective January 1, 2007, for purposes of determining contributions under the Plan, compensation received by the Employee on or after January 1, 2007, for payroll periods ending on or after December 28, 2006, shall not include payments made after the Employee's Separation from Service.
- 1.8 "<u>Deferral Percentage</u>" means the percentage of a Participant's Compensation for the Election Year to be deferred in accordance with an Election pursuant to Article II of this Plan.
- 1.9 "<u>Deferred Compensation</u>" means Compensation with respect to which a Participant has made an Election to defer receipt thereof in accordance with Article II of this Plan.
- 1.10 "<u>Deferred Compensation Reserve</u>" means the book reserve reflecting the total aggregate amounts credited to the individual accounts of Participants under Articles II and III of this Plan.

- 1.11 "Effective Date" means January 1, 2008, the effective date of this restatement of the Plan, except as otherwise indicated herein. The Plan was originally effective March 27, 1998.
  - 1.12 "Election" means an election made by a Participant in accordance with Article II of this Plan.
- 1.13 "Election Year" means, for an Employee, the calendar year for which a Participant makes an Election with respect to Compensation received during such calendar year pursuant to Article II of this Plan. "Election Year" means, for a Non-Employee Director, the one-year period that begins immediately following the first Annual Meeting of Shareholders which is subsequent to the Election period and ends on the next Annual Meeting of Shareholders.
- 1.14 "<u>Employee</u>" means any individual employed by the Company. Any Employee who, at the request and on the assignment of the Company specifically referencing this provision of the Plan, becomes an employee of another employer shall continue to be treated as an Employee for all purposes hereunder during the period of such assignment.
- 1.15 "<u>Fiscal Year</u>" means each year beginning on the first day of each fiscal year of Marriott International, Inc. and ending on the last day of each fiscal year of Marriott International, Inc. The fiscal year of Marriott International, Inc. is currently an annual period which varies from 52 to 53 weeks and ends on the Friday closest to December 31. A reference to a Fiscal Year preceding an Election Year means the Fiscal Year ending closest to the first day of the Election Year.
  - 1.16 "HR Officer" means the most senior human resources executive of the Company, as designated by the President of the Company.
- 1.17 "In-Service Withdrawal" means a distribution of Deferred Compensation and the earnings thereon, in accordance with a Participant's Election under Section 4.1, before a Participant incurs a Separation from Service from the Company.
- 1.18 "LTCI Compensation" means any compensation payable under a plan, agreement or award designated as a long term incentive or premium incentive plan, agreement or award.
- 1.19 "Non-Employee Director" means an individual who is not an Employee and (i) is a member of the Board of Directors of Marriott International, Inc., or (ii) has been elected to serve as such for a term which will begin at a subsequent point in time.
  - 1.20 "Participant" means an individual who meets the requirements of any of the following paragraphs (a) through (f):
- (a) Employees who are eligible to participate in the Retirement Savings Plan and have at least one Year of Service as of a date in the Election Year and Compensation, as defined below, greater than or equal to \$165,000 or such higher Compensation limitation as may be determined for such Election Year by the Administrator on advice of counsel; provided, however, that such Employee's Election shall be effective solely with respect to Compensation paid or payable on or after the date such Employee has completed one Year of Service.

For purposes of this Section 1.20(a), "Compensation" means:

With respect to Employees other than commissioned sales executive Employees of the Marriott Vacation Club International Division of the Company, the sum of the following: (i) the rate of base pay as of November 1 (or such other date as may be specified by the Administrator) immediately preceding the Election Year, annualized; (ii) the executive bonuses, commissions and management quarterly banquet awards received from January 1 through October 31 (or such other date as may be specified by the Administrator) of the year preceding the Election Year; and (iii) with respect to Employees who have review dates between October 31 (or such other date as may be specified by the Administrator) of the year preceding the Election Year and the last day of February of the Election Year, the annualized base pay as determined in (i), above, times 1.04.

With respect to commissioned sales executive Employees of the Marriott Vacation Club International division of the Company, the commissions received from January 1 through October 31 (or such other date as may be specified by the Administrator) of the year preceding the Election Year, annualized.

- (b) Select management or highly compensated employees of a business acquired by the Company who, prior to that acquisition, were covered by a nonqualified deferred compensation program of such acquired business;
  - (c) Employees with whom the Company has entered into a deferred compensation agreement under this Plan;
  - (d) Non-Employee Directors;
  - (e) Former Participants, terminated Participants, and their beneficiaries, as appropriate to the context; and
  - (f) Such other individuals as shall be designated by the HR Officer.

Except with respect to the Participants described in Section 1.20(d) through (f), in no event shall an individual be a Participant in this Plan unless the Administrator has invited such individual to participate in the Plan.

1.21 "<u>Permanent Disability</u>" means that the Participant, as a result of a disability, will be prevented on a permanent basis from engaging in any occupation for which he or she is reasonably qualified by education, training or experience as certified by a competent medical authority designated by the Named Fiduciary of the Retirement Savings Plan to make such determination. The foregoing shall include disability attributable to the permanent loss of or loss of use of a member or function of the body, or to the permanent disfigurement of the Participant. The determination of the existence of a Permanent Disability shall be made by the Administrator and shall be final and binding upon the Participant and all other parties.

- 1.22 "Plan" means the Marriott International, Inc. Executive Deferred Compensation Plan, as described herein and as may be amended from time to time.
- 1.23 "Reinstatement" or "Reinstated" means an Employee, upon being rehired by the Company, is credited with the same hire date as that Employee's hire date for his or her most recent period of continuous employment with the Company prior to being rehired.
- 1.24 "Retire" or "Retirement" means to have a Separation from Service, other than due to death or Permanent Disability, on or after (i) attainment of age fifty-five (55) and the completion of ten (10) Years of Service, or (ii) completion of 240 whole months of service with the Company, including Service, as defined in the Retirement Savings Plan, and service as a Non-Employee Director. A whole month of service is a monthly period that begins on the date of the month on which service began and ends on the date preceding the same date in the next month.
  - 1.25 "Retirement Savings Plan" means the Marriott International, Inc. Employees' Profit Sharing, Retirement and Savings Plan and Trust.
  - 1.26 "Separation from Service" means termination of service with the Company in any of the following circumstances:
    - (a) Where the Employee or Non-Employee Director voluntarily resigns;
    - (b) Where the Employee or Non-Employee Director voluntarily Retires;
    - (c) Where the Employee or Non-Employee Director is discharged;
    - (d) Where the Employee or Non-Employee Director terminates service with the Company on account of a Permanent Disability;
    - (e) Where the Employee or Non-Employee Director dies; or
    - (f) Where the Non-Employee Director is not re-elected to serve on the Board of Directors of the Company.
- 1.27 "<u>Subsidiary</u>" means either (a) a member of a controlled group of corporations of which the Company is a member as determined in accordance with the provisions of Code Section 414(b), or (b) an unincorporated trade or business which is under common control by or with the Company as determined in accordance with Section 414(c) of the Code.
- 1.28 "<u>Vested Portion</u>" of a Participant's Deferred Compensation Reserve account means (i) one hundred percent (100%) of the Deferred Compensation credited to the account, and earnings thereon, and (ii) the portion of the Company Accruals credited to the account, and earnings thereon, which have vested in accordance with the terms of Section 3.3 of the Plan.

1.29 "<u>Year of Service</u>" means, for Employees, a Year of Service as defined in the Retirement Savings Plan and, for Non-Employee Directors, a twelve (12) consecutive month period of service as a Non-Employee Director. If an Employee terminates employment with the Company after at least one Year of Service and subsequently resumes employment with the Company, the Employee's Years of Service, for eligibility purposes under this Plan, shall be determined in accordance with Article II of the Retirement Savings Plan.

## ARTICLE II

## PARTICIPANT ELECTIONS

# 2.1 <u>Deferred Compensation Reserve</u>.

The Company shall establish and maintain a book reserve (the "Deferred Compensation Reserve") to which it shall credit the amounts of Deferred Compensation determined in accordance with Section 2.3, Company Accruals under Section 3.2, as well as earnings allocated thereto under Section 3.5. The Deferred Compensation credited each Election Year shall be based on their Elections as provided in Sections 2.2. The Company shall maintain a separate Account under the Deferred Compensation Reserve with respect to each Participant.

#### 2.2 Elections.

- (a) Each Participant (other than a Participant under subsections 1.20(e)) shall have the option each calendar year to designate in an Election, in the form prescribed in Section 2.3, a percentage (the "Deferral Percentage"), specified in multiples of one percent (1%), of such Participant's Compensation for the pertinent Election Year, to be credited to the Deferred Compensation Reserve; provided, however, that the Administrator shall have the right to approve or disapprove such Election by any Participant, in whole or in part, in the sole discretion of the Administrator. The Administrator shall, in its discretion, establish a maximum Deferral Percentage for the Compensation with respect to which a Participant may make an Election for the Election Year (including LTCI Compensation, subject to the election requirements in (b) below). In accordance with procedures established by the Administrator, a Participant may make a separate election under this Section 2.2(a) with respect to regular pay and to bonus.
- (b) Elections described in Section 2.2(a) shall be made in accordance with procedures prescribed by the Administrator on or before (i) the last day of the calendar year immediately preceding the Election Year or (ii) such other earlier date as designated by the Administrator, provided such date precedes any service period during which the Participant earns the Compensation for which the election is made; provided, further, that an Election to have a portion or all of a Participant's LTCI Compensation for an Election Year credited to the Deferred Compensation Reserve shall be made on or before (i) the last business day of the calendar year preceding the calendar year which precedes the Election Year or (ii) such other date as may be designated by the Administrator that satisfies the election rules for performance-based compensation under Code section 409A(a)(4)(B)(iii). Notwithstanding the preceding sentence, effective January 1, 2005, with respect to Deferred Compensation subject to Code section 409A relating all or in part to services performed on or before December 31, 2005, an Election may be made any time on or before March 15, 2005; provided that on or before the date of such Election the subject Deferred Compensation has not been paid or become payable to the Participant. Late Elections shall be invalid.
- (c) Except as provided in Article IV, an Election shall be irrevocable with respect to all Compensation payable during an Election Year that is subject to the Election. A Participant's Election made as to an Election Year shall remain in effect for all subsequent Election Years unless the Participant notifies the Administrator, in accordance with procedures specified by the Administrator, of such Participant's desire to modify his or her Election.

(d) If an Employee is a Participant in accordance with Section 1.20(a) for an Election Year and incurs a Separation from Service, upon the subsequent Reinstatement of such Employee within the same Election Year, the Employee shall immediately be reinstated as a Participant and shall be subject to the same deferral Elections as were in effect immediately prior to such Employee's Separation from Service.

# 2.3 Form of Election.

- (a) Each Election shall be made on a form provided by the Administrator within the period described in Section 2.2(b), and shall designate a Deferral Percentage. Such Elections shall designate a distribution commencement date and manner of distribution in accordance with Article IV. If no designation is received by the Administrator within the prescribed time period, the Administrator shall select the time and manner of distribution within the period described in Section 2.2(b) and notify the Participant of such selection.
- (b) For purposes of this Section 2.3, Participants eligible to make Elections provided herein shall include only Participants described in Sections 1.20(a), (b), (c), (d) and (f), and shall exclude all other Participants.

## ARTICLE III

# PARTICIPANT ACCOUNTS

# 3.1 Individual Accounts.

The Administrator shall establish and maintain records reflecting each Participant's Account in the Deferred Compensation Reserve to which the Administrator shall credit Deferred Compensation in accordance with each Participant's Election pursuant to Section 2.3, Company Accruals pursuant to Section 3.2 and earnings pursuant to Section 3.5.

# 3.2 Company Accruals.

- (a) <u>Discretionary Company Accruals</u>. The Company may make discretionary Company Accruals for each Election Year to be allocated to the Deferred Compensation Reserve on behalf of Participants. In any Election Year for which the Company makes such discretionary Company Accrual, the Company Accrual shall be calculated as follows:
  - (i) for Participants whose Compensation is equal to or greater than a threshold dollar amount established for that Election Year by the Administrator in its sole discretion (which threshold shall apply to each future Election Year unless changed by the Administrator) but less than the compensation threshold established under the following subparagraph (ii): a percentage of the first three percent (3%) of Compensation deferred by the Participant under the Plan for the Election Year.
  - (ii) for Participants whose Compensation is equal to or greater than a threshold dollar amount established for that Election Year by the Administrator in its sole discretion (which threshold shall apply to each future Election Year unless changed by the Administrator): a percentage of the first six percent (6%) of Compensation deferred under the Plan for the Election Year.

Notwithstanding the preceding sentence, a Participant shall only be eligible for Company Accruals for Compensation earned during periods in which the Participant is eligible to participate in the Retirement Savings Plan.

- (b) <u>Additional discretionary Company Accruals</u> may be made by the Company from time to time. Such additional Company Accruals may be made in accordance with procedures established by the Company at the time such Company Accruals are allocated to a Participant's Account.
- (c) Company Accruals under this Section 3.2 shall be allocated only on behalf of Participants in the Plan who are actively employed (including Participants on approved leaves of absence) by the Company or serving as Non-Employee Directors as of the last day of the

Fiscal Year of the Company for which the allocation is made. Notwithstanding the preceding sentence, Participants who incur a Separation from Service before the last day of the Fiscal Year because they Retire, have a Permanent Disability, or die, or because they are employed by a business unit which is sold or otherwise disposed of on or after January 3, 1998, shall be eligible to have Company Accruals credited to the Deferred Compensation Reserve on their behalf in accordance with the provisions of Sections 3.2(a) and (b).

(d) Notwithstanding paragraphs (c) and (d) above, a Participant who incurs a Separation from Service during an Election Year and is within ninety (90) days Reinstated as an Employee or a Non-Employee Director prior to the end of such Election Year and remains employed as of the last day of the Fiscal Year shall be credited with Company Accruals in accordance with this Section 3.2 for such Election Year if such Participant otherwise satisfies the requirements of the first sentence of paragraph (c).

#### 3.3 Vesting.

- (a) <u>Deferred Compensation</u>. Participants shall be immediately vested in Deferred Compensation and the related earnings allocated to their account under the Deferred Compensation Reserve.
- (b) <u>Company Accruals</u>. For Company Accruals attributable to Deferred Compensation for periods prior to January 1, 2001, Participants shall be one hundred percent (100%) vested in Company Accruals allocated to their accounts under the Deferred Compensation Reserve. Participants shall become vested in Company Accruals allocated in accordance with Section 3.2(a) at the rate of twenty-five percent (25%) for each Year of Service of the Participant following the date on which such Company Accrual is allocated to the Participant's Account under the Deferred Compensation Reserve. For purposes of the preceding sentence, Company Accruals allocated in a given calendar year shall be deemed allocated on March 1 of such calendar year. Notwithstanding the foregoing of this Section 3.3(b), and subject to the approval of the HR Officer, a Participant shall become fully vested in Company Accruals and the related earnings allocated to the Participant's Separation from Service is due to Retirement, Death or Permanent Disability.
- (c) <u>Additional Discretionary Company Accruals</u>. Additional discretionary Company Accruals made under Section 3.2(c) shall vest in accordance with a schedule established by the Company at the time such Company Accruals are allocated to a Participant's Account.
- (d) <u>Forfeiture for Failure to Comply with Non-Competition Requirements</u>. All vesting on Company Accruals is subject to a Participant's compliance with the Company's Non-Competition Agreement. A Participant shall be deemed to comply with the Non-Competition Agreement if such Participant does not engage in activities in Competition with the business of the Company. "Competition" shall mean (i) engaging, individually or as an employee, consultant, owner (more than five percent (5%)) or agent of any entity, in or on behalf of any business engaged in significant competition (or that transacts or cooperates with another business in activities of significant competition) with any business operated by the Company or

with interests adverse to those of the Company; (ii) soliciting and hiring a key employee of the Company in another business, whether or not in significant competition with any business operated by the Company; or (iii) using or disclosing confidential or proprietary Company information, in each case, without the approval of the Company. Determination of whether or not particular activities are in competition will be made by the Company in its reasonable judgment.

# 3.4 Forfeitures.

The non-Vested Portion of a Participant's Account shall be forfeited upon the Participant's Separation from Service or if a Participant is found to have engaged in Competition with the Company. Forfeitures shall be applied to reduce the administrative expenses of the Plan.

# 3.5 Crediting of Earnings.

At the time a Participant makes an Election for the amount to be deferred for an Election Year in accordance with Section 3.2, such Participant may elect that a specified percentage of the Deferred Compensation be credited with hypothetical earnings in accordance with the performance of designated funds selected by the Company or its delegate ("Benchmark Funds"), as described in Appendix A. The Company shall credit such earnings to the Deferred Compensation Reserve on a daily basis. If a Participant does not make an allocation election, the Participant's account will be credited with the rate of return on the money market fund included in the Benchmark Funds. Once a Participant has allocated amounts in the Participant's Account to Benchmark Funds, the Participant may elect to change the allocation of all or a portion of his Account among the Benchmark Funds on a periodic basis in accordance with procedures established by the Administrator. Notwithstanding the foregoing of this Section 3.5, the hypothetical earnings credited to a Participant's or beneficiary's Account may be adjusted in accordance with Section 5.2.

# 3.6 Accounts Do Not Result in Property Rights.

- (a) The Deferred Compensation Reserve and the accounts maintained thereunder on behalf of each Participant are for administrative purposes only, and do not vest in the Participants any right, title or interest in such reserve or such accounts, except as expressly set forth in this Plan.
- (b) Title to and beneficial ownership of any assets, whether cash or investments which the Company may designate to make payments of Deferred Compensation hereunder, shall at all times remain in the Company, and no Participant shall have any property interest whatsoever in any specific assets of the Company.

# 3.7 No Assignment of Interests.

The rights of Participants or any other persons to the payment of amounts from the Deferred Compensation Reserve under this Plan shall not be assigned, transferred, pledged or encumbered except by will or by the laws of descent and distribution.

#### 3.8 Federal and State Taxes

Federal and state payroll taxes or state, local or foreign income taxes required to be withheld on Deferred Compensation credited to a Participant's Deferred Compensation Reserve shall be withheld from other Compensation paid to the Participant at the time of deferral. Notwithstanding the preceding sentence, if a Participant's other Compensation is insufficient to pay such amounts, the amount of Deferred Compensation credited to the Deferred Compensation Reserve on the Participant's account shall be reduced, at the time amounts are to be credited, to the extent necessary to cover all required withholding taxes.

## ARTICLE IV

## **DISTRIBUTIONS**

# 4.1 Election of Distribution.

(a) For each Election Year, a Participant shall designate in an Election made in accordance with Section 2.3 whether distribution of amounts credited to the Participant's Deferred Compensation Reserve for such Election Year as Deferred Compensation are to be distributed following Separation from Service or as an In-Service Withdrawal. A Participant may make a separate distribution election for each Election Year. Elections for distribution following Separation from Service will continue from Election Year to Election Year unless a new election is made by the Participant. A Participant must affirmatively elect an In-Service Withdrawal for an Election Year or the Participant shall be deemed to have elected a distribution following Separation from Service.

Notwithstanding the preceding paragraph, the Vested Portion of a Participant's Company Accruals and the earnings thereon shall become distributable only following such Participant's Separation from Service; such distribution shall be made to the Participant in the manner specified in paragraph (a) of Section 4.2.

(b) Each participant may elect, on a form provided by the Administrator, that distributions which are to be made to the Participant in installments following Separation from Service shall be deemed to come first from the money market fund included as a Benchmark Fund, in Appendix A, to the extent the Participant's hypothetical Account is invested in such money market fund. If a Participant does not make such an election, the distribution shall be deemed to come proportionally from each Benchmark Fund in which the Participant's Account is deemed to be invested. If a Participant makes such election and the amount allocated to the money market fund in the Participant's Account is less than the amount of the distribution, the remaining portion of the distribution shall be deemed to come proportionally from the remaining Benchmark Funds in which the Participant's Account is deemed to be invested. Any election under this subsection 4.1(b) shall be effective as soon as practicable after the election is received by the Administrator.

# 4.2 Form and Timing of Distribution.

(a) <u>Distribution Following Separation from Service</u>. Any amounts credited to the Participant's Account for which the Participant has elected distribution following Separation from Service may be distributed in any of the following forms, as elected by the Participant: (i) a lump sum cash payment within the ninety (90) days immediately following the Participant's Separation from Service; (ii) annual cash installments payable each January over a designated term not to exceed twenty (20) years, commencing in the January immediately following the year of the Participant's Separation from Service; or (iii) five (5) annual cash installments commencing in the sixth (6<sup>th</sup>) January following such Participant's Separation from Service and continuing in each of the four (4) immediately following Januaries. Notwithstanding the

preceding sentence, if the balance credited to a Participant's Deferred Compensation Reserve Account is less than \$10,000 upon a Participant's Separation from Service, such Participant's Deferred Compensation Reserve Account shall be paid to the Participant in a lump sum within the ninety (90) days immediately following the Participant's Separation from Service.

(b) <u>In-Service Withdrawal</u>. Subject to procedures established by the Administrator, at the time that a Participant makes an Election for an Election Year, the Participant may elect to receive an In-Service Withdrawal, occurring or beginning in a future calendar year specified by the Participant, of the Deferred Compensation attributable to that Election (specified as a dollar amount or as a percentage); provided, however, that the year in which such withdrawal begins shall be no earlier than the third (3<sup>rd</sup>) calendar year following the calendar year in which the Deferred Compensation is credited to the Participant's Account. The Participant may elect to have amounts subject to an In-Service Withdrawal election distributed in annual cash installments paid over a term of two (2) to five (5) years or as a single lump sum cash payment. Notwithstanding the preceding two sentences, if the balance credited to a Participant's Deferred Compensation Reserve is \$10,000 or less on the date an In-Service Withdrawal is scheduled to commence in installments, payment will be in the form of a single lump sum cash payment.

Notwithstanding the preceding paragraph, Company Accruals and related earnings are not available for scheduled In-Service Withdrawals.

- (c) <u>Installments</u>. For purposes of Section 4.2(a) and (b), the amount of any installment payment shall be computed as the Participant's current distributable interest divided by the remaining unpaid installments (including the installment being computed).
- (d) <u>Separation from Service Prior to Receipt of In-Service Withdrawal</u>. If a Participant incurs a Separation from Service prior to the date elected by the Participant for an In-Service Withdrawal, the portion of the Participant's Account subject to the In-Service Withdrawal election shall be distributed in a lump sum cash payment within the ninety (90) days immediately following the Participant's Separation from Service. If a Participant receiving scheduled in-service annual installment distributions incurs a Separation from Service prior to receiving the last of the installments, the Participant will receive the remaining installments in a lump sum within the ninety (90) days immediately following the Participant's Separation from Service.
- (e) <u>Failure to Elect Form of Distribution</u>. Notwithstanding paragraphs (a) through (d), amounts allocated to the Participant's Account for which no distribution election has been made shall be distributed in the form of a single lump sum cash payment made within the ninety (90) days immediately following the Participant's Separation from Service.
- (f) <u>Distribution Following Death of Participant</u>. If the Participant dies before distribution of his or her account has begun or after distribution has begun but before the Vested Portion of the Participant's Account is fully distributed, the undistributed Vested Portion of the account shall be distributed to the Participant's beneficiary in a single lump sum cash payment within the ninety (90) days immediately following the Participant's death. If a Participant fails to

designate a beneficiary in accordance with Section 4.5, or if the beneficiary designated by the Participant does not survive the Participant, the default beneficiary shall be determined in accordance with Section 4.5 and the distribution to such default beneficiary shall be in the form of a single lump sum as provided above, notwithstanding any designation by the Participant.

- (g) Non-Vested Amounts. Upon a Participant's Separation from Service, the Company shall have no further obligation to the Plan or to the Participant for the part of the Participant's account that is not the Vested Portion.
- (h) <u>Permissible Delays in Distribution</u>. Notwithstanding elections made under Section 4.1, distributions may be delayed in accordance with the following provisions of this Section 4.2(h), provided that any such distribution shall be made solely in the discretion of the Administrator without regard to the request, intent or wishes of any Participant or beneficiary:
  - (i) <u>Delay for Specified Employees</u>. Distributions on account of a Separation from Service of a Participant who is a Specified Employee (as defined as follows) shall be made or commence not before the date which is six (6) months following the Separation from Service, except in the event of the Participant's death. For this purpose, a Specified Employee is a person described under Treasury Regulation section 1.409A-1(i), applying the default rules thereunder.
  - (ii) <u>Section 162(m) Delays</u>. Subject to the requirements of Treasury Regulations section 1.409A-2(b)(7)(i), the Administrator, in its sole discretion, may delay distributions to a Participant to the extent necessary to avoid application of the deduction limitation under Code section 162(m).
  - (iii) <u>Violations of Law</u>. Subject to the requirements of Treasury Regulation section 1.409A-2(b)(7)(ii), the Administrator may delay distributions to a Participant or beneficiary to the extent that it reasonably anticipates that the distribution, if paid, will violate Federal securities laws or other applicable law.
- (i) <u>Permissible Accelerated Distributions</u>. Distributions shall not be accelerated except solely at the discretion of the Administrator, subject to the following provisions of this Section 4.2(i):
  - (i) <u>Employment Taxes</u>. To the extent that federal payroll taxes are required to be withheld on a Participant's Account as it vests, including, but not limited to, taxes due under Code section 3101, the Company will determine these amounts and collect them as follows:
    - (I) If the Participant is an Employee at the time the tax is determined, the tax will be deducted from the Employee's non-Deferred Compensation.

- (II) If the Participant is a current or former Non-Employee Director at the time Deferred Compensation or Company Accruals are credited to the Participant's account under this Plan, no tax shall be paid with respect to such amounts and no reduction to reflect such taxes shall be made in the amounts credited.
- (III) Otherwise, the Participant's Account shall be reduced to the extent permitted under Treasury Regulation section 1.409A-3(j)(4)(vi) as necessary to satisfy the amount of any applicable taxes payable.
- (ii) <u>Other Taxes</u>. State, local and foreign tax withholding may be satisfied a similar manner as under Section 4.2(i)(i) above, to the extent permitted under Treasury Regulation section 1.409A-3(j)(4)(xi).
- (iii) <u>Section 409A Inclusion</u>. If a determination is made by the Internal Revenue Service that the Plan or any part thereof fails to meet the requirements of Code section 409A, the Account balance of any Participant shall be immediately distributed to the Participant to the extent of the amount required to be included in the Participant's income as a result of the failure to comply with section 409A.
- (iv) <u>Violations of Ethics Laws</u>. A Participant's Account balance or portion thereof may be distributed at the sole discretion of the Administrator to the extent reasonably necessary to avoid the violation of an applicable Federal, state, local or foreign ethics law or conflict of interest law.
- 4.3 <u>Tax Impact</u>. The gross amount of any payment due in accordance with this subsection shall be reduced to reflect applicable federal and state income tax withholding prior to payment to the Participant or beneficiary.

# 4.4 Changes in Distribution Election.

- (a) Notwithstanding anything in Section 4.1 to the contrary, a Participant who is employed by the Company or serving on the Company's Board of Directors shall be entitled to change the manner of distribution of his or her account under Section 4.2(a) or (b), provided that such change shall be made (i) using a form provided by the Administrator, (ii) in accordance with procedures established by the Administrator, and (iii) subject to the following limitations: (A) any such change in distribution election must be made and be irrevocable at least twelve (12) months before the date the distribution originally was scheduled to occur or commence, (B) the only distribution elections which may be changed are lump sum distributions payable upon a Separation from Service and In-Service Withdrawals, and (C) all such changes in distribution elections must provide for distributions to be paid in five (5) installments commencing with the sixth (6<sup>th</sup>) January following Separation from Service, except upon death or Permanent Disability.
- (b) A separate change may be made with respect to each Election Year beginning on or after January 1, 2001, and with respect to the Participant's Account attributable

to Deferred Compensation and Company Accruals as of December 31, 2000. A request for change shall become effective on the first anniversary (the "Anniversary Date") of the date such request was received by the Administrator, provided such request shall be invalid if the Participant has a Separation from Service as described in Section 1.26 (but not including Section 1.26(d) or (e)) prior to the Anniversary Date, or, as to Deferred Compensation relating to any Election Year, if any amount of such Deferred Compensation for an Election Year would otherwise become distributable prior to the Anniversary Date.

# 4.5 Beneficiaries.

Each Participant may designate a beneficiary on a form, provided by the Administrator, to receive distributions made pursuant to Section 4.2. If no beneficiary is designated under this Plan, or if the beneficiary shall not survive the Participant, the Participant shall be deemed to have designated (i) the Participant's surviving spouse; or (ii) if the Participant is not married or the spouse died before the Participant, the Participant's estate.

# 4.6 Discharge of Obligation For Payment.

If a legal guardian or conservator is appointed for any person to whom any payment is payable under this Plan, then, upon proof to the Administrator of such appointment, amounts which would otherwise be paid under this Plan to such person shall be paid to the legal guardian or conservator. Any such payment shall be complete discharge of the liabilities of the Company under this Plan.

## ARTICLE V

# **ADMINISTRATION**

# 5.1 Administrator.

The Company shall appoint an Administrator who shall be responsible for the management, operation and administration of the Plan. Except as provided in Section 6.2, the Administrator shall have full power and authority to interpret, construe and administer this Plan and the Administrator's interpretations and construction thereof, and actions hereunder, including any valuation of the Deferred Compensation Reserve, or the amount or recipient of the payment to be made therefrom, shall be binding and conclusive on all persons for all purposes. The HR Officer's interpretations and construction thereof, and actions under those Sections shall be binding and conclusive on all persons. The Company shall not be liable to any person for any action taken or omitted in connection with the interpretation and administration of this Plan unless attributable to willful misconduct or lack of good faith by the Company.

#### 5.2 Expenses.

The Administrator may offset the Company's costs of administering the Plan by allocating a charge against the Deferred Compensation Reserve of each Participant in a manner to be determined by the Administrator.

## ARTICLE VI

## CLAIMS PROCEDURE

# 6.1 Initial Claims.

A Participant or a beneficiary of a Participant may submit a written claim for benefits under this Plan with the Administrator. The Administrator shall notify the claimant within a reasonable period of time but no later than ninety (90) days after the written claim is received by the Administrator whether the claim is wholly or partially denied, unless the claimant receives a written notice from the Administrator prior to the end of the ninety (90) day period stating that special circumstances require an extension of the time for the decision. Such extension shall not exceed a period of ninety (90) days from the end of the initial ninety (90) day period and such extension shall be in writing indicating the special circumstances requiring an extension of time and the date by which the Administrator expects to render the decision. The notice of the decision by the Administrator shall be in writing and in a manner calculated to be understood by the claimant, and, if a denial of the claim, must contain the following information: (i) the specific reason or reasons for the denial; (ii) the specific reference to pertinent provisions of the Plan on which the denial is based; (iii) if applicable, a description of any additional information or material necessary for the claimant to perfect the claim; and (iv) an explanation of the Plan's claim review procedures and the time limits applicable to such procedures, including a statement of the claimant's right to bring a civil action under section 502(a) of ERISA.

## 6.2 Appeals.

A claimant is entitled to request a final review by the HR Officer of any denial of the claim by the Administrator. The request for review must be submitted to the HR Officer in writing within sixty (60) days of the Participant's receipt of the Administrator's notice of denial. Absent a request for review within the sixty (60) day period, the claim will be deemed to be conclusively denied. The HR Officer shall provide the claimant, upon request and free of charge, reasonable access to, and copies of, all pertinent documents and shall afford the claimant the opportunity to submit issues, comments and other information relating to the claim in writing and the HR Officer shall render a decision in writing no later than sixty (60) days after receipt of a request for a review, provided that the HR Officer determines that special circumstances require an extension of the time for the decision. Such extension shall not exceed a period of sixty (60) days from the end of the initial sixty (60) day period and such extension will be in writing indicating the special circumstances requiring an extension of time and the date by which the HR Officer expects to render the decision on review. The claimant shall receive written notice of the HR Officer's decision and such written notice shall be written in a manner calculated to be understood by the claimant, indicating the specific reason or reasons for the decision; reference to the pertinent provisions of the Plan on which the decision is based; a statement that the claimant is entitled to receive, upon request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to the claimant's claim for benefits; and a statement of the claimant's right to bring an action under section 502(a) of ERISA.

## ARTICLE VII

## **MISCELLANEOUS**

# 7.1 Plan Not An Employment Contract.

Nothing contained herein shall be construed as conferring upon any Participant the right to continue in the employ of the Company as an Employee or in any other capacity.

## 7.2 No Trust Created.

Nothing contained in this Plan and no action taken pursuant to the provisions of this Plan shall create or be construed to create a trust of any kind, or a fiduciary relationship between the Company and any person, including any Participant or any other person. Any amounts which may be credited to the Deferred Compensation Reserve shall continue for all purposes to be a part of the general funds of the Company and no person other than the Company shall by virtue of the provisions of this Plan have any interest in such funds. To the extent that any person acquires a right to receive payments from the Company under this Plan, such right shall be no greater than the right of any unsecured general creditor of the Company.

# 7.3 Amendment or Termination of Plan.

- (a) The Board of Directors of the Company may amend the Plan at any time and from time to time, or terminate and liquidate the Plan pursuant to written resolutions adopted by such Board of Directors, provided that any termination of the Plan shall comply with the requirements of Treasury Regulation section 1.409A-3(j)(ix).
- (b) In no event will any such amendment or termination of the Plan have the effect of reducing the accrued account balance or the Vested Portion of any Participant's account under this Plan. The Board may delegate its authority to amend the Plan to the HR Officer or other Company representatives pursuant to written resolutions adopted by such Board of Directors.

## 7.4 Effect of Plan.

This Plan shall be binding upon and inure to the benefit of the Company, its successors and assigns, and the Participants and their heirs, beneficiaries, executors, administrators and legal representatives.

## 7.5 Severability.

If any provision of this Plan shall for any reason be invalid or unenforceable, the remaining provisions shall nevertheless remain in full force and effect.

# 7.6 <u>Applicable Law</u>.

This Plan shall be construed in accordance with and governed by the laws of the State of Maryland.

# APPENDIX A

# BENCHMARK FUNDS

As of January 1, 2008, the following Benchmark Funds are available for selection by participants:

Money Market Fund – Vanguard Money Market

Bond Fund – PIMCO Total Return

Balanced Fund – Vanguard Balanced Portfolio

S&P 500 Index – Fidelity VIP Index 500

Large Core Fund – Fidelity VIP Contrafund

Large Cap Value – Vanguard Diversified Value Portfolio

Large Growth Fund – GE Premier Growth Equity

Mid Core Fund – Vanguard Mid-Cap Index

Small Growth Fund – Royce Small Cap

Foreign Fund – Franklin Templeton International

The Company has the right to change the benchmark funds from time to time.

# EMPLOYEE NON-QUALIFIED STOCK OPTION AGREEMENT MARRIOTT INTERNATIONAL, INC. STOCK AND CASH INCENTIVE PLAN

THIS AGREEMENT is made on **<GRANT DATE>** (the "Award Date") by MARRIOTT INTERNATIONAL, INC. (the "Company") and **<NAME>** ("Employee").

WITNESSETH:

WHEREAS, the Company maintains the Marriott International, Inc. Stock and Cash Incentive Plan, as amended (the "Plan"); and

WHEREAS, the Company wishes to award to designated employees certain option awards as provided in Article 6 of the Plan ("Options" or "Awards"); and

WHEREAS, Employee has been approved by the Compensation Policy Committee (the "Committee") of the Company's Board of Directors (the "Board") to receive an award of Options under the Plan;

NOW, THEREFORE, it is agreed as follows:

- 1. **Prospectus**. Employee has been provided with, and hereby acknowledges receipt of, a Prospectus for the Plan dated <**DATE**>>, which contains, among other things, a detailed description of the Option award provisions of the Plan.
- 2. **Interpretation**. The provisions of the Plan are incorporated by reference and form an integral part of this Agreement. Except as otherwise set forth herein, capitalized terms used herein shall have the meanings given to them in the Plan. In the event of any inconsistency between this Agreement and the Plan, the terms of the Plan shall govern. A copy of the Plan is available from the Compensation Department of the Company upon request. All decisions and interpretations made by the Committee or its delegate with regard to any question arising hereunder or under the Plan shall be binding and conclusive. (The options granted pursuant to this Agreement are not intended to qualify as "incentive stock options" within the meaning of Section 422 of the Internal Revenue Code.)
- 3. **Grant of Options.** The Company hereby grants to Employee as of the Grant Date Options to purchase **«Grant»** shares of the Company's Common Stock (the "Option Shares"), subject to the terms and conditions of the Plan, Employee's acceptance of this Agreement and satisfaction of the tax provisions of the Company's International Assignment Policy ("IAP"), if applicable.
  - 4. Purchase Price. Subject to Paragraph 12 hereof, the purchase price per share of the Option Shares is << Option Price>> (the "Option Price").
- 5. **Waiting Period and Exercise Dates.** The Option Shares may not be purchased during the one-year period following the Grant Date (the "waiting period"). Following the waiting period, the Option Shares may be purchased in accordance with the following schedule: 25% of the Option Shares commencing on each of the first, second, third and fourth one-year anniversaries of the Grant Date. To the extent that the Options to purchase Option Shares are not exercised by Employee when they become initially exercisable, the Options shall not expire but shall be carried forward and shall be exercisable at any time thereafter; provided, however, that the Options shall not be exercisable after the expiration of ten (10) years from the Grant Date or sooner as set forth in paragraph 9, if applicable. Exercise of the Options shall not be dependent upon the prior or sequential exercise of any other options heretofore granted to Employee by the Company. Except as provided in Article 6 of the Plan and Paragraph 9 below, the Options may not be exercised at any time unless Employee shall then be an employee of the Company.
- 6. **Method of Exercising Options.** To exercise the Options, the person entitled to exercise the Options must provide a signed written notice or the equivalent to the Company or its designee, as prescribed in the administrative procedures of the Plan, stating the number of Option Shares with respect to which the Options are being exercised. The Options may be exercised by (a) payment of the Option Price for the Option Shares being purchased in accordance with procedures established by the Committee, (b) making provision for the satisfaction of the applicable withholding taxes, and (c) an undertaking to furnish and execute such documents as the Company deems necessary (i) to evidence such exercise, and (ii) to determine whether registration is then required to comply with the Securities Act of 1933 or any other law. Upon payment of the Option Price and provision for the satisfaction of the withholding taxes, the Company shall provide confirmation from the Plan record keeper that the transfer agent for the common stock of the Company is holding shares for the account of such person in a certificateless account. Pursuant to procedures, if any, that may be adopted by the Committee or its delegate, payment of the Option Price may be made by delivery of shares of the Company's common stock held by Employee for at least six months prior to the delivery, or by any other means that the Committee determines to be consistent with the Plan's purpose and applicable law.

- 7. **Rights as a Shareholder.** Employee shall have no rights as a shareholder with respect to any Option Shares covered by the Options granted hereby until the date of acquisition by Employee of such Option Shares. No adjustment shall be made for dividends or other rights for which the record date is prior to such date.
- 8. **Non-Assignability.** The Options shall not be assignable or transferable by Employee except by will or by the laws of descent and distribution. During Employee's lifetime, the Options may be exercised only by Employee or, in the event of incompetence, by Employee's legally appointed guardian.
- 9. Effect of Termination of Employment or Death. If Employee goes on leave of absence for a period of greater than twelve months (except a leave of absence approved by the Board of Directors or the Committee) or ceases to be an employee of the Company for any reason except death, the portion of the Options which is unexercisable on the date on which Employee ceased to be an Employee or has been on a leave of absence for over twelve months (except a leave of absence approved by the Board or Committee) shall expire on such date and any unexercised portion of the Options which was otherwise exercisable on such date shall expire at the earlier of (i) the expiration of the Options in accordance with the term for which the Options were granted, or (ii) three months from such date, except in the case of an Employee who is an "Approved Retiree" as defined below. If Employee is an Approved Retiree, then the Options shall expire at the sooner to occur of, (i) the expiration of such Options in accordance with their original term, (ii) the expiration of five years from the date of retirement, or (iii) with respect to Options granted less than one year before the date the Approved Retiree retires, such retirement date, except not with respect to that portion of the Options equal to such number of shares multiplied by the ratio of (a) the number of days between the Grant Date and the retirement date inclusive, over (b) the number of days in the twelve (12) month period following the Grant Date. In the event of the death of Employee without Approved Retiree status during the three (3) month period following termination of employment or a leave of absence over twelve (12) months (except a leave of absence approved by the Board or Committee), the Options shall be exercisable by Employee's personal representative, heirs or legatees to the same extent and during the same period that Employee could have exercised the Options if Employee had not died. In the event of the death of Employee while an employee of the Company or while an Approved Retiree, the Options (if the waiting period has elapsed) shall be exercisable in their entirety by Employee's personal representatives, heirs or legatees at any time prior to the expiration of one year from the date of the death of Employee, but in no event after the term for which the Options were granted. For purposes of this Agreement, an "Approved Retiree" is any optionee who (i) terminates employment by reason of a Disability, or (ii) (A) retires from employment with the Company with the specific approval of the Committee on or after such date on which the optionee has attained age 55 and completed 10 Years of Service, and (B) has entered into and has not breached an agreement to refrain from Engaging in Competition in form and substance satisfactory to the Committee; and if the Committee subsequently determines, in its sole discretion, that an Approved Retiree has violated the provisions of the Agreement to refrain from Engaging in Competition, or has engaged in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation, such Approved Retiree shall have ninety (90) days from the date of such finding within which to exercise any Options or portions thereof which are exercisable on such date, and any Options or portions thereof which are not exercised within such ninety (90) day period shall expire and any Options or portion thereof which are not exercisable on such date shall be cancelled on such date.
- 10. **Consent.** By executing this Agreement, Employee consents to the collection, maintenance and processing of Employee's personal information (such as Employee's name, home address, home telephone number and email address, social security number, assets and income information, birth date, hire date, termination date, other employment information, citizenship, marital status) by the Company and the Company's service providers for the purposes of (i) administering the Plan (including ensuring that the conditions of transfer are satisfied from the Award Date through the Exercise Date), (ii) providing Employee with services in connection with Employee's participation in the Plan, (iii) meeting legal and regulatory requirements and (iv) for any other purpose to which Employee may consent ("Permitted Purposes"). Employee's personal information will not be processed for longer than is necessary for such Permitted Purposes. Employee's personal information is collected from the following sources:
  - (a) from this Agreement, investor questionnaires or other forms that Employee submits to the Company or contracts that Employee enters into with the Company;
  - (b) from Employee's transactions with the Company, the Company's affiliates and service providers;
  - (c) from Employee's employment records with the Company; and
  - $\begin{tabular}{ll} (d) & from meetings, telephone conversations and other communications with Employee. \end{tabular}$

In addition, Employee further consents to the Company disclosing Employee's personal information to the Company's third party service providers and affiliates and other entities in connection with the services the Company provides related to Employee's participation in the Plan, including:

- (a) financial service providers, such as broker-dealers, custodians, banks and others used to finance or facilitate transactions by, or operations of, the Plan:
- (b) other service providers to the Plan, such as accounting, legal, or tax preparation services;
- (c) regulatory authorities; and
- (d) transfer agents, portfolio companies, brokerage firms and the like, in connection with distributions to Plan participants.

Where Employee's personal information is provided to such third parties, the Company requires (to the extent permitted by applicable law) that such parties agree to process Employee's personal information in accordance with the Company's instructions.

Employee's personal information is maintained on the Company's networks and the networks of the Company's service providers, which may be in the United States or other countries other than the country in which this Award was granted. Employee acknowledges and agrees that the transfer of Employee's personal information to the United States or other countries other than the country in which this Award was granted is necessary for the Permitted Purposes. To the extent (if any) that the provisions of the European Union's Data Protection Directive (Directive 95/46/EC of the European Parliament and of the Council) and/or applicable national legislation derived from such Directive apply, then by executing this Agreement Employee expressly consents to the transfer of Employee's personal information outside of the European Economic Area. Employee may access Employee's personal information to verify its accuracy, update Employee's personal information and/or request a copy of Employee's personal information by contacting Employee's local Human Resources representative. Employee may obtain account transaction information online or by contacting the Plan record keeper as described in the Plan enrollment materials. By accepting the terms of this Agreement, Employee further agrees to the same terms with respect to other Awards Employee received in any prior year under the Plan.

- 11. **No Additional Rights.** Benefits under this Plan are not guaranteed. The grant of Awards is a one-time benefit and does not create any contractual or other right or claim to any future grants of Awards under the Plan, nor does a grant of Awards guarantee future participation in the Plan. The value of Employee's Awards is an extraordinary item outside the scope of Employee's employment contract, if any. Employee's Awards are not part of normal or expected compensation for purposes of calculating any severance, resignation, redundancy, end-of-service payments, bonuses, long-term service awards, pension or retirement benefits (except as otherwise provided by the terms of any U.S.-qualified retirement or pension plan maintained by the Company or any of its subsidiaries), or similar payments. By accepting the terms of this Agreement, Employee further agrees to these same terms and conditions with respect to any other Awards Employee received in any prior year under the Plan.
- 12. **Recapitalization or Reorganization.** Certain events affecting the Common Stock of the Company and mergers, consolidations and reorganizations affecting the Company may affect the number or type of securities deliverable upon exercise of the Options or limit the remaining term over which these Options may be exercised.
- 13. **General Restriction.** In accordance with the terms of the Plan, the Company may limit or suspend the exercisability of the Options or the purchase or issuance of Option Shares thereunder under certain circumstances. Any delay caused thereby shall in no way affect the date of termination of the Options.
- 14. **Amendment of This Agreement.** The Board of Directors may at any time amend, suspend or terminate the Plan; provided, however, that no amendment, suspension or termination of the Plan or the Options shall adversely affect in any material way the Options without the written consent of Employee.
- 15. **Notices.** Notices hereunder shall be in writing, and if to the Company, may be delivered personally to the Compensation Department or such other party as designated by the Company or mailed to its principal office at 10400 Fernwood Road, Bethesda, Maryland 20817, addressed to the attention of the Stock Option Administrator (Department 935.40), and if to Employee, may be delivered personally or mailed to Employee at his or her address on the records of the Company.
- 16. **Successors and Assigns.** This Agreement shall bind and inure to the benefit of the parties hereto and the successors and assigns of the Company and, to the extent provided in Paragraph 9 above and the provisions of the Plan, to the personal representatives, legatees and heirs of Employee.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement to be effective as of the Grant Date.

MARRIOTT INTERNATIONAL, INC.

EMPLOYEE

Employee Name (Please Print)

By:
Executive Vice President, Global Human Resources

Employee Social Security Number (Please Print)

17. **No Effect on Employment.** Nothing contained in this Agreement shall be construed to limit or restrict the right of the Company to terminate Employee's employment at any time, with or without cause, or to increase or decrease Employee's compensation from the rate of compensation in existence at the

Employee Signature

# EXECUTIVE RESTRICTED STOCK UNIT AGREEMENT MARRIOTT INTERNATIONAL, INC. STOCK AND CASH INCENTIVE PLAN

THIS AGREEMENT is made on <<**GRANT DATE**>> (the "Award Date") by MARRIOTT INTERNATIONAL, INC. (the "Company") and <<**NAME**>> ("Employee").

WITNESSETH:

WHEREAS, the Company maintains the Marriott International, Inc. Stock and Cash Incentive Plan (the "Plan"); and

WHEREAS, the Company wishes to award to designated employees certain share-based awards as provided in Article 10 of the Plan; and

WHEREAS, Employee has been approved by the Compensation Policy Committee (the "Committee") of the Company's Board of Directors (the "Board") to receive an award of "Executive Restricted Stock Units" ("RSUs") under the Plan;

NOW, THEREFORE, it is agreed as follows:

- 1. **Prospectus.** Employee has been provided with, and hereby acknowledges receipt of, a Prospectus for the Plan dated >>**DATE** >>. which contains, among other things, a detailed description of the RSU award provisions of the Plan.
- 2. **Interpretation.** The provisions of the Plan are incorporated by reference and form an integral part of this Agreement. Except as otherwise set forth herein, capitalized terms used herein shall have the meanings given to them in the Plan. In the event of any inconsistency between this Agreement and the Plan, the terms of the Plan shall govern. A copy of the Plan is available from the Compensation Department of the Company upon request. All decisions and interpretations made by the Committee or its delegate with regard to any question arising hereunder or under the Plan shall be binding and conclusive.
- 3. **Grant of RSUs.** Subject to Employee's acceptance of this Agreement, and subject to satisfaction of the tax provisions of the Company's International Assignment Policy ("IAP"), if applicable, this award (the "Award") of <<# RSUs GRANTED>> RSUs is made as of the Grant Date.
- 4. **RSUs and Common Share Rights.** The RSUs awarded under this Agreement shall be recorded in a Company book-keeping account and shall represent Employee's unsecured right to receive from the Company the transfer of title to shares of Common Stock of the Company ("Common Shares") in accordance with the schedule of Vesting Dates set forth in paragraph 5 below, provided that Employee has satisfied the Conditions of Transfer set forth in paragraph 6 below and subject to the satisfaction of the provision on withholding taxes set forth in paragraph 9 below. On each such Vesting Date, if it occurs, or such later date(s) pursuant to procedures established by the Committee under Article 10 of the Plan, the Company shall reverse the book-keeping entry for all such related RSUs and transfer a corresponding number of Common Shares (which may be reduced by the number of shares withheld to satisfy withholding taxes as set forth in paragraph 9 below, if share reduction is the method utilized for satisfying the tax withholding obligation) to an individual brokerage account (the "Account") established and maintained in Employee's name. Employee shall have all the rights of a stockholder with respect to such Common Shares transferred to the Account, including but not limited to the right to vote the Common Shares, to sell, transfer, liquidate or otherwise dispose of the Common Shares, and to receive all dividends or other distributions paid or made with respect to the Common Shares from the time they are deposited in the Account. Employee shall have no voting, transfer, liquidation, dividend or other rights of a Common Share stockholder with respect to RSU shares prior to such time that the corresponding Common Shares are transferred, if at all, to Employee's Account.
  - 5. **Vesting in RSUs.** This Award shall vest in accordance with the following schedule:

Vesting Date	Vested Percentage of Award Shares
< <dates>&gt;</dates>	< <percentages>&gt;</percentages>

Notwithstanding the foregoing schedule, in the event that any such Vesting Date is a day on which stock of the Company is not traded on the New York Stock Exchange or another national exchange, then the Vesting Date shall be the next following day on which the stock of the Company is traded on the New York Stock Exchange or another national exchange.

- 6. **Conditions of Transfer.** With respect to any RSUs awarded to Employee, as a condition of Employee receiving a transfer of corresponding Common Shares in accordance with paragraph 4 above, Employee shall meet all of the following conditions during the entire period from the Award Date hereof through the Vesting Date relating to such RSUs:
  - (a) Employee must continue to be an active employee of the Company ("Continuous Employment");
  - (b) Employee must refrain from Engaging in Competition (as defined in Section 2.25 of the Plan) without first having obtained the written consent thereto from the Company ("Non-competition"); and
  - (c) Employee must refrain from committing any criminal offense or malicious tort relating to or against the Company or, as determined by the Committee in its discretion, engaging in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation ("No Improper Conduct"). The Company's determination as to whether or not particular conduct constitutes Improper Conduct shall be conclusive.

If Employee should fail to meet the requirements relating to (i) Continuous Employment, (ii) Non-competition, or (iii) No Improper Conduct, then Employee shall forfeit the right to vest in any RSUs that have not already vested as of the time such failure is determined, and Employee shall accordingly forfeit the right to receive the transfer of title to any corresponding Common Shares. The forfeiture of rights with respect to unvested RSUs (and corresponding Common Shares) shall not affect the rights of Employee with respect to any RSUs that already have vested nor with respect to any Common Shares the title of which has already been transferred to Employee's Account.

- 7. **Non –Assignability.** The RSUs shall not be assignable or transferable by Employee except by will or by the laws of descent and distribution. During Employee's lifetime, the RSUs may be exercised only by Employee or, in the event of incompetence, by Employee's legally appointed guardian.
  - 8. **Effect of Termination of Employment.** Notwithstanding the foregoing:
  - (a) In the event Employee's Continuous Employment is terminated prior to the relevant Vesting Date on account of death, and if Employee had otherwise met the requirements of Continuous Employment, Non-competition and No Improper Conduct from the Award Date through the date of such death, then Employee's unvested RSUs shall immediately vest in full upon death and Employee's rights hereunder with respect to any such RSUs shall inure to the benefit of Employee's executors, administrators, personal representatives and assigns.
  - (b) In the event Employee's Continuous Employment is terminated prior to the relevant Vesting Date on account of Employee's Disability (as defined in Section 2.19 of the Plan) and if Employee had otherwise met the requirements of Continuous Employment, Non-competition and No Improper Conduct from the Award Date through the date of such Disability, and provided that Employee continues to meet the requirements of Non-competition and No Improper Conduct, then Employee's rights hereunder with respect to any outstanding, unvested RSUs shall continue in the same manner as if Employee continued to meet the Continuous Employment requirement through the Vesting Dates related to the Award.

Except as set forth in this paragraph 8 above, no other transfer of rights with respect to RSUs shall be permitted pursuant to this Agreement.

- 9. **Taxes.** The transfer of Common Shares, pursuant to paragraphs 4 and 7 above, shall be subject to the further condition that the Company shall provide for the withholding of any taxes required by federal, state, or local law in respect of that Vesting Date by reducing the number of RSUs to be transferred to Employee's Account or by such other manner as the Committee shall determine in its discretion.
- 10. **Consent.** By executing this Agreement, Employee consents to the collection, maintenance and processing of Employee's personal information (such as Employee's name, home address, home telephone number and email address, social security number, assets and income information, birth date, hire date, termination date, other employment information, citizenship, marital status) by the Company and the Company's service providers for the purposes of (i) administering the Plan (including ensuring that the conditions of transfer are satisfied from the Award Date through the Vesting Date), (ii) providing Employee with services

in connection with Employee's participation in the Plan, (iii) meeting legal and regulatory requirements and (iv) for any other purpose to which Employee may consent ("Permitted Purposes"). Employee's personal information will not be processed for longer than is necessary for such Permitted Purposes. Employee's personal information is collected from the following sources:

- (a) from this Agreement, investor questionnaires or other forms that Employee submits to the Company or contracts that Employee enters into with the Company;
- (b) from Employee's transactions with the Company, the Company's affiliates and service providers;
- (c) from Employee's employment records with the Company; and
- (d) from meetings, telephone conversations and other communications with Employee.

In addition, Employee further consents to the Company disclosing Employee's personal information to the Company's third party service providers and affiliates and other entities in connection with the services the Company provides related to Employee's participation in the Plan, including:

- (a) financial service providers, such as broker-dealers, custodians, banks and others used to finance or facilitate transactions by, or operations of, the Plan:
- (b) other service providers to the Plan, such as accounting, legal, or tax preparation services;
- (c) regulatory authorities; and
- (d) transfer agents, portfolio companies, brokerage firms and the like, in connection with distributions to Plan participants.

Where Employee's personal information is provided to such third parties the Company requires (to the extent permitted by applicable law) that such parties, agree to process Employee's personal information in accordance with the Company's instructions.

Employee's personal information is maintained on the Company's networks and the networks of the Company's service providers, which may be in the United States or other countries other than the country in which this Award was granted. Employee acknowledges and agrees that the transfer of Employee's personal information to the United States or other countries other than the country in which this Award was granted is necessary for the Permitted Purposes. To the extent (if any) that the provisions of the European Union's Data Protection Directive (Directive 95/46/EC of the European Parliament and of the Council) and/or applicable national legislation derived from such Directive apply, then by executing this Agreement Employee expressly consents to the transfer of Employee's personal information outside of the European Economic Area. Employee may access Employee's personal information to verify its accuracy, update Employee's personal information and/or request copy of Employee's personal information by contacting Employee's local Human Resources representative. Employee may obtain account transaction information online or by contacting the Plan record keeper as described in the Plan enrollment materials. By accepting the terms of this Agreement, Employee further agrees to the same terms with respect to other Awards Employee received in any prior year under the Plan.

- 11. **No Additional Rights.** Benefits under this Plan are not guaranteed. The grant of Awards is a one-time benefit and does not create any contractual or other right or claim to any future grants of Awards under the Plan, nor does a grant of Awards guarantee future participation in the Plan. The value of Employee's Awards is an extraordinary item outside the scope of Employee's employment contract, if any. Employee's Awards are not part of normal or expected compensation for purposes of calculating any severance, resignation, redundancy, end-of-service payments, bonuses, long-term service awards, pension or retirement benefits (except as otherwise provided by the terms of any U.S.-qualified retirement or pension plan maintained by the Company), or similar payments. By accepting the terms of this Agreement, Employee further agrees to these same terms and conditions with respect to any other Awards Employee received in any prior year under the Plan.
- 12. **Amendment of This Agreement.** The Board of Directors may at any time amend, suspend or terminate the Plan; provided, however, that no amendment, suspension or termination of the Plan or the Award shall adversely affect the Award in any material way without written consent of Employee.
- 13. **Notices.** Notices hereunder shall be in writing, and if to the Company, may be delivered personally to the Compensation Department or such other party as designated by the Company or mailed to its principal office at 10400 Fernwood Road, Bethesda, Maryland 20817, addressed to the attention of the Stock Option Administrator (Department 935.40), and if to Employee, may be delivered personally or mailed to Employee at his or her address on the records of the Company.

- 14. **Successors and Assigns.** This Agreement shall bind and inure to the benefit of the parties hereto and the successors and assigns of the Company and, to the extent provided in paragraph 8 above and in the Plan, to the personal representatives, legatees and heirs of Employee.
- 15. **No Effect on Employment.** This agreement is not a contract of employment or otherwise a limitation on the right of the Company to terminate the employment of Employee or to increase or decrease Employee's compensation from the rate of compensation in existence at the time this Agreement is executed.

IN WITNESS WHEREOF, MARRIOTT INTERNATIONAL, INC. has caused this Agreement to be signed by its Executive Vice President, Global Human Resources, effective the day and year first hereinabove written.

MARRIOTT INTERNATIONAL, INC.	EMPLOYEE	
	<name></name>	
Executive Vice President, Global Human Resources	<ssn></ssn>	
	Signed Electronically	

# MI SHARES AGREEMENT MARRIOTT INTERNATIONAL, INC. STOCK AND CASH INCENTIVE PLAN

THIS AGREEMENT is made on **<GRANT DATE>** (the "Award Date") by MARRIOTT INTERNATIONAL, INC. (the "Company") and **<NAME>** ("Employee").

## WITNESSETH:

WHEREAS, the Company maintains the Marriott International, Inc. Stock and Cash Incentive Plan, as amended (the "Plan"); and

WHEREAS, the Company wishes to award to designated employees certain MI Share awards as provided in Article 9A of the Plan; and

WHEREAS, Employee has been approved by the Compensation Policy Committee (the "Committee") of the Company's Board of Directors (the "Board") to receive an award of MI Shares under the Plan;

NOW, THEREFORE, it is agreed as follows:

- 1. **Prospectus.** Employee has been provided with, and hereby acknowledges receipt of, a Prospectus for the Plan dated <**DATE**>>, which contains, among other things, a detailed description of the MI Share award provisions of the Plan.
- 2. **Interpretation.** The provisions of the Plan are incorporated by reference and form an integral part of this Agreement. Except as otherwise set forth herein, capitalized terms used herein shall have the meanings given to them in the Plan. In the event of any inconsistency between this Agreement and the Plan, the terms of the Plan shall govern. A copy of the Plan is available from the Compensation Department of the Company upon request. All decisions and interpretations made by the Committee or its delegate with regard to any question arising hereunder or under the Plan shall be binding and conclusive.
- 3. **Grant of MI Shares.** Subject to Employee's acceptance of this Agreement, and subject to satisfaction of the tax provisions of the Company's International Assignment Policy ("IAP"), if applicable, this award (the "Award") of <<# MI SHARES GRANTED>>MI Shares is made as of the Grant Date.
- 4. MI Share and Common Share Rights. The MI Shares awarded under this Agreement shall be recorded in a Company book-keeping account and shall represent Employee's unsecured right to receive from the Company the transfer of title to shares of Common Stock of the Company ("Common Shares") in accordance with the schedule of Vesting Dates set forth in paragraph 5 below, provided that Employee has satisfied the Conditions of Transfer set forth in paragraph 6 below and subject to the satisfaction of the provision on withholding taxes set forth in paragraph 9 below. On each such Vesting Date, if it occurs, the Company shall reverse the book-keeping entry for all such related MI Shares and transfer a corresponding number of Common Shares (which may be reduced by the number of shares withheld to satisfy withholding taxes as set forth in paragraph 9 below, if share reduction is the method utilized for satisfying the tax withholding obligation) to an individual brokerage account (the "Account") established and maintained in Employee's name. Employee shall have all the rights of a stockholder with respect to such Common Shares transferred to the Account, including but not limited to the right to vote the Common Shares, to sell, transfer, liquidate or otherwise dispose of the Common Shares, and to receive all dividends or other distributions paid or made with respect to the Common Shares from the time they are deposited in the Account. Employee shall have no voting, transfer, liquidation, dividend or other rights of a Common Share stockholder with respect to MI Shares prior to such time that the corresponding Common Shares are transferred, if at all, to Employee's Account.
- 5. **Vesting in MI Shares.** The MI Shares shall vest pro rata with respect to an additional 25 percent of the MI Shares granted hereunder on the 15<sup>th</sup> day of the month in which occurs the first, second, third and fourth anniversaries of the Grant Date, respectively. Notwithstanding the foregoing, in the event that any such 15<sup>th</sup> day of the month is a Saturday, Sunday or other day on which stock of the Company is not traded on the New York Stock Exchange or another national exchange, then the Vesting Date shall be the next following day on which the stock of the Company is traded on the New York Stock Exchange or another national exchange.
- 6. **Conditions of Transfer.** With respect to any MI Shares awarded to Employee, as a condition of Employee receiving a transfer of corresponding Common Shares in accordance with paragraph 4 above, Employee shall meet all of the following conditions during the entire period from the Award Date hereof through the Vesting Date relating to such MI Shares:
  - (a) Employee must continue to be an active employee of the Company ("Continuous Employment");

- (b) Employee must refrain from Engaging in Competition (as defined in Section 2.25 of the Plan) without first having obtained the written consent thereto from the Company ("Non-competition"); and
- (c) Employee must refrain from committing any criminal offense or malicious tort relating to or against the Company or, as determined by the Committee in its discretion, engaging in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation. ("No Improper Conduct"). The Company's determination as to whether or not particular conduct constitutes Improper Conduct shall be conclusive.

If Employee should fail to meet the requirements relating to (i) Continuous Employment, (ii) Non-competition, or (iii) No Improper Conduct, then Employee shall forfeit the right to vest in any MI Shares that have not already vested as of the time such failure is determined, and Employee shall accordingly forfeit the right to receive the transfer of title to any corresponding Common Shares. The forfeiture of rights with respect to unvested MI Shares (and corresponding Common Shares) shall not affect the rights of Employee with respect to any MI Shares that already have vested nor with respect to any Common Shares the title of which has already been transferred to Employee's Account.

7. **Non-Assignability.** The MI Shares shall not be assignable or transferable by Employee except by will or by the laws of descent and distribution. During Employee's lifetime, the MI Shares may be exercised only by Employee or, in the event of incompetence, by Employee's legally appointed guardian.

#### 8. Effect of Termination of Employment.

- (a) In the event Employee's Continuous Employment is terminated prior to the relevant Vesting Date on account of death, and if Employee had otherwise met the requirements of Continuous Employment, Non-competition and No Improper Conduct from the Award Date through the date of such death, then Employee's unvested MI Shares shall immediately vest in full upon death and Employee's rights hereunder with respect to any such MI Shares shall inure to the benefit of Employee's executors, administrators, personal representatives and assigns.
- (b) In the event Employee's Continuous Employment is terminated prior to the relevant Vesting Date on account of Employee's Disability or Retirement (as defined below), and if Employee had otherwise met the requirements of Continuous Employment, Non-competition and No Improper Conduct from the Award Date through the date of such Disability or Retirement, and provided that Employee continues to meet the requirements of Non-competition and No Improper Conduct, then Employee's rights hereunder with respect to any outstanding, unvested MI Shares shall continue in the same manner as if Employee continued to meet the Continuous Employment requirement through the Vesting Dates related to the Award, except not for that portion of MI Shares granted less than one year prior to Employee's termination equal to such number of shares multiplied by the ratio of (a) the number of days after the termination date and before the first anniversary of the Grant Date, over (b) the number of days in the twelve (12) month period following the Grant Date. For purposes of this Agreement, "Retirement" shall mean termination of employment on account of Disability (as defined in Section 2.19 of the Plan) or by retiring with the specific approval of the Committee on or after such date on which Employee has attained age 55 and completed ten (10) Years of Service.

Except as set forth in this paragraph 8 above, no other transfer of rights with respect to MI Shares shall be permitted pursuant to this Agreement.

- 9. **Taxes.** The transfer of Common Shares upon each Vesting Date, pursuant to paragraphs 4 and 6 above, shall be subject to the further condition that the Company shall provide for the withholding of any taxes required by federal, state, or local law in respect of that Vesting Date by reducing the number of MI Shares to be transferred to Employee's Account or by such other manner as the Committee shall determine in its discretion.
- 10. **Consent.** By executing this Agreement, Employee consents to the collection, maintenance and processing of Employee's personal information (such as Employee's name, home address, home telephone number and email address, social security number, assets and income information, birth date, hire date, termination date, other employment information, citizenship, marital status) by the Company and the Company's service providers for the purposes of (i) administering the Plan (including ensuring that the conditions of transfer are satisfied from the Award Date through the Vesting Date), (ii) providing Employee with services in connection with Employee's participation in the Plan, (iii) meeting legal and regulatory requirements and (iv) for any other purpose to which Employee may consent ("Permitted Purposes"). Employee's personal information will not be processed for longer than is necessary for such Permitted Purposes. Employee's personal information is collected from the following sources:
  - (a) from this Agreement, investor questionnaires or other forms that Employee submits to the Company or contracts that Employee enters into with the Company;

- (b) from Employee's transactions with the Company, the Company's affiliates and service providers;
- (c) from Employee's employment records with the Company; and
- (d) from meetings, telephone conversations and other communications with Employee.

In addition, Employee further consents to the Company disclosing Employee's personal information to the Company's third party service providers and affiliates and other entities in connection with the services the Company provides related to Employee's participation in the Plan, including:

- (a) financial service providers, such as broker-dealers, custodians, banks and others used to finance or facilitate transactions by, or operations of, the Plan:
- (b) other service providers to the Plan, such as accounting, legal, or tax preparation services;
- (c) regulatory authorities; and
- (d) transfer agents, portfolio companies, brokerage firms and the like, in connection with distributions to Plan participants.

Where Employee's personal information is provided to such third parties the Company requires (to the extent permitted by applicable law) that such parties, agree to process Employee's personal information in accordance with the Company's instructions.

Employee's personal information is maintained on the Company's networks and the networks of the Company's service providers, which may be in the United States or other countries other than the country in which this Award was granted. Employee acknowledges and agrees that the transfer of Employee's personal information to the United States or other countries other than the country in which this Award was granted is necessary for the Permitted Purposes. To the extent (if any) that the provisions of the European Union's Data Protection Directive (Directive 95/46/EC of the European Parliament and of the Council) and/or applicable national legislation derived from such Directive apply, then by executing this Agreement Employee expressly consents to the transfer of Employee's personal information outside of the European Economic Area. Employee may access Employee's personal information to verify its accuracy, update Employee's personal information and/or request a copy of Employee's personal information by contacting Employee's local Human Resources representative. Employee may obtain account transaction information online or by contacting the Plan record keeper as described in the Plan enrollment materials. By accepting the terms of this Agreement, Employee further agrees to the same terms with respect to other Awards Employee received in any prior year under the Plan.

- 11. **No Additional Rights.** Benefits under this Plan are not guaranteed. The grant of Awards is a one-time benefit and does not create any contractual or other right or claim to any future grants of Awards under the Plan, nor does a grant of Awards guarantee future participation in the Plan. The value of Employee's Awards is an extraordinary item outside the scope of Employee's employment contract, if any. Employee's Awards are not part of normal or expected compensation for purposes of calculating any severance, resignation, redundancy, end-of-service payments, bonuses, long-term service awards, pension or retirement benefits (except as otherwise provided by the terms of any U.S.-qualified retirement or pension plan maintained by the Company), or similar payments. By accepting the terms of this Agreement, Employee further agrees to these same terms and conditions with respect to any other Awards Employee received in any prior year under the Plan.
- 12. **Amendment of This Agreement.** The Board of Directors may at any time amend, suspend or terminate the Plan; provided, however, that no amendment, suspension or termination of the Plan or the Award shall adversely affect the Award in any material way without written consent of Employee.
- 13. **Notices.** Notices hereunder shall be in writing, and if to the Company, may be delivered personally to the Compensation Department or such other party as designated by the Company or mailed to its principal office at 10400 Fernwood Road, Bethesda, Maryland 20817, addressed to the attention of the Stock Option Administrator (Department 935.40), and if to Employee, may be delivered personally or mailed to Employee at his or her address on the records of the Company.

- 14. **Successors and Assigns.** This Agreement shall bind and inure to the benefit of the parties hereto and the successors and assigns of the Company and, to the extent provided in paragraph 8 above and in the Plan, to the personal representatives, legatees and heirs of Employee.
- 15. **No Effect on Employment.** This agreement is not a contract of employment or otherwise a limitation on the right of the Company to terminate the employment of Employee or to increase or decrease Employee's compensation from the rate of compensation in existence at the time this Agreement is executed.

IN WITNESS WHEREOF, MARRIOTT INTERNATIONAL, INC. has caused this Agreement to be signed by its Executive Vice President, Global Human Resources, effective the day and year first hereinabove written.

MARRIOTT INTERNATIONAL, INC.	EMPLOYEE	
	<name></name>	_
Executive Vice President, Global Human Resources	<ssn></ssn>	
	Signed Electronically	

# STOCK APPRECIATION RIGHT AGREEMENT MARRIOTT INTERNATIONAL, INC. STOCK AND CASH INCENTIVE PLAN

THIS AGREEMENT is made on **<GRANT DATE>** (the "Award Date") by MARRIOTT INTERNATIONAL, INC. (the "Company") and **<NAME>** ("Employee").

## WITNESSETH:

WHEREAS, the Company maintains the Marriott International, Inc. Stock and Cash Incentive Plan, as amended (the "Plan"); and

WHEREAS, the Company wishes to award to designated employees certain stock appreciation right awards ("SARs" or "Awards") as provided in Article 6 of the Plan; and

WHEREAS, Employee has been approved by the Compensation Policy Committee (the "Committee") of the Company's Board of Directors (the "Board") to receive an award of SARs under the Plan;

NOW, THEREFORE, it is agreed as follows:

- 1. **Prospectus.** Employee has been provided with, and hereby acknowledges receipt of, a Prospectus for the Plan dated <<**DATE>>**, which contains, among other things, a detailed description of the SAR provisions of the Plan.
- 2. **Interpretation.** The provisions of the Plan are incorporated by reference and form an integral part of this Agreement. Except as otherwise set forth herein, capitalized terms used herein shall have the meanings given to them in the Plan. In the event of any inconsistency between this Agreement and the Plan, the terms of the Plan shall govern. A copy of the Plan is available from the Compensation Department of the Company upon request. All decisions and interpretations made by the Committee or its delegate with regard to any question arising hereunder or under the Plan shall be binding and conclusive.
- 3. **Grant of SARs.** The Company hereby grants to Employee as of the Grant Date SARs on **«Grant»** shares of the Company's Common Stock (the "SAR Shares"), subject to the terms and conditions of the Plan, Employee's acceptance of this Agreement and satisfaction of the tax provisions of the Company's International Assignment Policy ("IAP"), if applicable. Under this Agreement, upon satisfying the conditions for exercising SARs as set forth in paragraphs 5 and 6 below, Employee shall receive a number of shares of Common Stock of the Company equal to the number of SAR shares that are being exercised under such SARs multiplied by the quotient of (a) the Final Value minus the Base Value, divided by (b) the Final Value.
- 4. **Base Value and Final Value.** Subject to Paragraph 12 hereof, the Base Value per share of the SAR Shares is <**Base Value** >> and the Final Value is the Fair Market Value of a Share of Common Stock of the Company as of the date the SARs are exercised.
- 5. **Waiting Period and Exercise Dates.** The SAR Shares may not be exercised during the one-year period following the Grant Date (the "waiting period"). Following the waiting period, the SAR Shares may be exercised in accordance with the following schedule: 25% of the SAR Shares commencing on each of the first, second, third and fourth one-year anniversaries of the Grant Date. To the extent that the SARs are not exercised by Employee when they become initially exercisable, the SARs shall not expire but shall be carried forward and shall be exercisable at any time thereafter; provided, however, that the SARs shall not be exercisable after the expiration of ten (10) years from the Grant Date or sooner as set forth in paragraph 9, if applicable. Exercise of the SARs shall not be dependent upon the prior or sequential exercise of any other SARs heretofore granted to Employee by the Company. Except as provided in Article 6 of the Plan and Paragraph 9 below, the SARs may not be exercised at any time unless Employee shall then be an employee of the Company.
- 6. **Method of Exercising SARs.** To exercise the SARs, the person entitled to exercise the SARs must provide a signed written notice or the equivalent to the Company or its designee, as prescribed in the administrative procedures of the Plan, stating the number of SAR Shares with respect to which the SARs are being exercised. The

SARs may be exercised by (a) making provision for the satisfaction of the applicable withholding taxes, and (b) an undertaking to furnish and execute such documents as the Company deems necessary (i) to evidence such exercise, and (ii) to determine whether registration is then required to comply with the Securities Act of 1933 or any other law. Upon satisfying the conditions for exercise including the provision for the satisfaction of the withholding taxes, the Company shall provide confirmation from the Plan record keeper that the transfer agent for the common stock of the Company is holding shares for the account of such person in a certificateless account. The exercise of the SARs may be made by any other means that the Committee determines to be consistent with the Plan's purpose and applicable law.

- 7. **Rights as a Shareholder.** Employee shall have no rights as a shareholder with respect to any SAR Shares covered by the SARs granted hereby until the date of acquisition by Employee of such SAR Shares. No adjustment shall be made for dividends or other rights for which the record date is prior to such date.
- 8. **Non-Assignability.** The SARs shall not be assignable or transferable by Employee except by will or by the laws of descent and distribution. During Employee's lifetime, the SARs may be exercised only by Employee or, in the event of incompetence, by Employee's legally appointed guardian.
- 9. **Effect of Termination of Employment or Death.** If Employee goes on leave of absence for a period of greater than twelve months (except a leave of absence approved by the Board of Directors or the Committee) or ceases to be an employee of the Company for any reason except death, the portion of the SARs which is unexercisable on the date on which Employee ceased to be an Employee or has been on a leave of absence for over twelve months (except a leave of absence approved by the Board or Committee) shall expire on such date and any unexercised portion of the SARs which was otherwise exercisable on such date shall expire at the earlier of (i) the expiration of the SARs in accordance with the term for which the SARs were granted, or (ii) three months from such date, except in the case of an Employee who is an "Approved Retiree" as defined below. If Employee is an Approved Retiree, then the SARs shall expire at the sooner to occur of (i) the expiration of such SARs in accordance with their original term, (ii) the expiration of five years from the date of retirement, or (iii) with respect to SARs granted less than one year before the date the Approved Retiree retires, such retirement date, except not with respect to that portion of the SARs equal to the number of such shares multiplied by the ratio of (a) the number of days between the Grant Date and the retirement date inclusive, over (b) the number of days in the twelve (12) month period following the Grant Date. In the event of the death of Employee without Approved Retiree status during the three (3) month period following termination of employment or a leave of absence over twelve (12) months (except a leave of absence approved by the Board or Committee), the SARs shall be exercisable by Employee's personal representative, heirs or legatees to the same extent and during the same period that Employee could have exercised the SAR if Employee had not died. In the event of the death of Employee while an employee of the Company or while an Approved Retiree, the SAR (if the waiting period has elapsed) shall be exercisable in its entirety by Employee's personal representatives, heirs or legatees at any time prior to the expiration of one year from the date of the death of Employee, but in no event after the term for which the SAR was granted. For purposes of this Agreement, an "Approved Retiree" is any SAR holder who (i) terminates employment by reason of a Disability, or (ii) (A) retires from employment with the Company with the specific approval of the Committee on or after such date on which the SAR holder has attained age 55 and completed 10 Years of Service, and (B) has entered into and has not breached an agreement to refrain from Engaging in Competition in form and substance satisfactory to the Committee; and if the Committee subsequently determines, in its sole discretion, that an Approved Retiree has violated the provisions of the Agreement to refrain from Engaging in Competition, or has engaged in willful acts or omissions or acts or omissions of gross negligence that are or potentially are injurious to the Company's operations, financial condition or business reputation, such Approved Retiree shall have ninety (90) days from the date of such finding within which to exercise any SARs or portions thereof which are exercisable on such date, and any SARs or portions thereof which are not exercised within such ninety (90) day period shall expire and any SARs or portion thereof which are not exercisable on such date shall be cancelled on such date.
- 10. **Consent.** By executing this Agreement, Employee consents to the collection, maintenance and processing of Employee's personal information (such as Employee's name, home address, home telephone number and email address, social security number, assets and income information, birth date, hire date, termination date, other employment information, citizenship, marital status) by the Company and the Company's service providers for the purposes of (i) administering the Plan (including ensuring that the conditions of transfer are satisfied from the

Award Date through the Exercise Date), (ii) providing Employee with services in connection with Employee's participation in the Plan, (iii) meeting legal and regulatory requirements and (iv) for any other purpose to which Employee may consent ("Permitted Purposes"). Employee's personal information will not be processed for longer than is necessary for such Permitted Purposes. Employee's personal information is collected from the following sources:

- (a) from this Agreement, investor questionnaires or other forms that Employee submits to the Company or contracts that Employee enters into with the Company;
- (b) from Employee's transactions with the Company, the Company's affiliates and service providers;
- (c) from Employee's employment records with the Company; and
- (d) from meetings, telephone conversations and other communications with Employee.

In addition, Employee further consents to the Company disclosing Employee's personal information to the Company's third party service providers and affiliates and other entities in connection with the services the Company provides related to Employee's participation in the Plan, including:

- (a) financial service providers, such as broker-dealers, custodians, banks and others used to finance or facilitate transactions by, or operations of, the Plan;
- (b) other service providers to the Plan, such as accounting, legal, or tax preparation services;
- (c) regulatory authorities; and
- (d) transfer agents, portfolio companies, brokerage firms and the like, in connection with distributions to Plan participants.

Where Employee's personal information is provided to such third parties the Company requires (to the extent permitted by applicable law) that such parties, agree to process Employee's personal information in accordance with the Company's instructions.

Employee's personal information is maintained on the Company's networks and the networks of the Company's service providers, which may be in the United States or other countries other than the country in which this Award was granted. Employee acknowledges and agrees that the transfer of Employee's personal information to the United States or other countries other than the country in which this Award was granted is necessary for the Permitted Purposes. To the extent (if any) that the provisions of the European Union's Data Protection Directive (Directive 95/46/EC of the European Parliament and of the Council) and/or applicable national legislation derived from such Directive apply, then by executing this Agreement Employee expressly consents to the transfer of Employee's personal information outside of the European Economic Area. Employee may access Employee's personal information to verify its accuracy, update Employee's personal information and/or request a copy of Employee's personal information by contacting Employee's local Human Resources representative. Employee may obtain account transaction information online or by contacting the Plan record keeper as described in the Plan enrollment materials. By accepting the terms of this Agreement, Employee further agrees to the same terms with respect to other Awards Employee received in any prior year under the Plan.

11. **No Additional Rights.** Benefits under this Plan are not guaranteed. The grant of Awards is a one-time benefit and does not create any contractual or other right or claim to any future grants of Awards under the Plan, nor does a grant of Awards guarantee future participation in the Plan. The value of Employee's Awards is an extraordinary item outside the scope of Employee's employment contract, if any. Employee's Awards are not part of normal or expected compensation for purposes of calculating any severance, resignation, redundancy, end-of-service payments, bonuses, long-term service awards, pension or retirement benefits (except as otherwise provided by the terms of any U.S.-qualified retirement or pension plan maintained by the Company or any of its subsidiaries), or similar payments. By accepting the terms of this Agreement, Employee further agrees to these same terms and conditions with respect to any other Awards Employee received in any prior year under the Plan.

- 12. **Recapitalization or Reorganization.** Certain events affecting the Common Stock of the Company and mergers, consolidations and reorganizations affecting the Company may affect the number or type of securities deliverable upon exercise of the SAR or limit the remaining term over which the SAR may be exercised.
- 13. **General Restriction.** In accordance with the terms of the Plan, the Company may limit or suspend the exercisability of the SARs or the purchase or issuance of SAR Shares thereunder under certain circumstances. Any delay caused thereby shall in no way affect the date of termination of the SARs.
- 14. **Amendment of This Agreement.** The Board of Directors may at any time amend, suspend or terminate the Plan; provided, however, that no amendment, suspension or termination of the Plan or the SARs shall adversely affect in any material way the SARs without the written consent of Employee.
- 15. **Notices.** Notices hereunder shall be in writing, and if to the Company, may be delivered personally to the Compensation Department or such other party as designated by the Company or mailed to its principal office at 10400 Fernwood Road, Bethesda, Maryland 20817, addressed to the attention of the SAR Administrator (Department 935.40), and if to Employee, may be delivered personally or mailed to Employee at his or her address on the records of the Company.
- 16. **Successors and Assigns.** This Agreement shall bind and inure to the benefit of the parties hereto and the successors and assigns of the Company and, to the extent provided in Paragraph 9 above and the provisions of the Plan, to the personal representatives, legatees and heirs of Employee.
- 17. **No Effect on Employment.** Nothing contained in this Agreement shall be construed to limit or restrict the right of the Company to terminate Employee's employment at any time, with or without cause, or to increase or decrease Employee's compensation from the rate of compensation in existence at the time this Agreement is executed.

**IN WITNESS WHEREOF,** the parties hereto have executed this Agreement to be effective as of the Grant Date.

MARRIOTT INTERNATIONAL, INC.	EMPLOYEE
	Employee Name (Please Print)
By: Executive Vice President, Global Human Resources	Employee Social Security Number (Please Print)
	Employee Signature

# STOCK APPRECIATION RIGHT AGREEMENT (FOR NON-EMPLOYEE DIRECTORS) MARRIOTT INTERNATIONAL, INC. STOCK AND CASH INCENTIVE PLAN

THIS AGREEMENT is made on **<GRANT DATE>** (the "Grant Date") by MARRIOTT INTERNATIONAL, INC. (the "Company") and **<NAME>** ("Director").

#### WITNESSETH:

WHEREAS, the Company maintains the Marriott International, Inc. Stock and Cash Incentive Plan, as amended (the "Plan"); and

WHEREAS, the Company wishes to award to designated directors certain stock appreciation right awards as provided in Article 12 of the Plan; and

WHEREAS, Director has been approved by the Compensation Policy Committee (the "Committee") of the Company's Board of Directors (the "Board") to receive an award of "Stock Appreciation Rights" under the Plan;

NOW, THEREFORE, it is agreed as follows:

- 1. **Prospectus.** Director has been provided with, and hereby acknowledges receipt of, a Prospectus for the Plan dated <<**DATE>>**, which contains, among other things, a detailed description of the stock appreciation right award provisions of the Plan.
- 2. **Interpretation.** The provisions of the Plan are incorporated by reference and form an integral part of this Agreement. Except as otherwise set forth herein, capitalized terms used herein shall have the meanings given to them in the Plan. In the event of any inconsistency between this Agreement and the Plan, the terms of the Plan shall govern. A copy of the Plan is available from the Compensation Department of the Company upon request. All decisions and interpretations made by the Committee or its delegate with regard to any question arising hereunder or under the Plan shall be binding and conclusive.
- 3. **Grant of SARs.** The Company hereby grants to Director as of the Grant Date stock appreciation rights (the "SARs") on <<**Grant Date**>> shares of the Company's Common Stock (the "SAR Shares"), subject to the terms and conditions of the Plan and Director's acceptance of this Agreement. Under this Agreement, upon satisfying the conditions for exercising SARs as set forth in paragraphs 5 and 6 below, Director shall receive a number of shares of Common Stock of the Company equal to the number of SAR shares that are being exercised under such SARs multiplied by the quotient of (a) the Final Value minus the Base Value, divided by (b) the Final Value.
- 4. **Base Value and Final Value.** Subject to Paragraph 12 hereof, the Base Value per share of the SAR Shares is **<<Base Value>>** and the Final Value is the Fair Market Value of a Share of Common Stock of the Company as of the date the SARs are exercised.
- 5. **Waiting Period and Exercise Dates.** The SAR Shares may not be exercised during the one-year period following the Grant Date (the "waiting period"). Following the waiting period, all or a portion of the SAR Shares may be exercised. To the extent that the SARs are not exercised by Director when they become initially exercisable, the SARs shall not expire but shall be carried forward and shall be exercisable at any time thereafter; provided, however, that the SARs shall not be exercisable after the expiration of ten (10) years from the Grant Date or sooner as set forth in paragraph 9, if applicable. Exercise of the SARs shall not be dependent upon the prior or sequential exercise of any other SARs heretofore granted to Director by the Company. Except as provided in Article 6 of the Plan and Paragraph 9 below, the SARs may not be exercised at any time unless Director shall then be a Director of the Company.
- 6. **Method of Exercising SAR.** To exercise the SARs, the person entitled to exercise the SARs must provide a signed written notice or the equivalent to the Company or its designee, as prescribed in the administrative procedures of the Plan, stating the number of SAR Shares with respect to which the SARs are being exercised. The SARs may be exercised by undertaking to furnish and execute such documents as the Company deems necessary (i) to evidence such exercise, and (ii) to determine whether registration is then required to comply with the Securities Act of 1933 or any other law. Upon satisfying the conditions for exercise, the Company shall, provide confirmation from the Plan record keeper that the transfer agent for the common stock of the Company is holding shares for the account of such person in a certificateless account. Pursuant to procedures, if any, that may be adopted by the Committee or its delegate, the exercise of the SARs may be by any other means that the Committee determines to be consistent with the Plan's purpose and applicable law.

- 7. **Rights as a Shareholder.** Director shall have no rights as a shareholder with respect to any SAR Shares covered by the SARs granted hereby until the date of acquisition by Director of such SAR Shares. No adjustment shall be made for dividends or other rights for which the record date is prior to such date.
- 8. **Non-Assignability.** The SARs shall not be assignable or transferable by Director except by will or by the laws of descent and distribution. During Director's lifetime, the SARs may be exercised only by Director or, in the event of incompetence, by Director's legally appointed guardian.
- 9. **Effect of Termination of Status as Director.** If Director ceases to be a Director of the Company for any reason except death, the SARs will continue to be exercisable until the expiration of such SARs in accordance with its original term. In the event of the death of Director, the SARs shall be exercisable by Director's personal representative, heirs or legatees at any time prior to the expiration of one year from the date of the death of Director, but in no event after the term for which the SARs were granted.
- 10. **Consent.** By executing this Agreement, Director consents to the collection, maintenance and processing of Director's personal information (such as Director's name, home address, home telephone number and email address, social security number, assets and income information, birth date, date of commencement and termination as a Director, citizenship, marital status) by the Company and the Company's service providers for the purposes of (i) administering the Plan (including ensuring that the conditions of transfer are satisfied from the Award Date through the Exercise Date), (ii) providing Director with services in connection with Director's participation in the Plan, (iii) meeting legal and regulatory requirements and (iv) for any other purpose to which Director may consent ("Permitted Purposes"). Director's personal information will not be processed for longer than is necessary for such Permitted Purposes. Director's personal information is collected from the following sources:
  - a. from this Agreement, investor questionnaires or other forms that Director submits to the Company or contracts that Director enters into with the Company;
  - b. from Director's transactions with the Company, the Company's affiliates and service providers;
  - c. from Director's service records with the Company; and
  - d. from meetings, telephone conversations and other communications with Director.

In addition, Director further consents to the Company disclosing Director's personal information to the Company's third party service providers and affiliates and other entities in connection with the services the Company provides related to Director's participation in the Plan, including:

- a. financial service providers, such as broker-dealers, custodians, banks and others used to finance or facilitate transactions by, or operations of, the Plan;
- b. other service providers to the Plan, such as accounting, legal, or tax preparation services;
- c. regulatory authorities; and
- d. transfer agents, portfolio companies, brokerage firms and the like, in connection with distributions to Plan participants.

Where Director's personal information is provided to such third parties, the Company requires (to the extent permitted by applicable law) that such parties agree to process Director's personal information in accordance with the Company's instructions.

Director's personal information is maintained on the Company's networks and the networks of the Company's service providers, which may be in the United States or other countries other than the country in which this Award was granted. Director acknowledges and agrees that the transfer of Director's personal information to the United States or other countries other than the country in which this Award was granted is necessary for the Permitted Purposes. To the extent (if any) that the provisions of the European Union's Data Protection Directive (Directive 95/46/EC of the European Parliament and of the Council) and/or applicable national legislation derived from such Directive apply, then by executing this Agreement Director' expressly consents to the transfer of Director's personal information outside of the European Economic Area. Director may access Director's personal information to verify its accuracy, update Director's personal information and/or by request a copy of Director's personal information by contacting the Senior Vice President, Compensation and Benefits. Director may obtain account transaction information online or by contacting the Plan record keeper as described in the Plan enrollment materials.

11. **No Additional Rights.** Benefits under this Plan are not guaranteed. The grant of Awards is a one-time benefit and does not create any contractual or other right or claim to any future grants of Awards under the Plan, nor does a grant of Awards guarantee future participation in the Plan.

- 12. **General Restriction.** In accordance with the terms of the Plan, the Company may limit or suspend the exercisability of the SARs or the purchase or issuance of SAR Shares thereunder under certain circumstances. Any delay caused thereby shall in no way affect the date of termination of the SARs.
- 13. **Amendment of This Agreement.** The Board of Directors may at any time amend, suspend or terminate the Plan; provided, however, that no amendment, suspension or termination of the Plan or the SARs shall adversely affect in any material way the SARs without the written consent of Director.
- 14. **Notices.** Notices hereunder shall be in writing, and if to the Company, may be delivered personally to the Compensation Department or such other party as designated by the Company or mailed to its principal office at 10400 Fernwood Road, Bethesda, Maryland 20817, addressed to the attention of the SARs Administrator (Department 935.40), and if to Director, may be delivered personally or mailed to Director at his or her address on the records of the Company.
- 15. **Successors and Assigns.** This Agreement shall bind and inure to the benefit of the parties hereto and the successors and assigns of the Company and, to the extent provided in Paragraph 9 above and the provisions of the Plan, to the personal representatives, legatees and heirs of Director.

**IN WITNESS WHEREOF,** the parties hereto have executed this Agreement to be effective as of the Grant Date.

MAF	RRIOTT INTERNATIONAL, INC.	DIRECTOR
By:		
	Executive Vice President, Global Human Resources	Director Signature

# **Summary of Marriott International, Inc. Director Compensation**

# **Non- Employee Directors**

### Cash Compensation

Annual Cash Retainer	
Vice Chairman	\$ 75,000
Other non-employee directors	\$ 60,000
Attendance Fees (each Board, Committee or shareholders meeting)	
Vice Chairman	\$1,562.50
Other non-employee directors	\$ 1,250
Annual Committee Cash Retainers:	
Chair of the Audit Committee	\$ 20,000
Chair of the Compensation Policy Committee	\$ 10,000
Chair of the Nominating and Corporate Governance	
Committee	\$ 10,000
Chair of the Committee for Excellence	\$ 10,000

Annual Cash Retainers and Attendance Fees are paid quarterly in cash and can be deferred or received in the form of Company stock units or stock appreciation rights at the director's election under the Company's Executive Deferred Compensation Plan or Stock and Cash Incentive Plan.

# **Equity Compensation**

Immediately before the annual meeting of shareholders each year, non-employee directors receive an annual award for a number of vested shares of Company Stock. Subject to the terms and limits in the Stock and Cash Incentive Plan, the market value of the awards will be \$100,000 for the Vice Chairman and \$80,000 for the other non-employee directors.

# Other Information

Non-employee directors' compensation is paid based on an annual period commencing with the annual shareholders meeting. The Company reimburses directors for travel expenses, other out-of-pocket costs they incur when they attend meetings and, for one meeting per year, attendance by spouses. To encourage our directors to visit and personally evaluate properties, the directors also receive complimentary rooms, food and beverages at Company-owned, operated or franchised hotels when on personal travel. The value of these benefits is reported to the directors as taxable compensation.

# **Employee Directors**

Officers of the Company are not paid for their service as directors.

#### Marriott International, Inc. Executive

### Officer Incentive Plan and Executive

#### Officer Individual Performance Plan

#### Marriott International, Inc. Annual Cash Incentives

To help motivate the attainment of annual objectives, the Company maintains the Marriott International, Inc. Executive Officer Incentive Plan and the Marriott International, Inc. Executive Officer Individual Performance Plan for the Chief Executive Officer and each of the other named executives. Under each plan, specific objectives are established annually for a minimum level, target level, and a maximum level of performance, and for each such objective, actual performance is measured against these levels in order to determine the actual payment.

The Executive Officer Incentive Plan is operated under the shareholder-approved Marriott International, Inc. Stock and Cash Incentive Plan. The plan rewards executives for the Company's achievement of Company financial objectives. Typically, the compensation committee has measured performance based upon earnings per share. Notwithstanding the achievement of financial or other performance goals, the compensation committee has the authority to adjust the amount payable under the plan.

The Individual Performance Plan emphasizes individual executive performance as well as measures of business/operating unit financial performance such as revenue growth relating to newly developed rooms and customer, owner/franchisee and associate satisfaction. Each year, the compensation committee selects the performance criteria and weighting of the performance objectives for each eligible executive. These criteria and their weighting vary among eligible executives.

Under the plans, the aggregate norm and maximum payments for each named executive are as follows:

	Norm (%) of Salary	Maximum (%) of Salary
Chairman and Chief Executive Officer	125	190
President and Chief Operating Officer	90	135
Executive Vice President Lodging Development	75	115
Executive Vice President, Chief Financial Officer and President- Continental European		
Lodging	75	115
President North American Lodging Operations and Global Brand Management	60	90

Of these amounts, 50% to 60% are payable under the Executive Officer Incentive Plan. No payment is made if performance fails to meet the minimum level for a particular objective.

# MARRIOTT INTERNATIONAL, INC. COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES

		F	iscal Year		
(\$ in millions, except ratio)	2007	2006	2005	2004	2003
Income from continuing operations before income taxes and minority interest	\$1,137	\$1,092	\$829	\$752	\$582
(Income) loss related to equity method investees	(15)	(3)	(36)	14	17
	1,122	1,089	793	766	599
Add/(deduct):					
Fixed charges	336	252	216	197	211
Interest capitalized	(49)	(32)	(30)	(16)	(25)
Distributed income of equity method investees	29	17	21	9	30
Minority interest in pretax loss	1		(2)		
Earnings available for fixed charges	\$1,439	\$1,326	\$998	\$956	\$815
Fixed charges:					
Interest expensed and capitalized (1)	\$ 233	\$ 156	\$136	\$115	\$135
Estimate of interest within rent expense	103	96	80	82	76
Total fixed charges	\$ 336	\$ 252	\$216	\$197	\$211
Ratio of earnings to fixed charges	4.3	5.3	4.6	4.9	3.9

<sup>(1) &</sup>quot;Interest expensed and capitalized" includes amortized premiums, discounts, and capitalized expenses related to indebtedness.

Exhibit 12

Country: Anguilla

Marriott Anguilla Licensing Company Inc.

The Ritz-Carlton Hotel Company Ltd. (Anguilla Branch)

Country: Antigua and Barbuda

The Ritz-Carlton Hotel Company of Antigua Limited

**Country: Argentina** 

Marriott Argentina Licensing Company S.A.

Marriott Argentina S.A.

Marriott International Hotels, Inc. (Argentina Branch)

Country: Armenia

Marriott Hotels International B.V. (Armenia Branch)

Country: Aruba

Aruba Surf Club N.V.

Cooperative Vereniging Marriott Vacation Club International of Aruba

Costa Del Sol Development Company N.V. Marriott Aruba Licensing Company AVV

Marriott Aruba N.V.

Marriott Resorts Hospitality of Aruba N.V. Marriott Vacation Club International of Aruba, N.V.

MVCI Finance CV

The Ritz-Carlton Hotel Company of Aruba N.V.

The Ritz-Carlton Hotel Holding Company of Aruba N.V.

Country: Australia

158 Ferny Avenue Pty Limited 30 Pitt Street Pty Limited 515 Queen Street PTY Ltd

Hotel Holdings and Services Pty Limited

Lonex Pty Limited

Marriott International Management Company B.V. (Australian Branch)

Renaissance Services B.V. (Australian Branch)

Country: Austria

Graz Marriott Hotelmanagement GmbH Marriott Hotel- Betriebsgesellschaft, mbH Marriott Vacation Club Timesharing GmbH

**Country: Bahamas** 

Marriott Bahamas Licensing Company Ltd.
Marriott Ownership Resorts (Bahamas) Limited
Marriott Resorts Hospitality (Bahamas) Limited
New World Hotels (Bahamas) Limited
RC Management Company Bahamas Limited

RC Rose Island Hotel Company Limited The Abaco Club RC, Ltd. (Joint Venture)

The Ritz-Carlton Hotel Company, Ltd. (Registration in the Bahamas)

**Country: Bahrain** 

Renaissance Services B.V. (Bahrain Representative Office)

Exhibit 21

**Country: Belgium** 

Luxury Hotels International Management Belgium SPRL Marriott Hotels International B.V. (Brussels Branch) Marriott Hotels International Limited (Belgium Branch) Renaissance Hotels International B.V. (Belgium Branch) Renaissance Hotels International Management Belgium SPRL

Renaissance Hotels International, S.A.

Country: Bermuda

Adamar International Lodging, Ltd. Crest Management Services, Limited Luxury Hotels International Lodging Ltd. Marriott International Services, Ltd. Renaissance International Lodging Ltd. The Ritz-Carlton Hotel Company, Ltd.

**Country: Brazil** 

Mar Hoteis de Sao Paulo Ltda. Marriott do Brasil Hotelaria Ltda. Operadora Sao Paulo Renaissance Ltda. Renaissance do Brasil Hoteleria Ltda. RHI do Brasil Hotelaria Ltda. Sao Paulo Alphaville Marriott Ltda.

Country: Canada

Marriott Hotels of Canada Ltd. Ritz Carlton Holdings Canada ULC

The Ritz-Carlton Hotel Company of Canada, Ltd.

**Country: Cayman** 

Luxury Hotels International Lodging Ltd. (Cayman Islands Branch)

Marriott Cayman Islands Licensing Company I, Ltd. Marriott Cayman Islands Licensing Company II, Ltd. Marriott Cayman Islands Licensing Company III, Ltd. Marriott Cayman Islands Licensing Company IV, Ltd. Marriott Cayman Islands Licensing Company IX, Ltd. Marriott Cayman Islands Licensing Company V, Ltd. Marriott Cayman Islands Licensing Company VI, Ltd. Marriott Cayman Islands Licensing Company VII, Ltd. Marriott Cayman Islands Licensing Company VIII, Ltd. Marriott Cayman Islands Licensing Company X, Ltd. RC Hotel Holding Company Limited (Cayman Islands Branch)

Renaissance Caribbean Limited

The Ritz-Carlton Hotel Company of the Cayman Islands, Ltd.

**Country: Chile** 

Hoteles de Chile S.A.

Hoteles de Chile S.A. (Joint Venture) Inversiones Hoteleras S.A. (Joint Venture) Marriott Chile Licensing Company Limitada

Marriott Chile S.A.

Marriott Inversiones y Servicios Limitada

MORI Chile S.A.

The Ritz-Carlton Hotel Company of Chile Limitada

The Ritz-Carlton Inversiones Limitada

Exhibit 21

**Country: China** 

Marriott Asia Pacific Management Limited (Beijing, China Branch - Rep. Office)

Marriott Management (Shanghai) Company, Limited

**Country: Czech Republic** 

Gestin Holding, A.S. (Joint Venture)

Marriott Hotels International B.V. (Czech Branch)

Marriott International Hotels, Inc. (Czech Republic Branch)

Penta Hotel Management GmbH (Czech Branch) Renaissance Hotels International B.V. (Czech Branch)

**Country: Denmark** 

Hotelinvest Kalvebod A/S - (Joint Venture)

Marriott Hotels Denmark A/S

Marriott Hotels International Limited (Denmark - Representative Office)

**Country: Dominican Republic** 

Marriott Hotels International B.V. (Dominican Republic Branch)

Transamerican Hoteles, S.A.

**Country: Ecuador** 

Amazonas H.O.T. S.A. (Joint Venture)

Marriott Ecuador Licensing Company MLC S.A.

Country: Egypt

Marriott Hotels International (Egypt) B.V. (Egypt Branch)
Marriott Hotels International B.V. (Egypt Branch)
Marriott Hurghada Management, Inc. (Egypt Branch)
Marriott Mirage City Management, Inc. (EgyptBranch)
Marriott Sharm El Sheikh Management (Egypt Branch)
Renaissance Hotels International B.V. (Egypt Branch)
Renaissance Services B.V. (Representative Office in Egypt)

The Ritz-Carlton Hotel Company of Egypt S.A.E.

The Ritz-Carlton Hotel Company, L.L.C. (Egyptian Branch)

Country: El Salvador

Luxury Hotels International Management Company B.V. (El Salvador Branch)

Marriott International, Inc. (El Salvador Branch)

**Country: France** 

Luxury Hotels International of France SAS

Lyon Real Estate Company SAS Marriott France Group Companies SAS

Marriott Hotels International Limited (French Branch - Liaison Office)

Marriott Management France SAS

MVCI France SAS

MVCI Holidays France SAS

Paris Arcueil Real Estate Company SAS Paris St. Denis Pleyel Real Estate Company SAS

Ramcap SAS Renthotel Paris SAS Roissy CYBM SAS

Toulouse Operating Company SAS

Exhibit 21

**Country: Germany** 

Berlin Marriott Hotelmanagement GmbH Bremen Marriott Hotelmanagement GmbH Cologne MH Operating Company GmbH Dusseldorf RC Operating Company GmbH Frankfurt Marriott Hotelmanagement GmbH Hamburg Marriott Hotelmanagement GmbH Leipzig Marriott Hotelmanagement GmbH

Marriott Hotel Holding GmbH Marriott Hotelmanagement GmbH

Middle Ring Properties GmbH Hotelbetriebsgesellschaft Munich CY Schwalther Operating Company GmbH

MVCI Holidays GmbH

Penta Hotel Managementgesellchaft mbH
The Ritz-Carlton Hotel Company (Berlin) GmbH
The Ritz-Carlton Hotel Company of Germany, GmbH
The Ritz-Carlton Hotel Management GmbH

**Country: Greece** 

Marriott Hotels Hellas, S.A.

**Country: Guam** 

International Hotel Licensing Company S.A.R.L. (Guam Branch)

Marriott Hotels International B.V. (Guam Branch)

Country: Guatemala

Marriott Guatemala Licensing S.A.

**Country: Honduras** 

Marriot De Honduras, Sociedad de Responsabilidad Limitada

**Country: Hong Kong** 

Club Resorts (Thailand) Holdings Limited

Courtyard China Hotels Limited

Luxury Hotels International of Hong Kong Limited

Marriott Hong Kong Limited

Marriott International Development, Limited Marriott Properties (International) Limited MVCI Asia Pacific (Hong Kong) Pte Limited MVCI Asia Pacific Finance Pte. Limited

Renaissance Hotels Limited

Renaissance Management Hong Kong Limited

RHIL Limited

The Ritz-Carlton Hotel Company Sales and Marketing B.V. (Hong Kong Branch)

The Ritz-Carlton Limited

Country: India

Marriott Hotels India Private Limited

**Country: Indonesia** 

P.T. Luxury Hotels International Indonesia P.T. Marriott International Indonesia

P.T. Ritz-Carlton Indonesia

Exhibit 21

**Country: Ireland** 

Marriott (Ireland) Investments Limited

Marriott Hotels International B.V. (Ireland Branch) Marriott Hotels International Limited (Irish Branch)

MVCI Ireland Ltd.

MVCI Irish Holding Company Limited

**MVCI** Services Limited

The Ritz-Carlton Hotel Company B.V. (Ireland Branch)

Torriam Hotel Operating Company Limited

**Country: Italy** 

Bulgari Hotels and Resorts Milano, S.r.l.

Luxury Hotels International Management Company B.V. (Italy Branch)

Luxury Hotels International Management Italy S.r.l.

MVCI Holidays S.r.l.

**Country: Jamaica** 

Marriott Jamaica Licensing Company Limited The Ritz-Carlton Hotel Company of Jamaica Limited

Country: Japan

Luxury Hotels International of Hong Kong Limited (Japan Branch)

Luxury Hotels International of Japan, Inc.

MVCI Asia Pacific (Hong Kong) Pte. Limited (Japan Branch)

Renaissance Services B.V. (Tokyo, Japan Branch)

The Ritz-Carlton Hotel Company Sales and Marketing (Tokyo, Japan Branch)

The Ritz-Carlton Property Management Company (Tokyo), Ltd.

**Country: Jersey, Channel Islands** 

Aberdeen Hotel Limited

Marriott European Holdings Limited - (Joint Venture)

**Country: Jordan** 

Business Tourism Company LLC - (Joint Venture) Marriott Hotels International B.V. (Jordan Branch) Marriott International Hotels, Inc. (Jordan Branch) Marriott Worldwide Corporation (Jordan Branch) Renaissance Services B.V. (Rep. Office in Jordan)

Country: Kazakhstan

Marriott Hotels International B.V. (Kazakhstan Branch) Renaissance Hotels International B.V. (Kazakhstan Branch)

**Country: Kuwait** 

Kuwait National Hotel & Tourism Company - (Joint Venture)

**Country: Lebanon** 

Marriott International Hotels, Inc. (Lebanon Branch)

MVCI Lebanon, S.A.R.L.

Country: Liberia

Renaissance Management Services Company Limited

Exhibit 21

**Country: Luxembourg** 

BRE/Munich Hotel S.a.r.l.

IHLC Investment Company S.a.r.l.

International Hotel Licensing Company S.a.r.l.

Marriott ECP GP SARL Marriott ECP LP I SARL Marriott ECP LP II SARL

Marriott International Treasury SCA

Country: Macau

MVCI Asia Pacific (Hong Kong) Pte Limited (Macau Branch)

Country: Malaysia

Renaissance Hotels International Corporation Limited (Malaysia Branch) Renaissance International Management Company B.V.(Malaysian Branch)

Renaissance Services B.V. (Malaysia Branch)

The Ritz-Carlton Hotel Company, L.L.C. (Malaysian Branch)

Country: Malta

Renaissance Hotels International B.V. (Malta Branch)

**Country: Mexico** 

Adquisiciones Cancun-Vallarta S. de R.L. de C.V. - (Joint Venture)

Elcrisa, S.A. de C.V. - (Joint Venture) Marriott Hotels, S.A. de C.V. Marriott Mexicana S.A. de C.V. Operadora Marriott, S.A. de C.V. Polsery, S.A. de C.V. - (Joint Venture) Promociones Marriott, S.A. de C.V.

R.M. Mexicana S.A.de C.V.

Renaissance P.V. Mexicana S.A. de C.V. Servimar, S.A. de C.V. - (Joint Venture)

The R.C. Management Company of Mexico, S.A. de C.V. The Ritz-Carlton Hotel Company of Mexico, S.A. de C.V.

**Country: Netherlands** 

Adamar Hotels International B.V.

Adamar International Management Company B.V.

Aruba Finance Holdings BV Bulgari Hotels & Resorts B.V. Diplomat Properties B.V. Luxury Hotels International B.V.

Luxury Hotels International Management Company B.V.

Marriott Dutch Acquisition Company B.V.

Marriott European Ventures B.V.

Marriott Hotels International (Egypt) B.V.

Marriott Hotels International B.V.

Marriott Hotels of Amsterdam, B.V.

Marriott International Finance Company B.V.

Marriott International Holding Company B.V.

Marriott International Licensing Company B.V.

Marriott International Management Company B.V.

Marriott Netherlands Group Companies N.V.

Marriott RHG Acquisition B.V.

Exhibit 21

Country: Netherlands cont'd.

MVCI Egypt B.V. MVCI Holdings B.V.

Penha Longa Hotel Management Company B.V. Renaissance do Brasil Hotelaria Holding Company B.V.

Renaissance Hotels International B.V.

Renaissance International Management Company B.V.

Renaissance Management B.V. Renaissance Services B.V. Renaissance Special Purposes B.V.

Rochapora Desenvolvimento Imobiliario Holding Company, B.V.

The Ritz-Carlton Hotel Company B.V.

The Ritz-Carlton Hotel Company Sales and Marketing B.V. The Ritz-Carlton International Licensing Company B.V. The Ritz-Carlton International Management Company B.V.

**Country: Netherlands Antilles** 

Adamar Netherlands Antilles N.V. La Defense Holding (Paris) N.V. Lux International Hotels N.V. Marriott Curacao N.V.

Marriott International Lodging N.V. Renaissance International Lodging N.V.

RHG Holding N.V.

The Ritz-Carlton Hotel Company N.V. Torriam International Lodging N.V.

**Country: New Zealand** 

Marriott International Services Ltd. (New Zealand Branch)

Country: Peru

Marriott Peru Licensing Company SAC

Marriott Peru S.A.C.

**Country: Philippines** 

Marriott International Hotels, Inc. (Philippine Branch) Porto Bello Cove Hotel Corporation (Joint Venture)

Renaissance Hotels International Corporation Limited (Philippine Branch)

**Country: Poland** 

LIM Joint Venture Ltd.

**Country: Portugal** 

Marriott Hotels International B.V. (Portugal Branch)

Penha Longa Hotel Management Company B.V. (Portuguese Branch)

**Country: Puerto Rico** 

MVCI Puerto Rico, Inc.

The Ritz-Carlton Hotel Company San Juan, Inc.

Country: Romania

Marriott International Hotels, Inc. Maryland USA Sucursala Bucuresti

(Romanian Branch)

Exhibit 21

Country: Russia

Intour Penta Ltd. - (Joint Venture)

Limited Liability Company "CYBM Voznesenkry Hotel Leasing" Limited Liability Company "Renaissance Samara Hotel Leasing" Limited Liability Company "Renaissance St. Petersburg Hotel Leasing"

The Ritz-Carlton Hotel Company B.V. (Russian Branch)

**Country: Saint Kitts and Nevis** 

Marriott St. Kitts Licensing Company Limited Marriott St. Kitts Management Company, Inc.

MVCI St. Kitts Company Limited

**Country: Saint Lucia** 

The Ritz-Carlton Hotel Company of St. Lucia Limited

**Country: Singapore** 

International Luxury Hotels (Singaport) Pte. Limited

MVCI Asia Pacific Pte. Ltd.

The Ritz-Carlton Hotel Company of Singapore PTE LTD.

The Ritz-Carlton Hotel Company Sales and Marketing B.V. (Singapore Branch) The Ritz-Carlton Residential Management Company of Singapore Pte. Limited

**Country: South Korea** 

Marriott Hotels International B.V. (Pusan, South Korea Branch)
Marriott Hotels International B.V. (Seoul MEA Branch)
Marriott Hotels International B.V. (Seoul, South Korea Branch)
Renaissance Services B.V. (South Korea Branch - Liaison Office)
The Ritz-Carlton Hotel Company B.V. (Seoul, South Korea Branch)

**Country: Spain** 

Marriott Hotels International Limited (Spain Branch)

Marriott Hotels, S.L. MVCI Espana, S.L. MVCI Holidays, S.L. MVCI Mallorca, S.L. MVCI Management, S.L.

MVCI Playa Andaluza Holidays, S.L.

R-C Spain, S.L.

Spa Son Antem, S.L. (Joint Venture)

**Country: Switzerland** 

International Hotel Licensing Company S.a.r.l. Luxembourg, Zurich Branch

Luxury Hotels International Management Switzerland GmbH

Marriott (Schweitz) GmbH

Marriott Switzerland Licensing Company II Sarl Marriott Switzerland Licensing Company Sarl

**Country: Thailand** 

Maikhao Ownership (Thailand) Ltd.

Maikhao Vacation Villas Limited - (Joint Venture)

Marriott Hotels (Thailand) Limited

Marriott Hotels (Thailand) Limited (Bangkok Thailand Branch)

MVCI (Thailand) Limited

Renaissance Holdings (Thailand) Limited

Renaissance Management Company (Thailand) Ltd

Exhibit 21

Country: Thailand cont'd.

Ritz-Carlton Holdings (Thailand) Ltd.

South Land Ltd.

The Ritz-Carlton Management Company (Thailand) Ltd.

**Country: Trinidad and Tobago** 

Marriott Trinidad & Tobago Limited

**Country: Tunisia** 

Renaissance Services B.V. (Tunisia Branch)

**Country: Turks and Caicos Isl** 

Marriott Turks & Caicos Licensing Company Limited Renaissance Caribbean Limited (Turks & Caicos Branch) The Ritz-Carlton Hotel Company of Turks & Caicos Ltd

**Country: United Arab Emirates** 

Marriott Hotels and Catering (Holdings) Limited (United Arab Emirates Branch)

Marriott Ownership Resorts, Inc. (Dubai Branch)

MVCI Europe Limited (United Arab Emirates – Sharjah Branch)

Renaissance Services B.V. (Dubai Branch)

**Country: United Kingdom** 

Cheshunt Hotel Limited

Cheshunt Hotel Operating Company Limited

Cheshunt Marriott Limited Consolidated Supplies Limited Financiere 47 Park Street Limited Fortyseven Park Street Limited GH Hotel Operating Company Limited

Marriott European Hotel Operating Company Limited - (Joint Venture)

Marriott Hotels (Reading) Limited

Marriott Hotels and Catering (Holdings) Limited

Marriott Hotels International Limited

Marriott Hotels Limited

Marriott In-Flight Services Limited

Marriott International Design & Construction Services, Inc. (UK Branch)

Marriott Restaurants Limited

Marriott UK Acquisition Company Limited Marriott UK Group Company Limited Marriott UK Management Company Limited Marriott V&A Hotel Operating Company Limited

MGRC Management Limited MGRC Marketing Limited MVCI Europe Limited

MVCI Management (Europe) Limited

Rarework Limited

Renaissance Manchester Hotel Operating Company Limited Renaissance Services B.V. (United Kingdom Branch) Ronevsorg Hotel Operating Company Limited

The Ritz-Carlton Hotel Limited

Exhibit 21

Country: Venezuela

Desarrolos Hotelco, C.A.

Marriott Hotels International B.V. (Caracas, Venezuela Branch)

Country: Virgin Islands-BR

Costa Del Sol Holdings LTD.
Dominican Hotels (B.V.I.) Ltd.
Ramasia International Limited
RC Abaco Holding Company Limited
RC Hotel Holding Company Limited
RC Rose Holding Company Limited

Renaissance Hotels International Corporation Limited Renaissance Hotels Marketing Services Limited

Country: Virgin Islands-U.S.

Cabrita Partners LLC
Marriott Ownership Resorts (St. Thomas), Inc.
Marriott U.S. Virgin Islands Licensing Company LLC
Muller Bay Holdings, LLC
RC Hotels (Virgin Islands), Inc.
The Ritz-Carlton Club, St. Thomas, Inc.

Exhibit 21

# Marriott International, Inc. Domestic Subsidiaries – State of Incorporation

State: Arizona

Camelback Country Club, Inc. (d/b/a Camelback Golf Club)

Marriott Rewards, LLC Vestar-Athens Tuson, L.L.C.

State: California

SJMEC, Inc. SJMFB, LLC

State: Delaware

Addison SHS, LLC

Aeropuerto Shareholder, Inc.

Baltimore Marriott Inner Harbor, L.L.C.

Camelback Properties Inn, Inc. Capitol Employment Services, Inc.

CBM Annex, Inc.

Centerline Georgia Investor LLC

Charleston Marriott, LLC

Chicago Hotel Services, LLC

City Center Annex Tenant Corporation

CNL Philadelphia Annex, LLC

Corporate General, Inc.

Courtyard Management Corporation

CR14 Tenant Corporation CRTM17 Tenant Corporation CTYD III Corporation Desert Ridge Resort, LLC

Detroit CY, LLC

Detroit Hotel Services, LLC

Detroit MHS, LLC

e-CRM Central, LLC

East Side Hotel Services, Inc.

Essex House Condominium Corporation

ExecuStay, LLC Fairfield FMC, LLC FMPC Acquisition, LLC

Franchise System Holdings, Inc.

Host Restaurants, Inc. Hunt Valley Courtyard, Inc.

Kauai Lagoons Holdings LLC LAX Properties, LLC LF, South Beach, LLC

Luxury Finance, LLC

Luxury Hotels International Design & Construction Services, Inc.

Marriott Acquisition 2002 Subsidiary, LLC

Marriott Acquisition 2002, LLC Marriott College Food, L.L.C.

Marriott Crystal City Manager, Inc.

Marriott CS Holdings, LLC

Marriott Distribution Holding Co.

Marriott Distribution Services, Inc.

Marriott Fifth Avenue, LLC

Marriott Hotel Services, Inc.

Marriott Hurghada Management, Inc.

Marriott International Administrative Services, Inc.

Exhibit 21

# Marriott International, Inc. Domestic Subsidiaries – State of Incorporation

State: Delaware cont'd.

Marriott International Capital Corporation

Marriott International Construction Services, Inc.

Marriott International Design & Construction Services, Inc.

Marriott International JBS Corporation

Marriott International Resorts, L. P.

Marriott International, Inc.

Marriott Kauai Ownership Resorts, Inc.

Marriott Magenta Holding Company, Inc.

Marriott Market Street Hotel, Inc.

Marriott Mexico City Partnership G.P.

Marriott Mexico City Partnership, G.P.

Marriott Mirage City Management, Inc.

Marriott Oak Brook Hills Services, LLC

Marriott Overseas Owners Services Corporation

Marriott Ownership Resorts Procurement, LLC

Marriott Ownership Resorts, Inc.

Marriott P.R. Management Corporation

Marriott Payroll Services, LLC

Marriott Ranch Properties, Inc.

Marriott Resort at Seaview, LLC

Marriott Resorts Sales Company, Inc.

Marriott Resorts, Travel Company, Inc.

Marriott Rewards Subsidiary, Inc.

Marriott Rewards, Inc.

Marriott Senior Holding Co.

Marriott Sharm El Sheikh Management, Inc.

Marriott Signal Capital, L.L.C.

Marriott Two Flags Member LLC

Marriott Two Flags, LP

Marriott Vacation Club Ownership II LLC

Marriott Vacation Club Ownership LLC 2002-1

Marriott Vacation Club Ownership, LLC

Marriott Vacation Properties of Florida, Inc.

Marriott Worldwide Management, Inc.

Marriott Worldwide Payroll, LLC

Marriott Worldwide Reservation Services, LLC

Marriott Worldwide Sales and Marketing, Inc.

Marriott's Desert Springs Development Corporation

Marriott's Greenbelt Hotel Services, Inc.

MC Lodging Investment Opportunities, Inc.

MENYC, LLC

MH Kapalua Venture, LLC

MHS Guam, Inc.

MHSFR II, LLC

MHSFR, LLC

MHSI Hawaii, LLC.

MI Boston Leaseco, LLC

MI CBM Investor, LLC

MI Distribution, LLC

MI Finance Company

MI Georgia Credits, LLC

MI GILVN, LLC

MI Holding, L.P.

Exhibit 21

# Marriott International, Inc. **Domestic Subsidiaries - State of Incorporation**

### State: Delaware cont'd.

MI Member, LLC

MI Procurement Holdings, LLC

MI Tenant LLC

MI TH4 INVESTOR, LLC

MI Western Investment, LLC

MICC SPE I Corp.

MICC (California), LLC

MORI Golf (Kauai), LLC

MORI Las Vegas I, LLC

MORI Member (Kauai), LLC

MORI Residences, Inc.

MORI SPC 2005-1 Corp.

MORI SPC 2005-2 Corp.

MORI SPC 2006-1 Corp.

MORI SPC 2006-2 Corp.

MORI SPC 2007-1 Corp.

MORI SPC Corp.

MORI SPC II, Inc.

MORI SPC III CORP.

MORI SPC Series Corp.

MORI SPC V Corp.

MORI SPC VI Corp.

MORI SPC VII Corp.

MRC I Funding Corporation

MTMG Corporation

MTSC, INC.

MVCO 2005-1 LLC

MVCO 2005-2 LLC

MVCO 2006-1 LLC

MVCO 2006-2 LLC

MVCO 2007-1 LLC

MVCO Series LLC

North Drury Lane Productions, LLC

R.C. Chronicle Building L.P.

RBF, LLC

RC Marriott II, Inc.

RC Marriott III, Inc.

RC Marriott, Inc.

RC Paradise Valley Development, LLC

RC Social Club of San Francisco, Inc.

RC-UK, Inc.

RCC (GP) Holdings LLC

RCC (LP) Holdings L.P.

RCDC Chronicle LLC

REN Boston Hotel Management LLC

REN Boston LP

REN Boston Waterfront Hotel, LLC

REN Hollywood, LLC

Renaissance Cleveland Hotel, LLC

Renaissance Cleveland IOSA, LLC

Renaissance Hollywood Payroll Company, LLC

Renaissance Hotel Holdings, Inc.

Renaissance Hotel Management Company, LLC

Exhibit 21

# Marriott International, Inc. Domestic Subsidiaries – State of Incorporation

State: Delaware cont'd.

Renaissance Hotel Operating Company

Renaissance International, Inc.

Renaissance Oakbrook Hotel, LLC

Renaissance Services, Inc.

Renaissance St. Louis Grand, LLC

Renaissance St. Louis Suites, LLC

Residence Inn by Marriott, LLC

RHG Investments, LLC

RHHI Acquisition Corp.

RHHI Investment Corp.

RHOC Consolidation, LLC

RI Mill Road Alexandria LLC

Ritz-Carlton (Virgin Islands), Inc.

RST4 Tenant LLC

SC Orlando, L.L.C.

Schaumberg/Oakbrook Marriott Hotels, LLC

Senior Living Limited Partnership

Shady Grove Courtyard, Inc.

SpringHill SMC, LLC

Square 369 Hotel Associates, LLC

Synthetic American Fuel Enterprises I, LLC

Synthetic American Fuel Enterprises II, LLC

The Cobalt Travel Company, L.L.C.

The Ritz-Carlton Development Company, Inc.

The Ritz-Carlton Hotel Company of Puerto Rico, Inc.

The Ritz-Carlton Hotel Company, L.L.C.

The Ritz-Carlton International Construction Services, Inc.

The Ritz-Carlton International Holdings II, LLC

The Ritz-Carlton International Holdings III, LLC

The Ritz-Carlton International Holdings, Inc.

The Ritz-Carlton International Licensing Company, LLC

The Ritz-Carlton Management Company, L.L.C.

The Ritz-Carlton Mexico Holdings, LLC

The Ritz-Carlton Sales Company, Inc.

The Ritz-Carlton Title Company, Inc. TownePlace Management, LLC

WEC 99C-1 LLC

WEC 99C-10 LLC

WEC 99C-11 LLC

WEC 99C-12 LLC

WEC 99C-13 LLC

WEC 99C-14 LLC

WEC 99C-2 LLC WEC 99C-3 LLC

WEC 99C-4 LLC

WEC 99C-5 LLC

WEC 99C-6 LLC

WEC 99C-7 LLC

WEC 99C-8 LLC

WEC 99C-9 LLC

Exhibit 21

# Marriott International, Inc. **Domestic Subsidiaries - State of Incorporation**

State: Florida

Eagle Tree Construction, LLC Marriott Resorts Title Company, Inc.

State: Georgia

The Dining Room Corporation

State: Hawaii

F. L. Insurance Corporation K D Kapule LLC Kauai Lagoons LLC Kauai Lagoons Vessels LLC KB Hotel Operator Inc. Marquis Insurance Corporation

State: Kansas

Kansas Hospitality Services, Inc.

State: Maryland

C1 Maryland Land Business Trust Columbia Courtyard, Inc. Marriott International Hotels, Inc. Marriott Worldwide Corporation MHS Realty Sales, Inc. MI Fulfillment Services, LLC MII Conference Center, Inc.

State: Massachusetts

MI Hotels of Massachusetts, Inc.

State: Nevada

Hard Carbon, LLC

Heavenly Resort Properties LLC MI Hotels of Las Vegas, Inc.

State: Oregon

Synthetic American Fuel Enterprises Holdings, Inc.

**State: South Carolina** 

Marriott Resorts Hospitality Corporation

State: Texas

Chaparral Club (Non-profit) Hospitality International, Inc. Marriott Claims Services Corporation MHSI Conference Centers of Texas, Inc. Residence Inn Club, Inc.

The Finish Line Club (Non-profit) The Gazebo Club (Non-profit)

The Legacy Park Club

State: Utah

Gambits, A Nonprofit Corporation (Incorporated Club)

State: West Virginia

West Virginia Marriott Hotels, Inc.

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Alabama

 Entity Name
 Assumed Name

 Marriott Ownership Resorts, Inc.
 Marriott Vacation Club International (MVCI)

State: Arizona

Entity Name	Assumed Name
Courtyard Management Corporation	Camelback CbM
Courtyard Management Corporation	Chandler Courtyard by Marriott
Courtyard Management Corporation	Phoenix Airport CbM
Courtyard Management Corporation	Phoenix Mesa CbM
Courtyard Management Corporation	Phoenix Metro Center CbM
Courtyard Management Corporation	Scottsdale CbM
Courtyard Management Corporation	Scottsdale Downtown Courtyard
Courtyard Management Corporation	Tuscon CbM
Desert Ridge Resort, LLC	Wild Fire Golf Course
Fairfield FMC, LLC	Flagstaff FibM
Fairfield FMC, LLC	Phoenix FibM
Fairfield FMC, LLC	Scottsdale FibM
Marriott Hotel Services, Inc.	Marriott Camelback Inn Resort
Marriott Hotel Services, Inc.	Scottsdale Marriott At McDowell Mountains
Marriott International, Inc.	Mountain Shadows Resort, Mountain Shadows,
	Marriott's Mountain Shadows Resort
Marriott Ownership Resorts, Inc.	MVCI
Marriott Rewards, Inc.	Marriott Rewards II, Inc.
Residence Inn by Marriott, LLC	Flagstaff RI
Residence Inn by Marriott, LLC	Phoenix Airport-Tempe RI

State: Arkansas

Residence Inn by Marriott, LLC

Residence Inn by Marriott, LLC

Residence Inn by Marriott, LLC

Entity Name	Assumed Name
Courtyard Management Corporation	Little Rock CbM

Phoenix RI

Tucson RI

Scottsdale RI

State: California

Entity Name	Assumed Name
Courtyard Management Corporation	Courtyard by Marriott
Courtyard Management Corporation	Novato Courtyard by Marriott
Courtyard Management Corporation	San Francisco Oyster Point Courtyard
CTYD III Corporation	Courtyard by Marriott
Fairfield FMC, LLC	Anaheim Fairfield Inn
Fairfield FMC, LLC	Buena Park FibM
Fairfield FMC, LLC	Fairfield Inn by Marriott
Fairfield FMC, LLC	Ontario FibM
Fairfield FMC, LLC	Placentia FibM
Fairfield FMC, LLC	Rancho Cordova FibM
Marriott Hotel Services, Inc.	Anaheim Marriott
Marriott Hotel Services, Inc.	Costa Mesa Marriott Suites
Marriott Hotel Services, Inc.	La Jolla Marriott Hotel
Marriott Hotel Services, Inc.	Los Angeles Airport Marriott
Marriott Hotel Services, Inc.	Marriott's Desert Springs Resort and Spa

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

# State: California cont'd.

Entity Name	Assumed Name
Marriott Hotel Services, Inc.	Monterey Marriott Hotel
Marriott Hotel Services, Inc.	Napa Valley Marriott Hotel
Marriott Hotel Services, Inc.	Newport Beach Marriott Hotel
Marriott Hotel Services, Inc.	Rancho Las Palmas Marriott Resort
Marriott Hotel Services, Inc.	Santa Clara Marriott Hotel
Marriott International, Inc.	Courtyard by Marriott
Marriott International, Inc.	Irvine Marriott Hotel
Marriott International, Inc.	La Jolla Marriott Hotel
Marriott International, Inc.	Los Angeles Airport Marriott
Marriott International, Inc.	Manhattan Beach Marriott
Marriott International, Inc.	San Diego Marriott Hotel Marina
Marriott Kauai Ownership Resorts, Inc.	Marriott Vacation Club International (MVCI)
Marriott Ownership Resorts, Inc.	Grand Residences by Marriott
Marriott Ownership Resorts, Inc.	MVCI
Marriott Ownership Resorts, Inc.	MVCI
Marriott Resorts Hospitality Corporation	Marriott Vacation Club International (MVCI)
Marriott Resorts, Travel Company, Inc.	Marriott Vacation Club International (MVCI)
Renaissance Hotel Management Company	The Lodge at Sonoma
Residence Inn by Marriott, LLC	Anaheim Residence Inn
Residence Inn by Marriott, LLC	Anaheim RI
Residence Inn by Marriott, LLC	Arcadia Residence Inn
Residence Inn by Marriott, LLC	Arcadia Residence Inn
Residence Inn by Marriott, LLC	Arcadia RI
Residence Inn by Marriott, LLC	Bakersfield Residence Inn
Residence Inn by Marriott, LLC	Bakersfield RI
Residence Inn by Marriott, LLC	Beverly Hills Residence Inn
Residence Inn by Marriott, LLC	Beverly Hills Residence Inn
Residence Inn by Marriott, LLC	Beverly Hills RI
Residence Inn by Marriott, LLC	Costa Mesa Residence Inn
Residence Inn by Marriott, LLC	Costa Mesa RI
Residence Inn by Marriott, LLC	Fountain Valley Residence Inn
Residence Inn by Marriott, LLC	Fountain Valley RI
Residence Inn by Marriott, LLC	Fremont RI
Residence Inn by Marriott, LLC	Irvine Residence Inn
Residence Inn by Marriott, LLC	Irvine RI
Residence Inn by Marriott, LLC	Kearney Mesa Residence Inn
Residence Inn by Marriott, LLC	Kearney Mesa RI
Residence Inn by Marriott, LLC	La Jolla Residence Inn
Residence Inn by Marriott, LLC	La Jolla RI
Residence Inn by Marriott, LLC	Long Beach Residence Inn
Residence Inn by Marriott, LLC	Long Beach RI
Residence Inn by Marriott, LLC	Manhattan Beach Residence Inn
Residence Inn by Marriott, LLC	Manhattan Beach Residence Inn
Residence Inn by Marriott, LLC	Manhattan Beach RI
Residence Inn by Marriott, LLC	Mira Mesa Residence Inn
Residence Inn by Marriott, LLC	MIRI Mesa Residence Inn
Residence Inn by Marriott, LLC	Mountain View Residence Inn
Residence Inn by Marriott, LLC	Mountain View RI

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

### State: California cont'd.

Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC The Ritz-Carlton Hotel Company, LLC The Ritz-Carlton Hotel Company, LLC

Residence Inn by Marriott, LLC

Placentia Residence Inn

Placentia RI

Pleasant Hills Residence Inn

Pleasant Hills RI Rancho Bernardo RI Residence Inn Cypress

Sacramento-Natomas Residence Inn

Sacramento-Natomas RI San Jose Residence Inn

San Jose RI

San Mateo Residence Inn

San Mateo RI San Ramon RI

San Ramon Residence Inn Silicon Valley I RI

Silicon Valley II Residence Inn

Silicon Valley II RI Torrance Residence Inn Torrance Residence Inn

Torrance RI
Ritz-Carlton Hotel

The Ritz-Carlton, Laguna Niguel

#### State: Colorado

Courtyard Management Corporation Courtyard Management Corporation Courtyard Management Corporation Marriott Hotel Services, Inc.

Marriott Kauai Ownership Resorts, Inc.

Marriott Ownership Resorts, Inc.
Marriott Ownership Resorts, Inc.

Marriott Resorts Hospitality Corporation Marriott Resorts Sales Company, Inc.

RBF, LLC

Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC Residence Inn by Marriott, LLC The Ritz-Carlton Hotel Company, L.L.C. Boulder CbM Denver Airport CbM Denver SE CbM

Denver West Marriott Hotel

MVCI

Grand Residences by Marriott

MVCI MVCI

Marriott Vacation Club International (MVCI)

RBF-Jupiter, LLC Boulder RI

Colorado Springs RI Denver Downtown RI Denver South RI The Ritz-Carlton, Denver

### **State: Connecticut**

# Entity Name

Courtyard Management Corporation Courtyard Management Corporation

Fairfield FMC, LLC Marriott Hotel Services, Inc. Marriott Ownership Resorts, Inc.

#### **Assumed Name**

Hartford CbM Norwalk CbM

Hartford Airport FibM (Windsor/Windsor Lock) Stamford Marriott Hotel (Stamford & Rocky Hill)

MVCI

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Delaware

**Entity Name Assumed Name** Courtyard Management Corporation Wilmington CbM Fairfield FMC, LLC Wilmington FibM Marriott Ownership Resorts, Inc. MVCI

Residence Inn by Marriott, LLC Wilmington RI

State: District Of Columbia

**Entity Name** Assumed Name

Marriott Hotel Services, Inc JW Marriott on Pennsylvania Avenue Renaissance Hotel Operating Company Renaissance Washington DC Hotel

The Ritz-Carlton Hotel Company, L.L.C. The Fairfax Club

State: Florida

Entity Name Assumed Name Courtvard Management Corporation Courtyard by Marriott Courtyard Management Corporation Courtyard Café Courtyard Management Corporation Courtyard Lobby Lounge Courtyard Management Corporation Courtyard Orlando Lake Buena Vista in the Marriott Village Courtyard Management Corporation Courtyard Pool Bar & Grill CTYD III Corporation Courtyard by Marriott Fairfield FMC, LLC Fairfield Inn Café Fairfield FMC, LLC Fairfield Inn Orlando Lake Buena Vista in the Marriott Village Fairfield FMC, LLC Fairfield Inn Pool Bar & Grill Fairfield FMC, LLC Gainesville FibM Fairfield FMC, LLC Miami West FibM Fairfield FMC, LLC Orlando International Drive FibM Fairfield FMC, LLC Winter Park FibM Marriott Hotel Services, Inc. 3030 Ocean Marriott Hotel Services, Inc. Cafe Waterside Marriott Hotel Services, Inc. Champions Marriott Hotel Services, Inc. Deco Blue Marriott Hotel Services, Inc. Fort Lauderdale Marina, Tampa Airport Marriott Hotel Services, Inc. IL Terrazzo Le Grande Blue Marriott Hotel Services, Inc. Marriott Hotel Services, Inc. Marriott South Beach Marriott Hotel Services, Inc. Miami Beach Marriott At South Beach Marriott Hotel Services, Inc. Miami International Airport Marriott

Marriott Hotel Services, Inc. Riva Restaurant SPA Waterside

Marriott Hotel Services, Inc. Marriott Hotel Services, Inc. Tampa Marriott Waterside

Marriott Hotel Services, Inc. The Club at Marriott's Harbor Beach Resort & Spa Marriott Hotel Services, Inc. The Spa at Marriott's Harbor Beach Resort

Marriott Hotel Services, Inc. The Spa Café Marriott Hotel Services, Inc. Tranquility Marriott International, Inc. Bleu Bar & Food Marriott International, Inc. Citron Patisserie

Marriott International, Inc. Citron, An American Brasserie

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

#### State: Florida cont'd.

Marriott International, Inc. Destinations by Marriott Marriott International, Inc. Doral Golf Resort & Spa Marriott International, Inc. Fairways Pub Marriott International, Inc. Hawk's Landing Steakhouse & Grille Marriott International, Inc. JW Marriott Orlando Grande Marriott International, Inc. Marriott International, Inc. Quench Bar & Grill Marriott International, Inc. Tampa Marriott Waterside Marriott International, Inc. The Ritz Carlton Golf Club Marriott International, Inc. The Ritz Carlton Spa-Cartia Masion de Beaute Marriott International, Inc. The Ritz-Carlton Orlando Grande Lakes Marriott International, Inc. The Signature Shop Marriott International, Inc. The Vineyard Grill Marriott International, Inc. Vitale Spa Café Marriott Ownership Resorts, Inc. Crowne Plaza Oceanfront Singer Island Marriott Ownership Resorts, Inc. Faldo Golf Institute by Marriott Marriott Ownership Resorts, Inc. Flagler's Marriott Ownership Resorts, Inc. Grande Pines Golf Club Marriott Ownership Resorts, Inc. Horizons by Marriott Vacation Club (HMVC) Marriott Ownership Resorts, Inc. International Golf Club Marriott Ownership Resorts, Inc. **MVCI** Marriott Ownership Resorts, Inc. The Pool Patio and Grill Marriott Resorts Hospitality Corporation Horizons By Marriott Vacation Club Marriott Resorts Hospitality Corporation Marriott's Legends Edge at Bay Point Marriott Resorts Hospitality Corporation Marriott's Villas at Doral Marriott Resorts Hospitality Corporation MVCI Marriott Resorts, Travel Company, Inc. MVCI RBF, LLC The Ritz-Carlton Golf Club & Spa, Jupiter Renaissance Hotel Management Company Eden Roc, A Renaissance Resort & Spa Renaissance Hotel Management Company Flagler's Renaissance Hotel Management Company Renaissance Singer Island Hotel Renaissance Hotel Management Company Tampa Renaissance Hotel Renaissance Hotel Management Company The Pool Patio and Grill The Renaissance Tampa International Plaza Hotel Renaissance Hotel Management Company Renaissance Hotel Operating Company Renaissance Orlando Resort Residence Inn by Marriott, LLC Boca Raton RI Residence Inn by Marriott, LLC Jacksonville RI Residence Inn by Marriott, LLC Lake Buena Vista RI Residence Inn by Marriott, LLC Pensacola RI Residence Inn by Marriott, LLC Residence Inn Melbourne Residence Inn by Marriott, LLC St. Petersburg RI SpringHill SMC, LLC Springhill Pool Bar & Grill SpringHill SMC, LLC Springhill Suites Orlando Lake Buena Vista in the Marriott Village

Springhill Suites Seasons

Americana Restaurant

DiLido Beach Club

Aria

Bun

Cantina

Cioppino

SpringHill SMC, LLC

The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton Hotel Company, L.L.C.

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

#### State: Florida cont'd.

The Ritz-Carlton Hotel Company, L.L.C. Lemonia The Ritz-Carlton Hotel Company, L.L.C. One L.R. The Ritz-Carlton Hotel Company, L.L.C. Pool Side Grill The Ritz-Carlton Hotel Company, L.L.C. Rum Bar The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton Golf Resort Naples The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton, Amelia Island The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton, Key Biscayne The Ritz-Carlton, Key Biscayne Spa The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton, South Beach The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton Hotel Company, L.L.C. The Sand Bar Grill TownePlace Management, LLC Miami Airport West Towneplace Suites

Miami Lakes Towneplace Suites

Atlanta Airport CbM

Courtyard Management Corporation

TownePlace Management, LLC

# State: Georgia

#### Assumed Name **Entity Name**

Courtyard Management Corporation Atlanta Airport South CbM Courtyard Management Corporation Atlanta Delk Road CbM Courtyard Management Corporation Atlanta Perimeter CbM Courtyard Management Corporation Augusta CbM Courtyard Management Corporation Columbus CbM Courtyard Management Corporation Cumberland Center CbM Courtyard Management Corporation Executive Park CbM Courtyard Management Corporation Gwinnet Mall CbM Courtyard Management Corporation Jimmy Carter CbM Courtyard Management Corporation Macon CbM Courtyard Management Corporation Midtown Atlanta CbM Courtyard Management Corporation Northlake CbM Courtyard Management Corporation Peachtree Corners CbM Courtyard Management Corporation Peachtree-Dunwoody CbM

Courtyard Management Corporation Roswell CbM Courtyard Management Corporation Savannah CbM Courtyard Management Corporation Windy Hill CbM, Fairfield FMC, LLC

Atlanta Gwinnett Mall FibM Fairfield FMC, LLC Atlanta Northlake FibM Marriott Hotel Services, Inc. Atlanta Norcross Marriott Hotel Marriott Hotel Services, Inc. Atlanta Perimeter Center Hotel Marriott Ownership Resorts, Inc. **MVCI** 

Marriott Resorts Hospitality Corporation MVCI Marriott Resorts, Travel Company, Inc. MVCI Residence Inn by Marriott, LLC Atlanta Airport RI

Residence Inn by Marriott, LLC Atlanta Alpharetta RI Residence Inn by Marriott, LLC Atlanta Buckhead RI Residence Inn by Marriott, LLC Atlanta Midtown RI Residence Inn by Marriott, LLC Atlanta Perimeter Mall RI

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

# State: Hawaii

Entity Name	Assumed Name
KB Hotel Operator Inc.	Kapalua Bay, A Renaissance Resort
Marriott Hotel Services, Inc.	J.W. Marriott Ihilani Resort & Spa
Marriott Hotel Services, Inc.	Koloh's Beach Bar & Grill
Marriott Hotel Services, Inc.	Kuhio Beach Grill
Marriott Hotel Services, Inc.	Marriott's Waiohai Beach Club
Marriott Hotel Services, Inc.	Maui Marriott Resort and Ocean Club
Marriott Hotel Services, Inc.	Moana Terrace
Marriott Hotel Services, Inc.	Waikiki Beach Hotel
Marriott Hotel Services, Inc.	Waikiki Beach Marriott Resort & Spa
Marriott Hotel Services, Inc.	Waikoloa Beach Marriott Resort
Marriott Hotel Services, Inc.	Wailea Marriott Resort
Marriott International, Inc.	Maui Marriott Resort
Marriott International, Inc.	Waikiki Beach Hotel
Marriott International, Inc.	Waikiki Beach Marriott Hotel
Marriott Kauai Ownership Resorts, Inc.	MVCI - Registration Number: 234405
Marriott Ownership Resorts, Inc.	Grand Residences by Marriott
Marriott Ownership Resorts, Inc.	Marriott's Waiohai Beach Resort
Marriott Ownership Resorts, Inc.	MVCI
The Ritz-Carlton Hotel Company, L.L.C.	The Ritz-Carlton, Kapalua

# State: Illinois

Entity Name	Assumed Name
Courtyard Management Corporation	Arlington Heights CbM,
Courtyard Management Corporation	Arlington Heights South CbM
Courtyard Management Corporation	Chicago Downtown CbM
Courtyard Management Corporation	Chicago-Highland Park CbM
Courtyard Management Corporation	Chicago/Deerfield CbM
Courtyard Management Corporation	Chicago/Lincolnshire CbM
Courtyard Management Corporation	Glenview CbM
Courtyard Management Corporation	Naperville CbM
Courtyard Management Corporation	O'Hare CbM
Courtyard Management Corporation	Oakbrook Terrace CbM
Courtyard Management Corporation	Rockford CbM,
Courtyard Management Corporation	St. Charles CbM
Courtyard Management Corporation	Waukegan CbM
Courtyard Management Corporation	Wood Dale CbM
CTYD III Corporation	Courtyard by Marriott
Fairfield FMC, LLC	Bloomington/Normal FibM
Fairfield FMC, LLC	Chicago Lansing FibM
Fairfield FMC, LLC	Glenview FibM
Fairfield FMC, LLC	Peoria FibM
Fairfield FMC, LLC	Rockford FibM
Fairfield FMC, LLC	Willowbrook FibM
Marriott Hotel Services, Inc.	Chicago Deerfield Marriott Suites
Marriott Hotel Services, Inc.	Chicago Marriott Downtown Hotel
Marriott Hotel Services, Inc.	Chicago Marriott Oakbrook Hotel
Marriott Hotel Services, Inc.	Lincolnshire Catering
Marriott Hotel Services, Inc.	Oak Brook Hills Marriott Resort

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Illinois cont'd.

Marriott International, Inc.

Chicago Marriott O'Hare

Marriott Ownership Resorts, Inc. HMVCI
Marriott Ownership Resorts, Inc. MVCI

Renaissance Hotel Operating Company

Renaissance Oak Brook Hotel, Renaissance Chicago Hotel

Residence Inn by Marriott, LLC
Deerfield RI

State: Indiana

Entity Name Assumed Name

Courtyard Management Corporation Courtyard by Marriott
CTYD III Corporation Courtyard by Marriott

Fairfield FMC, LLC
Fort Wayne FibM
Fairfield FMC, LLC
Indianapolis Castleton FibM
Fairfield FMC, LLC
Indianapolis College Park FibM
Pacidnes In by Marriett, LLC
Fort Wayne PL

Residence Inn by Marriott, LLC Fort Wayne RI
Residence Inn by Marriott, LLC Indianapolis North RI

State: Iowa

Entity Name Assumed Name

Courtyard Management CorporationDes Moines/Clive CbMCourtyard Management CorporationQuad Cities CbMCTYD III CorporationCourtyard by Marriott

Fairfield FMC, LLC Cedar Rapids FibM Fairfield FMC, LLC Des Moines FibM

State: Kentucky

Entity Name Assumed Name

Courtyard Management Corporation

CTYD III Corporation

Fairfield FMC, LLC

Florence FibM

Fairfield FMC, LLC

Louisville East FibM

Marriott Ownership Resorts, Inc.

Marriott Ownership Resorts, Inc.

Marriott Resorts Hospitality Corporation

Marriott Resorts, Travel Company, Inc.

MVCI One

Marriott Resorts, Travel Company, Inc.

MVCI Two

Residence Inn by Marriott, LLC

Lexington North RI

Residence Inn by Marriott, LLC

Louisville RI

State: Louisiana

 Entity Name
 Assumed Name

 Courtyard Management Corporation
 Baton Rouge CbM

Courtyard Management Corporation Courtyard by Marriott (Metarie, LA)

CTYD III Corporation Courtyard by Marriott

Renaissance Hotel Management Company Renaissance Pere Marquette Hotel

Residence Inn by Marriott, LLC Bossier City RI

SpringHill SMC, LLC Springhill Suites by Marriott

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Maine

Entity Name Assumed Name

Fairfield FMC, LLC Portland FibM

Marriott Resorts Title Company, Inc.

Marriott Resorts Title, Inc.

State: Maryland

Entity Name Assumed Name

Courtyard Management Corporation Courtyard by Marriott
CTYD III Corporation Courtyard by Marriott
Marriott Hotel Services, Inc. Bethesda Marriott Hotel

Marriott Hotel Services, Inc.

Washington Gaithersburg Marriott Hotel

Marriott International, Inc.

Courtyard by Marriott

Marriott Ownership Resorts, Inc.

Marriott Resorts Hospitality Corporation

MVCI

Marriott Resorts, Travel Company, Inc.

Residence Inn by Marriott, LLC

Residence Inn by Marriott, LLC

Bethesda RI

State: Massachusetts

Entity NameAssumed NameCourtyard Management CorporationBoston Tremont Courtyard

Courtyard Management Corporation Lowell CbM
Courtyard Management Corporation Milford CbM
Courtyard Management Corporation Stoughton CbM
Marriott Hotel Services, Inc. Marriott Long Wharf

Marriott Ownership Resorts, Inc. MVCI
Marriott Resorts Hospitality Corporation MVCI

Renaissance Hotel Management Company Renaissance Boston Waterfront Hotel

Renaissance Hotel Operating Company
Residence Inn by Marriott, LLC
Residence Inn by Marriott, LLC.
Boston-Westborough RI
Residence Inn by Marriott, LLC
Cambridge RI

Residence Inn by Marriott, LLC

Danvers RI (aka/ Boston-North Shore)

The Ritz-Carlton Hotel Company, L.L.C.

The Ritz-Carlton Boston Common

The Ritz-Carlton Hotel Company, L.L.C.

The Ritz-Carlton, Boston

State: Michigan

 Entity Name
 Assumed Name

 Courtyard Management Corporation
 Auburn Hills CbM

Courtyard Management Corporation
Warren CbM

Detroit Hotel Services, LLC
Detroit MHS, LLC
Detroit MHS, LLC
Detroit Marriott at Renaissance Center
Detroit MHS, LLC

Fairfield FMC, LLC Detroit Airport FibM Fairfield FMC, LLC Detroit Madison FibM

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Michigan cont'd.

Fairfield FMC, LLC

Fairfield FMC, LLC

Detroit Warren FibM

Detroit West FibM

Fairfield FMC, LLC

Kalamazoo FibM

Marriott Hotel Services, Inc.Detroit Metro Airport Marriott HotelMarriott Hotel Services, Inc.Detroit Romulus Marriott HotelMarriott International, Inc.Courtyard by Marriott, Fairfield Inn

Residence Inn by Marriott, LLC Ann Arbor RI Residence Inn by Marriott, LLC Dearborn RI Residence Inn by Marriott, LLC East Lansing RI Residence Inn by Marriott, LLC Grand Rapids RI Residence Inn by Marriott, LLC Kalamazoo RI Residence Inn by Marriott, LLC Southfield Michigan RI Residence Inn by Marriott, LLC Troy Central RI Residence Inn by Marriott, LLC Troy South RI Residence Inn by Marriott, LLC Warren RI

State: Minnesota

Entity Name Assumed Name

Courtyard Management Corporation Eden Prairie CbM
Courtyard Management Corporation Mendota Heights CbM
Marriott Hotel Services, Inc. Minneapolis City Center Marriott Hotel

Marriott Ownership Resorts, Inc.HMVCMarriott Ownership Resorts, Inc.MVCIMarriott Resorts Hospitality CorporationMVCIMarriott Resorts, Travel Company, Inc.MVCIResidence Inn by Marriott, LLCEden Prairie RI

State: Missouri

Entity Name Assumed Name

Courtyard Management Corporation Creve Coeur CbM Courtyard Management Corporation Earth City CbM Courtyard Management Corporation Kansas City Airport CbM Courtyard Management Corporation South Kansas City CbM Courtyard Management Corporation St. Louis CbM Courtyard Management Corporation St. Louis-Westport CbM **CRTM17** Tenant Corporation St. Louis Airport Marriott Hotel Fairfield FMC, LLC St. Louis Hazelwood FibM Marriott Hotel Services, Inc. Kansas City Airport Marriott Marriott Hotel Services, Inc. St. Louis Airport Marriott Marriott Hotel Services, Inc. St. Louis Pavilion Marriott Hotel Marriott Hotel Services, Inc. Tan-Tar-A Marriott Resort

Marriott Ownership Resorts, Inc.

Marriott Ownership Resorts, Inc.

HMVC

HMVC

Renaissance Hotel Operating Company

Residence Inn by Marriott, LLC
Residence Inn by Marriott, LLC
St. Louis Galleria RI
Residence Inn by Marriott, LLC
St. Louis Westport RI

Exhibit 21

Renaissance St. Louis Hotel Airport

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Nebraska

Entity Name Assumed Name

Marriott Ownership Resorts, Inc. HMVC

Residence Inn by Marriott, LLC Omaha Central RI

State: Nevada

Entity NameAssumed NameCourtyard Management CorporationCourtyard by MarriottCTYD III CorporationCourtyard by MarriottFairfield FMC, LLCLas Vegas FibM

Marriott Ownership Resorts, Inc.

MVCI

Marriott Ownership Resorts, Inc.

Marriott's Grand Chateau

Marriott Resorts Hospitality Corporation MVCI
Marriott Resorts, Travel Company, Inc. MVCI

MI Hotels of Las Vegas, Inc.

Courtyard by Marriott

Grand Chateau

Las Vegas Marriott Suites

MI Hotels of Las Vegas, Inc.

Mi Hotels of Las Vegas, Inc.

Marriott's Grand Chateau

Marriott Resorts Hospitality Corporation

MI Hotels of Las Vegas, Inc.

Residence Inn by Marriott
Residence Inn by Marriott, LLC

Residence Inn by Marriott, LLC

Las Vegas Hughes Center
Las Vegas RI

The Ritz-Carlton Hotel Company, L.L.C.

The Ritz-Carlton, Lake Las Vegas

State: New Hampshire

Entity Name Assumed Name

Courtyard Management Corporation

Fairfield FMC, LLC

Merrimack Fairfield FibM

Marriott Hotel Services, Inc.

Marriott Ownership Resorts, Inc.

MVCI

MVCI

Marriott Resorts Hospitality Corporation MVCI
Marriott Resorts, Travel Company, Inc. MVCI

State: New Jersey

Entity Name Assumed Name

Courtyard Management Corporation

CTYD III Corporation

Execustay, LLC

Marriott Hotel Services, Inc.

arriott Hotel Services, Inc.
Marriott Hotel Services, Inc.
Marriott Hotel Services, Inc.
Marriott Hotel Services, Inc.
Park Ridge Marriott Hotel
Marriott Hotel Services, Inc.
Princeton Marriott Hotel
Marriott Hotel Services, Inc.
Somerset Marriott Hotel

Marriott Hotel Services, Inc.

The Lafayette Yard Marriott Conference Hotel

Courtyard by Marriott

Marriott Ownership Resorts, Inc.HMVCMarriott Ownership Resorts, Inc.MVCI

Marriott Resort at Seaview, LLC
Marriott Resorts Hospitality Corporation
Marriott Resorts Hospitality Corporation
Marriott Resorts By Marriott Vacation Club

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: New Jersey cont'd.

Marriott Resorts Hospitality Corporation MVCI
Marriott Resorts, Travel Company, Inc. MVCI

Residence Inn by Marriott, LLC Residence Inn by Marriott II, LLC

Springhill SMC, LLC Springhill SMC II, LLC

Towneplace Management, LLC State: New Mexico Towneplace Management II, LLC

State. I've w Mexico

Entity Name Assumed Name

Courtyard Management Corporation Courtyard by Marriott
CTYD III Corporation Courtyard by Marriott

Residence Inn by Marriott, LLC

Santa Fe RI, Albuquerque RI

State: New York

 Entity Name
 Assumed Name

 Courtyard Management Corporation
 Fishkill CbM

Courtyard Management Corporation Midtown East Courtyard Courtyard Management Corporation Poughkeepsie CbM Courtyard Management Corporation Rochester CbM Courtyard Management Corporation Rye CbM Courtyard Management Corporation Syracuse CbM Courtyard Management Corporation Tarrytown CbM Fairfield FMC, LLC Lancaster FibM

Fairfield FMC, LLC

Marriott Hotel Services, Inc.

Marriott Hotel Services, Inc.

LaGuardia Marriott Hotel

Long Island Marriott Hotel

Marriott Hotel Services, Inc.

New York Brooklyn Marriott

Marriott Hotel Services, Inc. New York Marriott Financial Center Hotel

Marriott Hotel Services, Inc.

Westchester Marriott Hotel
Marriott International, Inc.

LaGuardia Marriott

Marriott International, Inc.

Long Island Marriott Hotel and Conference Center
Marriott International, Inc.

Marriott's Wind Watch Hotel and Golf Club

Marriott International, Inc.

New York Marriott East Side

Marriott International, Inc.

Mew York Marriott Financial Center Hotel

Marriott International, Inc.

New York Marriott Marquis Hotel

Marriott International, Inc. Westchester Marriott

Marriott Kauai Ownership Resorts, Inc. MVCI
Marriott Ownership Resorts, Inc. HMVC
Marriott Ownership Resorts, Inc. MVCI

Marriott Resorts Hospitality Corporation Horizons by Marriott Vacation Club

Residence Inn by Marriott, LLC East Syracuse RI

The Ritz-Carlton Hotel Company, L.L.C.

The Ritz-Carlton, Westchester

State: North Carolina

Entity Name Assumed Name

Courtyard Management CorporationCharlotte Arrowood CbMCourtyard Management CorporationCharlotte South Park CbMCourtyard Management CorporationCharlotte University CbMCourtyard Management CorporationFayetteville CbMCourtyard Management CorporationGreensboro CbM

Courtyard Management Corporation Raleigh Airport CbM

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

#### State: North Carolina cont'd.

Courtyard Management CorporationRaleigh CbMCourtyard Management CorporationRaleigh-Cary CbMCTYD III CorporationCourtyard by MarriottFairfield FMC, LLCCharlotte Airport FibMFairfield FMC, LLCCharlotte Northeast FibMFairfield FMC, LLCDurham FibM

Fairfield FMC, LLC Fayetteville FibM
Fairfield FMC, LLC Greensboro Highpoint FibM

Fairfield FMC, LLC

Fairfield FMC, LLC

Raleigh Northeast FibM

Fairfield FMC, LLC

Rocky Mount FibM

Fairfield FMC, LLC

Wilmington FibM

Marriott Ownership Resorts, Inc.MVCIMarriott Resorts Hospitality CorporationMVCIMarriott Resorts, Travel Company, Inc.MVCI

Residence Inn by Marriott, LLC Charlotte North RI
Residence Inn by Marriott, LLC Durham RI, Greensboro RI

Residence Inn by Marriott, LLC
Residence Inn by Marriott, LLC
Winston-Salem RI
WEC 99C-8 LLC
WEC 99C-8 LLC

#### State: Ohio

### Entity Name Assumed Name

Courtyard Management CorporationBlue Ash CbMCourtyard Management CorporationDayton Mall CbMCourtyard Management CorporationDublin CbMCourtyard Management CorporationToledo CbMCourtyard Management CorporationWorthington CbMFairfield FMC, LLCAkron FibM

Fairfield FMC, LLC
Fairfield FMC, LLC
Cincinnati Sharonville FibM
Fairfield FMC, LLC
Cleveland Brook Park FibM
Fairfield FMC, LLC
Cleveland Willoughby FibM
Fairfield FMC, LLC
Columbus North FibM
Fairfield FMC, LLC
Columbus West FibM

Fairfield FMC, LLC

Dayton FibM

Fairfield FMC, LLC

Toledo Holland FibM

Fairfield FMC, LLC Toledo Holland FibM Marriott International, Inc. Fairfield Inn Marriott Ownership Resorts, Inc. **HMVC** MVCI Marriott Ownership Resorts, Inc. Marriott Resorts Hospitality Corporation **MVCI** Marriott Resorts, Travel Company, Inc. MVCI Residence Inn by Marriott, LLC Akron RIbM Residence Inn by Marriott, LLC Blue Ash RIbM Residence Inn by Marriott, LLC Cincinnati North RIbM Residence Inn by Marriott, LLC Columbus East RIbM Residence Inn by Marriott, LLC Dayton North RIbM Residence Inn by Marriott, LLC Dayton South RIbM Residence Inn by Marriott, LLC Dublin Ohio RIbM

Exhibit 21

Toledo RIbM

Residence Inn by Marriott, LLC

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Oklahoma

Entity NameAssumed NameCourtyard Management CorporationOklahoma City CbMMarriott Hotel Services, Inc.Oklahoma City Marriott HotelResidence Inn by Marriott, LLCOklahoma City RI

State: Oregon

Entity NameAssumed NameCourtyard Management CorporationPortland CbMMarriott Hotel Services, Inc.Portland Marriott HotelMarriott Ownership Resorts, Inc.MVCIMarriott Resorts Hospitality CorporationMVCIMarriott Resorts, Travel Company, Inc.MVCI

State: Pennsylvania

**Entity Name** Assumed Name Courtyard Management Corporation Devon CbM Philadelphia CbM Courtyard Management Corporation Courtyard Management Corporation Pittsburgh CbM Courtyard Management Corporation Valley Forge CbM Courtyard Management Corporation Willow Grove CbM Fairfield FMC, LLC Harrisburg West FibM Fairfield FMC, LLC Pittsburgh/Warrendale FibM Marriott Hotel Services, Inc. Philadelphia Airport Marriott Hotel Marriott Hotel Services, Inc. Philadelphia Marriott Hotel Philadelphia Marriott West Marriott Hotel Services, Inc. Residence Inn by Marriott, LLC Berwyn RI Residence Inn by Marriott, LLC Greentree RI Residence Inn by Marriott, LLC Philadelphia Airport RI Residence Inn by Marriott, LLC Willow Grove RI

State: Puerto Rico

Entity Name Assumed Name

MVCI Puerto Rico, Inc.

Marriott Vacation Club International

State: Rhode Island

Entity NameAssumed NameCourtyard Management CorporationMiddletown CbMMarriott Ownership Resorts, Inc.MVCI

**State: South Carolina** 

**Entity Name Assumed Name** Courtyard Management Corporation Columbia NW CbM Fairfield FMC, LLC Greenville FibM Fairfield FMC, LLC Hilton Head FibM Marriott Ownership Resorts, Inc. MVCI MVCI Marriott Resorts Hospitality Corporation Marriott Resorts, Travel Company, Inc. MVCI Residence Inn by Marriott, LLC Columbia RI

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

#### State: Tennessee

Assumed Name **Entity Name** Courtyard Management Corporation Brentwood CbM Courtyard Management Corporation Chattanooga CbM Memphis Airport CbM Courtyard Management Corporation Courtyard Management Corporation Nashville Airport CbM Courtyard Management Corporation Park Avenue, Memphis CbM Fairfield FMC, LLC Chattanooga FibM Fairfield FMC, LLC Jackson FibM Johnson City FibM Fairfield FMC, LLC Marriott Hotel Services, Inc. Nashville Airport Marriott Hotel Fairfield Inn Marriott International, Inc. HMVC Marriott Ownership Resorts, Inc. Residence Inn by Marriott, LLC Maryland Farms RIbM Residence Inn by Marriott, LLC Memphis RIbM

#### State: Texas

Entity Name	Assumed Name
Courtyard Management Corporation	Addison CbM
Courtyard Management Corporation	Addison/Quorum Courtyard
Courtyard Management Corporation	Arlington CbM
Courtyard Management Corporation	Bedford CbM
Courtyard Management Corporation	Courtyard Dallas Central Expressway
Courtyard Management Corporation	Courtyard Houston Downtown
Courtyard Management Corporation	Dallas North Park CbM
Courtyard Management Corporation	Dallas Northeast CbM
Courtyard Management Corporation	Dallas Stemmons CbM
Courtyard Management Corporation	DFW Courtyard North
Courtyard Management Corporation	Fort Worth CbM
Courtyard Management Corporation	Houston Downtown CbM
Courtyard Management Corporation	Las Colinas CbM
Courtyard Management Corporation	LBJ at Josey CbM
Courtyard Management Corporation	Plano CbM
Courtyard Management Corporation	San Antonio Airport CbM
Courtyard Management Corporation	San Antonio CbM
Courtyard Management Corporation	San Antonio Medical Center CbM
Courtyard Management Corporation	The Campbell Club
CTYD III Corporation	Courtyard by Marriott
Fairfield FMC, LLC	Arlington Fairfield Suites
Marriott Hotel Services, Inc.	Dallas Marriott Quorum
Marriott Hotel Services, Inc.	Houston Airport Marriott
Marriott Kauai Ownership Resorts, Inc.	MVCI
Marriott Ownership Resorts, Inc.	HMVC
Marriott Ownership Resorts, Inc.	MVCI
Marriott Resorts Hospitality Corporation	MVCI
Marriott Resorts, Travel Company, Inc.	MVCI
Residence Inn by Marriott, LLC	Dallas Central Expressway RI
Residence Inn by Marriott, LLC	Dallas Market Center RI
Residence Inn by Marriott, LLC	Houston Astrodome RI
Residence Inn by Marriott, LLC	Houston Clear Lake RI
Residence Inn by Marriott, LLC	Houston Southwest RI

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

State: Texas cont'd.

Residence Inn by Marriott, LLC Lubbock RI Lubbock RI

Residence Inn by Marriott, LLC Residence Inn Houston Downtown

Residence Inn by Marriott, LLC Richardson RI
Residence Inn by Marriott, LLC San Antonio RI
Residence Inn by Marriott, LLC Tyler RI

SpringHill SMC, LLCAddison Springhill SuitesSpringHill SMC, LLCArlington Springhill SuitesSpringHill SMC, LLCFort Worth University

TownePlace Management, LLC Houston Clearlake TownePlace Suites

State: Utah

Entity Name Assumed Name

Marriott Kauai Ownership Resorts, Inc. MVCI

Marriott Ownership Resorts, Inc.
Marriott's Mountainside Resort
Marriott Ownership Resorts, Inc.
Marriott's Summit Watch Resort

Marriott Ownership Resorts, Inc.MVCIMarriott Resorts Hospitality CorporationMVCIMarriott Resorts, Travel Company, Inc.MVCI

Residence Inn by Marriott, LLC Residence Inn at the Cottonwoods

State: Vermont

Entity Name Assumed Name

Fairfield FMC, LLC Burlington Colchester FibM

State: Virginia

Entity Name Assumed Name

Courtyard Management Corporation

Courtyard Management CorporationCharlottesville CbMCourtyard Management CorporationCourtyard by MarriottCourtyard Management CorporationCourtyard by MarriottCourtyard Management CorporationDulles South CbMCourtyard Management CorporationDulles Town Center CbMCourtyard Management CorporationDunn Loring Fairfax Courtyard

Brookfield CbM

Courtyard Management Corporation Fair Oaks CbM
Courtyard Management Corporation Herndon CbM,
Courtyard Management Corporation Manassas CbM

Courtyard Management Corporation Richmond Innsbrook CbM

Courtyard Management Corporation Rosslyn CbM

Marriott Hotel Services, Inc.

Berry Hill Conference Center

Marriott Hotel Services, Inc. CC Bistro

Marriott Hotel Services, Inc.

Crystal City Marriott at Regan National

Marriott Hotel Services, Inc.

Crystal City Marriott Hotel

Marriott Hotel Services, Inc.

Marriott's Westfields Conference Center
Marriott Hotel Services, Inc.

Westfield's Golf Club (Fairfax file date)

Marriott Hotel Services, Inc. Westfield's Marriott

Marriott Ownership Resorts, Inc.

MVCI

Exhibit 21

"CbM" means "Courtyard by Marriott" / "RIbM" means "Residence Inn by Marriott" / "FibM" means "Fairfield Inn by Marriott" / "MVCI" means "Marriott Vacation Club International" / "HMVC" means "Horizons by Marriott Vacation Club"

Marriott Resorts Hospitality Corporation

Marriott Resorts Hospitality Corporation Tidewater's Sweets and Sundries

MVCI

Marriott Resorts, Travel Company, Inc. MVCI

State: Virginia cont'd.

Residence Inn by Marriott, LLC Herndon Residence Inn by Marriott

Residence Inn by Marriott, LLC Herndon RI

Residence Inn by Marriott, LLC Residence Inn Alexandria - Old Town Residence Inn by Marriott, LLC Residence Inn Alexandria-Old Town The Ritz-Carlton Hotel Company, L.L.C. The Ritz-Carlton, Tysons Corner

**State: Washington** 

**Entity Name** Assumed Name

Courtyard Management Corporation Courtyard by Marriott CTYD III Corporation Courtyard by Marriott Marriott Hotel Services, Inc. Seattle Waterfront Marriott

Marriott Ownership Resorts, Inc. MVCI Marriott Resorts Hospitality Corporation MVCI

Residence Inn by Marriott, , LLC Residence Inn Redmond Seattle Downtown Springhill Suites SpringHill SMC, LLC

SpringHill SMC, LLC Seattle South Renton Springhill TownePlace Management, LLC Seattle South Renton TownePlace Suites

State: Wisconsin

**Assumed Name Entity Name** 

Fairfield FMC, LLC Madison FibM Fairfield FMC, LLC Milwaukee FibM Green Bay RI

Residence Inn by Marriott, LLC

Exhibit 21 Page 32

# **Consent of Independent Registered Public Accounting Firm**

We consent to the incorporation by reference in the following Registration Statements of Marriott International, Inc., and in the related Prospectuses of our reports dated February 14, 2008, with respect to the consolidated financial statements of Marriott International, Inc., and the effectiveness of internal control over financial reporting of Marriott International, Inc., included in this Annual Report (Form 10-K) for the fiscal year ended December 28, 2007.

- 1. Form S-3 ASR No. 333-130212
- 2. Form S-8 No. 333-125216
- 3. Form S-8 No. 333-55350
- 4. Form S-8 No. 333-36388
- 5. Form S-8 No. 333-48407

/s/ Ernst & Young LLP

McLean, Virginia February 14, 2008

Exhibit 23

# Certification of Chief Executive Officer Pursuant to Rule 13a–14(a)

### I, J.W. Marriott, Jr., certify that:

- 1. I have reviewed this annual report on Form 10-K of Marriott International, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February 15, 2008

/s/ J.W. Marriott, Jr.

J.W. Marriott, Jr. Chairman of the Board and Chief Executive Officer (Principal Executive Officer)

Exhibit 31.1

# Certification of Chief Financial Officer Pursuant to Rule 13a–14(a)

#### I, Arne M. Sorenson, certify that:

- 1. I have reviewed this annual report on Form 10-K of Marriott International, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February 15, 2008

/s/ Arne M. Sorenson

Arne M. Sorenson Executive Vice President and Chief Financial Officer (Principal Financial Officer)

Exhibit 31.2

# Certification Pursuant to Rule 13a–14(b) and Section 906 of the Sarbanes-Oxley Act of 2002 (18 U.S.C. Section 1350 (a) and (b))

- I, J.W. Marriott, Jr., Chairman of the Board and Chief Executive Officer of Marriott International, Inc. (the "Company") certify that:
  - (1) the annual report on Form 10-K of the Company for the year ended December 28, 2007 (the "Annual Report") fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)); and
  - (2) the information contained in the Annual Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

February 15, 2008

/s/ J.W. Marriott, Jr.

J.W. Marriott, Jr.

Chairman of the Board and Chief Executive Officer (Principal Executive Officer)

I, Arne M. Sorenson, Executive Vice President and Chief Financial Officer of Marriott International, Inc. (the "Company") certify that:

- (1) the annual report on Form 10-K of the Company for the year ended December 28, 2007 (the "Annual Report") fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)); and
- (2) the information contained in the Annual Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

February 15, 2008

/s/ Arne M. Sorenson

Arne M. Sorenson

Executive Vice President and Chief Financial Officer

(Principal Financial Officer)

Exhibit 32